## Analysis and optimal control of a damage model with penalty

Dissertation
zur Erlangung des akademischen Grades eines
Doktors der Naturwissenschaften
(Dr. rer. nat.)

der Fakultät für Mathematik der Technischen Universität Dortmund vorgelegt von

Livia-Mihaela Susu

im März 2017

-				
1)	isse	rt a	at i	on

Analysis and optimal control of a damage model with penalty

Fakultät für Mathematik Technische Universität Dortmund

Erstgutachter: Prof. Dr. Christian Meyer

Zweitgutachter: Prof. Dr. Dorothee Knees

Tag der mündlichen Prüfung: 31. Mai 2017

#### Acknowledgments

I would like to express my gratitude to Prof. Dr. Christian Meyer for the opportunity to write this thesis and for his valuable guidance and interest in sharing his knowledge.

I thank my parents for the encouragement to study abroad and for their support.

My special thanks go to my husband, Dr. Thomas Betz, who helped me in so many aspects. Thank you particularly for your moral support and for the fruitful discussions on mathematics.

## Contents

1	Ana	alysis of the damage model with penalty	9
	1.1	Existence and uniqueness of solutions	10
		1.1.1 The elliptic system	11
		1.1.2 The evolution equation	26
	1.2	Improved regularity and Lipschitz continuity of the nonlocal damage	29
		1.2.1 Improved regularity	30
		1.2.2 Improved Lipschitz continuity	31
	1.3	Fréchet-differentiability of the elliptic system	40
		1.3.1 Differentiability of $\mathcal{U}$	41
		1.3.2 Differentiability of $\Phi$	46
2	Pen	alization limit	53
	2.1	Energy identity	54
	2.2	Limit analysis	60
		2.2.1 Uniform boundedness	61
		2.2.2 Passage to the limit in the elliptic system	67
		2.2.3 Passage to the limit in the energy identity	69
	2.3	A single-field gradient damage model	75
	2.4	$L^{\infty}$ -bound for the limit damage variable	81
		2.4.1 Passage to the limit in the time-discrete problem	82
		2.4.2 Uniform estimates for the discrete damage variables	86
	2.5	Comparison to classical partial damage models	92
3	Opt	imal control of the damage model with penalty	98
	3.1	The dependence of the elliptic system on the load	100
	3.2	The control-to-state operator ${\mathcal S}$	114
	3.3	The directional differentiability of $\mathcal{S}$	120
	3.4	Optimality system	127
1	Cor	relusions and outlook	147

5	Auxiliary results				
	5.1	Nemytskii operators $g$ and $g'$	149		
	5.2	Product and chain rule	150		
	5.3	Properties of the max-operator	153		
	5.4	Picard-Lindelöf's theorem in abstract function spaces	156		
	5.5	Miscellaneous	159		
Bi	bliog	graphy	164		

## Introduction

This thesis is devoted to the analysis of a viscous *two-field* gradient damage model. The particular feature thereof is that it contains two damage variables, which are connected through a penalty term in the free energy functional. While one damage variable provides a local character and carries the time evolution, the other one accounts for nonlocal effects, cf. [12].

The problem under consideration describes the evolution of damage in an elastic body under the influence of an external loading. It is inspired by the model presented in [12], which is widely used in computational mechanics. From a mathematical perspective, the damage model in [12] provides two main drawbacks. Firstly, the coupling between the damage evolution and balance of momentum equation is realized via the less regular one of the two damage variables. To guarantee satisfying solvability results, we slightly modify the original problem, by coupling these equations through the more regular damage variable. This enables the use of compact embeddings, which are essential for establishing solvability results. It turns out that this modification has only little influence in practice, as we later obtain that both damage variables coincide when the penalty term becomes  $\infty$ . Secondly, [12] deals with a rate-independent model where the corresponding dissipation functional is unbounded. In this situation, the existence of classical solutions (also known as differential solutions, see [52]) has been proven under smoothness and convexity assumptions on the reduced energy functional, see [59] (uniform convexity) and [44] (quadratic case), whereas uniqueness results can be found in [59], as well as in [8, 53] (generalization of the result in [59] under restrictive assumptions). On the other hand, in the nonconvex case, solutions may often be discontinuous is time, and thus, weak solvability concepts are required, such as global energetic solutions, see [52, Section 3.3] for the abstract main existence result. Another popular approach in this context consists of performing a viscous regularization. Weak solutions for the rate-independent model are then found via a vanishing viscosity analysis. For references and more details, see the introduction of Chapter 1. Since after modifying the original model as mentioned above, our reduced energy functional is not necessarily convex, we follow this standard procedure, namely, we add a viscosity term to the damage evolution, which turns the rate-independent model into a rate-dependent one.

For the viscous penalized damage model under consideration, see (P) below, we establish in this thesis the following main results:

• The well-posedness from a mathematical point of view. We prove the unique

solvability of (P), make statements about the regularity of the unique solution and derive a system of differential equations which characterizes it (Chapter 1). These results can be also found in [49];

- The viability of the penalty approach. We show that the two damage variables become equal when the penalty term tends to infinity. It turns out that the limit damage variable satisfies a version of a classical viscous damage model analyzed by [41]. We derive conditions for the data under which both penalized damage variables and their limit are almost everywhere bounded by a desired positive value (Chapter 2). These results can be also found in [51];
- An optimality system for a class of optimal control problems governed by (P),
  where the external load acts as control. We prove the existence and directional
  differentiability of the associated control-to-state operator. We derive necessary
  optimality conditions in primal form and show that these are equivalent to an
  optimality system, if strict complementarity is fulfilled (Chapter 3).

To the best of our knowledge, a damage model containing two damage variables has never been investigated so far with regard to a rigorous mathematical analysis, although these models are frequently used for numerical simulations cf. e.g. [67, 76, 46, 64, 83]. This concerns the existence and regularity of solutions, let alone the behaviour as the penalty term tends to  $\infty$  and the optimal control thereof.

#### Motivation

The problem analyzed throughout this thesis was inspired by a damage model proposed in [12]. Therein two damage variables are introduced, which the authors call 'local' and 'nonlocal' damage. Time-dependent volume and boundary forces, denoted by  $\ell$ , are applied during the process upon an elastic body which has a part of its boundary clamped. The body is described by the domain  $\Omega \subset \mathbb{R}^N$ ,  $N \in \{2,3\}$ , on which we impose mild smoothness assumptions, see Assumption 0.5 below. The load induces a certain displacement  $\boldsymbol{u}:[0,T]\times\Omega\to\mathbb{R}^N$ , as well as local and nonlocal damage. The latter one is denoted by  $\varphi:[0,T]\times\Omega\to\mathbb{R}$ , while the local damage is called  $d:[0,T]\times\Omega\to\mathbb{R}$ . Its values measure the degree of the material rigidity loss. Therefore, d(t,x)=0 means that the body is completely sound, while  $d(t,x)=\infty$  means that the body is fully damaged.

In [12] the free energy function is enhanced by including a gradient term and a term which penalizes the difference between local and nonlocal damage. To be precise, the energy functional  $\hat{\mathcal{E}}: [0,T] \times V \times H^1(\Omega) \times L^2(\Omega) \to \mathbb{R}$  according to [12, (4) and (5)] is given by

$$\hat{\mathcal{E}}(t, \boldsymbol{u}, \varphi, d) := \frac{1}{2} \int_{\Omega} g(d) \mathbb{C}\varepsilon(\boldsymbol{u}) : \varepsilon(\boldsymbol{u}) \ dx - \langle \ell(t), \boldsymbol{u} \rangle_{V} + \frac{\alpha}{2} \|\nabla \varphi\|_{2}^{2} + \frac{\beta}{2} \|\varphi - d\|_{2}^{2},$$

where V is an appropriate Sobolev space and  $\varepsilon(\boldsymbol{u}) = \frac{1}{2}(\nabla \boldsymbol{u} + \nabla \boldsymbol{u}^{\top})$  is the linearized strain tensor. Note that the prefactor  $\gamma_1$  from [12] is entirely numerically motivated

and thus, not considered here. The function g cf. [12, (2)] is supposed to be smooth and it measures the influence of the damage on the elastic behaviour of the body, see Assumption 0.6 below.  $\mathbb{C}$  is the elasticity tensor, which is assumed to be coercive and bounded, see Assumption 0.7 below. The parameters  $\alpha, \beta > 0$  are weighting parameters for the gradient regularization and for the penalization, respectively, see [12, Section 2] for more details.

The model in [12] can be formulated as follows. At (almost) all time points, the displacement and the nonlocal damage are supposed to minimize the stored energy, i.e.,

$$(\boldsymbol{u}(t),\varphi(t)) \in \mathop{\arg\min}_{(\boldsymbol{u},\varphi) \in V \times H^1(\Omega)} \hat{\mathcal{E}}(t,\boldsymbol{u},\varphi,d(t)) \quad \text{ f.a.a. } t \in (0,T). \tag{1}$$

Further, the evolution of the local damage is modeled by the differential inclusion

$$-\partial_d \hat{\mathcal{E}}(t, \boldsymbol{u}(t), \varphi(t), d(t)) \in \partial \mathcal{R}_1(\dot{d}(t)) \quad \text{f.a.a. } t \in (0, T),$$

where the function  $\mathcal{R}_1$  denotes the dissipated energy. This is given by

**Definition 0.1** (Dissipation functional). The dissipation  $\mathcal{R}_1: L^2(\Omega) \to [0, \infty]$  is defined as

$$\mathcal{R}_1(\eta) := \begin{cases} r \int_{\Omega} \eta \ dx & \text{if } \eta \geq 0 \ \text{a.e. in } \Omega, \\ \infty & \text{otherwise,} \end{cases}$$

where r > 0 stands for the fracture toughness of the material.

Due to the positive homogeneity of  $\mathcal{R}_1$ , the considered model is *rate-independent*, which means that the values of the damage do not depend on the rate with which  $\ell$  changes in time. As a consequence, one ignores inertial effects and assumes a slow external process. Definition 0.1 also tells us that the damage can only increase in time, in view of (2). Thus, we deal with an irreversible damage process, i.e., healing is not allowed.

We observe that in (1) one solves (at almost all time points) a convex optimization problem. Thus, the global minimum is (at almost all time points) characterized via the necessary and sufficient conditions. An easy computation shows that these are in fact given by [12, (6) and (7)]. On the other hand, (2) is equivalent to [12, (18) combined with (19) and (20)]. To see this, we refer to Subsection 1.1.2, where a similar result is proven by employing the positive homogeneity of  $\mathcal{R}_1$ . Altogether, we remark that the system (1)-(2) describes indeed the damage model introduced in [12].

#### The mathematical model

Because of theoretical reasons, we modify the energy functional  $\hat{\mathcal{E}}$  such that the function g depends on the nonlocal damage  $\varphi$  instead of the local damage d. This modification is motivated by the fact that the local damage possesses less regularity, so that working further with the functional  $\hat{\mathcal{E}}$  would lead to undesirable solvability results. Instead, we consider the energy functional

**Definition 0.2** (Energy functional). The stored energy  $\mathcal{E}: [0,T] \times V \times H^1(\Omega) \times L^2(\Omega) \to \mathbb{R}$  is given by

$$\mathcal{E}(t, \boldsymbol{u}, \varphi, d) := \frac{1}{2} \int_{\Omega} g(\varphi) \mathbb{C}\varepsilon(\boldsymbol{u}) : \varepsilon(\boldsymbol{u}) \ dx - \langle \ell(t), \boldsymbol{u} \rangle_{V} + \frac{\alpha}{2} \|\nabla \varphi\|_{2}^{2} + \frac{\beta}{2} \|\varphi - d\|_{2}^{2}.$$

**Remark 0.3.** As the penalty approach aims to minimize the deviation between  $\varphi$  and d, we expect the two models to yield similar results, at least for large values of  $\beta$ . This is confirmed by the limit analysis for  $\beta \to \infty$  in Chapter 2, which shows that both damage variables equal in the limit.

We will also work with a different dissipation functional, namely a viscous regularization of the functional from Definition 0.1. Although weak solvability results for rate-independent damage processes with nonconvex energy functional as in our case may be proven, one can neither expect the solutions to be unique nor smooth in time, see [52] for more details. To overcome this issue, we add an  $L^2$ -viscosity term in the dissipation functional, which leads to a rate-dependent process, since the dissipation loses its positive homogeneity.

**Definition 0.4** (Viscous dissipation functional). We define  $\mathcal{R}_{\delta}: L^2(\Omega) \to [0, \infty]$  as

$$\mathcal{R}_{\delta}(\eta) := \begin{cases} r \int_{\Omega} \eta \ dx + \frac{\delta}{2} \|\eta\|_{2}^{2} & \text{if } \eta \geq 0 \ \text{a.e. in } \Omega, \\ \infty & \text{otherwise,} \end{cases}$$

where  $\delta > 0$  denotes the viscosity parameter.

Note that replacing  $\mathcal{R}_1$  by  $\mathcal{R}_{\delta}$  corresponds to smoothing the indicator functional in the dual formulation of the damage evolution by means of the Moreau-Yosida regularization, see Remark 1.31 below for more details.

To summarize, the viscous damage model with penalty arising from the above considerations reads:

$$\left\{ \begin{aligned} & (\boldsymbol{u}(t), \varphi(t)) \in \mathop{\arg\min}_{(\boldsymbol{u}, \varphi) \in V \times H^1(\Omega)} \mathcal{E}(t, \boldsymbol{u}, \varphi, d(t)), \\ & 0 \in \partial \mathcal{R}_{\delta}(\dot{d}(t)) + \partial_d \mathcal{E}(t, \boldsymbol{u}(t), \varphi(t), d(t)) \quad \text{f.a.a. } t \in (0, T), \end{aligned} \right\}$$
 (P)

with the initial condition  $d(0) = d_0$  a.e. in  $\Omega$ .

#### Notation

In all what follows, T>0 is fixed and  $\Omega\subset\mathbb{R}^N$  is a bounded domain, where  $N\in\{2,3\}$  denotes the spatial dimension. By bold-face case letters we denote vector-valued variables and vector-valued spaces. The Frobenius norm on  $\mathbb{R}^{N\times N}$ , as well as the euclidean norm on  $\mathbb{R}^N$  are denoted by  $|\cdot|$ , whereas the inducing scalar product in  $\mathbb{R}^{N\times N}$  is represented by  $(\cdot:\cdot)$ . Let X and Y be Banach spaces. The open ball in X around  $X \in X$  with radius X > 0 is denoted by  $X \in X$ . The space of linear and bounded operators from X to X is called X0, and if X1 is called X2. The dual of the space

X will be denoted by  $X^*$  and for the dual pairing between X and  $X^*$  we write  $\langle .,.\rangle_X$ . If X is compactly embedded in Y, we write  $X \hookrightarrow \hookrightarrow Y$  and  $X \stackrel{d}{\hookrightarrow} Y$  means that X is dense in Y. Let  $s \in [1,\infty]$  and s' be the conjugate exponent of s, i.e., 1/s + 1/s' = 1. By  $\|\cdot\|_s$  we abbreviate the notation for the  $L^s(\Omega)$ -norm and by  $(\cdot,\cdot)_2$ , the  $L^2(\Omega)$ -scalar product. For frequently used function spaces we introduce the following abbreviations:

$$\begin{split} \boldsymbol{W}_{\widetilde{\Gamma}}^{1,s}(\Omega) &:= \{\boldsymbol{v} \in \boldsymbol{W}^{1,s}(\Omega) : \boldsymbol{v}|_{\widetilde{\Gamma}} = \boldsymbol{0}\}, \\ \boldsymbol{W}_{D}^{1,s}(\Omega) &:= \boldsymbol{W}_{\Gamma_{D}}^{1,s}(\Omega), \\ \boldsymbol{W}_{0}^{1,s}(\Omega) &:= \boldsymbol{W}_{\Gamma}^{1,s}(\Omega), \\ \boldsymbol{V} &:= \boldsymbol{W}_{D}^{1,2}(\Omega), \\ \boldsymbol{W}^{-1,s}(\Omega) &:= \boldsymbol{W}_{D}^{1,s'}(\Omega)^{*}, \\ \boldsymbol{W}_{0}^{1,s}(0,T;\boldsymbol{X}) &:= \{\boldsymbol{z} \in \boldsymbol{W}^{1,s}(0,T;\boldsymbol{X}) : \boldsymbol{z}(0) = \boldsymbol{0}\}, \\ \boldsymbol{W}_{T}^{1,s}(0,T;\boldsymbol{X}) &:= \{\boldsymbol{z} \in \boldsymbol{W}^{1,s}(0,T;\boldsymbol{X}) : \boldsymbol{z}(T) = \boldsymbol{0}\}, \\ \boldsymbol{H}_{0}^{1}(0,T;\boldsymbol{X}) &:= \boldsymbol{W}_{0}^{1,2}(0,T;\boldsymbol{X}), \end{split}$$

where  $\widetilde{\Gamma}$ ,  $\Gamma_D$  and  $\Gamma$  stand for a part of the boundary, the Dirichlet boundary and the entire boundary of  $\Omega$ , respectively. For the precise assumption on  $\Gamma_D$  see Assumption 0.5 below. By div :  $L^s(\Omega; \mathbb{R}^{N \times N}_{sym}) \to W^{1,s'}(\Omega)^*$  we denote the distributional vector-valued divergence, i.e.,

$$\langle \operatorname{div} \boldsymbol{\sigma}, \boldsymbol{v} \rangle := -\int_{\Omega} \boldsymbol{\sigma} : \varepsilon(\boldsymbol{v}) \, dx \quad \forall \, \boldsymbol{v} \in \boldsymbol{W}^{1,s'}(\Omega)$$

and  $\Delta:W^{1,s}(\Omega)\to W^{1,s'}(\Omega)^*$  is the distributional Laplace operator, i.e.,

$$\langle \Delta \phi, \psi \rangle := -\int_{\Omega} \nabla \phi \nabla \psi \, dx \quad \forall \, \psi \in W^{1,s'}(\Omega).$$

Throughout the thesis, c and C denote generic positive constants, which depend only on the given data, unless otherwise specified. Time derivatives are frequently denoted by a dot.  $\mathbb{R}^+$  and  $\mathbb{N}^+$  represent the sets of positive real and natural numbers, respectively. By  $\chi_M$  we denote the characteristic function associated to the set M. For  $a, b \in \mathbb{R}$  we write  $a \perp b$  if ab = 0. For  $f : [0,T] \times \Omega \to \mathbb{R} \cup \{\infty\}$  and  $a \in \mathbb{R}$  the abbreviation  $f \geq a$  means that  $f(t,x) \geq a$  f.a.a.  $(t,x) \in (0,T) \times \Omega$ . Given  $A: X \to Y$ , the symbol  $A^{-1}y$  stands for the inverse at a fixed point  $y \in Y$  and does not necessarily mean that  $A: X \to Y$  is invertible. For a mapping  $A: X \to L^1(0,T;Y)$  and  $U \subset Y$  we say that  $A(M)(t) \in U$  if and only if  $A(x)(t) \in U$  for all  $x \in M \subset X$ , where  $t \in [0,T]$ . Finally, we point out that the notation is not strictly consistent, but it is always clear from the context. To improve the readability, we sometimes neglect the subscripts when working with dual pairings, norms, scalar products and variables as long as it is clear from the context which one is meant. Nevertheless, to emphasize the dependency on the penalty term  $\beta$  and on the load  $\ell$ , respectively, the problem (P) is renamed  $(P_{\beta})$  in Chapter 2 and  $(P_{\ell})$  in Chapter 3.

### General assumptions

**Assumption 0.5.** The domain  $\Omega \subset \mathbb{R}^N$ ,  $N \in \{2,3\}$ , is a bounded Lipschitz domain, see [22, Chap. 1.2]. Its boundary is denoted by  $\Gamma$  and consists of two disjoint measurable parts  $\Gamma_N$  and  $\Gamma_D$  such that  $\Gamma = \Gamma_N \cup \Gamma_D$ . While  $\Gamma_N$  is an open subset,  $\Gamma_D$  is a closed subset of  $\Gamma$ . Moreover,  $\Gamma_D$  is assumed to have positive measure.

In addition, the set  $\Omega \cup \Gamma_N$  is regular in the sense of Gröger, cf. [23, Definition 2]. That is, for every point  $\mathbf{x} \in \Gamma$ , there exists an open neighborhood  $\mathcal{U}_{\mathbf{x}} \subset \mathbb{R}^N$  of  $\mathbf{x}$  and a bi-Lipschitz map (a Lipschitz continuous and bijective map with Lipschitz continuous inverse)  $\Psi_{\mathbf{x}} : \mathcal{U}_{\mathbf{x}} \to \mathbb{R}^N$  such that  $\Psi_{\mathbf{x}}(\mathbf{x}) = \mathbf{0} \in \mathbb{R}^N$  and  $\Psi_{\mathbf{x}}(\mathcal{U}_{\mathbf{x}} \cap (\Omega \cup \Gamma_N))$  equals one of the following sets:

$$E_1 := \{ \mathbf{y} \in \mathbb{R}^N : |y| < 1, \ y_N < 0 \},$$
  

$$E_2 := \{ \mathbf{y} \in \mathbb{R}^N : |y| < 1, \ y_N \le 0 \},$$
  

$$E_3 := \{ \mathbf{y} \in E_2 : y_N < 0 \text{ or } y_1 > 0 \}.$$

A detailed characterization of Gröger regular sets in two and three spatial dimensions is given in [24, Section 5].

**Assumption 0.6.** The function 
$$g : \mathbb{R} \to [\epsilon, 1]$$
, where  $\epsilon \in (0, 1]$ , satisfies  $g \in C^{1,1}(\mathbb{R})$ .

With a little abuse of notation, the Nemytskii operators associated to g and g', considered with different domains and ranges, will be denoted by the same symbol.

The coefficient function g measures how the elastic properties of the body are preserved depending on the value of the damage. Since with increasing damage, the material becomes weaker, it would make sense to impose that g is monotonically decreasing. This property of g is needed e.g. if one aims to show that the nonlocal damage variable admits just non-negative values, as the local damage variable does. For this, it suffices in fact that g decreases only on its negative domain, cf. Remark 2.40 on page 92 below. However, since we do not need this result in our analysis, we do not require here that g has this property.

Letting the smoothness assumption aside, we shortly compare the assumptions we made on g with those in [12, (2)]. Firstly, we observe that in [12] it holds g(0) = 1, i.e., the elastic properties of the body are not affected at all (pure elastic behaviour) when no damage is present. This precise condition is however not needed for the upcoming analysis and from a mathematical point of view, it would have actually sufficed to impose that  $g(\cdot) \leq \epsilon_1$  (instead of  $g(\cdot) \leq 1$ ) with some arbitrary (but fixed)  $\epsilon_1 \geq \epsilon$ . Secondly,  $\lim_{\varphi \to \infty} g(\varphi) = 0$  is assumed in [12, (2)], which means that complete material rigidity loss occurs in the case of complete damage. By contrast, we impose the condition  $g(\cdot) \geq \epsilon > 0$ . This is essential for the solvability of (P) or to be more precise, for the coercivity of the bilinear form associated with the balance of momentum equation (1.11), see also proof of Lemma 1.3 below. Thus, as in the most mathematical literature, we investigate a partial damage model. For methods for dealing with complete desintegration

we refer here to [27, Chapter 6], see also [68] for rate-dependent complete damage systems in thermoviscoelastic materials.

**Assumption 0.7.** The fourth-order tensor  $\mathbb{C} \in L^{\infty}(\Omega; \mathcal{L}(\mathbb{R}^{N \times N}_{\mathrm{sym}}))$  is symmetric and uniformly coercive, i.e., there is a constant  $\gamma_{\mathbb{C}} > 0$  such that

$$\mathbb{C}(x)\boldsymbol{\sigma}: \boldsymbol{\sigma} \ge \gamma_{\mathbb{C}}|\boldsymbol{\sigma}|^2 \quad \forall \, \boldsymbol{\sigma} \in \mathbb{R}_{\mathrm{sym}}^{N \times N} \, and \, f.a.a. \, \, x \in \Omega.$$
 (4)

**Assumption 0.8.** The load  $\ell$  is a mapping from [0,T] to  $V^*$ . The initial damage is supposed to satisfy  $d_0 \in L^2(\Omega)$ .

For convenience of the reader, the often used notations are collected in the following table:

Variables, functionals, (solution) operators, integrability exponents and sets  $\,$ 

Symbol	Meaning	Definition
$\mathcal{R}_1$	Dissipation functional	Definition 0.1, p. 3
$\mathcal{E}$	Stored energy functional	Definition 0.2, p. 4
$\mathcal{R}_{\delta}$	Viscous dissipation functional	Definition 0.4, p. 4
$A_{\varphi}$	Linear elliptic operator in (1.25a)	Definition 1.2, p. 12
$m{W}_D^{1,p}(\Omega)$	Space regularity of the optimal displacement	Lemma 1.3, p. 12
U	Solution operator of (1.25a)	Definition 1.8, p. 15
B	Linear part in (1.25b)	Definition 1.15, p. 18
F	Nonlinear part in (1.25b)	Definition 1.15, p. 18
Φ	Solution operator of (1.25b)	Definition 1.24, p. 24
$W^{1,q}(\Omega)$	Space regularity of the optimal nonlocal damage	Theorem 1.37, p. 31
$\widetilde{\mathcal{E}}$	Reduced energy functional	Definition 2.1, p. 55
$\widetilde{\mathcal{E}}$	Energy functional without penalty	(2.47), p. 70
$\widetilde{\mathcal{I}}$	Reduced energy functional without penalty	Definition 2.19, p. 70
$\widetilde{\mathcal{R}}_1$	Dissipation functional after passing to the limit $\beta \to \infty$	(2.76), p. 78
$\widetilde{\mathcal{R}}_{\delta}$	Viscous dissipation functional after passing to the limit $\beta \to \infty$	Definition 2.21, p. 71
$\mathcal{B}_M$	Set of admissible loads (time-independent case)	Definition 3.3, p. 103
$\mathfrak{B}_M$	Set of admissible loads (time-dependent case)	Definition 3.21, p. 118
$\mathcal{S}$	Control-to-state operator	Definition 3.22, p. 119
£	Control set	Assumption 3.29, p. 128
$\mathcal J$	Objective functional	Assumption 3.33, p. 129
u	Displacement	
$\varphi$ $d$	Nonlocal damage	
d	Local damage	

### Chapter 1

# Analysis of the damage model with penalty

In this chapter we show that the problem (P) admits a unique solution, and thus, that the modified damage model with penalty is well posed from a mathematical point of view. Moreover, we investigate the regularity of the unique solution on different levels. We also derive an equivalent formulation of (P), which consists of an elliptic system for the displacement field and the nonlocal damage and an operator differential equation for the local damage. The problem under consideration is analyzed throughout this chapter in the context of a smooth given load and a fixed penalization parameter. The herein achieved results constitute the basis for the rest of the thesis. In Chapter 2 we employ in particular a final existence result for (P), see Theorem 1.62 below, as well as various intermediate results. Concerning Chapter 3, we mention Sections 3.1 and 3.2, which rely on arguments similar to the proofs given in the present chapter.

Let us put our work into perspective. Numerous damage models have been addressed by many authors under different aspects. In [4, 5, 6, 16, 26] various viscous damage models are analyzed regarding existence of solutions. The concept of viscosity plays an important role in the mathematical treatment of rate-independent damage models, as the vanishing viscosity approach is a prominent method to find solutions for rate-independent problems, such as balanced viscosity (BV) solutions, see [55, 56], and parametrized solutions, see [13, 41, 42, 61]. These are obtained by considering a viscous dissipation in the initial model, which becomes rate-dependent, and by then letting the viscosity parameter tend to zero. We further refer here to [1, 39, 43, 45, 47, 48, 54, 78] and the references therein. There are also other solution concepts for evolutionary equations in the context of rate-independent damage modelling. An overview thereof can be found in [52], in the framework of generalized gradient systems. A particular attention is given in the literature to discontinuous solutions, such as energetic solutions and BV solutions, which were introduced in [58, 60] and [54, 55], respectively.

Unlike in the above contributions, we deal with a viscous damage model containing two damage variables. To the best of our knowledge, such a model has not been investigated so far with regard to a rigorous mathematical analysis, although these models are frequently used for numerical simulations, cf. e.g. [46, 64, 67, 76, 83]. We approach the problem (P) in this chapter mostly by classical arguments, such as elements of calculus of variations, a contraction argument, boot strapping and the implicit function theorem. An essential tool in the context of proving the unique solvability of (P) is the  $W^{1,p}$ -theory with p > 2 from [29].

### Outline of the chapter

The chapter is organized as follows. In Section 1.1 we first address the existence and uniqueness of solutions for the minimization problem in (P) and show that it is equivalent to an elliptic system. Based on these results, we then deal with the complete model including the evolutionary equation for the local damage. This turns out to be equivalent to an operator differential equation and the unique solvability of (P) follows by standard arguments. Sections 1.2 and 1.3 are devoted to improve the regularity of the solution. In Section 1.2 we show a higher spatial regularity result for the nonlocal damage, as well as a corresponding Lipschitz continuity condition. In Section 1.3 we prove that the solution operators of the elliptic system are continuously Fréchet-differentiable. This finally allows us to establish that the overall solution of (P) is continuously differentiable in appropriate spaces, as a mapping in time.

#### 1.1 Existence and uniqueness of solutions

In this section we mainly focus on finding unique solutions  $\boldsymbol{u}, \varphi, d$  to the problem (P) for a given load  $\ell$ . For this purpose, we first show that the optimization problem in (P) admits solutions for fixed t and d. However, the existence cannot be demonstrated by the direct method of the calculus of variations, since in the first place the displacement  $\boldsymbol{u}$  and the nonlocal damage  $\varphi$  do not provide sufficient regularity, see also Definition 0.2 and Lemma 5.2. Therefore we proceed as follows. Starting from

$$\min_{(\boldsymbol{u},\varphi)\in V\times H^1(\Omega)} \mathcal{E}(t,\boldsymbol{u},\varphi,d) = \min_{\varphi\in H^1(\Omega)} \min_{\boldsymbol{u}\in V} \mathcal{E}(t,\boldsymbol{u},\varphi,d), \tag{1.1}$$

we first show that, for every  $\varphi \in H^1(\Omega)$ , the problem  $\min_{u \in V} \mathcal{E}(t, u, \varphi, d)$  admits a unique solution, denoted by  $\mathcal{U}(t, \varphi)$ , which possesses improved regularity. In the second part of Subsection 1.1.1 this allows us to show existence of solutions for the outer optimization problem on the right-hand side in (1.1), see also Remark 1.13 below. Such solutions will turn out to satisfy (together with the corresponding optimal displacement) the elliptic system (1.25) below, as necessary optimality condition. As this system is uniquely solvable under suitable assumptions, we obtain the unique solvability of the optimization problem in (P), with solutions characterized by (1.25). After concluding the uniqueness, the Lipschitz continuity of the resulting solution maps is proven. Finally, by means of the latter ones, the problem (P) can be reduced to an evolutionary equation, which is in fact an ordinary differential equation in Banach space. This aspect is addressed in Subsection 1.1.2, where one establishes the unique solvability of the reduced problem (P). The section ends with a first existence and uniqueness result for (P), which will be improved in the upcoming sections.

#### 1.1.1 The elliptic system

Throughout this section we work with a fixed  $(t,d) \in [0,T] \times L^2(\Omega)$  and deal with the optimization problem

$$\min_{(\boldsymbol{u},\varphi)\in V\times H^1(\Omega)} \mathcal{J}(\boldsymbol{u},\varphi),\tag{1.2}$$

where  $\mathcal{J}: V \times H^1(\Omega) \to \mathbb{R}$  is defined as

$$\mathcal{J}(\boldsymbol{u},\varphi) := \mathcal{E}(t,\boldsymbol{u},\varphi,d),\tag{1.3}$$

This means in view of Definition 0.2 that for all  $(\boldsymbol{u}, \varphi) \in V \times H^1(\Omega)$  it holds

$$\mathcal{J}(\boldsymbol{u},\varphi) = \frac{1}{2} \int_{\Omega} g(\varphi) \mathbb{C}\varepsilon(\boldsymbol{u}) : \varepsilon(\boldsymbol{u}) \ dx - \langle \ell(t), \boldsymbol{u} \rangle_{V} + \frac{\alpha}{2} \|\nabla \varphi\|_{2}^{2} + \frac{\beta}{2} \|\varphi - d\|_{2}^{2}.$$

#### **Displacement**

As indicated above, we first fix  $\varphi \in H^1(\Omega)$  and investigate the problem

$$\min_{\boldsymbol{u}\in V} \mathcal{J}(\boldsymbol{u},\varphi). \tag{1.4}$$

The unique solvability of (1.4) can be followed by very standard arguments, while proving the improved regularity of the optimal displacement requires some preparatory work. The key tool therefor is the next result, which is based on Assumption 0.5:

**Lemma 1.1.** [29, Theorem 1.1, Proposition 1.2] Let  $\{b_{\iota}\}$  be a family of nonlinearities, so that each  $b_{\iota} \colon \Omega \times \mathbb{R}^{N \times N}_{\mathrm{sym}} \to \mathbb{R}^{N \times N}_{\mathrm{sym}}$  satisfies

$$b_{\iota}(\cdot, \mathbf{0}) \in L^{\infty}(\Omega; \mathbb{R}_{\text{sym}}^{N \times N}),$$
 (1.5a)

$$b_{\iota}(\cdot, \boldsymbol{\sigma})$$
 is measurable, (1.5b)

$$(b_t(x,\sigma) - b_t(x,\varsigma)) : (\sigma - \varsigma) > m|\sigma - \varsigma|^2, \tag{1.5c}$$

$$|b_{\iota}(x, \boldsymbol{\sigma}) - b_{\iota}(x, \boldsymbol{\varsigma})| \le \overline{m} |\boldsymbol{\sigma} - \boldsymbol{\varsigma}| \tag{1.5d}$$

f.a.a.  $x \in \Omega$  and all  $\boldsymbol{\sigma}, \boldsymbol{\varsigma} \in \mathbb{R}^{N \times N}_{\text{sym}}$  with constants  $\underline{m}, \overline{m} > 0$  independent of  $\iota$ . Furthermore, let  $B_{p,\iota} \colon \boldsymbol{W}^{1,p}_D(\Omega) \to \boldsymbol{W}^{-1,p}(\Omega)$  be defined as

$$\langle B_{p,\iota}(\boldsymbol{u}), \boldsymbol{v} \rangle := \int_{\Omega} b_{\iota}(\cdot, \varepsilon(\boldsymbol{u})) : \varepsilon(\boldsymbol{v}) \ dx \quad \forall \, \boldsymbol{v} \in \boldsymbol{W}_{D}^{1,p'}(\Omega).$$

Then there exists p > 2 such that for all  $\bar{p} \in [2, p]$ , all operators  $B_{\bar{p},\iota}$  are continuously invertible. Moreover, their inverses are globally Lipschitz continuous and share a uniform Lipschitz constant independent of  $\bar{p} \in [2, p)$  and  $\iota$ .

The above result will be applied in the upcoming lemma for the following operator(s):

**Definition 1.2.** Given  $\varphi \in L^1(\Omega)$ , we define the linearity  $A_{\varphi}: V \to V^*$  as

$$\langle A_{\varphi} \boldsymbol{u}, \boldsymbol{v} \rangle_{V} := \int_{\Omega} g(\varphi) \mathbb{C} \varepsilon(\boldsymbol{u}) : \varepsilon(\boldsymbol{v}) \ dx \quad \forall \, \boldsymbol{v} \in V.$$

The operator  $A_{\varphi}$  considered with different domains and ranges will be denoted by the same symbol for the sake of convenience. Note that  $A_{\varphi}$  is well defined in view of Hölder's inequality combined with Assumption 0.6 and  $\mathbb{C} \in L^{\infty}(\Omega; \mathcal{L}(\mathbb{R}^{N \times N}_{\text{sym}}))$ , cf. Assumption 0.7.

**Lemma 1.3.** There exists p > 2 such that, for all  $\bar{p} \in [2, p]$  and all  $\varphi \in L^1(\Omega)$ , the operator  $A_{\varphi} : W_D^{1,\bar{p}}(\Omega) \to W^{-1,\bar{p}}(\Omega)$  is continuously invertible. Moreover, there exists a constant c > 0, independent of  $\varphi$  and  $\bar{p} \in [2, p)$ , such that

$$||A_{\varphi}^{-1}h||_{\mathbf{W}_{D}^{1,\bar{p}}(\Omega)} \le c ||h||_{\mathbf{W}^{-1,\bar{p}}(\Omega)} \quad \forall h \in \mathbf{W}^{-1,\bar{p}}(\Omega)$$
 (1.6)

holds for all  $\varphi \in L^1(\Omega)$  and  $\bar{p} \in [2, p]$ .

*Proof.* The result follows by applying Lemma 1.1. To this end, we define the family of functions  $\{b_{\varphi}\}_{{\varphi}\in L^1(\Omega)},\ b_{\varphi}:\Omega\times\mathbb{R}^{N\times N}_{\mathrm{sym}}\to\mathbb{R}^{N\times N}_{\mathrm{sym}}$ , by

$$b_{\varphi}(x, \boldsymbol{\sigma}) := g(\varphi(x))\mathbb{C}(x)\boldsymbol{\sigma} \tag{1.7}$$

and verify (1.5) therefor. The condition (1.5a) is obviously fulfilled as  $\mathbb{C}(x) \in \mathcal{L}(\mathbb{R}^{N \times N}_{\mathrm{sym}})$  f.a.a.  $x \in \Omega$ , and thus,  $\mathbb{C}(\cdot)\mathbf{0} = \mathbf{0}$ , while (1.5b) follows by Assumptions 0.6 and 0.7. From the latter ones we also deduce the uniform coercivity, i.e., (1.5c) and boundedness, i.e., (1.5d), with  $\underline{m} = \epsilon \gamma_{\mathbb{C}} > 0$  and  $\overline{m} = \|\mathbb{C}\|_{\infty}$ , respectively. Since these constants are independent of  $\varphi$ , we can now apply Lemma 1.1, which yields the existence of a p > 2 such that, for all  $\bar{p} \in [2, p]$  and all  $\varphi \in L^1(\Omega)$ , the operator  $A_{\varphi} : \mathbf{W}_D^{1,\bar{p}}(\Omega) \to \mathbf{W}^{-1,\bar{p}}(\Omega)$  is continuously invertible, as a result of (1.7). Note that since  $A_{\varphi} \in \mathcal{L}(\mathbf{W}_D^{1,\bar{p}}(\Omega), \mathbf{W}^{-1,\bar{p}}(\Omega))$ , the global Lipschitz continuity of its inverse holds true anyway. However, Lemma 1.1 gives us the additional information that the norm of  $A_{\varphi}^{-1} \in \mathcal{L}(\mathbf{W}^{-1,\bar{p}}(\Omega), \mathbf{W}_D^{1,\bar{p}}(\Omega))$  can be estimated independently of  $\varphi$  and  $\bar{p} \in [2, p)$ . This finalizes the proof.

Remark 1.4. In view of Definition 1.2, the operator  $A_{\varphi} \in \mathcal{L}(W_D^{1,\zeta}(\Omega), W^{-1,\zeta}(\Omega))$  is the adjoint of  $A_{\varphi} \in \mathcal{L}(W_D^{1,\zeta'}(\Omega), W^{-1,\zeta'}(\Omega))$  for any  $\zeta \in (1,\infty)$  and any  $\varphi \in L^1(\Omega)$ . This follows by the definition of the adjoint, where one employs the reflexivity of  $W_D^{1,\zeta}(\Omega)$ . Therefore, in view of Lemma 1.3,  $A_{\varphi} : W_D^{1,\zeta}(\Omega) \to W^{-1,\zeta}(\Omega)$  is continuously invertible for  $p' \leq \zeta \leq 2$  as well, and

$$\|A_{\varphi}^{-1}\|_{\mathcal{L}(\boldsymbol{W}^{-1,\zeta}(\Omega),\boldsymbol{W}_{D}^{1,\zeta}(\Omega))} = \|A_{\varphi}^{-1}\|_{\mathcal{L}(\boldsymbol{W}^{-1,\zeta'}(\Omega),\boldsymbol{W}_{D}^{1,\zeta'}(\Omega))} \le c,$$

where c > 0 is independent of  $\varphi$  and  $\zeta' \in [2, p)$ , and thus of  $\zeta \in (p', 2]$ .

In the rest of the thesis, p > 2 stands for the exponent given by Lemma 1.3.

We can now state the assumption for the load, which is supposed to hold throughout this entire chapter and the next one. This reads as follows: **Assumption 1.5.** For the applied volume and boundary load we require

$$\ell \in C^{0,1}([0,T]; \mathbf{W}^{-1,p}(\Omega)),$$

see also Assumptions 1.14 and 1.17.1 below.

We point out that the choice of the space for the load is not random at all, as explained in Remark 1.26 below.

The next lemma covers some differentiability properties of the energy functional  $\mathcal{E}$ . These will be in particular needed when deriving optimality conditions for the optimal displacement and optimal nonlocal damage.

**Lemma 1.6.** Then the functional  $\mathcal{E}$  is partially Fréchet-differentiable w.r.t.  $\mathbf{u}$  and d on  $[0,T] \times V \times H^1(\Omega) \times L^2(\Omega)$ , and the partial derivatives are given by

$$\partial_{\boldsymbol{u}} \mathcal{E}(t, \boldsymbol{u}, \varphi, d) = A_{\varphi} \boldsymbol{u} - \ell(t) \quad \text{in } V^*,$$
 (1.8)

$$\partial_d \mathcal{E}(t, \boldsymbol{u}, \varphi, d) = \beta(d - \varphi) \quad \text{in } L^2(\Omega).$$
 (1.9)

Furthermore, if considered as a mapping on  $[0,T] \times \boldsymbol{W}_{D}^{1,r}(\Omega) \times H^{1}(\Omega) \times L^{2}(\Omega)$  with r > 2 for N = 2 and r > 12/5 for N = 3, then  $\mathcal{E}$  is also partially Fréchet-differentiable w.r.t.  $\varphi$ . The partial derivative reads

$$\partial_{\varphi} \mathcal{E}(t, \boldsymbol{u}, \varphi, d)(\delta \varphi) = \frac{1}{2} \int_{\Omega} g'(\varphi) \mathbb{C}\varepsilon(\boldsymbol{u}) : \varepsilon(\boldsymbol{u}) \delta \varphi \ dx + \int_{\Omega} \alpha \nabla \varphi \cdot \nabla \delta \varphi + \beta(\varphi - d) \delta \varphi \ dx \quad \forall \, \delta \varphi \in H^{1}(\Omega).$$
(1.10)

*Proof.* Let  $(t, \varphi, d) \in [0, T] \times H^1(\Omega) \times L^2(\Omega)$  be arbitrary, but fixed. To prove the partial Fréchet-differentiability w.r.t.  $\boldsymbol{u}$  we first observe that the mapping

$$\boldsymbol{u} \mapsto \frac{1}{2} \int_{\Omega} g(\varphi) \mathbb{C} \varepsilon(\boldsymbol{u}) : \varepsilon(\boldsymbol{u}) \ dx$$

is Fréchet-differentiable on V by means of product rule. Keep in mind that g maps  $H^1(\Omega)$  to  $L^{\infty}(\Omega)$ , cf. Lemma 5.1, and  $\varepsilon \in \mathcal{L}(V; L^2(\Omega; \mathbb{R}^{N \times N}))$ . In view of Definition 1.2 and since  $\ell(t) \in V^*$ , cf. Assumption 0.8, we get (1.8).

The partial Fréchet-differentiability w.r.t. d follows immediately from the Fréchet-differentiability of  $\|\cdot\|_2^2$  on  $L^2(\Omega)$ .

Let now  $(t, \boldsymbol{u}, d) \in [0, T] \times \boldsymbol{W}_{D}^{1,r}(\Omega) \times L^{2}(\Omega)$  be arbitrary, but fixed. On account of Hölder's inequality with (r-2)/r + 2/r = 1 we deduce that the linear functional

$$L^{\frac{r}{r-2}}(\Omega) \ni w \mapsto \frac{1}{2} \int_{\Omega} w \, \mathbb{C}\varepsilon(\boldsymbol{u}) : \varepsilon(\boldsymbol{u}) \, dx \in \mathbb{R}$$

is bounded and thus an element of  $L^{r/(r-2)}(\Omega)^*$ . Moreover, the conditions on r and Sobolev embeddings imply  $H^1(\Omega) \hookrightarrow L^s(\Omega)$  with some s > r/(r-2) so that, in view of Lemma 5.3, g is Fréchet-differentiable from  $H^1(\Omega)$  to  $L^{r/(r-2)}(\Omega)$ . The identity (1.10) then follows from chain rule, in combination with the Fréchet-differentiability of  $\|\nabla \cdot\|_2^2$  and  $\|\cdot\|_2^2$  on  $H^1(\Omega)$ .

We are now finally in the position to state the unique solvability of (1.4), as well as the (improved) regularity of the optimal solution:

**Proposition 1.7** (Existence and uniqueness of the optimal displacement). Let Assumption 1.5 hold. Then, for every  $\varphi \in H^1(\Omega)$ , the optimization problem (1.4) is convex and admits a unique solution  $\bar{\boldsymbol{u}} \in \boldsymbol{W}_D^{1,p}(\Omega)$ , which is characterized by

$$\langle A_{\varphi}\bar{\boldsymbol{u}}, \boldsymbol{v} \rangle_{\boldsymbol{W}_{D}^{1,p'}(\Omega)} = \langle \ell(t), \boldsymbol{v} \rangle_{\boldsymbol{W}_{D}^{1,p'}(\Omega)} \quad \forall \, \boldsymbol{v} \in \boldsymbol{W}_{D}^{1,p'}(\Omega).$$
 (1.11)

*Proof.* Let  $\varphi \in H^1(\Omega)$  be arbitrary, but fixed and define the functional

$$f_{\varphi}: V \ni \boldsymbol{u} \mapsto \mathcal{J}(\boldsymbol{u}, \varphi) \in \mathbb{R},$$

which is just the objective in (1.4). First we establish that  $f_{\varphi}$  is strictly convex, by checking if

$$f_{\varphi}(u_1) - f_{\varphi}(u_2) > f'_{\varphi}(u_2)(u_1 - u_2) \quad \forall u_1, u_2 \in V, u_1 \neq u_2,$$
 (1.12)

holds true. In view of (1.3) and (1.8), (1.12) is equivalent to

$$\int_{\Omega} g(\varphi) \mathbb{C}[\varepsilon(\boldsymbol{u}_1) - \varepsilon(\boldsymbol{u}_2)] : [\varepsilon(\boldsymbol{u}_1) - \varepsilon(\boldsymbol{u}_2)] \ dx > 0, \quad \forall \, \boldsymbol{u}_1, \boldsymbol{u}_2 \in V, \, \boldsymbol{u}_1 \neq \boldsymbol{u}_2, \quad (1.13)$$

where we also employed Definitions 0.2 and 1.2. Since (1.13) can be deduced from Assumptions 0.6 and 0.7, we obtain that  $f_{\varphi}$  is indeed strictly convex. Thus, if existent, the solution  $\bar{u}$  of (1.4) is unique and, since we optimize on V,  $\bar{u}$  is characterized by

$$A_{\varphi}\bar{\boldsymbol{u}} = \ell(t) \quad \text{in } V^*, \tag{1.14}$$

in view of (1.8). Lemma 1.3 now gives the (unique) solvability as well as the improved regularity of  $\bar{\boldsymbol{u}}$ , which is guaranteed by Assumption 1.5. In light of  $V \stackrel{d}{\hookrightarrow} \boldsymbol{W}_D^{1,p'}(\Omega)$  and (1.14) combined with Definition 1.2, the unique solution  $\bar{\boldsymbol{u}}$  is (also) characterized by (1.11). This completes the proof.

The unique solvability of (1.11) leads to the following

**Definition 1.8** (Solution operator of (1.11)). Under Assumption 1.5, we define the operator  $\mathcal{U}: [0,T] \times H^1(\Omega) \to \boldsymbol{W}^{1,p}_D(\Omega)$  by

$$\mathcal{U}(t,\varphi) := A_{\varphi}^{-1}\ell(t).$$

As an immediate consequence of Lemma 1.3 one obtains the following

Corollary 1.9. If Assumption 1.5 holds true, then there exists a constant c > 0, independent of t and  $\varphi$  such that

$$\|\mathcal{U}(t,\varphi)\|_{\boldsymbol{W}_{D}^{1,p}(\Omega)} \leq c \quad \forall (t,\varphi) \in [0,T] \times H^{1}(\Omega).$$

We also observe that c is independent of  $\beta$ , which will be crucial in Chapter 2.

The constant c depends on the given data, and in particular on  $\|\ell\|_{C([0,T];\mathbf{W}^{-1,p}(\Omega))}$ . In Chapter 3 this will play an essential role, as therein the load is no longer fixed, but a variable. A special attention to this situation is given in Section 3.1, where counterpart results (of the ones in this chapter) are derived. For example, the estimate corresponding to the one in Corollary 1.9 is given by (3.5), see page 102 below. We refer here also to Section 3.2. As in this chapter and in the next one,  $\ell$  remains the whole time fixed, any dependence of the upcoming constants on the load is not of concern.

We now address the continuity properties of the solution operator  $\mathcal{U}$ . We begin with the Lipschitz continuity.

**Proposition 1.10** (Lipschitz continuity of  $\mathcal{U}$ ). Suppose that Assumption 1.5 is fulfilled and let  $r \in [2p/(p-2), \infty]$  be given. Then there exists L > 0 such that for all  $\varphi_1, \varphi_2 \in H^1(\Omega) \cap L^r(\Omega)$  and all  $t_1, t_2 \in [0, T]$  it holds

$$\|\mathcal{U}(t_1, \varphi_1) - \mathcal{U}(t_2, \varphi_2)\|_{\mathbf{W}_D^{1,\pi}(\Omega)} \le L(\|\varphi_1 - \varphi_2\|_r + |t_1 - t_2|),$$
 (1.15)

where  $1/\pi = 1/p + 1/r \in [1/p, 1/2]$ .

*Proof.* The proof follows the lines of the proof of [41, Lemma 2.5]. For simplicity, we abbreviate in what follows  $\mathbf{u}_i := \mathcal{U}(t_i, \varphi_i)$ , i = 1, 2. The idea is to write down an equation of the type (1.11) which characterizes  $\mathbf{u}_1 - \mathbf{u}_2$  and apply (1.6) therefor. This is done by subtracting the equations associated with  $\mathbf{u}_i$ , i = 1, 2, which yields

$$A_{\varphi_1}(\boldsymbol{u}_1 - \boldsymbol{u}_2) = (A_{\varphi_2} - A_{\varphi_1})\boldsymbol{u}_2 + \ell(t_1) - \ell(t_2) \quad \text{in } \boldsymbol{W}^{-1,p}(\Omega).$$
 (1.16)

Let us first notice that for given  $\mu, \rho, \tau \ge 1$  such that  $1/\mu = 1/\rho + 1/\tau$ , Hölder's inequality and Assumption 0.7 imply

$$\|\mathbb{C}\varepsilon(\boldsymbol{u}):\varepsilon(\boldsymbol{w})\|_{\mu} \leq C\|\boldsymbol{u}\|_{\boldsymbol{W}_{D}^{1,\rho}(\Omega)}\|\boldsymbol{w}\|_{\boldsymbol{W}_{D}^{1,\tau}(\Omega)} \quad \forall \, \boldsymbol{u} \in \boldsymbol{W}_{D}^{1,\rho}(\Omega), \forall \, \boldsymbol{w} \in \boldsymbol{W}_{D}^{1,\tau}(\Omega). \quad (1.17)$$

We now apply Hölder's inequality with  $1/r + 1/p + 1/\pi' = 1$  to the first term on the right-hand side in (1.16). This gives together with Lemma 5.1, (1.17), and Corollary 1.9 the following estimate

$$\|(A_{\varphi_{2}} - A_{\varphi_{1}})\boldsymbol{u}_{2}\|_{\boldsymbol{W}^{-1,\pi}(\Omega)} \leq C \|g(\varphi_{1}) - g(\varphi_{2})\|_{r} \|\boldsymbol{u}_{2}\|_{\boldsymbol{W}_{D}^{1,p}(\Omega)}$$

$$\leq C \|\varphi_{1} - \varphi_{2}\|_{r}.$$
(1.18)

Since  $0 \le 1/r \le (p-2)/(2p)$ , by assumption, it holds  $\pi \in [2, p]$ . Thus, we are allowed to apply estimate (1.6) to  $A_{\varphi_1}$  when considered as an operator from  $\boldsymbol{W}_D^{1,\pi}(\Omega)$  to  $\boldsymbol{W}^{-1,\pi}(\Omega)$ . Therewith we deduce by means of (1.16) and (1.18) the estimate

$$\|\boldsymbol{u}_1 - \boldsymbol{u}_2\|_{\boldsymbol{W}_{\mathcal{D}}^{1,\pi}(\Omega)} \le L(\|\varphi_1 - \varphi_2\|_r + |t_1 - t_2|),$$

where we used  $\ell \in C^{0,1}([0,T]; \mathbf{W}^{-1,\pi}(\Omega))$ , as a result of Assumption 1.5. Note that the constant L > 0 is independent of  $(t_i, \varphi_i)$ , i = 1, 2. This completes the proof.

We finish the discussion concerning the optimal displacement with a result which is essential for proving the existence of minimizers for (1.2).

**Lemma 1.11** (Continuity of  $\mathcal{U}$ ). Suppose that Assumption 1.5 holds. Let  $\{(t_n, \varphi_n)\} \subset [0, T] \times H^1(\Omega)$  and  $(t, \varphi) \in [0, T] \times H^1(\Omega)$  be given such that  $(t_n, \varphi_n) \to (t, \varphi)$  in  $\mathbb{R} \times L^1(\Omega)$  as  $n \to \infty$ . Then  $\mathcal{U}(t_n, \varphi_n) \to \mathcal{U}(t, \varphi)$  in  $\mathbf{W}_{D}^{1,s}(\Omega)$  as  $n \to \infty$  for every  $s \in [2, p)$ .

*Proof.* We again abbreviate  $\boldsymbol{u} := \mathcal{U}(t,\varphi)$  and  $\boldsymbol{u}_n := \mathcal{U}(t_n,\varphi_n)$ , where  $n \in \mathbb{N}$ . By subtracting the equations associated with  $\boldsymbol{u}$  and  $\boldsymbol{u}_n$  we obtain for all  $n \in \mathbb{N}$ 

$$A_{\varphi}(\boldsymbol{u} - \boldsymbol{u}_n) = (A_{\varphi_n} - A_{\varphi})\boldsymbol{u}_n + \ell(t) - \ell(t_n) \quad \text{in } \boldsymbol{W}^{-1,p}(\Omega).$$
 (1.19)

Completely analogously to (1.18), one derives for all  $n \in \mathbb{N}$  the estimate

$$\|(A_{\varphi_n} - A_{\varphi})\boldsymbol{u}_n\|_{\boldsymbol{W}^{-1,s}(\Omega)} \le C \|g(\varphi_n) - g(\varphi)\|_{\varrho} \|\boldsymbol{u}_n\|_{\boldsymbol{W}_D^{1,p}(\Omega)}, \tag{1.20}$$

with  $\varrho \in [1, \infty)$  such that  $1/\varrho + 1/p + 1/s' = 1$ . Notice that the existence of  $\varrho$  is due to  $1/s' \in [1/2, 1/p')$ . Lemma 5.2, Corollary 1.9, (1.20) and Assumption 1.5 now lead to

$$\|(A_{\varphi_n} - A_{\varphi})\boldsymbol{u}_n + \ell(t) - \ell(t_n)\|_{\boldsymbol{W}^{-1,s}(\Omega)} \to 0 \text{ as } n \to \infty.$$

In view of (1.19), applying (1.6) to  $A_{\varphi}: W_D^{1,s}(\Omega) \to W^{-1,s}(\Omega)$  then gives the assertion.

#### Nonlocal damage

We now turn our attention to the outer optimization problem on the right-hand side of (1.1), i.e., we investigate

$$\min_{\varphi \in H^1(\Omega)} \mathcal{J}(\mathcal{U}(t,\varphi),\varphi). \tag{1.21}$$

For convenience of the reader we recall that  $\mathcal{J}: V \times H^1(\Omega) \to \mathbb{R}$  is given by

$$\mathcal{J}(\boldsymbol{u},\varphi) = \frac{1}{2} \int_{\Omega} g(\varphi) \mathbb{C}\varepsilon(\boldsymbol{u}) : \varepsilon(\boldsymbol{u}) \ dx - \langle \ell(t), \boldsymbol{u} \rangle_{V} + \frac{\alpha}{2} \|\nabla \varphi\|_{2}^{2} + \frac{\beta}{2} \|\varphi - d\|_{2}^{2}.$$

Note that  $\varphi \mapsto \mathcal{J}(\mathcal{U}(t,\varphi),\varphi)$  is not necessarily convex, and thus, we cannot directly derive a characterization of a (possible existent) solution of (1.21). As we will see, even deriving necessary optimality conditions is without further ado not possible. The next result covers only the solvability of (1.21). As already mentioned at the beginning of this section, the uniqueness can be first deduced after writing down the necessary optimality conditions.

**Proposition 1.12** (Existence of the optimal nonlocal damage). Suppose that Assumption 1.5 holds. Then, the optimization problem (1.21) admits at least one solution, and therefore (1.2) possesses a solution as well.

*Proof.* By means of Definitions 1.2 and 1.8, the objective in (1.21) can be rewritten as

$$f: H^1(\Omega) \to \mathbb{R}, \quad f(\varphi) := \mathcal{J}(\mathcal{U}(t,\varphi),\varphi) = -\frac{1}{2} \langle \ell(t), \mathcal{U}(t,\varphi) \rangle + \frac{\alpha}{2} \|\nabla \varphi\|_2^2 + \frac{\beta}{2} \|\varphi - d\|_2^2.$$

The existence of solutions for (1.21) follows by classical arguments of the direct method of variational calculus. To this end, notice that f is radially unbounded because of Corollary 1.9. Moreover, it is weakly lower semicontinuous. To see this, consider a sequence  $\{\varphi_n\} \subset H^1(\Omega)$  with  $\varphi_n \rightharpoonup \varphi$  in  $H^1(\Omega)$  as  $n \to \infty$ . The compact embedding  $H^1(\Omega) \hookrightarrow \hookrightarrow L^1(\Omega)$  and Lemma 1.11 then imply

$$\mathcal{U}(t,\varphi_n) \to \mathcal{U}(t,\varphi) \text{ in } V \text{ as } n \to \infty.$$

This together with the weak lower semicontinuity of  $\|\nabla\cdot\|_2^2$  and  $\|\cdot\|_2^2$  on  $H^1(\Omega)$  gives that f is indeed weakly lower semicontinuous. Since  $H^1(\Omega)$  is a reflexive Banach space, a standard argument yields now that (1.21) admits solutions. In consequence, so does (1.2), as the set of solutions for (1.2) is given by  $\{(\mathcal{U}(t,\varphi),\varphi): \varphi \text{ solves } (1.21)\}$ , which is straight forward to see.

Remark 1.13. We point out that the continuity of the operator  $\mathcal{U}(t,\cdot)$  in Lemma 1.11 is crucial for proving the existence of solutions for (1.21). One should however notice that, if we assume that the optimal displacement has only V-regularity, then the result in Lemma 1.11 holds true for any  $s \in [p',2)$ , see the proof thereof and Remark 1.4. Since by assumption  $\ell(t) \in \mathbf{W}^{-1,p}(\Omega) \hookrightarrow (\mathbf{W}_D^{1,s}(\Omega))^*$ , Proposition 1.12 can still be concluded in the exact same way. Note that this alternative proof does not (directly) use the improved regularity of the optimal displacement. However, it uses Lemma 1.3 (in form of Remark 1.4) and Assumption 1.5, which were the tools needed for establishing the improved regularity in Proposition 1.7. Hence, these are the essential ingredients for proving the existence of the optimal nonlocal damage.

As the upcoming analysis shows, the  $\mathbf{W}_{D}^{1,p}(\Omega)$ -regularity of the optimal displacement is also crucial for deriving necessary optimality conditions, and thus, for proving the uniqueness of solutions for (1.21).

Next we concentrate on deriving necessary optimality conditions for the optimal nonlocal damage. For this purpose one has to differentiate the function  $\mathcal{J}$  w.r.t.  $\varphi$  at  $(\mathcal{U}(t,\bar{\varphi}),\bar{\varphi})$ , where  $\bar{\varphi}$  is a solution of (1.21). This can be done in view of Lemma 1.6 and Definition 1.8 only under the following additional

**Assumption 1.14.** From now on we assume that, in case of N=3, the assertion in Lemma 1.3 holds with p>12/5.

The existence of p in Assumption 1.14 is ensured by further conditions on the data, which are stated in Remark 1.27 below. We emphasize that one can go without this additional assumption, if one replaces the  $H^1$ -seminorm in the energy functional in Definition 0.2 by the  $H^{3/2}$ -seminorm, see Remark 1.28 below for more details.

The following definition will be useful in the sequel:

**Definition 1.15** (The linear and nonlinear part of (1.25b)). Suppose that Assumptions 1.5 and 1.14 are fulfilled. Then we define the mappings  $B: H^1(\Omega) \to H^1(\Omega)^*$  and  $F: [0,T] \times H^1(\Omega) \to H^1(\Omega)^*$  by

$$\langle B\varphi, \psi \rangle_{H^1(\Omega)} := \int_{\Omega} \alpha \nabla \varphi \cdot \nabla \psi + \beta \varphi \psi \, dx \quad \forall \, \psi \in H^1(\Omega), \tag{1.22}$$

$$\langle F(t,\varphi),\psi\rangle_{H^1(\Omega)} := \frac{1}{2} \int_{\Omega} g'(\varphi) \mathbb{C}\varepsilon(\mathcal{U}(t,\varphi)) : \varepsilon(\mathcal{U}(t,\varphi))\psi \ dx \quad \forall \, \psi \in H^1(\Omega).$$
 (1.23)

We emphasize that F is well defined. To see this, first note that the Sobolev embedding  $H^1(\Omega) \hookrightarrow L^{p/(p-2)}(\Omega)$  holds true, since p > 2 for N = 2 and since, thanks to Assumption 1.14, p > 12/5 in case of N = 3. Then, by Assumptions 0.6 and 0.7, and in view of Definition 1.8, applying Hölder's inequality with 1/(p/2) + 1/(p/(p-2)) = 1 yields that F has range in  $L^{p/(p-2)}(\Omega)^*$  and thus, in  $H^1(\Omega)^*$ . On account of  $1 \le p/(p-2) < \infty$ , we can identify  $F(t,\varphi)$  with the term  $1/2 g'(\varphi) \mathbb{C} \varepsilon(\mathcal{U}(t,\varphi)) : \varepsilon(\mathcal{U}(t,\varphi)) \in L^{p/2}(\Omega)$  for all  $(t,\varphi) \in [0,T] \times H^1(\Omega)$ .

With a little abuse of notation, the operators B and F considered with different domains and ranges will be denoted by the same symbol.

Now we can state the necessary optimality conditions as follows:

**Proposition 1.16.** Under Assumptions 1.5 and 1.14, every local minimizer  $(\bar{\boldsymbol{u}}, \bar{\varphi})$  of (1.2) fulfills  $\bar{\boldsymbol{u}} = \mathcal{U}(t, \bar{\varphi}) \in \boldsymbol{W}_D^{1,p}(\Omega)$  and

$$B\bar{\varphi} + F(t,\bar{\varphi}) = \beta d \quad in \ H^1(\Omega)^*,$$
 (1.24)

which is equivalent to the following optimality system:

$$-\operatorname{div} g(\bar{\varphi})\mathbb{C}\varepsilon(\bar{\boldsymbol{u}}) = \ell(t) \quad \text{in } \boldsymbol{W}^{-1,p}(\Omega), \tag{1.25a}$$

$$-\alpha \Delta \bar{\varphi} + \beta \bar{\varphi} + \frac{1}{2} g'(\bar{\varphi}) \mathbb{C} \varepsilon(\bar{\boldsymbol{u}}) : \varepsilon(\bar{\boldsymbol{u}}) = \beta d \quad \text{in } H^1(\Omega)^*.$$
 (1.25b)

*Proof.* The local optimality of  $(\bar{u}, \bar{\varphi})$  in particular implies that  $\bar{u}$  is a local minimizer of

$$\min_{\boldsymbol{u}\in V}\mathcal{J}(\boldsymbol{u},\bar{\varphi}),$$

which is a convex problem according to Proposition 1.7. Therefore,  $\bar{u}$  is a global minimizer of this problem, and Proposition 1.7 yields  $\bar{u} = \mathcal{U}(t, \bar{\varphi})$ .

Similarly, the local optimality of  $(\bar{\boldsymbol{u}}, \bar{\varphi})$  also implies that  $\bar{\varphi}$  is a local minimizer of

$$\min_{\varphi \in H^1(\Omega)} \mathcal{J}(\bar{\boldsymbol{u}}, \varphi). \tag{1.26}$$

Thanks to the improved regularity of  $\bar{\boldsymbol{u}}$  by Proposition 1.7 and thanks to Assumption 1.14, one can differentiate  $\mathcal{J}$  w.r.t.  $\varphi$  at  $(\bar{\boldsymbol{u}}, \bar{\varphi})$  by means of Lemma 1.6. Thus, we can write  $\partial_{\varphi} \mathcal{J}(\bar{\boldsymbol{u}}, \bar{\varphi}) = 0$  as necessary optimality condition for a local minimizer of (1.26). In view of (1.10), Definition 1.15 and  $\bar{\boldsymbol{u}} = \mathcal{U}(t, \bar{\varphi})$ , this is equivalent to (1.24). The equivalence to (1.25) follows directly from the definitions of  $A_{\bar{\varphi}}$ , B, and F.

From Propositions 1.12 and 1.16 we know that (1.24) has at least one solution. In the following we aim for showing that this solution is unique, which will give in turn the unique solvability of (1.2). Unfortunately, Assumption 1.14 does not suffice to prove the uniqueness of solutions to (1.25). In order to show strong monotonicity of the operator on the left-hand side of (1.24), we additionally need that  $H^1(\Omega) \hookrightarrow L^r(\Omega)$  with r > 2p/(p-2), see proof of Lemma 1.20 below for more details. This motivates the first part of the following

#### **Assumption 1.17.** From now on we require:

- 1. The assertion of Lemma 1.3 holds for some p > N.
- 2. The penalization parameter β is sufficiently large, depending only on the given data, see (1.36), (1.102), (2.38) and (3.17) on page 106 below, as well as Remark 1.22 and Remark 3.9, p. 107, for Chapters 1, 2 and for Chapter 3, respectively.

Note that Assumption 1.17.1 is automatically fulfilled if N=2, see Lemma 1.3. For N=3 this is guaranteed by imposing additional conditions on the data, see Remark 1.27 below for more details. Moreover, as in case of Assumption 1.14, one does not need Assumption 1.17.1, if one replaces the  $H^1$ -seminorm in the energy functional by the  $H^{3/2}$ -seminorm, see Remark 1.28 below.

Assumption 1.17.2 is not restrictive at all, since  $\beta$  is a penalization parameter, which is supposed to be large anyway and will be send to  $\infty$  in the next chapter. Of course, the dependency of  $\beta$  on the given data, and in particular on the load, cf. Remark 1.22, does not affect the analysis in this chapter or in Chapter 2, while in Chapter 3, the dependency on the load asks for a new threshold, namely (3.17) on page 106 below, see Remark 1.22 for more details.

We can now start the discussion of the uniqueness of the optimal nonlocal damage. The main idea is to show the strong monotonicity of the operator on the left-hand side in (1.24), by making use of the fact that  $\beta$  is large, which allows us to absorb (possible bounded) terms. We begin with the Lipschitz continuity of the mapping F. For later purposes, we prove a slightly more general result.

**Lemma 1.18.** Let  $r \geq 2p/(p-2)$  and define s via 1/s + 2/p + 1/r = 1. Under Assumptions 1.5 and 1.14 we have for all  $t_1, t_2 \in [0, T]$ ,  $\varphi_1, \varphi_2 \in H^1(\Omega) \cap L^r(\Omega)$  and  $\psi \in L^s(\Omega)$  the following estimate

$$|\langle F(t_1, \varphi_1) - F(t_2, \varphi_2), \psi \rangle| \le C (\|\varphi_1 - \varphi_2\|_r + |t_1 - t_2|) \|\psi\|_s$$

with a constant C > 0 depending only on the given data.

*Proof.* We again abbreviate  $u_i := \mathcal{U}(t_i, \varphi_i)$  for i = 1, 2. First thing to notice is that  $s \in [p/(p-2), 2p/(p-2)]$ , by definition. Hence  $\psi \in L^s(\Omega) \hookrightarrow L^{p/(p-2)}(\Omega)$ . Now, the

definition of F in (1.23) implies

$$|\langle F(t_{1}, \varphi_{1}) - F(t_{2}, \varphi_{2}), \psi \rangle|$$

$$\leq \int_{\Omega} |(g'(\varphi_{1}) - g'(\varphi_{2}))\mathbb{C}\varepsilon(\boldsymbol{u}_{1}) : \varepsilon(\boldsymbol{u}_{1})\psi| dx$$

$$+ \int_{\Omega} |g'(\varphi_{2})[\mathbb{C}\varepsilon(\boldsymbol{u}_{1}) : \varepsilon(\boldsymbol{u}_{1}) - \mathbb{C}\varepsilon(\boldsymbol{u}_{2}) : \varepsilon(\boldsymbol{u}_{2})]\psi| dx.$$

$$(1.27)$$

We discuss the two terms on the right-hand side of (1.27) separately:

(i) In view of (1.17) and Corollary 1.9 we have

$$\|\mathbb{C}\varepsilon(\boldsymbol{u}_1):\varepsilon(\boldsymbol{u}_1)\|_{\frac{p}{2}} \le c,$$
 (1.28)

where c > 0 is a constant independent of  $(t_1, \varphi_1)$ . In addition, the function  $g' : L^r(\Omega) \to L^r(\Omega)$  is Lipschitz continuous according to Lemma 5.1. Thus, applying Hölder's inequality with 1/r + 2/p + 1/s = 1 for the first term on the right-hand side in (1.27) gives

$$\int_{\Omega} \left| (g'(\varphi_1) - g'(\varphi_2)) \mathbb{C}\varepsilon(\boldsymbol{u}_1) : \varepsilon(\boldsymbol{u}_1)\psi \right| dx \le C_1 \|\varphi_1 - \varphi_2\|_r \|\psi\|_s, \tag{1.29}$$

where  $C_1 > 0$  depends only on the given data.

(ii) Define  $\pi$  and  $\omega$  through  $1/\pi = 1/p + 1/r$  and  $1/\omega = 1/p + 1/\pi$ , respectively. Then (1.17), Corollary 1.9, and Proposition 1.10 result in

$$\|\mathbb{C}\varepsilon(\boldsymbol{u}_{1}) : \varepsilon(\boldsymbol{u}_{1}) - \mathbb{C}\varepsilon(\boldsymbol{u}_{2}) : \varepsilon(\boldsymbol{u}_{2})\|_{\omega} = \|\mathbb{C}[\varepsilon(\boldsymbol{u}_{1}) + \varepsilon(\boldsymbol{u}_{2})] : [\varepsilon(\boldsymbol{u}_{1}) - \varepsilon(\boldsymbol{u}_{2})]\|_{\omega}$$

$$\leq C_{2}\|\boldsymbol{u}_{1} + \boldsymbol{u}_{2}\|_{\boldsymbol{W}_{D}^{1,p}(\Omega)}\|\boldsymbol{u}_{1} - \boldsymbol{u}_{2}\|_{\boldsymbol{W}_{D}^{1,\pi}(\Omega)} \quad (1.30)$$

$$\leq C_{2}(\|\varphi_{1} - \varphi_{2}\|_{r} + |t_{1} - t_{2}|).$$

Now, in light of Assumption 0.6, Hölder's inequality with  $1/\omega + 1/s = 1$  yields

$$\int_{\Omega} |g'(\varphi_2)[\mathbb{C}\varepsilon(\boldsymbol{u}_1):\varepsilon(\boldsymbol{u}_1) - \mathbb{C}\varepsilon(\boldsymbol{u}_2):\varepsilon(\boldsymbol{u}_2)]\psi| dx 
\leq C_2(\|\varphi_1 - \varphi_2\|_r + |t_1 - t_2|)\|\psi\|_s.$$
(1.31)

Notice that  $C_2 > 0$  depends only on the given data. Inserting (1.29) and (1.31) in (1.27) finally gives the assertion.

We observe that, if p > N, then

$$r := \frac{2p}{p-2} \in \left(2, \frac{2N}{N-2}\right),\tag{1.32}$$

so that Sobolev embeddings give in turn  $H^1(\Omega) \hookrightarrow L^r(\Omega)$ . Since by construction, r satisfies 2/r + 2/p = 1, Lemma 1.18 is applicable with s = r, which results in the following

Corollary 1.19. Under Assumptions 1.5 and 1.17.1 it holds

$$|\langle F(t_1, \varphi_1) - F(t_2, \varphi_2), \psi \rangle| \le C (\|\varphi_1 - \varphi_2\|_{\frac{2p}{p-2}} + |t_1 - t_2|) \|\psi\|_{\frac{2p}{p-2}}$$
  
$$\forall t_1, t_2 \in [0, T], \ \forall \varphi_1, \varphi_2, \psi \in H^1(\Omega),$$

where C > 0 depends only on the given data.

The following lemma is an essential tool for proving the unique solvability of (1.2).

**Lemma 1.20.** Let m > 0 be given. Then, under Assumption 1.17.1, it holds

$$m\|\varphi\|_{\frac{2p}{p-2}}^2 \leq k \|\varphi\|_2^2 + \widetilde{c}(k) \|\varphi\|_{H^1(\Omega)}^2 \quad \forall \, \varphi \in H^1(\Omega) \ and \ \forall \, k > 0,$$

where  $\widetilde{c}: \mathbb{R}^+ \to \mathbb{R}^+$  is a function depending on m, p and N, which satisfies  $\widetilde{c}(k) \searrow 0$  as  $k \nearrow \infty$ .

*Proof.* For convenience we again set r:=2p/(p-2). First note that, because of Assumption 1.17.1, there is an index  $\varrho$  such that  $r\in(2,\varrho)$  and  $H^1(\Omega)\hookrightarrow L^\varrho(\Omega)$ . For instance, take  $\varrho=(2p+1)/(p-2)$  for N=2 and  $\varrho=6$  if N=3, see also (1.32). Therefore, there exists  $\theta\in(0,1)$  such that  $1/r=\theta/2+(1-\theta)/\varrho$ , whence by Lyapunov's inequality we have

$$m\|\varphi\|_{r}^{2} \leq m\|\varphi\|_{2}^{2\theta}\|\varphi\|_{\varrho}^{2-2\theta} \leq mC\|\varphi\|_{2}^{2\theta}\|\varphi\|_{H^{1}(\Omega)}^{2-2\theta} \quad \forall \varphi \in H^{1}(\Omega).$$
 (1.33)

Note that C is the embedding constant, and thus, it depends on  $\varrho$  and N, or to be more precise, on p and N. Thanks to the generalized Young inequality, (1.33) can be continued as

$$m\|\varphi\|_r^2 \le k \|\varphi\|_2^2 + \frac{(vk)^{1-w}m^wC^w}{w} \|\varphi\|_{H^1(\Omega)}^2 \quad \forall k > 0, \ \forall \varphi \in H^1(\Omega),$$

where  $v := 1/\theta$  and  $w := 1/(1-\theta)$ . We remark that v and w depend on p and N, as well. Note that the mapping

$$\widetilde{c}: \mathbb{R}^+ \ni k \mapsto \frac{(vk)^{1-w} m^w C^w}{w} \in \mathbb{R}^+$$

satisfies  $\widetilde{c}(k) \searrow 0$  as  $k \nearrow \infty$ , since  $k \mapsto k^{1-w}$  does so, in view of w > 1. The proof is now complete.

We are now finally in the position to state the 'strong monotonicity' of B + F, from which the uniqueness of the optimal nonlocal damage will follow easily.

**Lemma 1.21.** Let Assumptions 1.5 and 1.17 be satisfied. Then, for all  $t_1, t_2 \in [0, T]$  and all  $\varphi_1, \varphi_2 \in H^1(\Omega)$ ,  $\varphi_1 \neq \varphi_2$ , it holds

$$\frac{\langle B(\varphi_1 - \varphi_2) + F(t_1, \varphi_1) - F(t_2, \varphi_2), \varphi_1 - \varphi_2 \rangle_{H^1(\Omega)}}{\|\varphi_1 - \varphi_2\|_{H^1(\Omega)}} \ge \alpha/2\|\varphi_1 - \varphi_2\|_{H^1(\Omega)} - C|t_1 - t_2|,$$

where C > 0 depends only on the given data.

*Proof.* Let  $(t_i, \varphi_i)_{i=1,2} \in [0, T] \times H^1(\Omega)$  be arbitrary, but fixed with  $\varphi_1 \neq \varphi_2$ . Then, Corollary 1.19 and Lemma 1.20 yield

$$|\langle F(t_1, \varphi_1) - F(t_2, \varphi_2), \varphi_1 - \varphi_2 \rangle_{H^1(\Omega)}| \le k \|\varphi_1 - \varphi_2\|_2^2 + \widetilde{c}(k) \|\varphi_1 - \varphi_2\|_{H^1(\Omega)}^2 + C |t_1 - t_2| \|\varphi_1 - \varphi_2\|_{H^1(\Omega)} \quad \forall k > 0,$$
(1.34)

with  $\tilde{c}$  as in Lemma 1.20. Note that we also employed that  $H^1(\Omega) \hookrightarrow L^{2p/(p-2)}(\Omega)$ , in view of Assumption 1.17.1. Using the definition of B in (1.22) we further infer from (1.34) that for all k > 0 it holds

$$\frac{\langle B(\varphi_{1} - \varphi_{2}) + F(t_{1}, \varphi_{1}) - F(t_{2}, \varphi_{2}), \varphi_{1} - \varphi_{2} \rangle}{\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)}} \ge (\alpha - \widetilde{c}(k)) \|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)} - C|t_{1} - t_{2}| + (\beta - \alpha - k) \frac{\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)}}{\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)}}.$$
(1.35)

Keeping in mind the properties of  $\widetilde{c}$ , we can choose k > 0 (large enough) such that  $\alpha - \widetilde{c}(k) \ge \alpha/2$ . Furthermore, if

$$\beta \ge \alpha + k,\tag{1.36}$$

cf. Assumption 1.17.2, then (1.35) gives the assertion.

Remark 1.22. Note that in the above proof, Lemma 1.20 is applied with m = C, where C > 0 is the constant from Corollary 1.19. An inspection of the proof of Lemma 1.18 shows that this depends on the (supremum) norm of the load, see e.g. (1.28) and Corollary 1.9. Hence, the function  $\tilde{c}$  depends, cf. Lemma 1.20 and cf. Remark 1.27 below, only on the given data, and in particular on the load, which implies in view of (1.36) that  $\beta$  does so as well.

This becomes a central concern in Chapter 3, where the load is no longer fixed, but a variable. Roughly speaking, this is the reason why in Section 3.1 below, the (time-independent) variable loads have to be uniformly bounded to guarantee unique solvability of the time-independent version of the minimization problem in (P). The threshold (1.36) is then replaced by (3.17) (see page 106 below), where M is the (given) constant which uniformly bounds the variable loads. We refer here also to Section 3.2 below, where one deals with the same problem in the time-dependent case.

The unique solvability of the minimization problem in (P) is covered by the following

**Theorem 1.23** (Unique solvability of (1.2)). Under Assumptions 1.5 and 1.17, the optimization problem (1.2) admits a unique solution, which is characterized by (1.25). Consequently, the unique optimal nonlocal damage is characterized by (1.24).

*Proof.* We focus on showing a Lipschitz estimate for solutions of (1.24), as this will be also needed for later purposes. Then, the desired assertion follows immediately. To this end, let  $\varphi_i$  denote solutions of (1.24) associated with given  $(t_i, d_i) \in [0, T] \times L^2(\Omega)$ , i = 1, 2. Note that the existence thereof is ensured by Propositions 1.12 and 1.16. By

assuming  $\varphi_1 \neq \varphi_2$ , we obtain from Lemma 1.21 and Cauchy-Schwarz inequality the estimate

$$\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)} \leq \frac{2}{\alpha} \left( \frac{\langle B(\varphi_{1} - \varphi_{2}) + F(t_{1}, \varphi_{1}) - F(t_{2}, \varphi_{2}), \varphi_{1} - \varphi_{2} \rangle_{H^{1}(\Omega)}}{\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)}} + C|t_{1} - t_{2}| \right)$$

$$= \frac{2}{\alpha} \left( \beta \frac{(d_{1} - d_{2}, \varphi_{1} - \varphi_{2})_{2}}{\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)}} + C|t_{1} - t_{2}| \right)$$

$$\leq C \left( \|d_{1} - d_{2}\|_{2} + |t_{1} - t_{2}| \right).$$

$$(1.37)$$

Note that the estimate (1.37) holds trivially also for  $\varphi_1 = \varphi_2$ . If we set  $t_1 = t_2$  and  $d_1 = d_2$ , then (1.37) implies uniqueness for (1.24), and thus the unique solvability of (1.25), cf. Proposition 1.16. As this system constitutes the necessary optimality condition for (1.2), which admits global solutions by Proposition 1.12, we deduce that (1.2) is uniquely solvable too, and that every local minimizer must be the global minimizer. This completes the proof.

The unique solvability of (1.24) leads to

**Definition 1.24** (Solution operator of (1.24)). Let Assumptions 1.5 and 1.17 be fulfilled. We define the operator  $\Phi : [0,T] \times L^2(\Omega) \to H^1(\Omega)$  as

$$\Phi(t,d) := (B + F(t,\cdot))^{-1}(\beta d).$$

As a result of (1.37) we have the following

Corollary 1.25 (Lipschitz continuity of  $\Phi$ ). Under Assumptions 1.5 and 1.17, there exists a constant L > 0 such that

$$\|\Phi(t_1, d_1) - \Phi(t_2, d_2)\|_{H^1(\Omega)} \le L(\|d_1 - d_2\|_2 + |t_1 - t_2|)$$
(1.38)

holds true for all  $t_1, t_2 \in [0, T]$  and all  $d_1, d_2 \in L^2(\Omega)$ .

To summarize our results so far, we have proven that, under Assumptions 1.5 and 1.17, the optimization problem

$$\min_{(\boldsymbol{u},\varphi)\in V\times H^1(\Omega)} \mathcal{E}(t,\boldsymbol{u},\varphi,d)$$

is uniquely solvable for any  $(t,d) \in [0,T] \times L^2(\Omega)$ . The optimal displacement is given by  $\mathcal{U}(t,\bar{\varphi}) \in \mathbf{W}_D^{1,p}(\Omega)$ , where  $\bar{\varphi} = \Phi(t,d) \in H^1(\Omega)$  is the optimal nonlocal damage.

We conclude this subsection with some remarks concerning the assumptions we made on the load and on p, respectively.

Remark 1.26. We point out that it was necessary in Assumption 1.5 to impose that  $\ell$  has range in  $\mathbf{W}^{-1,p}(\Omega)$ , in order to obtain the improved regularity of the optimal displacement, or better said, maximal regularity therefor, see Definition 1.8 and Lemma 1.3. Since the threshold for  $\beta$  depends on  $\|\ell\|_{C([0,T];\mathbf{W}^{-1,p}(\Omega))}$ , cf. Remark 1.22, it makes sense to work with a space for the load so that the 'maximal value' of  $\ell$  is independent of the variable t (at least almost everywhere). For example, in Section 3.2 below one finds  $L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega))$  as a proper (more general) choice for the space for the load. As we are interested in solution operators which are smooth w.r.t. time, and later even continuously differentiable, we require in Assumption 1.5 that the load is (for the beginning) Lipschitz continuous.

Remark 1.27. We emphasize that by the results of [29], the maximal value of p depends only on the domain, the partition of its boundary, and the boundedness and monotonicity constants of the stress-strain relation, which in our case is depicted via (1.7). Roughly speaking, a sufficiently smooth domain, no mixed boundary conditions and a smaller difference between the above mentioned constants account for larger values for p. We refer here to the proof of Lemma 1.1 in [29]. Thus, from the proofs of Lemmata 1.1 and 1.3 we infer that p depends on  $\epsilon \gamma_{\mathbb{C}}$ ,  $\|\mathbb{C}\|_{\infty}$ , and the domain. We moreover infer that the existence of a p fulfilling Assumption 1.17.1 is ensured provided that the values  $\underline{m} = \epsilon \gamma_{\mathbb{C}}$  and  $\overline{m} = \|\mathbb{C}\|_{\infty}$  are close enough to each other and if the domain is smooth enough  $(C^1$ -boundary) with  $\Gamma_D$  as entire boundary.

Recall that, in the two-dimensional case, Assumption 1.17.1 is automatically fulfilled.

Remark 1.28. Alternatively to Assumption 1.17.1 one can proceed as in [41] and use in case of N=3 the Sobolev-Slobodeckij space  $H^{3/2}(\Omega)$  for the nonlocal damage. To this end, one replaces the gradient term in the energy functional by the seminorm generated by [41, (2.4b)] so that the radial unboundedness in the proof of Proposition 1.12 is still ensured, and thus, the existence of the optimal nonlocal damage. Note that the rest of the proof remains unaffected, as  $H^{3/2}(\Omega)$  (endowed with the same norm as in [41]) is a reflexive Banach space.

The advantage thereof is that the embedding  $H^{3/2}(\Omega) \hookrightarrow L^r(\Omega)$  holds for every  $r \in [1,\infty)$  in the three-dimensional case, which means that there is no longer need for making extra assumptions on p if N=3. This is shown by a closer inspection of the preceding analysis, which reveals that the embedding  $H^1(\Omega) \hookrightarrow L^r(\Omega)$  for all  $r \in [1,\infty)$  in case of N=2 is the key ingredient to prove unique solvability for (1.2) without any additional assumptions on the integrability exponent p. Thus, working with  $H^{3/2}(\Omega)$  instead of  $H^1(\Omega)$  in three dimensions allows to do the same in case of N=3, so that no extra assumptions on p are required. However, we chose not to work with  $H^{3/2}(\Omega)$ , as the bilinear form associated with the  $H^{3/2}(\Omega)$ -seminorm is difficult to realize in numerical computations.

#### 1.1.2 The evolution equation

This subsection is devoted to prove existence and uniqueness for the complete damage model (P). We suppose that Assumptions 1.5 and 1.17 hold true in what follows. Then,

with the results from Subsection 1.1.1 at hand, the problem (P) can be reformulated as

$$-\partial_d \mathcal{E}(t, \boldsymbol{u}(t), \varphi(t), d(t)) \in \partial \mathcal{R}_{\delta}(\dot{d}(t)) \quad \text{f.a.a. } t \in (0, T), \quad d(0) = d_0, \tag{1.39}$$

where  $\mathbf{u} = \mathcal{U}(\cdot, \varphi(\cdot))$  and  $\varphi = \Phi(\cdot, d(\cdot))$ , while  $d : [0, T] \to L^2(\Omega)$  is the local damage. Moreover, due to (1.9), the evolutionary equation (1.39) reads

$$-\beta(d(t) - \varphi(t)) \in \partial \mathcal{R}_{\delta}(\dot{d}(t)) \quad \text{f.a.a. } t \in (0, T), \quad d(0) = d_0. \tag{1.40}$$

We approach the reduced problem (P) by showing that (1.40) is equivalent to the following operator differential equation, which can then be solved by standard arguments.

**Theorem 1.29** (Operator differential equation). Let Assumptions 1.5 and 1.17 hold true. Then, the evolution equation (1.40) is equivalent to

$$\dot{d}(t) = \frac{1}{\delta} \max(-\beta(d(t) - \varphi(t)) - r) \quad \text{f.a.a.} \ t \in (0, T), \quad d(0) = d_0, \tag{1.41}$$

where  $\varphi = \Phi(\cdot, d(\cdot))$ .

*Proof.* First we note that  $\mathcal{R}_{\delta}$  is the sum of two convex functionals, namely  $\mathcal{R}_{1}$  and  $\frac{\delta}{2} \| \cdot \|_{2}^{2}$ . As the latter one is Fréchet-differentiable, we can apply the sum rule for convex subdifferentials, so that

$$\partial \mathcal{R}_{\delta}(\eta) = \partial \mathcal{R}_{1}(\eta) + \delta \eta$$
 for all  $\eta \in L^{2}(\Omega)$ .

Further, since  $\mathcal{R}_1$  is positively homogeneous, we have for all  $\xi, \eta \in L^2(\Omega)$  with  $\eta \geq 0$  the equivalence

$$\xi \in \partial \mathcal{R}_1(\eta) \quad \Longleftrightarrow \quad \begin{cases} (\xi, \eta)_2 = \mathcal{R}_1(\eta), \\ (\xi, v)_2 \le \mathcal{R}_1(v) \quad \forall \, v \in L^2(\Omega). \end{cases}$$
 (1.42)

This can be easily seen by writing down the definition of the subdifferential and testing therein with 0 and  $2\eta$ , respectively.

In view of the above, we can now rewrite the evolution in (1.40) as

$$(-\beta(d(t) - \varphi(t)) - \delta \dot{d}(t), \dot{d}(t))_2 = \mathcal{R}_1(\dot{d}(t)), \tag{1.43a}$$

$$(-\beta(d(t) - \varphi(t)) - \delta \dot{d}(t), v)_2 \le \mathcal{R}_1(v) \quad \forall v \in L^2(\Omega)$$
(1.43b)

for almost all  $t \in (0,T)$ . Note that from (1.40) we know that  $\dot{d}(t) \in \text{dom}(\mathcal{R}_1)$ , i.e.,  $\dot{d}(t) \geq 0$  f.a.a.  $t \in (0,T)$ . As a result of Definition 0.1 and fundamental lemma of the calculus of variations we further obtain

$$(1.43b) \iff -\beta(d(t) - \varphi(t)) - \delta \dot{d}(t) - r \le 0 \text{ a.e. in } \Omega, \text{ f.a.a. } t \in (0, T).$$

By employing again Definition 0.1, we also get

$$(1.43a) \quad \Longleftrightarrow \quad \big(\underbrace{-\beta(d(t)-\varphi(t))-\delta\dot{d}(t)-r}_{\leq 0},\underbrace{\dot{d}(t)}_{\geq 0}\big)_2 = 0 \quad \text{f.a.a. } t \in (0,T).$$

Therefore, the system (1.43) and thus, the evolution in (1.40), is equivalent to the following complementarity system

$$0 \le \delta \dot{d}(t) \perp -\beta(d(t) - \varphi(t)) - r - \delta \dot{d}(t) \le 0 \text{ a.e. in } \Omega, \text{ f.a.a. } t \in (0, T), \tag{1.44}$$

where we used  $\delta > 0$  for the left inequality. Since the max-function is a well known complementarity function, (1.44) gives the assertion.

Remark 1.30. There are multiple equivalent formulations of the penalized damage model in reduced form. An overview thereof is given in Proposition 2.7 on page 59 below. The operator differential equation (1.41) is certainly one of the most significant, as it has various advantages. Firstly, it may provide a useful starting point for a numerical solution of (P), as one could solve (the discretized version of) (1.41) by means of a semi-smooth Newton algorithm, see also  $(P_k^{\beta,\tau})$  on page 82 below. Moreover, it facilitates deriving necessary optimality conditions in Chapter 3, due to the directional differentiability of the operator max, cf. Lemma 5.6. Besides, by describing the evolution of the local damage via (1.41), the unique solvability of (1.39), and thus of (P), follows easily by Picard-Lindelöf's theorem, as we will next see. Furthermore, (1.41) will be the starting point for performing a time-discretization for (P) in Subsection 2.4 below, which will lead to finding an  $L^{\infty}$ -bound for the local damage. However, (1.41) does not help us when it comes to passing to the limit  $\beta \to \infty$  in Chapter 2, as explained at the beginning of Section 2.1 below. For this reason, one derives therein a new equivalent formulation for (1.39), i.e., the energy identity, by means of which the passage to the limit is possible.

Remark 1.31. The result in Theorem 1.29 can be as well proven by employing the dual formulation of (1.40), see (2.12) (a.e. in (0,T)), p. 59 below, and by showing that the conjugate functional  $\mathcal{R}^*_{\delta}$  equals the  $\delta$ -Moreau-Yosida regularization of the indicator functional associated to the set  $\{v \in L^2(\Omega) : v \leq r\}$ . According to [28, Lemma 4.1(a)], the latter one is Fréchet-differentiable. Moreover, its derivative can be expressed by means of the max-function, so that (2.12) (a.e. in (0,T)), p. 59, becomes (1.41).

One could also prove Theorem 1.29 by rewriting (1.40) as a minimization problem, see (2.14), p. 59 below. As this is convex,  $\dot{d}(t)$  can be described a.e. in (0,T) by the necessary and sufficient optimality condition, which is nothing else as the complementarity system (1.44), i.e., (1.41).

The unique solvability of the problem (P) in reduced form is covered by the following

**Theorem 1.32** (Existence and uniqueness for the evolution equation). Under Assumptions 1.5 and 1.17 there exists a unique function  $d \in C^{1,1}([0,T];L^2(\Omega))$  satisfying (1.39), and thus, (1.40) and (1.41), for all  $t \in [0,T]$ .

*Proof.* Theorem 1.29 tells us that (1.39) and (1.40) are equivalent to the operator differential equation given by (1.41). We intend to solve the latter one by means of Picard-Lindelöf's theorem. For this purpose, we define the mapping  $f:[0,T]\times L^2(\Omega)\to L^2(\Omega)$  as

$$f(t,d) := \frac{1}{\delta} \max(-\beta(d - \Phi(t,d)) - r). \tag{1.45}$$

Of course, f is well defined in view of Lemma 5.6.(i). Due to the Lipschitz continuity of max :  $L^2(\Omega) \to L^2(\Omega)$ , cf. Lemma 5.6.(i), and (1.38), it holds for all  $(t_1, d_1), (t_2, d_2) \in [0, T] \times L^2(\Omega)$ 

$$||f(t_1, d_1) - f(t_2, d_2)||_2 \le \frac{\beta}{\delta} (||\Phi(t_1, d_1) - \Phi(t_2, d_2)||_{H^1(\Omega)} + ||d_1 - d_2||_2)$$

$$\le \frac{\beta}{\delta} (L+1) ||d_1 - d_2||_2 + \frac{\beta}{\delta} L |t_1 - t_2|,$$
(1.46)

where L is the Lipschitz constant of  $\Phi$ . Therefore, f is globally Lipschitz continuous, and we can conclude with [14, Theorem 7.2.6] that there exists a unique  $d \in C^1([0,T];L^2(\Omega))$  satisfying

$$\dot{d}(t) = f(t, d(t)) \quad \forall t \in [0, T], \quad d(0) = d_0.$$
 (1.47)

In particular, it holds  $d \in C^{0,1}([0,T];L^2(\Omega))$ , which in view of (1.46) means that  $t \mapsto f(t,d(t)) \in C^{0,1}([0,T];L^2(\Omega))$  as well, and thus,  $d \in C^{1,1}([0,T];L^2(\Omega))$ , on account of (1.47). This together with (1.45) finally gives the assertion.

Remark 1.33 (Improved space regularity for the local damage). An inspection of the above proof shows the following: If we assume that the initial datum is more regular, in the sense that  $d_0 \in L^{\rho}(\Omega)$ , where  $\rho \in (2, \infty]$ , and if  $\Phi : [0, T] \times L^2(\Omega) \to L^{\rho}(\Omega)$  is Lipschitz continuous, then  $d \in C^{1,1}([0,T];L^{\rho}(\Omega))$ . Note that this is due to the fact that  $\max : L^{\rho}(\Omega) \to L^{\rho}(\Omega)$  is well defined and Lipschitz continuous, as a consequence of the Lipschitz continuity of  $\max : \mathbb{R} \to \mathbb{R}$ .

Hence, if  $d_0 \in L^r(\Omega)$ , then the local damage belongs to  $C^{1,1}([0,T];L^r(\Omega))$ , where  $r < \infty$  and r = 6 for N = 2 and N = 3, respectively, as a result of the embedding  $H^1(\Omega) \hookrightarrow L^r(\Omega)$ . As in the next section the Lipschitz continuity of the nonlocal damage improves, the same holds for the space regularity of d, see Remark 1.46 below.

Let us point out that the Lipschitz continuity of the local damage d readily transfers to  $\varphi = \Phi(\cdot, d(\cdot))$  and  $\boldsymbol{u} = \mathcal{U}(\cdot, \varphi(\cdot))$ , as explained in the sequel. First of all, (1.38) and the Lipschitz continuity of d imply the Lipschitz continuity of  $\varphi$ . Further, in view of  $H^1(\Omega) \hookrightarrow L^r(\Omega)$  with  $r \in [2p/(p-2), \infty)$  and  $r \in [2p/(p-2), 6]$  for N=2 and N=3, respectively, we obtain from Proposition 1.10 that  $\boldsymbol{u} \in C^{0,1}([0,T]; \boldsymbol{W}_D^{1,\pi}(\Omega))$ , with  $\pi \in [2,p)$  for N=2 and  $\pi \in [2,6p/(p+6)]$  in case of N=3. Note that the intervals  $[2p/(p-2),\infty)$  and [2p/(p-2),6] are not empty thanks to p>2 for N=2 and  $p\geq 3$  for N=3, respectively.

We observe that as long as  $\ell \in C^{0,1}([0,T]; \mathbf{W}^{-1,p}(\Omega))$ , one cannot expect more time regularity for  $\mathbf{u}$  and  $\varphi$ , see also Remark 3.16 on page 115 below. The time regularity thereof can be however further improved, provided that the load is more regular in time, as we will see in Section 1.3 below, in the case of a continuously differentiable load.

To summarize our results so far, we have proven that, under Assumptions 1.5 and 1.17, there exists a unique solution  $(\boldsymbol{u}, \varphi, d)$  for the penalized damage model (P) satisfying  $\boldsymbol{u} \in C^{0,1}([0,T]; \boldsymbol{W}_D^{1,\pi}(\Omega))$  with  $\pi$  as above,  $\boldsymbol{u}(t) \in \boldsymbol{W}_D^{1,p}(\Omega)$  for all  $t \in [0,T]$ ,

 $\varphi \in C^{0,1}([0,T];H^1(\Omega))$  and  $d \in C^{1,1}([0,T];L^2(\Omega))$ , and the following system of differential equations:

$$-\operatorname{div} g(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{u}(t)) = \ell(t) \quad \text{in } \boldsymbol{W}^{-1,p}(\Omega), \quad (1.48a)$$

$$-\operatorname{div} g(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{u}(t)) = \ell(t) \quad \text{in } \boldsymbol{W}^{-1,p}(\Omega), \quad (1.48a)$$

$$-\alpha\Delta\varphi(t) + \beta\,\varphi(t) + \frac{1}{2}\,g'(\varphi(t))\mathbb{C}\,\varepsilon(\boldsymbol{u}(t)) : \varepsilon(\boldsymbol{u}(t)) = \beta d(t) \quad \text{in } H^1(\Omega)^*, \quad (1.48b)$$

$$\dot{d}(t) - \frac{1}{\delta}\max(-\beta(d(t) - \varphi(t)) - r) = 0, \quad d(0) = d_0 \quad (1.48c)$$

$$\dot{d}(t) - \frac{1}{\delta} \max(-\beta(d(t) - \varphi(t)) - r) = 0, \qquad d(0) = d_0$$
 (1.48c)

for all  $t \in [0, T]$ .

#### Improved regularity and Lipschitz continuity of the non-1.2 local damage

In this section we show that the optimal nonlocal damage possesses higher regularity and satisfies a corresponding Lipschitz condition. The key tool therefor is a known result of  $W^{1,q}$ -theory, with q>2, of which we can make use mainly thanks to the space regularity of the nonlinearity in (1.48b). These new findings can be used to improve many results throughout this work, see Remarks 1.46, 1.61, 1.63, 3.12 and 3.23 below. Of course, Assumptions 1.5 and 1.17 are supposed to hold throughout this section.

#### 1.2.1 Improved regularity

The starting point for proving the improved regularity is the equation (1.24), which characterizes the optimal nonlocal damage, cf. Theorem 1.23. Applying a classical boot strapping argument therefor will then easily give the result. However, we first need to do a little preparatory work, of which we also make use in the upcoming subsection. For convenience, we introduce

**Definition 1.34.** We define the operator  $-\Delta + I : H^1(\Omega) \to H^1(\Omega)^*$  by

$$\langle (-\Delta + I)w, \psi \rangle_{H^1(\Omega)} := \int_{\Omega} \nabla w \cdot \nabla \psi + w \psi \, dx \quad \forall \, \psi \in H^1(\Omega).$$

The operator  $-\Delta + I$  considered with different domains and ranges will be denoted by the same symbol for the sake of simplicity.

The key result mentioned at the beginning of the section is covered by the following

**Lemma 1.35.** [23, Theorem 3, Lemma 1, Definition 4, Remark 6] Let  $\widetilde{\Gamma}_N \subset \Gamma$  be given so that  $\Omega \cup \widetilde{\Gamma}_N$  is regular in the sense of Gröger cf. [23, Definition 2], see also Assumption 0.5. Then there exists q>2 such that for all  $\nu\in[2,q]$  the operator  $-\Delta+I:W^{1,\nu}_{\Gamma\backslash\widetilde{\Gamma}_N}(\Omega)\to$  $W^{1,\nu'}_{\Gamma\backslash\widetilde{\Gamma}_N}(\Omega)^*$  is continuously invertible. The norms of their inverses are bounded by a constant depending only on q, the domain  $\Omega$  and  $\widetilde{\Gamma}_N$  at most.

In the remaining of the subsection we work with an arbitrary, but fixed  $(t,d) \in [0,T] \times L^2(\Omega)$  and use for simplicity the notations  $\bar{\varphi} := \Phi(t,d)$  and

$$f := \beta(d - \bar{\varphi}) + \alpha \,\bar{\varphi} - F(t, \bar{\varphi}). \tag{1.49}$$

We consider the equation

$$(-\Delta + I)w = \frac{1}{\alpha}f \quad \text{in } H^1(\Omega)^*, \tag{1.50}$$

which is solved by  $\bar{\varphi}$ , in view of (1.49) and Definition 1.24. Then, taking advantage of the fact that f possesses higher regularity than  $H^1(\Omega)^*$ , we show by means of Lemma 1.35 that  $\bar{\varphi} \in W^{1,q}(\Omega)$ , where q > 2.

The regularity of the linear form on the right-hand side in (1.50) is given by the following

**Lemma 1.36.** Under Assumptions 1.5 and 1.17, it holds  $f \in W^{1,\varrho'}(\Omega)^*$ , where

$$\frac{1}{\rho} = \max\left\{\frac{2}{p} - \frac{1}{N}, \frac{1}{2} - \frac{1}{N}\right\} < \frac{1}{N}.\tag{1.51}$$

*Proof.* From (1.49) and Sobolev embeddings we infer that  $f \in L^2(\Omega) \cap L^{\frac{p}{2}}(\Omega)$  (see Definition 1.15), which implies  $f \in L^r(\Omega)$ , where  $r := \min\{2, p/2\}$ . Thus, the idea of the proof is to find  $\varrho \in (2, \infty)$  (as large as possible) so that

$$L^r(\Omega) \hookrightarrow W^{1,\varrho'}(\Omega)^*.$$
 (1.52)

To this end, we employ Sobolev embeddings, which tell us in view of  $\varrho' \in (1,2)$  that  $W^{1,\varrho'}(\Omega) \hookrightarrow L^{\frac{N\varrho'}{N-\varrho'}}(\Omega)$ , whence  $L^{\frac{N\varrho}{N+\varrho}}(\Omega) \hookrightarrow W^{1,\varrho'}(\Omega)^*$ , as  $\frac{N\varrho'}{N-\varrho'} \in [1,\infty)$ . The embedding (1.52) is then guaranteed if

$$r \ge \frac{N\varrho}{N+\varrho},$$

which is equivalent to

$$\frac{1}{\rho} \ge \frac{1}{r} - \frac{1}{N}.$$

In light of the definition of r, this gives the identity in (1.51), as we searched for the largest value for  $\varrho$ . From Assumption 1.17.1 and N < 4 we finally deduce  $\varrho > N$ .

We can now conclude the main result of this subsection:

**Theorem 1.37** (Improved regularity of the nonlocal damage). Suppose that Assumptions 1.5 and 1.17 hold true. Then, there exists a q > 2 such that  $\Phi(t, d) \in W^{1,q}(\Omega)$  for every  $(t, d) \in [0, T] \times L^2(\Omega)$ .

Proof. In view of Lemma 1.36, we want to apply Lemma 1.35 for  $\widetilde{\Gamma}_N = \Gamma$ , which means that we first have to verify that  $\overline{\Omega}$  is regular in the sense of Gröger. Thanks to [24, Theorem 5.2, 5.4], this is indeed the case, since  $\Omega$  is a bounded Lipschitz domain, cf. Assumption 0.5. Thus, by virtue of Lemma 1.35, there exists  $q_{\Omega} > 2$  such that for all  $\nu \in [2, q_{\Omega}]$  the operator  $-\Delta + I : W^{1,\nu}(\Omega) \to W^{1,\nu'}(\Omega)^*$  is continuously invertible. Hence, in view of Lemma 1.36, (1.50) admits a unique solution  $\varphi \in W^{1,q}(\Omega)$ , where  $q := \min\{q_{\Omega}, \varrho\} > 2$ , with  $\varrho$  given by (1.51). Since  $\bar{\varphi}$  solves (1.50) as well, we obtain  $\varphi = \bar{\varphi}$ , which completes the proof.

#### 1.2.2 Improved Lipschitz continuity

As a consequence of the higher regularity of the solution of (1.24), one expects that  $\Phi$  satisfies a corresponding Lipschitz condition. For this reason, we investigate in what follows the  $W^{1,q}(\Omega)$ -Lipschitz continuity of the solution map of (1.24). Throughout this subsection, q,  $q_{\Omega}$  and  $\varrho$  stand for the numbers given by Theorem 1.37, Lemma 1.35, where  $\widetilde{\Gamma}_N := \Gamma$ , and Lemma 1.36, respectively. We work with  $(t_i, d_i) \in [0, T] \times L^2(\Omega)$  arbitrary, but fixed and define  $\varphi_i := \Phi(t_i, d_i) \in W^{1,q}(\Omega)$ , where i = 1, 2. In addition, we abbreviate

$$\iota := \frac{1}{\alpha} (\beta(d_1 - d_2) - (\beta - \alpha)(\varphi_1 - \varphi_2) - (F(t_1, \varphi_1) - F(t_2, \varphi_2))). \tag{1.53}$$

Note that  $\iota \in W^{1,\varrho'}(\Omega)^*$ , on account of Lemma 1.36. Moreover, notice that, by construction,  $\varphi_1 - \varphi_2$  solves

$$(-\Delta + I)w = \iota.$$

Thus, according to Lemma 1.35 applied for  $\widetilde{\Gamma}_N = \Gamma$ , the following estimate holds true

$$\|\varphi_1 - \varphi_2\|_{W^{1,\mu}(\Omega)} \le c\|\iota\|_{W^{1,\mu'}(\Omega)^*} \quad \forall \, 2 \le \mu \le q = \min\{q_\Omega, \varrho\}.$$
 (1.54)

Unfortunately, the desired Lipschitz continuity condition cannot be directly proven by setting  $\mu = q$  in (1.54), as one cannot directly derive an estimate of the form  $\|\iota\|_{W^{1,q'}(\Omega)^*} \leq L(\|d_1 - d_2\|_2 + |t_1 - t_2|)$ . However, by applying a finite number of boot strapping steps, it is possible to prove the result, as we will next see. The starting point therefor is the Lipschitz estimate (1.38).

The main idea in each step is to search for  $\nu \in [2, \varrho]$  as large as possible such that  $\|\iota\|_{W^{1,\nu'}(\Omega)^*} \leq C(\|d_1 - d_2\|_2 + |t_1 - t_2|)$  holds. Such an estimate will be established by using the Lipschitz continuity result for the solution map  $\Phi$ , which was proven in the previous boot strapping step. Then, applying (1.54) with  $\mu = \min\{q_{\Omega}, \nu\}$  leads to an improved Lipschitz continuity result, as we will later see.

In the proof of Theorem 1.45 below we establish that the number of boot strapping steps, which we hereafter call m, is finite, see (1.67). Therein we also show that, starting from the  $H^1(\Omega)$ -Lipschitz continuity of  $\Phi$ , the arising sequence of Lipschitz exponents reads

$$\{2 = \nu_0(p), ..., \nu_{m-1}(p), \min\{q_{\Omega}, \nu_m(p)\} = q\},\$$

with  $\nu_j(p)_{j=0,\dots,m}$  as in Definition 1.39 below. As confirmed by Lemma 1.41 below, the result improves in each step. Now, assuming that the  $W^{1,\nu_n(p)}(\Omega)$ -Lipschitz continuity of  $\Phi$  has already been proven, the n+1-st boot strapping step consists of

(i) showing the estimate

$$\|\iota\|_{W^{1,\nu_{n+1}(p)'}(\Omega)^*} \le C(\|\varphi_1 - \varphi_2\|_{W^{1,\nu_n(p)}(\Omega)} + \|d_1 - d_2\|_2 + |t_1 - t_2|), \quad (1.55)$$

see Lemma 1.44 below,

- (ii) using the  $W^{1,\nu_n(p)}(\Omega)$ -Lipschitz continuity of  $\Phi$  in (1.55),
- (iii) applying (1.54) for  $\mu = \nu_{n+1}(p) \leq q$ , which yields the  $W^{1,\nu_{n+1}(p)}(\Omega)$ -Lipschitz continuity.

We point out that the number of steps m, depends on p, N and  $q_{\Omega}$ , in view of (1.67) and (1.59). The formula (1.59) is in particular essential in three dimensions and gives, in light of (1.68), the maximal number of required steps. Thus, in the two-dimensional case, the  $W^{1,q}(\Omega)$ -Lipschitz continuity is established in maximal two steps, cf. (1.59). From Lemma 1.40 below we further deduce that small values of p account for small values of  $\nu_n(p)$ . Thus, the closer p is to N, the more difficult it is to come closer to q, which is also confirmed by (1.59), since  $n^*$  is larger for smaller values of p.

The subsection is structured as follows. First we lay the foundations for proving (1.55) by introducing the sequence  $\{\nu_n(p)\}_n$  (Definitions 1.38-1.39) and by enumerating some useful properties thereof (Lemmata 1.40-1.42). Afterwards, we proceed towards showing the main result by first deriving an estimate of the type (1.55) in a more general form (Lemma 1.43), of which we make use to establish Lemma 1.44. The subsection ends with the statement of the desired result (Thereom 1.45) and some comments regarding its consequences (Remark 1.46).

**Definition 1.38.** We define the sequence  $\{a_n\}_{n\in\mathbb{N}^+}$  as

$$a_n := \frac{4Nn}{4n - N + 2}.$$

Note that  $\{a_n\}_{n\in\mathbb{N}^+}$  is strictly decreasing in three dimensions and that it satisfies

$$\begin{cases} a_n = 2 & \text{for } N = 2, \\ 4 = a_1 \ge a_n > 3 & \text{for } N = 3. \end{cases}$$
 (1.56)

for all  $n \in \mathbb{N}^+$ . This can be easily shown by straight forward computation.

In the next definition we give a general formula for  $\nu_n(p)$ , by means of which the Lipschitz exponent achieved in the *n*-th step can be exactly specified, see the proof of Theorem 1.45 below.

**Definition 1.39.** Let Assumption 1.17.1 hold true. Given  $n \in \mathbb{N}$ ,  $n \geq 2$ , we define

$$\nu_n(p) := \begin{cases} \frac{2Np}{4Nn - (4n-N)p} & \text{if } p \in (N, a_{n-1}), \\ \in (N, \varrho) & \text{if } p = a_{n-1}, \\ \varrho & \text{if } p \in (a_{n-1}, \infty). \end{cases}$$

Further, define  $\nu_0(p) := 2$  and

$$\nu_1(p) := \begin{cases} \in (N, \varrho) & \text{if } p \in (N, 4] \\ \varrho & \text{if } p \in (4, \infty) \end{cases}, \quad \text{for } N = 2,$$

$$\nu_1(p) := \begin{cases} \frac{6p}{12 - p} & \text{if } p \in (N, 6) \\ \varrho & \text{if } p \in [6, \infty) \end{cases}, \quad \text{for } N = 3.$$

The notation  $:= \in (N, \varrho)$ ' stands for  $:= \kappa$ , where  $\kappa \in (N, \varrho)$  is given'.

Observe that for  $n \geq 2$  and N = 2 it holds  $\nu_n(p) = \varrho$ , due to (1.56) and p > 2. We have chosen however to define  $\nu_n(p)$ ,  $n \geq 2$ , by a formula which can be used for both space dimensions, in order to distinguish as least as possible between the cases N = 2 and N = 3 in the upcoming analysis.

The following lemma has a key role when making the connection between the Lipschitz exponents in Lemma 1.44 below.

**Lemma 1.40.** Under Assumption 1.17.1, it holds for all  $n \in \mathbb{N}^+$ 

$$2 < \nu_n(p) < N \quad \text{if } p \in (N, a_n), \\
\nu_n(p) = N \quad \text{if } p = a_n, \\
N < \nu_n(p) \le \varrho \quad \text{if } p \in (a_n, \infty).$$
(1.57)

Consequently, the embedding  $W^{1,\nu_n(p)}(\Omega) \hookrightarrow L^{S_n(p)}(\Omega)$  is true, where

$$S_n(p) = \begin{cases} \frac{N\nu_n(p)}{N - \nu_n(p)} & \text{if } p \in (N, a_n), \\ \kappa & \text{if } p = a_n, \\ \infty & \text{if } p \in (a_n, \infty), \end{cases}$$

$$(1.58)$$

with any  $\kappa \in [1, \infty)$ .

*Proof.* We only focus on proving (1.57), as the last statement is just a result of Sobolev embeddings. For n=1 and N=2, (1.57) is immediately given by (1.56) and Definition 1.39, since p>2. In case of n=1 and N=3, we observe that straight forward computation leads to the fact that  $(3,6)\ni p\mapsto \frac{6p}{12-p}\in (2,6)$  is strictly increasing, which ensures (1.57) in view of  $a_1=4$  and Definition 1.39. Note that  $\nu_1(p)>2$  is ensured

by p>3, cf. Assumption 1.17.1. Let now  $n\in\mathbb{N},\ n\geq 2$  be arbitrary, but fixed. For N=2 the system (1.57) is then automatically fulfilled, due to (1.56). In fact,  $\nu_n(p)=\varrho$  by Definition 1.39. Let now N=3. Again, we see that  $\nu_n(p)>2$ , as a result of Assumption 1.17.1. From Definition 1.39 we further deduce  $\nu'_n(p)=\frac{8N^2n}{(4Nn-(4n-N)p)^2}>0$  for  $p\in(N,a_{n-1})$ , and thus,  $\nu_n$  is strictly increasing on  $(N,a_{n-1})$ . Recall that for N=3 we have  $a_{n-1}>a_n>N$ . With  $\nu_n(a_n)=\frac{8N^2n}{8Nn}=N$  and by means of Definition 1.39 we then conclude (1.57). This completes the proof.

The next result confirms that the Lipschitz continuity of  $\Phi$  improves in each boot strapping step.

**Lemma 1.41.** Under Assumption 1.17.1 it holds  $\nu_n(p) \ge \nu_{n-1}(p)$  for all  $n \in \mathbb{N}^+$ . The equality is satisfied only if  $\nu_{n-1}(p) = \varrho$ .

*Proof.* One can easily see that  $\nu_1(p) > \nu_0(p)$  in view of Definition 1.39. Note that in the two-dimensional case it holds  $\nu_n(p) = \varrho \ge \nu_{n-1}(p)$  for all  $n \in \mathbb{N}, n \ge 2$ , since p > 2. In the three-dimensional case we further obtain

$$\frac{6p}{24 - 5p} = \nu_2(p) > \nu_1(p) = \frac{6p}{12 - p} \quad \text{if } p \in (3, 4),$$

$$(3, \varrho) \ni \nu_2(p) > \nu_1(p) = 3 \quad \text{if } p = a_1 = 4,$$

$$\varrho = \nu_2(p) \ge \nu_1(p) \quad \text{if } p \in (4, \infty).$$

Let now  $n \in \mathbb{N}$ ,  $n \geq 3$ , be arbitrary, but fixed and N = 3. With  $a_{n-1} < a_{n-2}$  and (1.57) we conclude

$$\frac{6p}{12n - (4n - 3)p} = \nu_n(p) > \nu_{n-1}(p) = \frac{6p}{12(n - 1) - (4n - 7)p} \quad \text{if } p \in (3, a_{n-1}),$$

$$(3, \varrho) \ni \nu_n(p) > \nu_{n-1}(p) = 3 \quad \text{if } p = a_{n-1},$$

$$\varrho = \nu_n(p) \ge \nu_{n-1}(p) \quad \text{if } p \in (a_{n-1}, \infty).$$

This gives the desired assertion.

An inspection of the proof of Theorem 1.45 below shows that, if  $q_{\Omega}$  is small enough, the number of steps m can be considerably reduced in the three dimensional case, while in the two-dimensional case we have  $m \leq 2$  anyway. We refer here to (1.67) and (1.68), respectively. The situation is far more challenging in case of N=3 and  $q_{\Omega}>\varrho$ . Here  $a_n$  plays an important role, as it tells us for given p and n, if one should expect  $W^{1,q}(\Omega)$ -Lipschitz continuity in the near future, see Lemma 1.40. Moreover, it facilitates treating different situations separately. One sees in Lemma 1.40 that the smaller p, the smaller the improvement of the Lipschitz-exponent in the n-th step. However, since  $\{a_n\}$  is strictly decreasing, the interval  $(3,a_n)$  becomes smaller with each step and regardless of how small the difference between p and 3 is, one can achieve  $\varrho$  after a finite number of boot strapping steps, in light of Definition 1.39. This result is stated for both dimensions in the following

**Lemma 1.42.** Under Assumption 1.17.1, it holds  $\nu_{n^*}(p) = \varrho$ , where

$$n^* = \left[\frac{p(N+2) - 4N}{4(p-N)}\right] + 1. \tag{1.59}$$

*Proof.* First note that  $\frac{p(N+2)-4N}{4(p-N)} \ge 1 \Leftrightarrow p(N-2) \ge 0$ . Thus,  $n^* \ge 2$ . On account of Definition 1.38 we have

$$a_{n^*-1} = \frac{4N(n^*-1)}{4n^*-N-2}.$$

By using (1.59) in the above identity we then obtain

$$a_{n^*-1} < p.$$

This leads in view of Definition 1.39 to  $\nu_{n^*}(p) = \varrho$ , whence the desired assertion.

We can now proceed towards proving (1.55). Prior to this, it is useful to derive a more general estimate, of which we make use in the proof of Lemma 1.44 below.

**Lemma 1.43.** Let Assumptions 1.5, 1.17 hold and let  $r \in [\frac{2p}{p-2}, \infty]$  be given such that  $\varphi_1, \varphi_2 \in L^r(\Omega)$ . Then the following estimate holds true

$$\|\iota\|_{W^{1,\nu'}(\Omega)^*} \le L(\|\varphi_1 - \varphi_2\|_r + \|d_1 - d_2\|_2 + |t_1 - t_2|), \tag{1.60}$$

where L > 0 and

$$\frac{1}{\nu} = \max\left\{\frac{1}{r} + \frac{2}{p} - \frac{1}{N}, \frac{1}{2} - \frac{1}{N}\right\}. \tag{1.61}$$

*Proof.* The idea of the proof is to search for the largest  $\nu \in (2, \varrho]$  such that (1.60) is fulfilled. Since  $\nu' < 2 \le N$ , Sobolev embeddings imply  $W^{1,\nu'}(\Omega) \hookrightarrow L^{\frac{N\nu'}{N-\nu'}}(\Omega)$ . Then, by applying Lemma 1.18 and Hölder's inequality, we have in view of (1.53) for all  $\psi \in W^{1,\nu'}(\Omega)$ 

$$|\langle \iota, \psi \rangle_{W^{1,\nu'}(\Omega)}| \leq C(\|\varphi_1 - \varphi_2\|_r + |t_1 - t_2| + \|d_1 - d_2\|_2 + \|\varphi_1 - \varphi_2\|_2)\|\psi\|_{\frac{N\nu'}{N-\nu'}}$$

$$\leq L(\|\varphi_1 - \varphi_2\|_r + \|d_1 - d_2\|_2 + |t_1 - t_2|)\|\psi\|_{W^{1,\nu'}(\Omega)},$$

$$(1.62)$$

provided that

$$\frac{N - \nu'}{N\nu'} + \frac{2}{p} + \frac{1}{r} \le 1, 
\frac{1}{2} + \frac{N - \nu'}{N\nu'} \le 1.$$
(1.63)

Note that the second inequality in (1.62) follows from  $r \ge 2p/(p-2) \ge 2$ , or alternatively, from (1.38). The system (1.63) is further equivalent to

$$\left\{ \begin{array}{l} \frac{1}{\nu'} - \frac{1}{N} + \frac{2}{p} + \frac{1}{r} \le 1, \\ \frac{1}{2} + \frac{1}{\nu'} - \frac{1}{N} \le 1 \end{array} \right\} \Longleftrightarrow \left\{ \begin{array}{l} \frac{1}{r} + \frac{2}{p} - \frac{1}{N} \le \frac{1}{\nu}, \\ \frac{1}{2} - \frac{1}{N} \le \frac{1}{\nu}. \end{array} \right.$$

By taking the definition of the  $\|\cdot\|_{W^{1,\nu'}(\Omega)^*}$ -norm into account and by keeping in mind that we search for  $\nu$  as large as possible, we now deduce that (1.60) holds true for  $\nu$  as in (1.61).

The next lemma is the essential tool for proving the  $W^{1,q}(\Omega)$ -Lipschitz continuity, as it connects the Lipschitz exponents of  $\Phi$  achieved in consecutive boot strapping steps by (1.55).

**Lemma 1.44.** Let Assumptions 1.5, 1.17 hold and let  $n \in \mathbb{N}$  be given with  $\nu_n(p) \leq q$ . Then there exists L > 0 such that

$$\|\iota\|_{W^{1,\nu_{n+1}(p)'}(\Omega)^*} \le L(\|\varphi_1 - \varphi_2\|_{W^{1,\nu_n(p)}(\Omega)} + \|d_1 - d_2\|_2 + |t_1 - t_2|).$$

*Proof.* The idea of the whole proof is to find  $r \geq 2p/(p-2)$  as large as possible such that  $W^{1,\nu_n(p)}(\Omega) \hookrightarrow L^r(\Omega)$  and then use (1.61) in order to obtain  $\nu$  for which the estimate

$$\|\iota\|_{W^{1,\nu'}(\Omega)^*} \le L(\|\varphi_1 - \varphi_2\|_r + \|d_1 - d_2\|_2 + |t_1 - t_2|)$$

$$\le L(\|\varphi_1 - \varphi_2\|_{W^{1,\nu_n(p)}(\Omega)} + \|d_1 - d_2\|_2 + |t_1 - t_2|)$$
(1.64)

is true. Note that once r is found, it holds  $\varphi_1, \varphi_2 \in L^r(\Omega)$ , since  $\varphi_1, \varphi_2 \in W^{1,q}(\Omega) \hookrightarrow W^{1,\nu_n(p)}(\Omega)$ , by assumption. Thus, when applying Lemma 1.43 in what follows, we only check that  $r \geq 2p/(p-2)$ . In view of Lemma 1.40, we treat the cases n = 0 and  $n \in \mathbb{N}^+$  separately.

(i) n = 0.

If N=2, the embedding  $H^1(\Omega) \hookrightarrow L^r(\Omega)$  is satisfied for any  $r \in [2p/(p-2), \infty)$  and therefore (1.64) holds in view of Lemma 1.43 when

$$\frac{1}{\nu} = \max\left\{\frac{1}{r} + \frac{2}{p} - \frac{1}{2}, 0\right\} \quad \forall r \in [2p/(p-2), \infty).$$

We now search for  $r \in [2p/(p-2), \infty)$  so that  $\nu = \nu_1(p)$ . If  $p \le 4$ , we have  $\frac{1}{r} + \frac{2}{p} - \frac{1}{2} > \frac{2}{p} - \frac{1}{2} \ge 0$ , in which case  $\frac{1}{r} + \frac{2}{p} - \frac{1}{2} = \frac{1}{\nu} > \frac{1}{\varrho} = \frac{2}{p} - \frac{1}{2}$ , by (1.51). In view of  $\frac{1}{N} > \frac{1}{\varrho}$ , we then choose r as large as possible such that  $\frac{1}{N} > \frac{1}{\nu} > \frac{1}{\varrho}$ . If p > 4, we have  $\frac{2}{p} - \frac{1}{2} < 0$ , so that  $\varrho = \infty$  by (1.51). In view of the definition of  $\nu_1(p)$  we choose r large enough such that  $\frac{1}{r} + \frac{2}{p} - \frac{1}{2} \le 0 = \frac{1}{\varrho}$ , which leads to  $\nu = \infty = \varrho$ . Altogether, it follows from the above that  $\nu = \nu_1(p)$ , in light of Definition 1.39.

In three dimensions we obtain  $r = 6 \ge 2p/(p-2)$  and (1.61) reads

$$\frac{1}{\nu} = \max\left\{\frac{1}{6} + \frac{2}{p} - \frac{1}{3}, \frac{1}{2} - \frac{1}{3}\right\}$$
$$= \max\left\{\frac{2}{p} - \frac{1}{6}, \frac{1}{6}\right\}.$$

If p < 6, we arrive at  $\nu = \nu_1(p)$ , since  $\frac{2}{p} - \frac{1}{6} > \frac{1}{6}$ . In case of  $p \ge 6$  we obtain from (1.51) that  $\frac{1}{\nu} = \frac{1}{6} = \frac{1}{\varrho}$ , which gives the assertion in view of Definition 1.39.

(ii)  $n \in \mathbb{N}^+$  arbitrary, but fixed.

On account of Lemma 1.40 we set  $r := S_n(p)$  and first check that  $r \ge 2p/(p-2)$ . To this end, note that, if N = 3,  $x \mapsto Nx/(N-x)$  is strictly increasing on [2, N), which in view of  $\nu_n(p) \in (2, N)$  leads to  $S_n(p) > 2N/(N-2) > 2p/(p-2)$  if  $p \in (N, a_n)$ , see also (1.32). Note further that, due to p > N, the interval  $[2p/(p-2), \infty)$  is not empty, so that, in light of (1.58) we have  $r \ge 2p/(p-2)$  in both dimensions. Now Lemma 1.43 ensures that (1.64) holds for

$$\frac{1}{\nu} = \max\left\{\frac{1}{S_n(p)} + \frac{2}{p} - \frac{1}{N}, \frac{1}{2} - \frac{1}{N}\right\} 
= \begin{cases}
\max\left\{\frac{N - \nu_n(p)}{N\nu_n(p)} + \frac{2}{p} - \frac{1}{N}, \frac{1}{2} - \frac{1}{N}\right\} & \text{if } p \in (N, a_n), \\
\max\left\{\frac{1}{\kappa} + \frac{2}{p} - \frac{1}{N}, \frac{1}{2} - \frac{1}{N}\right\} & \text{if } p = a_n, \\
\max\left\{\frac{1}{\infty} + \frac{2}{p} - \frac{1}{N}, \frac{1}{2} - \frac{1}{N}\right\} & \text{if } p \in (a_n, \infty),
\end{cases}$$
(1.65)

where  $\kappa < \infty$  can be chosen large enough. We distinguish between

•  $p \in (N, a_n)$ : This situation can occur only in three dimensions, see (1.56). In view of (1.57) and (1.56) we have  $\frac{1}{\nu_n(p)} + \frac{2}{p} - \frac{2}{N} > \frac{1}{N} + \frac{2}{p} - \frac{2}{N} = \frac{2}{p} - \frac{1}{N} > \frac{2}{a_n} - \frac{1}{N} \ge \frac{2}{a_1} - \frac{1}{N} = \frac{1}{2} - \frac{1}{N}$ . This implies

$$\frac{1}{\nu} = \frac{1}{\nu_n(p)} + \frac{2}{p} - \frac{2}{N} \Longleftrightarrow \nu = \frac{2Np}{4N(n+1) - (4n+4-N)p},$$

in view of the definition of  $\nu_n(p)$  and by keeping in mind that  $a_1 = 4 < 6$  if n = 1 and  $a_n < a_{n-1}$  in case of  $n \ge 2$ . Thus, by Definition 1.39, we now have  $\nu = \nu_{n+1}(p)$ .

•  $p = a_n$ : This can take place only in the three-dimensional case, in view of (1.56), which also yields  $\frac{2}{a_n} \ge \frac{1}{2}$ . This means that

$$\frac{1}{\nu} = \frac{1}{\kappa} + \frac{2}{a_n} - \frac{1}{N} \quad \text{with } \kappa < \infty, \tag{1.66}$$

and  $\frac{1}{\varrho} = \frac{2}{p} - \frac{1}{N}$ , on account of (1.65) and (1.51), respectively. Since  $\kappa < \infty$ , we have  $\frac{1}{\nu} > \frac{1}{\varrho}$ . Moreover, due to (1.56), it holds  $\frac{2}{N} - \frac{2}{a_n} > 0$  and thus, it is possible to choose  $\kappa$  (large enough) such that  $\frac{1}{\kappa} \in \left(0, \frac{2}{N} - \frac{2}{a_n}\right)$ . Then,  $\frac{1}{\kappa} + \frac{2}{a_n} - \frac{1}{N} < \frac{1}{N}$  and hence, in view of (1.66), we get  $\nu > N$ . Summarizing, we have  $\nu \in (N, \varrho)$  and Definition 1.39 combined with (1.64) finally give the desired estimate.

•  $p \in (a_n, \infty)$ : In this case, the assertion is immediately given by (1.65), (1.51) and Definition 1.39.

We are now in the position to establish the main result of this subsection:

**Theorem 1.45** (Improved Lipschitz continuity of  $\Phi$ ). Let Assumptions 1.5 and 1.17 hold. Then, there exists L > 0 such that the estimate

$$\|\Phi(t_1, d_1) - \Phi(t_2, d_2)\|_{W^{1,q}(\Omega)} \le L(\|d_1 - d_2\|_2 + |t_1 - t_2|)$$

holds true for all  $t_1, t_2 \in [0, T]$  and all  $d_1, d_2 \in L^2(\Omega)$ , where q > 2 is given by Theorem 1.37.

*Proof.* Throughout this proof,  $(t_i, d_i) \in [0, T] \times L^2(\Omega)$  is arbitrary, but fixed and  $\varphi_i$  denotes  $\Phi(t_i, d_i)$  for i = 1, 2. We show that after executing a finite number of boot strapping steps starting from the  $W^{1,\nu_0(p)}(\Omega)$ -Lipschitz continuity of  $\Phi$ , one obtains the  $W^{1,q}(\Omega)$ -Lipschitz continuity thereof. As it will turn out, the number of steps is given by

$$m := \begin{cases} \bar{n} & \text{if } \min\{q_{\Omega}, \varrho\} = q_{\Omega}(=q), \\ n^* & \text{if } \min\{q_{\Omega}, \varrho\} = \varrho(=q), \end{cases}$$
 (1.67)

where

- $\bar{n} := \min\{n \in \mathbb{N} : \nu_n(p) \ge q_\Omega\},\$
- $n^*$  as in (1.59), i.e.,  $\nu_{n^*}(p) = \varrho$ .

We remark that

$$m \le n^*, \tag{1.68}$$

since  $n^* \geq \bar{n}$  if  $q_{\Omega} \leq \varrho$ . Moreover, note that  $\{n \in \mathbb{N} : \nu_n(p) \geq q_{\Omega}\} \neq \emptyset$  if  $q_{\Omega} \leq \varrho$ . Otherwise, we have  $\nu_n(p) < q_{\Omega} \leq \varrho$  for all  $n \in \mathbb{N}$ , which is in contradiction with Lemma 1.42. Additionally, the latter one yields on account of (1.67)

$$\begin{cases}
\nu_m(p) = \nu_{\bar{n}}(p) \ge q_{\Omega} & \text{if } \min\{q_{\Omega}, \varrho\} = q_{\Omega}(=q), \\
\nu_m(p) = \nu_{n^*}(p) = \varrho & \text{if } \min\{q_{\Omega}, \varrho\} = \varrho(=q),
\end{cases}$$
(1.69)

whence

$$\min\{\nu_m(p), q_{\Omega}\} = q. \tag{1.70}$$

With Lemma 1.41, (1.69) and from the definition of  $\bar{n}$  we further imply

$$\nu_j(p) \le q \quad \text{for } j \in \{0, ..., m-1\},$$

which allows us to make use of Lemma 1.44 in what follows. Let us assume that  $m \ge 2$ . Then, after applying  $j \in [1, m-1]$  times Lemma 1.44 and (1.54), we have the improved Lipschitz condition

$$\|\varphi_{1} - \varphi_{2}\|_{W^{1,\nu_{j}(p)}(\Omega)} \leq c\|\iota\|_{W^{1,\nu_{j}(p)'}(\Omega)^{*}}$$

$$\leq L(\|\varphi_{1} - \varphi_{2}\|_{W^{1,\nu_{j-1}(p)}(\Omega)} + \|d_{1} - d_{2}\|_{2} + |t_{1} - t_{2}|)$$

$$\leq L(\|\varphi_{1} - \varphi_{2}\|_{W^{1,\nu_{0}(p)}(\Omega)} + \|d_{1} - d_{2}\|_{2} + |t_{1} - t_{2}|)$$

$$\leq L(\|d_{1} - d_{2}\|_{2} + |t_{1} - t_{2}|),$$

$$(1.71)$$

where in the last inequality we used the already known  $H^1(\Omega)$ -Lipschitz continuity of  $\Phi$ , cf. (1.38). We set j := m - 1 in (1.71) and apply one last time Lemma 1.44 and (1.54). This gives in turn

$$\begin{split} \|\varphi_{1} - \varphi_{2}\|_{W^{1,q}(\Omega)} &\leq c \|\iota\|_{W^{1,q'}(\Omega)^{*}} \\ &\leq c \|\iota\|_{W^{1,\nu_{m}(p)'}(\Omega)^{*}} \\ &\leq L(\|\varphi_{1} - \varphi_{2}\|_{W^{1,\nu_{m-1}(p)}(\Omega)} + \|d_{1} - d_{2}\|_{2} + |t_{1} - t_{2}|) \\ &\leq L(\|d_{1} - d_{2}\|_{2} + |t_{1} - t_{2}|), \end{split}$$

where we also employed (1.70). Note that the above estimate holds true in case of m = 1 as well. This completes the proof.

Remark 1.46. The results in Theorems 1.37 and 1.45 lead to an improvement of the existence result at the end of Section 1.1 in both space dimensions, and especially if N=2, as in this case we have  $W^{1,q}(\Omega) \hookrightarrow L^{\infty}(\Omega)$ , thanks to q>2. To be more precise, one can now state that the solution  $(\mathbf{u}, \varphi, d)$  of the damage model (P) satisfies  $\mathbf{u} \in C^{0,1}([0,T]; \mathbf{W}^{1,p}_D(\Omega)), \ \varphi \in C^{0,1}([0,T]; W^{1,q}(\Omega))$  and  $d \in C^{1,1}([0,T]; L^2(\Omega))$  in two dimensions. The improved result for the displacement is due to the fact that  $\mathcal{U}: [0,T] \times W^{1,q}(\Omega) \to \mathbf{W}^{1,p}_D(\Omega)$  is Lipschitz continuous, in view of Proposition 1.10 applied for  $r=\infty$ .

Moreover, as a result of Remark 1.33, it holds  $d \in C^{1,1}([0,T];L^{\infty}(\Omega))$ , if, in addition,  $d_0 \in L^{\infty}(\Omega)$ . This means that if N=2 and  $d_0 \in L^{\infty}(\Omega)$ , then there exists c>0 so that the local and nonlocal damage fulfill  $d(t,x), \varphi(t,x) \leq c$  a.e. in  $(0,T) \times \Omega$ , as a consequence of  $C^{1,1}([0,T];L^{\infty}(\Omega)), C^{0,1}([0,T];W^{1,q}(\Omega)) \hookrightarrow L^{\infty}((0,T) \times \Omega)$ . However, the constant c depends on  $\beta$ . This is no longer the case in Subsection 2.4 below, where the same estimate holds true in both dimensions and with c independent of  $\beta$  (under additional nonrestrictive assumptions). Furthermore, as explained in Remark 1.61 below, the results in this section yield in two dimensions the best differentiability result one can expect for  $\mathcal{U}$ , as well as for  $\Phi$ , in the sense that both operators can be differentiated without norm gap.

### 1.3 Fréchet-differentiability of the elliptic system

This section is dedicated to studying the differentiability properties of the solution operators  $\mathcal{U}$  and  $\Phi$ , which were already introduced in Definitions 1.8 and 1.24. To be more specific, we aim to show that  $\mathcal{U}$  and  $\Phi$  are continuously Fréchet-differentiable. This is essential for deriving the results in the next chapter, in particular Lemmata 2.2 and 2.20, which will be needed when applying chain rule, see Sections 2.1 and 2.3 below. As we will see at the end of this section, these new findings also ensure the continuously differentiability in time of the optimal displacement and optimal nonlocal damage, which leads to an improved existence result for (P).

We begin by observing that in light of (1.25) the time dependency of  $\mathcal{U}$  and  $\Phi$  is due only to the time dependency of  $\ell$ . Moreover, an inspection thereof shows that, within

our scope of proving the continuously differentiability (with respect to time) of  $\mathcal{U}$  and  $\Phi$ , it is necessary to impose the following additional assumption

**Assumption 1.47.** From now on we assume that the applied volume and boundary load satisfies

$$\ell \in C^1([0,T]; \mathbf{W}^{-1,p}(\Omega)),$$

see also Assumptions 1.17.1 and 2.29 below.

Assumptions 1.5 and 1.17 are supposed to hold throughout this section as well.

#### 1.3.1 Differentiability of $\mathcal{U}$

By relying on standard arguments, we establish in the sequel that  $\mathcal{U}$  possesses continuous partial derivatives, which then gives in turn the main result of this subsection.

**Lemma 1.48** (Partial differentiability of  $\mathcal{U}$  w.r.t. time). Under Assumption 1.47, the operator  $\mathcal{U}$  is partially differentiable w.r.t. time. Its partial derivative  $\partial_t \mathcal{U}$  belongs to  $C([0,T] \times H^1(\Omega); V)$  and satisfies the equation

$$A_{\varphi}(\partial_t \mathcal{U}(t,\varphi)) = \dot{\ell}(t) \quad \text{for all } (t,\varphi) \in [0,T] \times H^1(\Omega).$$
 (1.72)

Proof. Let  $\varphi \in H^1(\Omega)$  be arbitrary, but fixed. From Definition 1.2 and Lemma 1.3 we know that  $A_{\varphi}^{-1} \in \mathcal{L}(\boldsymbol{W}^{-1,p}(\Omega), \boldsymbol{W}_D^{1,p}(\Omega))$  and therefore, it is continuously Fréchet-differentiable. By employing Definition 1.8, Assumption 1.47, and chain rule, we thus obtain that  $\mathcal{U}(\cdot,\varphi):[0,T]\to \boldsymbol{W}_D^{1,p}(\Omega)$  is differentiable and the derivative fulfills (1.72). Completely analogously to the proof of Lemma 1.11 one deduces in view of Assumption 1.47 that

$$\partial_t \mathcal{U}(t_n, \varphi_n) \to \partial_t \mathcal{U}(t, \varphi) \quad \text{in } V,$$
 (1.73)

for any sequence  $\{(t_n, \varphi_n)\} \subset [0, T] \times H^1(\Omega)$  which converges towards  $(t, \varphi) \in [0, T] \times H^1(\Omega)$  as  $n \to \infty$ . This completes the proof.

Note that as a consequence of (1.6) and (1.72), one obtains on account of Assumption 1.47 the following estimate

$$\|\partial_t \mathcal{U}(t,\varphi)\|_{\mathbf{W}_{\Omega}^{1,p}(\Omega)} \le c \quad \forall (t,\varphi) \in [0,T] \times H^1(\Omega), \tag{1.74}$$

where c > 0 is independent of t and  $\varphi$ .

For a better overview, the partial differentiability w.r.t.  $\varphi$  and the continuity of the partial derivative are proven separately.

**Lemma 1.49** (Partial differentiability of  $\mathcal{U}$  w.r.t.  $\varphi$ ). Let Assumptions 1.5 and 1.17.1 be fulfilled. Then there exists an index  $\nu \in (2,p)$  such that, for every  $t \in [0,T]$ , the map  $\mathcal{U}(t,\cdot): H^1(\Omega) \to \mathbf{W}_D^{1,\nu}(\Omega)$  is Fréchet-differentiable. Moreover, for all  $t \in [0,T]$  and all  $\varphi, \delta \varphi \in H^1(\Omega)$  it holds

$$A_{\omega}(\partial_{\omega}\mathcal{U}(t,\varphi)(\delta\varphi)) = \operatorname{div}\left(g'(\varphi)(\delta\varphi)\mathbb{C}\varepsilon(\mathcal{U}(t,\varphi))\right) \text{ in } \mathbf{W}^{-1,\nu}(\Omega). \tag{1.75}$$

*Proof.* We first investigate (1.75) by showing the existence of the solution operator  $\mathcal{W}$  in (1.79) below, which will be the candidate for the partial derivative of  $\mathcal{U}$  w.r.t.  $\varphi$ . Prior to this, it is useful to define some integrability exponents, of which we make use throughout this proof. We set r:=2p/(p-2) and recall that Assumption 1.17.1 guarantees the existence of an index  $\varrho$  such that  $r\in(2,\varrho)$  and  $H^1(\Omega)\hookrightarrow L^\varrho(\Omega)$ , as shown at the beginning of the proof of Lemma 1.20. Since  $\varrho>r$ , there exists another index  $\kappa$  with  $r<\kappa<\varrho$ , say  $\kappa=(r+\varrho)/2$ . Then one defines  $\nu$  through

$$\frac{1}{\nu} = \frac{1}{\kappa} + \frac{1}{p}.\tag{1.76}$$

Notice that in view of  $\kappa \in (r, \infty)$  we have

$$\nu \in (2, p). \tag{1.77}$$

Let now  $t \in [0,T]$  and  $\varphi, \delta \varphi \in H^1(\Omega)$  be arbitrary, but fixed. For the right-hand side in (1.75), Hölder's inequality with  $1/\nu' + 1/\kappa + 1/p = 1$ , in combination with Assumption 0.6,  $H^1(\Omega) \hookrightarrow L^{\kappa}(\Omega)$  and Corollary 1.9 imply

$$\|\operatorname{div}\left(g'(\varphi)(\delta\varphi)\mathbb{C}\varepsilon(\mathcal{U}(t,\varphi))\right)\|_{\boldsymbol{W}^{-1,\nu}(\Omega)} \leq \|g'(\varphi)\|_{\infty} \|\delta\varphi\|_{\kappa} \|\mathbb{C}\varepsilon(\mathcal{U}(t,\varphi))\|_{p} \\ \leq C\|\delta\varphi\|_{\kappa}.$$
(1.78)

Due to (1.77), Lemma 1.3 is applicable with the exponent  $\nu$ , which gives in turn that the operator

$$W: H^{1}(\Omega) \ni \delta\varphi \mapsto A_{\varphi}^{-1} \operatorname{div} \left( g'(\varphi)(\delta\varphi) \mathbb{C}\varepsilon(\mathcal{U}(t,\varphi)) \right) \in \mathbf{W}_{D}^{1,\nu}(\Omega)$$
 (1.79)

is well defined. Moreover, it is linear and bounded, on account of (1.6) and (1.78), i.e.,

$$\mathcal{W} \in \mathcal{L}(H^1(\Omega), \mathbf{W}_D^{1,\nu}(\Omega)). \tag{1.80}$$

This operator is the candidate for the partial derivative of  $\mathcal{U}$  w.r.t.  $\varphi$  at the point  $(t, \varphi)$ . For this reason, we now consider the remainder term

$$R_{\varphi}(\delta\varphi) := \mathcal{U}(t, \varphi + \delta\varphi) - \mathcal{U}(t, \varphi) - \mathcal{W}(\delta\varphi) \tag{1.81}$$

and show an appropriate estimate therefor, which will ultimately give the claim. By employing Definitions 1.2, 1.8 and (1.79) we obtain

$$A_{\varphi}(R_{\varphi}(\delta\varphi)) = A_{\varphi} \big( \mathcal{U}(t, \varphi + \delta\varphi) \big) - \ell(t) - A_{\varphi} \big( \mathcal{W}(\delta\varphi) \big)$$

$$= A_{\varphi} \big( \mathcal{U}(t, \varphi + \delta\varphi) \big) - A_{\varphi + \delta\varphi} \big( \mathcal{U}(t, \varphi + \delta\varphi) \big)$$

$$- \operatorname{div} \big( g'(\varphi)(\delta\varphi) \mathbb{C}\varepsilon (\mathcal{U}(t, \varphi + \delta\varphi)) \big) + \operatorname{div} \big( g'(\varphi)(\delta\varphi) \mathbb{C}\varepsilon (\mathcal{U}(t, \varphi + \delta\varphi)) \big)$$

$$- \operatorname{div} \big( g'(\varphi)(\delta\varphi) \mathbb{C}\varepsilon (\mathcal{U}(t, \varphi)) \big)$$

$$= \operatorname{div} \big( \big( g(\varphi + \delta\varphi) - g(\varphi) - g'(\varphi)(\delta\varphi) \big) \mathbb{C}\varepsilon (\mathcal{U}(t, \varphi + \delta\varphi)) \big)$$

$$=: r_{\varphi}(\delta\varphi)$$

$$+ \operatorname{div} \big( g'(\varphi)(\delta\varphi) \mathbb{C}\varepsilon (\mathcal{U}(t, \varphi + \delta\varphi) - \mathcal{U}(t, \varphi)) \big).$$

Let us now define s via  $1/s = 1/\nu - 1/\varrho$ . Relying on (1.76), (1.77) and  $\varrho > \kappa$ , we infer that  $1/2 > 1/s > 1/\nu - 1/\kappa = 1/p$ , whence

$$2 < s < p$$
.

Applying Hölder's inequality with  $1/\nu = 1/\kappa + 1/p = 1/\varrho + 1/s$ , in combination with Corollary 1.9, Assumption 0.6 and  $H^1(\Omega) \hookrightarrow L^{\varrho}(\Omega)$  then gives

$$\begin{split} \|A_{\varphi}(R_{\varphi}(\delta\varphi))\|_{\boldsymbol{W}^{-1,\nu}(\Omega)} &\leq C \|r_{\varphi}(\delta\varphi)\|_{\kappa} \|\mathcal{U}(t,\varphi+\delta\varphi)\|_{\boldsymbol{W}_{D}^{1,p}(\Omega)} \\ &+ C \|g'(\varphi)\|_{\infty} \|\delta\varphi\|_{\varrho} \|\mathcal{U}(t,\varphi+\delta\varphi) - \mathcal{U}(t,\varphi)\|_{\boldsymbol{W}_{D}^{1,s}(\Omega)} \\ &\leq C \big( \|r_{\varphi}(\delta\varphi)\|_{\kappa} + \|\delta\varphi\|_{H^{1}(\Omega)} \|\mathcal{U}(t,\varphi+\delta\varphi) - \mathcal{U}(t,\varphi)\|_{\boldsymbol{W}_{D}^{1,s}(\Omega)} \big). \end{split}$$

Together with (1.6), this further implies

$$||R_{\varphi}(\delta\varphi)||_{\boldsymbol{W}_{D}^{1,\nu}(\Omega)} \leq C(||r_{\varphi}(\delta\varphi)||_{\kappa} + ||\delta\varphi||_{H^{1}(\Omega)}||\mathcal{U}(t,\varphi+\delta\varphi) - \mathcal{U}(t,\varphi)||_{\boldsymbol{W}_{D}^{1,s}(\Omega)}).$$
(1.82)

Moreover, we recall that  $H^1(\Omega) \hookrightarrow L^{\varrho}(\Omega)$  with  $\varrho > \kappa$ , which allows us to deduce from Lemma 5.3 that  $g: H^1(\Omega) \to L^{\kappa}(\Omega)$  is Fréchet-differentiable. Together with Lemma 1.11 and (1.82), one concludes

$$\frac{\|R_{\varphi}(\delta\varphi)\|_{\boldsymbol{W}_{D}^{1,\nu}(\Omega)}}{\|\delta\varphi\|_{H^{1}(\Omega)}} \leq C\left(\frac{\|r_{\varphi}(\delta\varphi)\|_{\kappa}}{\|\delta\varphi\|_{H^{1}(\Omega)}} + \|\mathcal{U}(t,\varphi+\delta\varphi) - \mathcal{U}(t,\varphi)\|_{\boldsymbol{W}_{D}^{1,s}(\Omega)}\right) \to 0$$
as  $\|\delta\varphi\|_{H^{1}(\Omega)} \to 0$ .

As a result of (1.80) and (1.81), this yields the Fréchet-differentiability of  $\mathcal{U}(t,\cdot)$ :  $H^1(\Omega) \to W^{1,\nu}_D(\Omega)$  and  $\partial_{\varphi}\mathcal{U}(t,\varphi) = \mathcal{W}$ . In view of (1.79), the proof is now complete.

Let us now derive an estimate which will be needed in the next subsection, in the proof of Lemma 1.59 below. We observe that analogously to (1.78), Hölder's inequality with 1/2 + 1/p + 1/r = 1, where again r = 2p/(p-2), leads to

$$\|\operatorname{div}\left(g'(\varphi)(\delta\varphi)\mathbb{C}\varepsilon(\mathcal{U}(t,\varphi))\right)\|_{V^*} \leq C\|\delta\varphi\|_r \quad \forall t \in [0,T], \ \forall \varphi, \delta\varphi \in H^1(\Omega).$$

Here we used that  $H^1(\Omega) \hookrightarrow L^r(\Omega)$ , by Assumption 1.17.1. Therefore, we deduce from (1.75) and (1.6) the following

**Lemma 1.50.** Under Assumptions 1.5 and 1.17.1, there exists C > 0 so that

$$\|\partial_{\varphi}\mathcal{U}(t,\varphi)(\delta\varphi)\|_{V} \leq C \|\delta\varphi\|_{r} \quad \forall t \in [0,T], \ \forall \varphi, \delta\varphi \in H^{1}(\Omega),$$

where r = 2p/(p-2).

To prove the continuously Fréchet-differentiability of  $\mathcal{U}$ , we also have to establish

**Lemma 1.51** (Continuity of  $\partial_{\varphi}\mathcal{U}$ ). Under Assumptions 1.5 and 1.17.1 the operator  $\partial_{\varphi}\mathcal{U}: [0,T] \times H^1(\Omega) \to \mathcal{L}(H^1(\Omega),V)$  is continuous.

Proof. Let  $(t_i, \varphi_i)_{i=1,2} \in [0, T] \times H^1(\Omega)$  and  $\delta \varphi \in H^1(\Omega)$  be arbitrary, but fixed with  $\delta \varphi \neq 0$ . Further, let us abbreviate  $\mathbf{u}_i' := \partial_{\varphi} \mathcal{U}(t_i, \varphi_i) \delta \varphi$  and  $\mathbf{u}_i := \mathcal{U}(t_i, \varphi_i)$  for i = 1, 2. Moreover, define  $f_1 := A_{\varphi_2} \mathbf{u}_2' - A_{\varphi_1} \mathbf{u}_2' \in V^*$  and  $f_2 := A_{\varphi_1} \mathbf{u}_1' - A_{\varphi_2} \mathbf{u}_2' \in V^*$  such that

$$A_{\omega_1}(\mathbf{u}_1' - \mathbf{u}_2') = f_1 + f_2. \tag{1.83}$$

Our purpose is to find an appropriate estimate for  $f_1$  and  $f_2$ , respectively, which will then give the assertion in view of (1.6).

Thanks to Lemma 1.49, there is an index  $\nu \in (2, p)$  such that  $\mathcal{U}(t_2, \cdot) : H^1(\Omega) \to \mathbf{W}_D^{1,\nu}(\Omega)$  is Fréchet-differentiable. We set  $\rho = 2\nu/(\nu - 2) \in [1, \infty)$ . From Definition 1.2 and Hölder's inequality with  $1/\rho + 1/\nu + 1/2 = 1$  we then have

$$||f_1||_{V^*} \le C_1 ||g(\varphi_2) - g(\varphi_1)||_{\rho} ||u_2'||_{W_D^{1,\nu}(\Omega)} \le C_1 ||g(\varphi_2) - g(\varphi_1)||_{\rho} ||\delta\varphi||_{H^1(\Omega)},$$

where for the last inequality we used (1.75), (1.78) and (1.6) applied for  $\nu$ . Due to Lemma 5.2, this gives

$$\sup_{\substack{\delta\varphi\in H^1(\Omega)\\\delta\varphi\neq 0}} \frac{\|f_1\|_{V^*}}{\|\delta\varphi\|_{H^1(\Omega)}} \to 0 \quad \text{as } \varphi_1 \to \varphi_2 \text{ in } H^1(\Omega).$$
(1.84)

It remains to find an estimate for  $f_2$ , which will lead to a similar result. From the definitions of  $u_i$  and  $u'_i$  it follows that

$$A_{\varphi_i} \mathbf{u}_i' = \operatorname{div} \left( g'(\varphi_i)(\delta \varphi) \mathbb{C} \varepsilon(\mathbf{u}_i) \right) \quad \text{for } i = 1, 2,$$

in view of equation (1.75). This allows us to rewrite  $f_2$  as

$$f_2 = \operatorname{div} \left( g'(\varphi_1)(\delta \varphi) \mathbb{C} \varepsilon(\boldsymbol{u}_1) \right) - \operatorname{div} \left( g'(\varphi_1)(\delta \varphi) \mathbb{C} \varepsilon(\boldsymbol{u}_2) \right)$$
  
+ 
$$\operatorname{div} \left( g'(\varphi_1)(\delta \varphi) \mathbb{C} \varepsilon(\boldsymbol{u}_2) \right) - \operatorname{div} \left( g'(\varphi_2)(\delta \varphi) \mathbb{C} \varepsilon(\boldsymbol{u}_2) \right).$$

We again abbreviate r:=2p/(p-2) and as at the beginning of the proof of Lemma 1.49, we infer that there is an index  $\varrho$  such that  $r\in(2,\varrho)$  and  $H^1(\Omega)\hookrightarrow L^\varrho(\Omega)$ . Moreover, we define s via  $1/s=1/2-1/\varrho$  and observe that  $s\in(2,p)$ , as a result of  $r>\varrho$  and 1/2-1/r=1/p. By applying Hölder's inequality with  $1/\varrho+1/s+1/2=1$  and 1/r+1/p+1/2=1, one then arrives at

$$||f_{2}||_{V^{*}} \leq C_{2}||g'(\varphi_{1})(\delta\varphi)||_{\varrho}||\mathbf{u}_{1} - \mathbf{u}_{2}||_{\mathbf{W}_{D}^{1,s}(\Omega)}$$

$$+ ||(g'(\varphi_{2}) - g'(\varphi_{1}))(\delta\varphi)||_{r}||\mathbf{u}_{2}||_{\mathbf{W}_{D}^{1,p}(\Omega)}$$

$$\leq C_{2}||\delta\varphi||_{H^{1}(\Omega)} (||\mathbf{u}_{1} - \mathbf{u}_{2}||_{\mathbf{W}_{D}^{1,s}(\Omega)} + ||g'(\varphi_{2}) - g'(\varphi_{1})||_{\mathcal{L}(H^{1}(\Omega),L^{r}(\Omega))}).$$

Note that for the second inequality we used Assumption 0.6, the embedding  $H^1(\Omega) \hookrightarrow L^{\varrho}(\Omega) \hookrightarrow L^r(\Omega)$ , the fact that  $g: H^1(\Omega) \to L^r(\Omega)$  is Fréchet-differentiable, cf. Lemma 5.3, and Corollary 1.9. Lemma 1.11 and the continuity of  $g': H^1(\Omega) \to \mathcal{L}(H^1(\Omega), L^r(\Omega))$ , according to Lemma 5.3, now ensure that

$$\sup_{\substack{\delta\varphi\in H^1(\Omega)\\\delta\varphi\neq 0}} \frac{\|f_2\|_{V^*}}{\|\delta\varphi\|_{H^1(\Omega)}} \to 0 \quad \text{as } (t_1,\varphi_1) \to (t_2,\varphi_2) \text{ in } \mathbb{R} \times H^1(\Omega).$$
(1.85)

Altogether, it follows from (1.83), (1.84), (1.85) and (1.6) that

$$\sup_{\substack{\delta\varphi\in H^1(\Omega)\\\delta\varphi\neq 0}}\frac{\|\boldsymbol{u}_1'-\boldsymbol{u}_2'\|_V}{\|\delta\varphi\|_{H^1(\Omega)}}\leq C\sup_{\substack{\delta\varphi\in H^1(\Omega)\\\delta\varphi\neq 0}}\frac{\|f_1+f_2\|_{V^*}}{\|\delta\varphi\|_{H^1(\Omega)}}\to 0$$

for  $(t_1, \varphi_1) \to (t_2, \varphi_2)$  in  $\mathbb{R} \times H^1(\Omega)$ . In light of the definitions of  $u'_1$  and  $u'_2$ , this completes the proof.

Remark 1.52. We point out that it was crucial that the partial derivative of  $\mathcal{U}$  w.r.t.  $\varphi$  (at any point, in any direction) has higher regularity than V, in order to obtain the continuity of  $\partial_{\varphi}\mathcal{U}:[0,T]\times H^1(\Omega)\to \mathcal{L}(H^1(\Omega),V)$ . This norm gap arises in the estimate for  $||f_1||_{V^*}$  in the proof of Lemma 1.51 and is a result of the fact that g is not necessarily continuous from  $H^1(\Omega)$  to  $L^{\infty}(\Omega)$ , but to  $L^{\rho}(\Omega)$ , where  $\rho\in[1,\infty)$ . However, by choosing  $\rho$  large enough, one can show that there is an index  $\xi\in(2,\nu)$ , i.e.,  $\xi$  defined via  $1/\xi=1/\rho+1/\nu\in(1/\nu,1/2)$ , so that  $||f_1||_{V^*}$  can be replaced by  $||f_1||_{\mathbf{W}^{-1,\xi}(\Omega)}$  in (1.84). Note that  $\nu\in(2,p)$  denotes here the index from Lemma 1.49. With the exact same arguments as in the proof of Lemma 1.51, one then estimates  $||f_2||_{\mathbf{W}^{-1,\xi}(\Omega)}$  correspondingly, which gives in turn that  $\partial_{\varphi}\mathcal{U}$  is continuous as a mapping from  $[0,T]\times H^1(\Omega)$  to  $\mathcal{L}(H^1(\Omega),\mathbf{W}_D^{1,\xi}(\Omega))$ , with  $\xi\in(2,\nu)$ , where  $\nu$  is the index from Lemma 1.49.

We are now in the position to state the main result of this subsection.

**Proposition 1.53** (Fréchet-differentiability of the operator  $\mathcal{U}$ ). Under Assumptions 1.17.1 and 1.47 it holds  $\mathcal{U} \in C^1([0,T] \times H^1(\Omega); V)$ .

Proof. From Lemma 1.11 we know that  $\mathcal{U} \in C([0,T] \times H^1(\Omega); V)$ , while Lemmata 1.48, 1.49 and 1.51 tell us that  $\mathcal{U}$  possesses partial derivatives with  $\partial_t \mathcal{U} \in C([0,T] \times H^1(\Omega); V)$  and  $\partial_{\varphi} \mathcal{U} \in C([0,T] \times H^1(\Omega); \mathcal{L}(H^1(\Omega), V))$ , respectively. Hence, we can apply [9, Theorem 3.7.1], which gives  $\mathcal{U} \in C^1((0,T) \times H^1(\Omega); V)$ . Note that, for every  $\varphi \in H^1(\Omega)$ , the derivative of  $\mathcal{U}$  can be continuously extended to  $(0,\varphi)$  and  $(T,\varphi)$ , due to the continuity of  $\partial_t \mathcal{U}$  and  $\partial_{\varphi} \mathcal{U}$  at  $(0,\varphi)$  and  $(T,\varphi)$ , respectively.

Remark 1.54. The result in Proposition 1.53 is not optimal, since actually it holds  $\mathcal{U} \in C^1([0,T] \times H^1(\Omega); \mathbf{W}_D^{1,\xi}(\Omega))$  with an index  $\xi \in (2,\nu)$ , where  $\nu$  is given by Lemma 1.49. To see this, first note that  $\mathcal{U} : [0,T] \times H^1(\Omega) \to \mathbf{W}_D^{1,p}(\Omega)$  is differentiable w.r.t. time, see proof of Lemma 1.48. Moreover, the convergence (1.73) holds true in  $\mathbf{W}_D^{1,\nu}(\Omega)$  as well, by Lemma 1.11, since  $\nu \in (2,p)$ . Hence,  $\partial_t \mathcal{U} \in C([0,T] \times H^1(\Omega); \mathbf{W}_D^{1,\nu}(\Omega))$ . In addition, cf. Remark 1.52,  $\partial_{\varphi}\mathcal{U} : [0,T] \times H^1(\Omega) \to \mathcal{L}(H^1(\Omega), \mathbf{W}_D^{1,\xi}(\Omega))$  is continuous, and thus, by the same arguments as in the proof of Proposition 1.53, we deduce that  $\mathcal{U}$  belongs indeed to  $C^1([0,T] \times H^1(\Omega); \mathbf{W}_D^{1,\xi}(\Omega))$ .

Working with these integrability exponents would lead to a lack of overview in the upcoming analysis. As Proposition 1.53 proves to be sufficient for differentiating  $\Phi$  (without norm gap), we do not employ in what follows the above mentioned improved differentiability results.

### 1.3.2 Differentiability of $\Phi$

Although it is possible to differentiate the operator  $\Phi$  by means of the same standard arguments used for differentiating  $\mathcal{U}$ , we choose not to do so, in view of the more complicated structure of the equation for the nonlocal damage, see (1.24) and (1.25b). Instead, we employ the implicit function theorem, see e.g. [84, Theorem 4.B(d)].

For this purpose, we introduce the following

**Definition 1.55.** Let Assumptions 1.5 and 1.14 be fulfilled. We define the mapping  $\Psi : [0,T] \times L^2(\Omega) \times H^1(\Omega) \to H^1(\Omega)^*$  by  $\Psi(t,d,\varphi) := B\varphi + F(t,\varphi) - \beta d$ . Note that B and F are given by Definition 1.15.

By construction, it holds  $\Psi(t, d, \Phi(t, d)) = 0$  for all  $(t, d) \in [0, T] \times L^2(\Omega)$ , in light of Definition 1.24. Our goal is to prove that  $\Psi$  satisfies all the premises of the implicit function theorem. First we focus on showing its continuously Fréchet-differentiability. To this end, we have to differentiate the Nemytskii operator g', in view of Definition 1.55. Therefor we need the following

**Assumption 1.56.** In addition to Assumption 0.6, we require that  $g \in C^2(\mathbb{R})$ .

The differentiability of the nonlinear part of  $\Psi$  is covered by

**Lemma 1.57** (Fréchet-differentiability of F). Let Assumptions 1.17.1, 1.47 and 1.56 hold. Then the function  $F: [0,T] \times H^1(\Omega) \to H^1(\Omega)^*$  is continuously Fréchet-differentiable. Its derivative at  $(t,\varphi) \in [0,T] \times H^1(\Omega)$  in direction  $(\delta t, \delta \varphi) \in \mathbb{R} \times H^1(\Omega)$  is given by

$$\langle F'(t,\varphi)(\delta t,\delta\varphi),z\rangle_{H^{1}(\Omega)} = \frac{1}{2} \int_{\Omega} g''(\varphi)(\delta\varphi)\mathbb{C}\varepsilon(\mathcal{U}(t,\varphi)) : \varepsilon(\mathcal{U}(t,\varphi))z \,dx + \int_{\Omega} g'(\varphi)\mathbb{C}\varepsilon(\mathcal{U}(t,\varphi)) : \varepsilon(\mathcal{U}'(t,\varphi)(\delta t,\delta\varphi))z \,dx \quad \forall z \in H^{1}(\Omega).$$

$$(1.86)$$

*Proof.* We begin by writing F as a product, where one of the factors is a product as well. We then prove the result in two steps, by applying Lemma 5.4 for these. For this purpose, one defines

$$\mathcal{H}: (0,T) \times H^1(\Omega) \to L^{p/2}(\Omega), \quad \mathcal{H}(t,\varphi) := \mathbb{C}\varepsilon(\mathcal{U}(t,\varphi)) : \varepsilon(\mathcal{U}(t,\varphi))$$
 (1.87)

and

$$P_{1}: L^{\infty}(\Omega) \times L^{p/2}(\Omega) \to H^{1}(\Omega)^{*},$$

$$\langle P_{1}(y_{1}, y_{2}), z \rangle_{H^{1}(\Omega)} := \frac{1}{2} \int_{\Omega} y_{1} \cdot y_{2} \cdot z \, dx \quad \forall z \in H^{1}(\Omega),$$
(1.88)

such that

$$F:(t,\varphi)\mapsto P_1(g'(\varphi),\mathcal{H}(t,\varphi)).$$
 (1.89)

With a little abuse of notation, the mapping  $P_1$  considered with different domains will be denoted by the same symbol. Notice that Hölder's inequality ensures that  $\mathcal{H}$  and  $P_1$  are indeed well defined, on account of Definition 1.8 and since  $H^1(\Omega) \hookrightarrow L^{2p/(p-2)}(\Omega)$  by Assumption 1.17.1, respectively.

One now proves the assertion by first applying the product rule from Lemma 5.4 to  $\mathcal{H}$  in form (1.92), see below (step (i)), and then to F in form (1.89) (step (ii)).

(i) We first show that  $\mathcal{H}$  is continuous as an operator with range in  $L^{\omega}(\Omega)$  and continuously Fréchet-differentiable, if considered as an operator with range in  $L^{\varrho}(\Omega)$ . Here  $\omega$  and  $\varrho$  are defined through

$$\frac{1}{\omega} = \frac{1}{p} + \frac{1}{s}$$
 and  $\frac{1}{\rho} = \frac{1}{2} + \frac{1}{s}$ , (1.90)

where  $s \in (2, p)$  is arbitrary, but fixed. Notice that due to the latter one, there holds  $1 < \varrho < \omega < p/2$ . Concerning the continuity, we estimate similarly to (1.30) by using (1.90), which gives for all  $(t_i, \varphi_i)_{i=1,2} \in (0, T) \times H^1(\Omega)$ 

$$\|\mathcal{H}(t_1,\varphi_1) - \mathcal{H}(t_2,\varphi_2)\|_{\omega} \leq C \|\mathcal{U}(t_1,\varphi_1) + \mathcal{U}(t_2,\varphi_2)\|_{\mathbf{W}_{D}^{1,p}(\Omega)} \|\mathcal{U}(t_1,\varphi_1) - \mathcal{U}(t_2,\varphi_2)\|_{\mathbf{W}_{D}^{1,s}(\Omega)}.$$

Lemma 1.11 in combination with Corollary 1.9 then imply the desired continuity of  $\mathcal{H}$ . To prove the differentiability, consider the mapping

$$P_2: \mathbf{W}_D^{1,s}(\Omega) \times V \to L^{\varrho}(\Omega), \quad P_2(\mathbf{u}, \mathbf{v}) := \mathbb{C}\varepsilon(\mathbf{u}) : \varepsilon(\mathbf{v}),$$
 (1.91)

such that

$$\mathcal{H}: (t,\varphi) \mapsto P_2(\mathcal{U}(t,\varphi), \mathcal{U}(t,\varphi)).$$
 (1.92)

With (1.90) and (1.17) we infer that  $P_2$  is a well defined product. In light of Lemma 5.4, we set

$$U := (0,T) \times H^1(\Omega), \quad X := \mathbb{R} \times H^1(\Omega), \quad W := L^{\varrho}(\Omega),$$
  
 $P := P_2, \quad f_i := \mathcal{U}, \quad Y_i := \mathbf{W}_D^{1,s}(\Omega), \quad Z_i := V, \quad i = 1, 2.$ 

From Lemma 1.11 and Proposition 1.53 we know that  $\mathcal{U}:(0,T)\times H^1(\Omega)\to \boldsymbol{W}_D^{1,s}(\Omega)$  is continuous and  $\mathcal{U}:(0,T)\times H^1(\Omega)\to V$  is continuously Fréchet-differentiable, respectively. Hence, we can apply Lemma 5.4 to (1.92) giving in turn that  $\mathcal{H}:(0,T)\times H^1(\Omega)\to L^\varrho(\Omega)$  is continuously Fréchet-differentiable with

$$\mathcal{H}'(t,\varphi)(\delta t,\delta\varphi) = 2\mathbb{C}\varepsilon(\mathcal{U}(t,\varphi)) : \varepsilon(\mathcal{U}'(t,\varphi)(\delta t,\delta\varphi))$$
(1.93)

for all  $(t, \varphi) \in (0, T) \times H^1(\Omega)$  and all  $(\delta t, \delta \varphi) \in \mathbb{R} \times H^1(\Omega)$ .

(ii) The result from the previous step allows us now to prove the continuously Fréchetdifferentiability of F. We again apply the product rule from Lemma 5.4, this time to (1.89). To fix the setting, let  $\kappa > 2p/(p-2)$  satisfy

$$H^1(\Omega) \hookrightarrow L^{\kappa}(\Omega),$$
 (1.94)

e.g. let  $\kappa := (2p+1)/(p-2)$  for N=2 and  $\kappa := 6$  if N=3, see also the beginning of the proof of Lemma 1.20. Since the interval  $(1/p, 1/2 - 1/\kappa)$  is not empty, the number  $s \in (2, p)$  from step (i) can be chosen such that

$$\frac{1}{p} < \frac{1}{s} < \frac{1}{2} - \frac{1}{\kappa} < \frac{1}{2},\tag{1.95}$$

which is assumed in the following. Given s and  $\kappa$ , we then define  $\rho$  and  $\tau$  via

$$\frac{1}{\rho} = \frac{1}{2} - \frac{1}{s} - \frac{1}{\kappa} \quad \text{and} \quad \frac{1}{\tau} = \frac{1}{2} - \frac{1}{p}, \text{ respectively.}$$
 (1.96)

Because of (1.95), these indices satisfy

$$1 \le \rho < \infty \quad \text{and} \quad 1 \le \tau < \kappa.$$
 (1.97)

In view of Lemma 5.4, we then set

$$\begin{split} U := (0,T) \times H^1(\Omega), \quad X := \mathbb{R} \times H^1(\Omega), \quad W := H^1(\Omega)^*, \\ P = P_1, \quad f_1 := g', \quad Y_1 := L^{\rho}(\Omega), \quad Z_1 := L^{\tau}(\Omega), \\ f_2 := \mathcal{H}, \quad Y_2 := L^{\omega}(\Omega), \quad Z_2 := L^{\varrho}(\Omega), \end{split}$$

where we considered g' with domain U, with a little abuse of notation. From step (i) we already know that  $f_2 = \mathcal{H}$  fulfills the required continuity and differentiability conditions. Moreover, due to (1.97) and (1.94), Assumption 1.56 together with Lemmata 5.2 and 5.3 yields that  $f_1 = g'$  is continuous from  $H^1(\Omega)$  to  $L^{\rho}(\Omega)$  and continuously Fréchet-differentiable from  $H^1(\Omega)$  to  $L^{\tau}(\Omega)$ . Further, the mapping  $P_1$  from (1.88) considered with domains  $L^{\tau}(\Omega) \times L^{\omega}(\Omega)$  and  $L^{\rho}(\Omega) \times L^{\varrho}(\Omega)$  is well defined, in view of Hölder's inequality applied with  $1/\tau + 1/\omega + 1/\kappa < 1$  and  $1/\rho + 1/\varrho + 1/\kappa = 1$ , respectively. Note that therefor we used (1.94), (1.95) and (1.96) combined with (1.90), as well as (1.97). As a byproduct, the bilinear form  $P_1$  satisfies

$$||P_1(y_1, y_2)||_{H^1(\Omega)^*} \le C||y_1||_{\tau}||y_2||_{\omega} \quad \forall (y_1, y_2) \in L^{\tau}(\Omega) \times L^{\omega}(\Omega),$$
  
$$||P_1(y_1, y_2)||_{H^1(\Omega)^*} \le C||y_1||_{\varrho}||y_2||_{\varrho} \quad \forall (y_1, y_2) \in L^{\varrho}(\Omega) \times L^{\varrho}(\Omega),$$

and is therefore continuous on the required spaces. Thus, Lemma 5.4 gives the continuous Fréchet-differentiability of  $F:(0,T)\times H^1(\Omega)\to H^1(\Omega)^*$  and (1.86), as a result of (1.88), (1.87) and (1.93). Note that the derivative of F can be extended continuously to  $(0,\varphi)$  and  $(T,\varphi)$  for every  $\varphi\in H^1(\Omega)$  due to Lemmata 1.11, 5.2, 5.3 and Proposition 1.53. The proof is now complete.

With Lemma 1.57 at hand, the continuously Fréchet-differentiability of  $\Psi$  follows immediately from Definition 1.55, by keeping in mind that  $B \in \mathcal{L}(H^1(\Omega), H^1(\Omega)^*)$ .

Corollary 1.58 (Fréchet-differentiability of  $\Psi$ ). Under Assumptions 1.17.1, 1.47 and 1.56 it holds  $\Psi \in C^1([0,T] \times L^2(\Omega) \times H^1(\Omega); H^1(\Omega)^*)$ .

The last result required for the application of the implicit function theorem is the following

**Lemma 1.59.** Under Assumptions 1.5, 1.17 and 1.56 the operator  $\partial_{\varphi}\Psi(t,d,\varphi): H^1(\Omega) \to H^1(\Omega)^*$  is bijective for all  $(t,d,\varphi) \in [0,T] \times L^2(\Omega) \times H^1(\Omega)$ .

*Proof.* Throughout this proof let  $(t, d, \varphi) \in [0, T] \times L^2(\Omega) \times H^1(\Omega)$  be arbitrary, but fixed. On account of Definition 1.55 we have to show that for every  $h \in H^1(\Omega)^*$  the equation

$$B\delta\varphi + \partial_{\varphi}F(t,\varphi)\delta\varphi = h \tag{1.98}$$

admits a unique solution  $\delta \varphi \in H^1(\Omega)$ . Clearly, when deriving  $\Psi$  w.r.t.  $\varphi$ , one does not need Assumption 1.47, as this was above necessary only when employing the differentiability of  $\mathcal{U}$  with respect to time. We prove the unique solvability of (1.98) by means of Lax-Milgram's lemma. Thanks to  $B, \partial_{\varphi} F(t, \varphi) \in \mathcal{L}(H^1(\Omega), H^1(\Omega)^*)$  one obtains

$$|\langle B\delta\varphi + \partial_{\varphi}F(t,\varphi)(\delta\varphi), z\rangle_{H^{1}(\Omega)}| \leq C||\delta\varphi||_{H^{1}(\Omega)}||z||_{H^{1}(\Omega)} \quad \forall \, \delta\varphi, z \in H^{1}(\Omega), \tag{1.99}$$

whence the boundedness of the bilinear form  $B + \partial_{\varphi} F(t,\varphi) : H^1(\Omega) \times H^1(\Omega) \to \mathbb{R}$ . Note that (at first glance) C > 0 depends on  $(t,\varphi)$ , which in this context is not problematic. Anyway, an estimate similar to (1.100) below shows that C actually depends only on the given data. In order to prove the coercivity of  $B + \partial_{\varphi} F(t,\varphi)$ , we follow the ideas of the proof of Lemma 1.21. From (1.86) we read

$$\langle \partial_{\varphi} F(t,\varphi)z, z \rangle_{H^{1}(\Omega)} = \frac{1}{2} \int_{\Omega} g''(\varphi) z \mathbb{C} \varepsilon (\mathcal{U}(t,\varphi)) : \varepsilon (\mathcal{U}(t,\varphi)) z \, dx$$
$$+ \int_{\Omega} g'(\varphi) \mathbb{C} \varepsilon (\mathcal{U}(t,\varphi)) : \varepsilon (\partial_{\varphi} \mathcal{U}(t,\varphi)(z)) z \, dx \quad \forall z \in H^{1}(\Omega).$$

As frequently mentioned before, Assumption 1.17.1 guarantees that  $H^1(\Omega) \hookrightarrow L^r(\Omega)$  with r := 2p/(p-2), see also (1.32). Thus, Hölder's inequality with 1/r + 2/p + 1/r = 1 and 1/p + 1/2 + 1/r = 1, respectively, leads to

$$|\langle \partial_{\varphi} F(t,\varphi)z, z \rangle_{H^{1}(\Omega)}| \leq C(\|g''(\varphi)\|_{\infty} \|z\|_{r} \|\mathbb{C}\varepsilon(\mathcal{U}(t,\varphi)) : \varepsilon(\mathcal{U}(t,\varphi))\|_{\frac{p}{2}} \|z\|_{r}$$

$$+ \|g'(\varphi)\|_{\infty} \|\varepsilon(\mathcal{U}(t,\varphi))\|_{p} \|\varepsilon(\partial_{\varphi}\mathcal{U}(t,\varphi)(z))\|_{2} \|z\|_{r})$$

$$\leq C \|z\|_{r}^{2} \quad \text{for all } z \in H^{1}(\Omega),$$

$$(1.100)$$

where for the first estimate we also used Lemma 5.1. Notice that the second estimate follows from  $g', g'' \in L^{\infty}(\mathbb{R})$ , cf. Assumptions 0.6 and 1.56, respectively, Corollary 1.9, and Lemma 1.50. By Lemma 1.20, (1.100) can be continued as follows

$$|\langle \partial_{\varphi} F(t,\varphi)z, z \rangle_{H^{1}(\Omega)}| \leq k \|z\|_{2}^{2} + \widetilde{c}(k) \|z\|_{H^{1}(\Omega)}^{2} \quad \forall z \in H^{1}(\Omega) \text{ and } \forall k > 0,$$
 (1.101)

where  $\tilde{c}: \mathbb{R}^+ \to \mathbb{R}^+$  depends on C from (1.100), p, and N, and satisfies  $\tilde{c}(k) \searrow 0$  as  $k \nearrow \infty$ . Note that due to Remark 1.27,  $\tilde{c}$  depends only on the given data. Further, (1.101) and the definition of B in (1.22) imply for all k > 0 that

$$\langle Bz + \partial_{\varphi} F(t,\varphi)z, z \rangle_{H^{1}(\Omega)} \geq (\alpha - \widetilde{c}(k)) \|z\|_{H^{1}(\Omega)}^{2} + (\beta - \alpha - k) \|z\|_{2}^{2} \quad \forall z \in H^{1}(\Omega).$$

Taking into account the characteristics of  $\widetilde{c}$ , we can choose k > 0 sufficiently large such that  $\alpha - \widetilde{c}(k) \ge \alpha/2$ . Moreover, if we require

$$\beta \ge \alpha + k,\tag{1.102}$$

cf. Assumption 1.17.2, we finally arrive at

$$\langle Bz + \partial_{\varphi} F(t, \varphi)z, z \rangle_{H^1(\Omega)} \ge \alpha/2 \|z\|_{H^1(\Omega)}^2 \quad \forall z \in H^1(\Omega),$$
 (1.103)

i.e., the coercivity of  $B + \partial_{\varphi} F(t, \varphi)$ . In view of (1.99) and (1.103), Lax-Milgram's lemma now gives the unique solvability of (1.98), whence the desired assertion.

We now have all the necessary tools to state the main result of this subsection.

**Proposition 1.60** (Fréchet-differentiability of the operator  $\Phi$ ). Let Assumptions 1.17, 1.47 and 1.56 hold true. Then  $\Phi \in C^1([0,T] \times L^2(\Omega); H^1(\Omega))$  and its derivative at  $(t,d) \in [0,T] \times L^2(\Omega)$  in direction  $(\delta t, \delta d) \in \mathbb{R} \times L^2(\Omega)$  satisfies

$$B\Phi'(t,d)(\delta t,\delta d) + F'(t,\varphi)(\delta t,\Phi'(t,d)(\delta t,\delta d)) = \beta \delta d, \qquad (1.104)$$

where we abbreviate  $\varphi := \Phi(t, d)$ .

Proof. Let  $(t,d) \in (0,T) \times L^2(\Omega)$  be arbitrary, but fixed and  $\varphi := \Phi(t,d)$ . We apply the implicit function theorem to  $\Psi$  given by Definition 1.55, cf. e.g. [84, Theorem 4.B(d)]. Due to Corollary 1.58 and Lemma 1.59,  $\Psi : (0,T) \times L^2(\Omega) \times H^1(\Omega) \to H^1(\Omega)^*$  is continuously Fréchet-differentiable and  $\partial_{\varphi}\Psi(t,d,\varphi) : H^1(\Omega) \to H^1(\Omega)^*$  is bijective. Thus, the implicit function theorem is applicable. This implies that  $\Phi$  is continuously Fréchet-differentiable from  $(0,T) \times L^2(\Omega)$  to  $H^1(\Omega)$ . Moreover, its derivative is given by

$$\Phi'(t,d)(\delta t,\delta d) = -[\partial_{\varphi}\Psi(t,d,\varphi)]^{-1}\partial_{(t,d)}\Psi(t,d,\varphi)(\delta t,\delta d) \quad \forall (\delta t,\delta d) \in \mathbb{R} \times L^{2}(\Omega),$$

which is equivalent to (1.104) in view of Definition 1.55.

It remains to prove that the derivative can be continuously extended to (0,d) and (T,d) for any  $d \in L^2(\Omega)$ . From Corollary 1.58 we know that  $\partial_{(t,d)}\Psi$  and  $\partial_{\varphi}\Psi$  can be continuously extended to  $(0,d,\varphi)$ , where  $\varphi$  denotes  $\Phi(0,d)$  in the rest of the proof. Furthermore, in light of Lemma 1.59, we are allowed to define

$$\Phi'(0,d)(\delta t,\delta d):=-[\partial_{\varphi}\Psi(0,d,\varphi)]^{-1}\partial_{(t,d)}\Psi(0,d,\varphi)(\delta t,\delta d)\quad\forall\,(\delta t,\delta d)\in\mathbb{R}\times L^2(\Omega),$$

which is equivalent to (1.104) in case of t=0. Note that the inversion  $A \mapsto A^{-1}$  is continuous from the set of isomorphisms in  $\mathcal{L}(H^1(\Omega), H^1(\Omega)^*)$  to  $\mathcal{L}(H^1(\Omega)^*, H^1(\Omega))$ , see e.g. [74, Ch. III.8], which now yields the continuity of  $\Phi'$  at (0,d), in view of the continuity of  $\partial_{(t,d)}\Psi$ ,  $\partial_{\varphi}\Psi$  at  $(0,d,\varphi)$  and of  $\Phi$  at (0,d), cf. (1.38). The continuity of  $\Phi'$  at (T,d) can then be shown in the exact same way.

Remark 1.61. We point out that in two dimensions one can show, by proceeding as in Subsection 1.3.1, that  $\mathcal{U} \in C^1([0,T] \times W^{1,q}(\Omega); \mathbf{W}_D^{1,p}(\Omega))$ , with q > 2 given by Theorem 1.37. This is mainly due to the Sobolev embedding  $W^{1,q}(\Omega) \hookrightarrow L^{\infty}(\Omega)$ , the continuously Fréchet-differentiability of  $g: W^{1,q}(\Omega) \to L^{\infty}(\Omega)$ , as a result of Lemma 5.1, Assumption 0.6 and [21, Theorem 7], combined with the fact that  $\mathcal{U}: [0,T] \times W^{1,q}(\Omega) \to \mathbf{W}_D^{1,p}(\Omega)$  is Lipschitz continuous, see Remark 1.46.

If in addition to Assumption 1.56 it holds  $g'' \in C^{0,1}(\mathbb{R})$ , then  $\Phi \in C^1([0,T] \times L^2(\Omega); W^{1,q}(\Omega))$ , in case of N=2. To see this, one follows the lines of the above proofs, with the goal to apply the implicit function theorem to  $\Psi: [0,T] \times L^2(\Omega) \times W^{1,q}(\Omega) \to W^{1,q'}(\Omega)^*$ . By arguing as in the proof of Lemma 1.36, one can show that actually  $\Psi$  maps  $[0,T] \times L^2(\Omega) \times W^{1,q}(\Omega)$  to  $W^{1,\varrho'}(\Omega)^*$ , and thus it is well defined, as  $\varrho \geq q$ , see proof of Theorem 1.37. A closer inspection of the proof of Lemma 1.57 then shows that  $F \in C^1([0,T] \times W^{1,q}(\Omega); W^{1,q'}(\Omega)^*)$ , due to  $\mathcal{H} \in C^1([0,T] \times W^{1,q}(\Omega); L^{p/2}(\Omega))$  and since  $g': W^{1,q}(\Omega) \to L^{\infty}(\Omega)$  is continuously Fréchet-differentiable, by assumption and [21, Theorem 7]. Thus,  $\Psi \in C^1([0,T] \times L^2(\Omega) \times W^{1,q}(\Omega); W^{1,q'}(\Omega)^*)$ , as  $B \in \mathcal{L}(W^{1,q}(\Omega), W^{1,q'}(\Omega)^*)$  and  $L^2(\Omega) \hookrightarrow W^{1,\varrho'}(\Omega)^* \hookrightarrow W^{1,q'}(\Omega)^*$ , see (1.52). Further, we know from Lemma 1.59 that  $\partial_{\varphi}\Psi(t,d,\varphi): H^1(\Omega) \to W^{1,q'}(\Omega)^*$  is bijective for all  $(t,d,\varphi) \in [0,T] \times L^2(\Omega) \times W^{1,q}(\Omega)$ . Now, to ensure that  $\partial_{\varphi}\Psi(t,d,\varphi)^{-1}$  maps  $W^{1,q'}(\Omega)^*$  to  $W^{1,q}(\Omega)$ , we rely again on a boot strapping argument similar to Subsection 1.2.2. In each boot strapping step, the regularity of the solution of (1.98) (with right-hand side in  $W^{1,q'}(\Omega)^*$ ) improves, and as in Subsection 1.2.2, the desired regularity is achieved in maximal two steps. With all these results at hand, one can prove Proposition 1.60 in this new setting completely analogously.

We end this chapter with a final theorem on the regularity of the solution of the viscous penalized damage model:

**Theorem 1.62.** Let Assumptions 1.17, 1.47 and 1.56 be fulfilled. Then there exists a unique solution  $(\boldsymbol{u}, \varphi, d)$  of the problem (P), satisfying  $d \in C^{1,1}([0,T]; L^2(\Omega)), \varphi \in C^{0,1}([0,T]; W^{1,q}(\Omega)) \cap C^1([0,T]; H^1(\Omega)), \boldsymbol{u} \in C([0,T]; W^{1,s}_D(\Omega)) \cap C^1([0,T]; V),$  with q > 2 given by Theorem 1.37 and  $s \in (2,p)$ . Moreover, this satisfies the system of differential equations (1.48).

Proof. At the end of Section 1.1 we already established that the unique solution of (P) satisfies (1.48), as well as a first regularity result, and in particular the regularity of the local damage d, see also Theorem 1.32. We recall that the optimal nonlocal damage and the optimal displacement are given by  $\varphi = \Phi(\cdot, d(\cdot))$  and  $\boldsymbol{u} = \mathcal{U}(\cdot, \varphi(\cdot))$ , respectively. The additional regularity for  $\varphi$  follows from Theorem 1.45, the regularity of d and Proposition 1.60 combined with chain rule, while for  $\boldsymbol{u}$  we employ Lemma 1.11, the fact that  $\varphi \in C([0,T];H^1(\Omega)) \cap C^1((0,T);H^1(\Omega))$ , and Proposition 1.53, as well as chain rule. The resulting derivatives can be continuously extended to t=0 and t=T, as d does so, and in view of the continuity of d,  $\varphi$ , and Propositions 1.60, 1.53. Therewith all the desired assertions are proven.

**Remark 1.63.** From Remark 1.61 and with arguments similar to the proof of Theorem 1.62 we infer that, in case of N=2, the unique solution  $(\boldsymbol{u},\varphi,d)$  of the problem (P) satisfies  $d\in C^{1,1}([0,T];L^2(\Omega)), \ \varphi\in C^1([0,T];W^{1,q}(\Omega)), \ with \ q>2$  given by Theorem 1.37, and  $\boldsymbol{u}\in C^1([0,T];\boldsymbol{W}^{1,p}_D(\Omega)),$  provided that  $g''\in C^{0,1}(\mathbb{R})$  holds true.

## Chapter 2

# Penalization limit

In this chapter we establish the viability of the penalty approach. To be more precise, we study the behaviour of the penalized damage model for  $\beta$  approaching  $\infty$  and show that in the limit both damage variables coincide. Moreover, we prove that the resulting limit model is in accordance with a class of classical damage models introduced in [17]. For convenience of the reader, we recall here the damage model with penalty

$$\left\{
\begin{aligned}
(\boldsymbol{u}(t), \varphi(t)) &\in \underset{(\boldsymbol{u}, \varphi) \in V \times H^{1}(\Omega)}{\operatorname{arg \, min}} \mathcal{E}(t, \boldsymbol{u}, \varphi, d(t)), \\
0 &\in \partial \mathcal{R}_{\delta}(\dot{d}(t)) + \partial_{d} \mathcal{E}(t, \boldsymbol{u}(t), \varphi(t), d(t)), \quad d(0) = d_{0}
\end{aligned}
\right\}$$
(P<sub>\beta</sub>)

for all  $t \in [0, T]$ . This was investigated regarding its unique solvability in the previous chapter, in the context where the penalization parameter was fixed. To emphasize the dependency on the penalty term of its unique solution, we denote this throughout this chapter by  $(\mathbf{u}_{\beta}, \varphi_{\beta}, d_{\beta})$ , unless otherwise specified.

The passage to the limit  $\beta \to \infty$  is performed by means of an equivalent reformulation of  $(P_{\beta})$  in terms of an energy identity. The latter one plays an essential role in the present chapter, as it not only ensures the convergence of the penalized solutions, but it is also crucial for deriving the energy inequality which characterizes the limit damage variable. In combination with well known convex analysis results, this allows for deriving a single-field gradient damage model, that can be ultimately transformed into a classical partial damage model. To make sure that these two viscous damage models are compatible with each other, we also show, by employing a time-discretization technique for  $(P_{\beta})$ , that the limit damage variable is bounded a.e. in  $(0,T) \times \Omega$ .

The energy inequality, often met in the form of an identity, plays a crucial role in the context of rate-independent damage models, as various notions of solutions are based on it, such as for example energetic and local solutions, see [57] for an overview. It is also an essential tool for the vanishing viscosity limit analysis as performed in [41]. The energy inequality turns out to be very useful also in the context of rate-dependent damage models when it comes to proving existence of viscous solutions. We mention here the contributions of [41] (classical viscous solutions for a gradient damage model), [42] (weak viscous solutions) and [26] (weak viscous solutions for Cahn-Larché systems coupled

with damage). In all these contributions, the existence of viscous solutions is shown by passing to the limit in a time-discretization scheme coupled with (a discrete version of) the energy inequality. An alternative approach consists of regularizing the viscous damage model, see [40]. The resulting (weak) solutions are then described in terms of an energy inequality. Our final result shows that the existence of viscous solutions may also be obtained via penalization. In other words, we show the existence of solutions (for a version of the viscous model in [41]) by following a similar approach, but unlike in the above mentioned contributions, we do not regularize nor discretize in time, but penalize.

### Outline of the chapter

Within our scope of passing to the limit  $\beta \to \infty$  in  $(P_{\beta})$ , we derive in Section 2.1 an equivalent formulation of the evolution in  $(P_{\beta})$ , namely the energy identity. This will be the starting point for the limit analysis, which is performed in Section 2.2 in two steps. We first prove that the solution of  $(P_{\beta})$  is bounded by a constant independent of  $\beta$  in suitable spaces, which ensures the existence of the limit variables. We then pass to the limit in the elliptic system which characterizes the minimization problem in  $(P_{\beta})$ , whence we deduce that the local damage equals the nonlocal damage for  $\beta = \infty$ . We also pass to the limit in the energy identity, which results in an energy inequality that describes the evolution of the *single* limit damage variable. Based on these results, Section 2.3 deals with deriving a single-field gradient damage model in terms of an evolutionary equation and addresses the unique solvability thereof. In Section 2.4 we prove, under rather nonrestrictive assumptions, that the limit damage variable belongs to  $L^{\infty}((0,T)\times\Omega)$ . By means thereof, we establish in Section 2.5 that the single-field gradient damage model is a version of a viscous damage model analyzed in [41]. Note that throughout this chapter, Assumptions 1.17, 1.47 and 1.56 are supposed to hold true, in order to guarantee the unique solvability of  $(P_{\beta})$  and the  $C^1$ -regularity in time of the solution, cf. Theorem 1.62.

### 2.1 Energy identity

The purpose of this section is to derive a characterization of the optimal local damage, which allows us to pass to the limit in  $(P_{\beta})$  as  $\beta \to \infty$ . In Subsection 1.1.2 we have already seen that the evolutionary equation (1.39) can be equivalently expressed as an operator differential equation. There are also other various equivalent formulations thereof. These are collected in Proposition 2.7 below, where one sees that, excepting (2.13), they all feature the term  $\beta(d_{\beta} - \varphi_{\beta})$ . This provides a significant disadvantage, as  $\beta(d_{\beta} - \varphi_{\beta})$  is not necessarily uniformly bounded w.r.t.  $\beta$  in suitable spaces. Hence, in the presence thereof, a passage to the limit  $\beta \to \infty$  in the penalized damage model is not to be expected.

By contrast, the formulation (2.13), also known as *energy identity*, has the advantage of not containing the above mentioned problematic term. This is eliminated in the process of deriving (2.13), via a convex analysis result and chain rule, see (2.2b), (2.7) and

(2.10) below. Thereby one obtains an alternative description of the evolution (1.40), cf. Proposition 2.5 and Remark 2.6 below. As it will turn out, all the addends therein, and in particular those involving  $d_{\beta}$ , are uniformly bounded w.r.t. the penalization parameter, which is essential for the upcoming analysis. We refer here to the proof of Lemma 2.11 in Section 2.2 below. For this reason, the energy identity constitutes the starting point for proving the uniform boundedness w.r.t.  $\beta$  of the overall solution of  $(P_{\beta})$  in the next section. Moreover, it is the fundamental tool for studying the limit behaviour  $\beta \to \infty$  of the local damage  $d_{\beta}$ , see Subsection 2.2.3 below.

We show the energy identity by following the lines of the proof of [41, Proposition 3.2]. This consists of two main steps: rewriting the differential inclusion, in our case (1.40), as a Fenchel identity and using chain rule to substitute the left-hand side in (1.40). For the latter one, we first need to introduce the reduced energy functional and investigate its differentiability properties. We recall that, for a fixed pair  $(t, d) \in [0, T] \times L^2(\Omega)$ , the functional  $\mathcal{E}$  admits a unique global minimizer on  $V \times H^1(\Omega)$ , namely  $(\mathcal{U}(t, \Phi(t, d)), \Phi(t, d))$ , cf. Theorem 1.23. This gives rise to the following

**Definition 2.1** (Reduced energy functional). Let Assumptions 1.5 and 1.17 hold. Then the reduced energy functional  $\mathcal{I}: [0,T] \times L^2(\Omega) \to \mathbb{R}$  is given by

$$\mathcal{I}(t,d) := \mathcal{E}(t,\mathcal{U}(t,\Phi(t,d)),\Phi(t,d),d).$$

On account of Definitions 0.2, 1.2 and 1.8 we can rewrite the reduced energy functional at all  $(t,d) \in [0,T] \times L^2(\Omega)$  as

$$\mathcal{I}(t,d) = \frac{1}{2} \langle A_{\Phi(t,d)} \big( \mathcal{U}(t,\Phi(t,d)) \big), \mathcal{U}(t,\Phi(t,d)) \rangle_{V} - \langle \ell(t), \mathcal{U}(t,\Phi(t,d)) \rangle_{V} 
+ \frac{\alpha}{2} \| \nabla \Phi(t,d) \|_{2}^{2} + \frac{\beta}{2} \| \Phi(t,d) - d \|_{2}^{2} 
= -\frac{1}{2} \langle \ell(t), \mathcal{U}(t,\Phi(t,d)) \rangle_{V} + \frac{\alpha}{2} \| \nabla \Phi(t,d) \|_{2}^{2} + \frac{\beta}{2} \| \Phi(t,d) - d \|_{2}^{2},$$
(2.1)

which will come in handy in the sequel. In particular, this reformulation allows us to establish

**Lemma 2.2** (Fréchet-differentiability of  $\mathcal{I}$ ). Under Assumptions 1.17, 1.47 and 1.56, it holds  $\mathcal{I} \in C^1([0,T] \times L^2(\Omega))$  with

$$\partial_t \mathcal{I}(t,d) = -\langle \dot{\ell}(t), \mathcal{U}(t,\Phi(t,d)) \rangle_V,$$
 (2.2a)

$$\partial_d \mathcal{I}(t,d) = \beta(d - \Phi(t,d))$$
 (2.2b)

at all  $(t,d) \in [0,T] \times L^2(\Omega)$ .

*Proof.* First note that the mapping

$$f:[0,T]\times L^2(\Omega)\to\mathbb{R},\quad f(t,d):=\langle \ell(t),\mathcal{U}(t,\Phi(t,d))\rangle_V$$

can be seen as product of the functions  $[0,T] \ni t \mapsto \ell(t) \in V^*$  and  $[0,T] \times L^2(\Omega) \ni (t,d) \mapsto \mathcal{U}(t,\Phi(t,d)) \in V$ . The latter one is continuously Fréchet-differentiable, thanks

to Propositions 1.53 and 1.60. Together with Assumption 1.47 we obtain  $f \in C^1([0,T] \times L^2(\Omega))$ , as a result of product rule. From the continuous Fréchet-differentiability of  $\|\cdot\|_2^2: L^2(\Omega) \to \mathbb{R}$  and  $\nabla: H^1(\Omega) \to L^2(\Omega; \mathbb{R}^N)$ , combined with Proposition 1.60, we can now deduce in view of (2.1) that  $\mathcal{I} \in C^1([0,T] \times L^2(\Omega))$ . Notice that the derivative of  $\mathcal{I}$  can be continuously extended to (0,d) and (T,d) for any  $d \in L^2(\Omega)$ , as  $\Phi'$  does so and since  $\mathcal{U}'$  can be continuously extended to  $(0,\varphi)$  and  $(T,\varphi)$  for any  $\varphi \in H^1(\Omega)$ , cf. Proposition 1.53.

To prove (2.2), let  $(t,d) \in [0,T] \times L^2(\Omega)$  be arbitrary, but fixed. By product and chain rule we obtain from (2.1) for all  $(\delta t, \delta d) \in \mathbb{R} \times L^2(\Omega)$ 

$$\mathcal{I}'(t,d)(\delta t,\delta d) = -\frac{1}{2} \langle \dot{\ell}(t)\delta t, \mathcal{U}(t,\Phi(t,d)) \rangle_{V} - \frac{1}{2} \langle \ell(t), \mathcal{U}'(t,\Phi(t,d))(\delta t,\delta\varphi) \rangle_{V}$$

$$+ \alpha (\nabla \Phi(t,d), \nabla \delta \varphi)_{2} + \beta (\Phi(t,d) - d, \delta \varphi - \delta d)_{2},$$
(2.3)

where we abbreviate  $\delta \varphi = \Phi'(t, d)(\delta t, \delta d)$ . In view of (1.72) tested with  $\mathcal{U}(t, \Phi(t, d))$  and Definitions 1.2 and 1.8 we have

$$\langle \dot{\ell}(t), \mathcal{U}(t, \Phi(t, d)) \rangle_{V} = \langle A_{\Phi(t, d)} \partial_{t} \mathcal{U}(t, \Phi(t, d)), \mathcal{U}(t, \Phi(t, d)) \rangle_{V} = \langle \ell(t), \partial_{t} \mathcal{U}(t, \Phi(t, d)) \rangle_{V},$$

whence

$$-\frac{1}{2}\langle\dot{\ell}(t),\mathcal{U}(t,\Phi(t,d))\rangle_{V} - \frac{1}{2}\langle\ell(t),\partial_{t}\mathcal{U}(t,\Phi(t,d))\rangle_{V} = -\langle\dot{\ell}(t),\mathcal{U}(t,\Phi(t,d))\rangle_{V}. \tag{2.4}$$

Relying on Definition 1.8 and (1.75) tested with  $\mathcal{U}(t,\Phi(t,d)) \in V$ , we further obtain

$$-\frac{1}{2}\langle \ell(t), \partial_{\varphi} \mathcal{U}(t, \Phi(t, d)) \delta \varphi \rangle_{V} + \alpha (\nabla \Phi(t, d), \nabla \delta \varphi)_{2} + \beta (\Phi(t, d) - d, \delta \varphi - \delta d)_{2}$$

$$= -\frac{1}{2} \langle \operatorname{div} \left( g'(\Phi(t, d)) (\delta \varphi) \mathbb{C} \varepsilon (\mathcal{U}(t, \Phi(t, d))), \mathcal{U}(t, \Phi(t, d)) \rangle_{V} + \alpha (\nabla \Phi(t, d), \nabla \delta \varphi)_{2} \right.$$

$$+ \beta (\Phi(t, d) - d, \delta \varphi - \delta d)_{2}$$

$$= \langle F(t, \Phi(t, d)), \delta \varphi \rangle_{H^{1}(\Omega)} + \langle B \Phi(t, d), \delta \varphi \rangle_{H^{1}(\Omega)} - \beta (d, \delta \varphi)_{2} + \beta (d - \Phi(t, d), \delta d)_{2}$$

$$= \beta (d - \Phi(t, d), \delta d)_{2} \quad \forall \, \delta d \in L^{2}(\Omega),$$

$$(2.5)$$

where for the last two equalities we used Definitions 1.15 and 1.24, respectively. Inserting (2.4) and (2.5) in (2.3) finally leads to

$$\mathcal{I}'(t,d)(\delta t,\delta d) = -\langle \dot{\ell}(t)\delta t, \mathcal{U}(t,\Phi(t,d))\rangle_V + \beta(d-\Phi(t,d),\delta d)_2 \quad \forall (\delta t,\delta d) \in \mathbb{R} \times L^2(\Omega),$$
 which gives the desired assertion.

As an immediate consequence of Lemma 2.2, we have

**Corollary 2.3** (Differentiability of  $\mathcal{I}(\cdot, d(\cdot))$ ). Let  $d \in C^1([0, T]; L^2(\Omega))$  be given. Under Assumptions 1.17, 1.47 and 1.56, the map  $[0, T] \ni t \mapsto \mathcal{I}(t, d(t))$  is continuously differentiable and it holds

$$\frac{d}{dt}\mathcal{I}(t,d(t)) = \partial_t \mathcal{I}(t,d(t)) + (\partial_d \mathcal{I}(t,d(t)),\dot{d}(t))_2 \quad \forall t \in [0,T].$$

Proof. The result follows easily by chain rule, Lemma 2.2 and since the map  $h:[0,T]\ni t\mapsto (t,d(t))\in [0,T]\times L^2(\Omega)$  is continuously differentiable, in light of  $d\in C^1([0,T];L^2(\Omega))$ . From  $\dot{h}(t)=(1,\dot{d}(t))$  for all  $t\in [0,T]$  we deduce the desired identity, which completes the proof.

Note that in view of (2.2b) and Theorem 1.32, the optimal local damage d can also be characterized by

$$-\partial_d \mathcal{I}(t, d(t)) \in \partial \mathcal{R}_{\delta}(\dot{d}(t)) \quad \forall t \in [0, T], \quad d(0) = d_0. \tag{2.6}$$

The evolutionary equation (2.6) and Corollary 2.3 will be essential for proving the *energy* identity in Proposition 2.5 below. Prior to this, we state a last auxiliary result.

**Lemma 2.4.** Let Assumptions 1.5, 1.17 and 1.56 hold true. Then, for any  $t \in [0,T]$ , we have

$$\mathcal{R}_{\delta}^* \left( -\partial_d \mathcal{I}(t, d(t)) \right) = \frac{\delta}{2} ||\dot{d}(t)||_2,$$

where d is the unique solution of (2.6).

*Proof.* Note that here we can go without Assumption 1.47, since this is not needed for the existence of the derivative of  $\mathcal{I}$  w.r.t. d, see also proof of Lemma 2.2. We also observe that, due to the characterization of d via (1.41), it holds  $\dot{d}(t) \geq 0$  for all  $t \in [0, T]$ . Since  $\mathcal{R}_{\delta}$  is convex and proper, see Definition 0.4, we obtain from (2.6) by means of e.g. [69, Theorem 23.5] the following Fenchel identity

$$\mathcal{R}_{\delta}(\dot{d}(t)) + \mathcal{R}_{\delta}^{*}(-\partial_{d}\mathcal{I}(t, d(t))) = (-\partial_{d}\mathcal{I}(t, d(t)), \dot{d}(t))_{2} \quad \forall t \in [0, T].$$
 (2.7)

Using Definitions 0.1, 0.4 and (2.2b) in (2.7) leads to

$$\mathcal{R}_{\delta}^{*}\left(-\partial_{d}\mathcal{I}(t,d(t))\right) = (-\beta(d(t) - \varphi(t)), \dot{d}(t))_{2} - \mathcal{R}_{1}(\dot{d}(t)) - \frac{\delta}{2} \|\dot{d}(t)\|_{2}^{2} \quad \forall t \in [0,T], (2.8)$$

where we abbreviate  $\varphi = \Phi(\cdot, d(\cdot))$ . Finally, we deduce from (1.43a) that (2.8) can be continued as

$$\mathcal{R}_{\delta}^* \left( -\partial_d \mathcal{I}(t, d(t)) \right) = \delta \|\dot{d}(t)\|_2^2 - \frac{\delta}{2} \|\dot{d}(t)\|_2 \quad \forall t \in [0, T],$$

which gives the assertion.

The main result of this section reads as follows:

**Proposition 2.5** (The energy identity). Under Assumptions 1.17, 1.47 and 1.56, the unique solution d of (2.6), and thus of (1.39), fulfills for all  $0 \le s \le t \le T$  the energy identity

$$\int_{s}^{t} \mathcal{R}_{\delta}(\dot{d}(\tau))d\tau + \int_{s}^{t} \mathcal{R}_{\delta}^{*}(-\partial_{d}\mathcal{I}(\tau, d(\tau)))d\tau + \mathcal{I}(t, d(t)) = \mathcal{I}(s, d(s)) + \int_{s}^{t} \partial_{t}\mathcal{I}(\tau, d(\tau))d\tau$$
(2.9)

and the initial datum condition  $d(0) = d_0$ .

*Proof.* First we recall that, cf. Theorem 1.32, d belongs to  $C^{1,1}([0,T];L^2(\Omega))$ , so that Corollary 2.3 yields the continuity of  $\frac{d}{dt}\mathcal{I}(\cdot,d(\cdot))$ . In combination with (2.7), this also gives the identity

$$\mathcal{R}_{\delta}(\dot{d}(\tau)) + \mathcal{R}_{\delta}^{*}(-\partial_{d}\mathcal{I}(\tau, d(\tau))) = \partial_{t}\mathcal{I}(\tau, d(\tau)) - \frac{d}{dt}\mathcal{I}(\tau, d(\tau)) \quad \forall \tau \in [0, T].$$
 (2.10)

Since  $\dot{d} \geq 0$ , as a result of (1.41), we obtain by Definition 0.4 and  $\dot{d} \in C([0,T]; L^2(\Omega))$ , that the map  $[0,T] \ni \tau \mapsto \mathcal{R}_{\delta}(\dot{d}(\tau)) \in \mathbb{R}$  is continuous. From Lemma 2.4 and Corollary 2.3 we deduce the continuity in time of the other terms in (2.10), and therefore, the integrability thereof. From (2.10) we now arrive at

$$\int_{s}^{t} \mathcal{R}_{\delta}(\dot{d}(\tau)) d\tau + \int_{s}^{t} \mathcal{R}_{\delta}^{*}(-\partial_{d}\mathcal{I}(\tau, d(\tau))) d\tau + \mathcal{I}(t, d(t)) = \mathcal{I}(s, d(s)) + \int_{s}^{t} \partial_{t}\mathcal{I}(\tau, d(\tau)) d\tau$$

for all  $0 \le s \le t \le T$ , which completes the proof, as the initial datum condition is automatically fulfilled.

Remark 2.6. We point out that the energy identity is just another equivalent formulation of the evolution in (2.6). To be more precise, one can show as in the proof of [41, Proposition 3.2], that if  $d \in C^1([0,T];L^2(\Omega))$  satisfies (2.9) and  $d(0) = d_0$ , then d is the unique solution of (2.6). This follows by Corollary 2.3 combined with the definition of the Fenchel conjugate, i.e.  $\mathcal{R}^*_{\delta}(\xi) = \sup_{v \in L^2(\Omega)}(\xi,v)_2 - \mathcal{R}_{\delta}(v) \ \forall \xi \in L^2(\Omega)$ , which give in turn

$$\mathcal{R}_{\delta}^{*}(-\partial_{d}\mathcal{I}(t,d(t))) \geq (-\partial_{d}\mathcal{I}(t,d(t)),\dot{d}(t))_{2} - \mathcal{R}_{\delta}(\dot{d}(t)) 
= \partial_{t}\mathcal{I}(t,d(t)) - \frac{d}{dt}\mathcal{I}(t,d(t)) - \mathcal{R}_{\delta}(\dot{d}(t)) \quad \forall t \in [0,T].$$
(2.11)

Using (2.9) for s = 0 and t = T then leads to equality in (2.11) at almost all  $t \in (0, T)$ , which by convex analysis (see e.g. [69, Theorem 23.5]) is equivalent to the evolutionary equation in (2.6) f.a.a.  $t \in (0,T)$ . The latter one holds however for any  $t \in [0,T]$ , in view of the regularity of d and the equivalency of (2.6) (f.a.a.  $t \in (0,T)$ ) with (1.41), cf. Theorem 1.29. Since throughout this chapter we need only the implication stated in Proposition 2.5, we do not go here into more details.

We end this section with a result which collects the various equivalent formulations of the reduced problem  $(P_{\beta})$ , namely (1.39).

**Proposition 2.7.** Suppose that Assumptions 1.17, 1.47 and 1.56 hold true. Let  $d \in C^1([0,T];L^2(\Omega))$  and denote  $\Phi(\cdot,d(\cdot))$  by  $\varphi$ . Then, the following are equivalent:

(i) d solves the reduced problem  $(P_{\beta})$  in primal form, i.e.,

$$-\beta(d(t) - \varphi(t)) \in \partial \mathcal{R}_{\delta}(\dot{d}(t)) \Leftrightarrow -\partial_{d} \mathcal{I}(t, d(t)) \in \partial \mathcal{R}_{\delta}(\dot{d}(t)) \quad \forall t \in [0, T];$$

(ii) d solves the reduced problem  $(P_{\beta})$  in dual form, i.e.,

$$\dot{d}(t) \in \partial \mathcal{R}_{\delta}^{*} \left(-\beta(d(t) - \varphi(t))\right) \quad \forall t \in [0, T]; \tag{2.12}$$

(iii) d solves the operator differential equation

$$\dot{d}(t) = \frac{1}{\delta} \max \left( -\beta (d(t) - \varphi(t)) - r \right) \quad \forall t \in [0, T];$$

(iv) d satisfies the following complementarity system at all  $t \in [0,T]$ 

$$0 \le \dot{d}(t) \perp -\beta(d(t) - \varphi(t)) - r - \delta \dot{d}(t) \le 0$$
 a.e. in  $\Omega$ ;

(v) d fulfills for all  $0 \le s \le t \le T$  the energy identity

$$\int_{s}^{t} \mathcal{R}_{\delta}(\dot{d}(\tau))d\tau + \int_{s}^{t} \mathcal{R}_{\delta}^{*}(-\partial_{d}\mathcal{I}(\tau, d(\tau)))d\tau + \mathcal{I}(t, d(t)) = \mathcal{I}(s, d(s)) + \int_{s}^{t} \partial_{t}\mathcal{I}(\tau, d(\tau))d\tau;$$
(2.13)

(vi) for all  $t \in [0,T]$ ,  $\dot{d}(t)$  is the unique solution of the minimization problem

$$\min_{\substack{v \in L^2(\Omega) \\ v > 0}} \mathcal{R}_{\delta}(v) + (\beta(d(t) - \varphi(t)), v)_2 ; \qquad (2.14)$$

(vii) d satisfies for all  $t \in [0,T]$  the variational inequality of first and second kind

$$\delta \int_{\Omega} \dot{d}(t)(v - \dot{d}(t)) \, dx + \int_{\Omega} \beta(d(t) - \varphi(t))(v - \dot{d}(t)) \, dx$$
$$+ \mathcal{R}_{1}(v) - \mathcal{R}_{1}(\dot{d}(t)) \ge 0 \quad \forall v \in L^{2}(\Omega) : v \ge 0 \text{ a.e. in } \Omega.$$
(2.15)

*Proof.* The equivalence in (i) is due to (2.2b), as already mentioned above. The equivalence  $(i) \Leftrightarrow (ii)$  follows by convex analysis, see e.g. [69, Corollary 23.5.1], as  $\mathcal{R}_{\delta}$  is convex, lower semicontinuous and proper. Note that the lower semicontinuity is implied by the continuity of the norms, see Definition 0.4. Moreover, we have  $(i) \Leftrightarrow (iii) \Leftrightarrow (iv) \Leftrightarrow (v)$ , as a result of Theorem 1.29, the proof thereof, and Proposition 2.5 combined with Remark 2.6.

Further, the implication  $(i) \Rightarrow (vii)$  is a consequence of the equivalence of (1.40) with (1.43) at any  $t \in [0,T]$ , see the proof of Theorem 1.29. On the other hand, from (vii) one deduces by Definition 0.1 that  $\dot{d} \geq 0$ . Testing (2.15) at all  $t \in [0,T]$  with 0 and  $2\dot{d}(t)$ , respectively, yields (1.43a), whence we have (1.43b), and thus, (1.40) at all  $t \in [0,T]$ . Note that therefor we used again the definition of  $\mathcal{R}_1$ , as well as its positive homogenity. Thus, it holds  $(i) \Leftrightarrow (vii)$ . We observe that (2.15) is a VI of first and second kind, as  $\mathcal{R}_1$  restricted on its domain has linear behaviour.

It remains to prove  $(i) \Leftrightarrow (vi)$ . The fact that d(t) solves the minimization problem (2.14) for all  $t \in [0, T]$  is equivalent with (i), in light of the definition of the subdifferential. Regarding the unique solvability of (2.14) for all  $t \in [0, T]$ , we rely on Definition 0.4, which tells us that the objective functional is strictly convex. The proof is now complete.

### 2.2 Limit analysis

This section proves the viability of the penalty approach, that is, it shows that one can pass to the limit  $\beta \to \infty$  in  $(P_{\beta})$  and obtain a damage model that features only one damage variable. The latter one is nothing else as the limit of the local and nonlocal damage, respectively, as  $\beta \to \infty$ . In the next section we will see that the damage model without penalty is equivalent to a version of a classical viscous partial damage model which was analyzed by [41].

In the first part of the present section we entirely focus on finding bounds independent of  $\beta$ , in suitable spaces, for the local and nonlocal damage. Note that for the displacement such a bound is already given by Corollary 1.9. Altogether, Subsection 2.2.1 allows us to find weakly convergent subsequences. The limiting behaviour thereof, as  $\beta$  approaches  $\infty$ , is studied in Subsections 2.2.2 and 2.2.3. As it will turn out, the limit problem reduces to an *energy inequality*, which is in fact an *energy identity*, cf. Remark 2.27 below.

### 2.2.1 Uniform boundedness

Throughout this subsection  $(\boldsymbol{u}, \varphi, d)$  denotes the unique solution of the problem  $(P_{\beta})$ , that is,

$$\mathbf{u}(t) = \mathcal{U}(t, \varphi(t)) = A_{\varphi(t)}^{-1} \ell(t), \tag{2.16a}$$

$$\varphi(t) = \Phi(t, d(t)) = (B + F(t, \cdot))^{-1}(\beta d(t)), \qquad (2.16b)$$

$$\dot{d}(t) = \frac{1}{\delta} \max\left(-\beta(d(t) - \varphi(t)) - r\right), \quad d(0) = d_0 \tag{2.16c}$$

for all  $t \in [0, T]$ , cf. Theorem 1.62, see also the definitions of the operators  $\mathcal{U}$  and  $\Phi$ , as well as Definitions 1.2 and 1.15. For later purposes, it is sufficient to recall here that  $(\boldsymbol{u}, \varphi, d) \in C^1([0, T]; V) \times C^1([0, T]; H^1(\Omega)) \times C^{1,1}([0, T]; L^2(\Omega))$ . The starting point for deriving the upcoming results is the *energy identity* in Proposition 2.5. To make use thereof, we need the following additional assumption, which is rather self-evident in many practical applications.

**Assumption 2.8.** In the rest of this chapter we assume that at the beginning of the process the body is completely sound, i.e.,  $d_0 = 0$  and that there is no load acting upon the body at the initial time, i.e.,  $\ell(0) = 0$ .

Remark 2.9 (Active damage process). To ensure the dynamic of the system, i.e.,  $\dot{d} \not\equiv 0$  on [0,T], one can impose various assumptions on the coefficient function g and on the load. For example, one can show by way of contradiction that  $\dot{d} \not\equiv 0$ , if there is a time point t for which  $\ell(t) \not= 0$  and if g is monotonically decreasing on  $(-\infty,0)$  and in addition, satisfies  $g'(x) \leq -C \ \forall x \in [0,r/\beta]$ , where C>0 is a constant which can be computed depending on the given data. Let us mention that the larger the value of  $\|\ell\|_{C([0,T];V^*)}$ , the smaller the constant C, and thus, a mild assumption on g is guaranteed when one requires that the maximal value of the load is large. We also observe that with increasing g the assumption on g becomes less restrictive. Recall that the monotonicity assumption is in any case reasonable from a practical point of view.

**Remark 2.10** (Inactive damage process). We remark that if g'(0) = 0 is allowed, then the solution of  $(P_{\beta})$  is given by  $(\mathcal{U}(t,0),0,0)$  at any time point  $t \in [0,T]$ , which in particular means that even if we assume that  $\|\ell\|_{C([0,T];V^*)}$  is very large, the body remains sound the whole time.

As a first consequence of Assumption 2.8, we obtain in view of (2.16), by employing the linearity of  $A_{\varphi(0)}^{-1}$ , as well as (1.22) and (1.23), that

$$\mathbf{u}(0) = \varphi(0) = 0, \tag{2.17a}$$

$$\dot{d}(0) = 0. {(2.17b)}$$

This allows us to establish

**Lemma 2.11** (Boundedness of the local damage). Let Assumptions 1.17, 1.47, 1.56 and 2.8 hold. Then there exists a constant C > 0 independent of  $\beta$ , such that

$$||d||_{H^1(0,T;L^2(\Omega))} \le C,$$

where d is the optimal local damage associated to  $(P_{\beta})$ .

*Proof.* The result follows mainly from Proposition 2.5. In order to see this, we first set s := 0 and t := T in (2.9) and employ the fact that  $\dot{d} \geq 0$ , Definition 0.4, Lemma 2.4, (2.1) and (2.2a), as well as Assumption 2.8 and (2.17a). These yield the following identity

$$\underbrace{r \int_{0}^{T} \|\dot{d}(\tau)\|_{1} d\tau}_{\geq 0} + \underbrace{\delta \int_{0}^{T} \|\dot{d}(\tau)\|_{2}^{2} d\tau}_{\geq 0} - \frac{1}{2} \langle \ell(T), \boldsymbol{u}(T) \rangle_{V} + \underbrace{\frac{\alpha}{2} \|\nabla \varphi(T)\|_{2}^{2}}_{\geq 0} + \underbrace{\frac{\beta}{2} \|\varphi(T) - d(T)\|_{2}^{2}}_{\geq 0} \\
= \int_{0}^{T} \langle -\dot{\ell}(\tau), \boldsymbol{u}(\tau) \rangle_{V} d\tau. \tag{2.18}$$

Besides, thanks to Corollary 1.9 and Assumption 1.47, one has

$$\int_{0}^{T} \langle -\dot{\ell}(\tau), \boldsymbol{u}(\tau) \rangle_{V} d\tau + \frac{1}{2} \langle \ell(T), \boldsymbol{u}(T) \rangle_{V} \leq \int_{0}^{T} ||\dot{\ell}(\tau)||_{V^{*}} ||\boldsymbol{u}(\tau)||_{V} d\tau + \frac{1}{2} ||\ell(T)||_{V^{*}} ||\boldsymbol{u}(T)||_{V} \\
\leq C, \tag{2.19}$$

with C > 0 independent of  $\beta$ . Inserting (2.19) in (2.18) leads now to

$$\int_0^T \|\dot{d}(\tau)\|_2^2 d\tau \le C,\tag{2.20}$$

where C > 0 is a constant independent of  $\beta$ . From (2.20) and d(0) = 0, by assumption, we infer the desired assertion from Poincaré-Friedrich's inequality, see (5.39). This completes the proof.

In the rest of this subsection we focus on finding a constant C>0 independent of  $\beta$  such that the estimate

$$\|\varphi\|_{H^1(0,T;H^1(\Omega))}^2 \le C \tag{2.21}$$

holds true. In view of (2.17a) and Poincaré-Friedrich's inequality, see (5.39), we only need to show that there is C > 0 independent of  $\beta$  such that

$$\|\dot{\varphi}\|_{L^2(0,T;H^1(\Omega))}^2 = \int_0^T \|\dot{\varphi}(\tau)\|_2^2 d\tau + \int_0^T \|\nabla \dot{\varphi}(\tau)\|_2^2 d\tau \le C. \tag{2.22}$$

The starting point herefor is the equation which characterizes the time derivative of the nonlocal damage, see (2.23) and (2.24) below. By means thereof, we first bound the second addend in (2.22) by  $\|\dot{\varphi}\|_{L^2(0,T;L^2(\Omega))}$ , see Lemma 2.14 below. Relying again on (2.24), one then finds C > 0 independent of  $\beta$  such that  $\|\dot{\varphi}\|_{L^2(0,T;L^2(\Omega))}^2 \leq C$  holds, see Lemma 2.15 below. This will then have as immediate consequence (2.22), and thus (2.21).

Since  $\varphi = \Phi(\cdot, d(\cdot))$ , we get by Proposition 1.60 and chain rule that the (time) derivative of  $\varphi$  satisfies at all  $t \in [0, T]$ 

$$B\dot{\varphi}(t) + \frac{d}{dt}F(t,\varphi(t)) = \beta\dot{d}(t) \quad \text{in } H^1(\Omega)^*.$$
 (2.23)

Testing (2.23) with  $\dot{\varphi}(t)$ , integrating over [0,T], and using (1.22) lead to

$$\int_0^T \alpha \|\nabla \dot{\varphi}(\tau)\|_2^2 d\tau = \beta \int_0^T (\dot{d}(\tau) - \dot{\varphi}(\tau), \dot{\varphi}(\tau))_2 d\tau - \int_0^T \langle \frac{d}{dt} F(\tau, \varphi(\tau)), \dot{\varphi}(\tau) \rangle d\tau.$$
(2.24)

Note that it is indeed possible to integrate over time, since all the above integrands are continuous, in light of  $\dot{\varphi} \in C([0,T];H^1(\Omega)), \dot{d} \in C([0,T];L^2(\Omega))$  and Lemma 1.57.

The next two results yield suitable estimates for the first and second term on the right-hand side in (2.24), respectively, which will then allow us to deduce Lemma 2.14.

**Lemma 2.12.** Under Assumptions 1.17, 1.47, 1.56 and 2.8, it holds

$$\int_0^T (\dot{d}(\tau) - \dot{\varphi}(\tau), \dot{\varphi}(\tau))_2 d\tau \le 0.$$

*Proof.* The proof follows mainly by the arguments employed for the proof of [40, (4.14)]. From Theorem 1.62 we recall that  $\dot{d}$ , d and  $\varphi$  are Lipschitz continuous from [0,T] to  $L^2(\Omega)$ . Hence, the mapping  $f:[0,T]\to L^2(\Omega)$  defined as

$$f(t) := \delta \dot{d}(t) + \beta (d(t) - \varphi(t)) + r \tag{2.25}$$

belongs to  $W^{1,\infty}(0,T;L^2(\Omega))$ , in view of [77, Proposition 2.28b)], and thus, it is almost everywhere differentiable, cf. [81, Theorem 3.1.40]. Let now  $t \in (0,T)$  be arbitrary, but

fixed and h small enough such that  $t + h \in (0, T)$ . From Proposition 2.7,  $(i) \Rightarrow (iv)$ , we deduce

$$(f(t+h), \dot{d}(t))_2 \ge 0,$$
 (2.26a)

$$(f(t), \dot{d}(t))_2 = 0,$$
 (2.26b)

where for the above inequality we used  $\dot{d}(t) \geq 0$ . Relying on (2.26) one arrives at

$$\begin{cases} \frac{1}{h}(f(t+h) - f(t), \dot{d}(t))_2 \ge 0 & \text{for } h > 0, \\ \frac{1}{h}(f(t+h) - f(t), \dot{d}(t))_2 \le 0 & \text{for } h < 0. \end{cases}$$

Passing to the limits  $h \searrow 0$  and  $h \nearrow 0$ , respectively, and keeping in mind the fact that f is almost everywhere differentiable further implies

$$(\dot{f}(t), \dot{d}(t))_2 = 0$$
 f.a.a.  $t \in (0, T)$ . (2.27)

With (2.25) we rewrite (2.27) as

$$\delta(\ddot{d}(t), \dot{d}(t))_2 + (\beta(\dot{d}(t) - \dot{\varphi}(t)), \dot{d}(t))_2 = 0,$$

which can be continued as

$$\frac{\delta}{2} \frac{d}{dt} \|\dot{d}(t)\|_{2}^{2} + \beta \|\dot{d}(t) - \dot{\varphi}(t)\|_{2}^{2} + \beta (\dot{d}(t) - \dot{\varphi}(t), \dot{\varphi}(t))_{2} = 0$$
(2.28)

for almost all  $t \in (0,T)$ , see also [81, Lemma 3.1.43]. Due to  $\dot{f} \in L^{\infty}(0,T;L^{2}(\Omega))$  and  $\dot{d} \in C([0,T];L^{2}(\Omega))$  we can integrate (2.27), i.e., (2.28), over [0,T]. This finally yields

$$\frac{\delta}{2} \|\dot{d}(T)\|_{2}^{2} - \frac{\delta}{2} \|\dot{d}(0)\|_{2}^{2} + \beta \int_{0}^{T} \|\dot{d}(\tau) - \dot{\varphi}(\tau)\|_{2}^{2} d\tau + \beta \int_{0}^{T} (\dot{d}(\tau) - \dot{\varphi}(\tau), \dot{\varphi}(\tau))_{2} d\tau = 0,$$

which on account of (2.17b) gives the assertion.

**Lemma 2.13.** Let Assumptions 1.17, 1.47 and 1.56 hold true. Then, for all  $t \in [0, T]$ , it holds

$$-\langle \frac{d}{dt} F(t, \varphi(t)), \dot{\varphi}(t) \rangle_{H^1(\Omega)} \le \widetilde{c}(k) \|\dot{\varphi}(t)\|_{H^1(\Omega)}^2 + k \|\dot{\varphi}(t)\|_2^2 + Ck \quad \forall k > 0,$$

where  $\tilde{c}: \mathbb{R}^+ \to \mathbb{R}^+$  is a function independent of  $\beta$ , which satisfies  $\tilde{c}(k) \searrow 0$  as  $k \nearrow \infty$ , and C > 0 is a constant independent of  $\beta$ .

*Proof.* Let  $t \in [0,T]$  be arbitrary, but fixed. From Lemma 1.57 we know that  $F \in C^1([0,T] \times H^1(\Omega); H^1(\Omega)^*)$ , so that chain rule gives

$$\langle \frac{d}{dt} F(t, \varphi(t)), \dot{\varphi}(t) \rangle_{H^{1}(\Omega)} = \langle \partial_{t} F(t, \varphi(t)), \dot{\varphi}(t) \rangle_{H^{1}(\Omega)} + \langle \partial_{\varphi} F(t, \varphi(t)) \dot{\varphi}(t), \dot{\varphi}(t) \rangle_{H^{1}(\Omega)},$$
(2.29)

in view of  $\varphi \in C^1([0,T];H^1(\Omega))$ . From (1.86) we read

$$\langle \partial_t F(t, \varphi(t)), \dot{\varphi}(t) \rangle = -\langle \operatorname{div} \left( g'(\varphi(t)) \dot{\varphi}(t) \mathbb{C} \varepsilon(\boldsymbol{u}(t)) \right), \partial_t \mathcal{U}(t, \varphi(t)) \rangle_V.$$

Due to p > N, by Assumption 1.17.1, we have  $H^1(\Omega) \hookrightarrow L^{\frac{2p}{p-2}}(\Omega)$ , see (1.32). Now Hölder's inequality with (p-2)/2p+1/p+1/2=1 and (3) yield

$$|\langle \partial_t F(t, \varphi(t)), \dot{\varphi}(t) \rangle_{H^1(\Omega)}| \leq \|g'(\varphi(t))\|_{\infty} \|\dot{\varphi}(t)\|_{\frac{2p}{p-2}} \|\boldsymbol{u}(t)\|_{\boldsymbol{W}_D^{1,p}(\Omega)} \|\partial_t \mathcal{U}(t, \varphi(t))\|_V$$

$$\leq C \|\dot{\varphi}(t)\|_{H^1(\Omega)}, \tag{2.30}$$

with C > 0 independent of  $\beta$ . Note that for the last estimate we used Corollary 1.9 and (1.74), where the constant depends only on the norm of  $A_{\varphi}^{-1}$  (see (1.6)) and  $\dot{\ell}$ , and it is thus independent of  $\beta$ . On account of the generalized Young inequality, (2.30) can be continued as follows

$$\left| \langle \partial_t F(t, \varphi(t)), \dot{\varphi}(t) \rangle_{H^1(\Omega)} \right| \le \frac{1}{4k} \| \dot{\varphi}(t) \|_{H^1(\Omega)}^2 + Ck \quad \forall k > 0.$$
 (2.31)

For the second term in (2.29), one obtains from (1.101) the estimate

$$|\langle \partial_{\varphi} F(t, \varphi(t)) \dot{\varphi}(t), \dot{\varphi}(t) \rangle_{H^{1}(\Omega)}| \le k ||\dot{\varphi}(t)||_{2}^{2} + \widetilde{c}_{1}(k) ||\dot{\varphi}(t)||_{H^{1}(\Omega)}^{2} \quad \forall k > 0,$$
 (2.32)

where  $\widetilde{c_1}: \mathbb{R}^+ \to \mathbb{R}^+$  satisfies  $\widetilde{c_1}(k) \searrow 0$  as  $k \nearrow \infty$  and depends on the constant C from (1.100), p and N. A close inspection of the proof of (1.100) and in particular of the proof of Lemma 1.50 shows that  $\widetilde{c_1}$  is a function independent of  $\beta$ . By inserting (2.31) and (2.32) in (2.29) we get

$$-\langle \frac{d}{dt}F(t,\varphi(t)),\dot{\varphi}(t)\rangle_{H^{1}(\Omega)} \leq \underbrace{\left(\frac{1}{4k} + \widetilde{c}_{1}(k)\right)}_{=:\widetilde{c}(k)} \|\dot{\varphi}(t)\|_{H^{1}(\Omega)}^{2} + k\|\dot{\varphi}(t)\|_{2}^{2} + Ck \quad \forall k > 0.$$

This gives the assertion, in view of the properties of  $\widetilde{c}_1$  and since C was independent of  $\beta$ .

As a result of Lemmata 2.12 and 2.13 we obtain

**Lemma 2.14** (Boundedness of the gradient). Let Assumptions 1.17, 1.47, 1.56 and 2.8 hold. Then there exist constants  $C_1, C_2 > 0$  independent of  $\beta$  such that

$$\int_0^T \|\nabla \dot{\varphi}(\tau)\|_2^2 d\tau \le C_1 \int_0^T \|\dot{\varphi}(\tau)\|_2^2 d\tau + C_2.$$

*Proof.* Applying Lemmata 2.12 and 2.13 for the right-hand side in (2.24) leads to

$$\int_{0}^{T} \alpha \|\nabla \dot{\varphi}(\tau)\|_{2}^{2} d\tau \leq \widetilde{c}(k) \int_{0}^{T} \|\dot{\varphi}(\tau)\|_{H^{1}(\Omega)}^{2} d\tau + k \int_{0}^{T} \|\dot{\varphi}(\tau)\|_{2}^{2} d\tau + Ck \quad \forall k > 0, \quad (2.33)$$

with  $\tilde{c}$  as in Lemma 2.13 and C > 0 independent of  $\beta$ . We employ the properties of  $\tilde{c}$  and set k (large enough) such that  $\tilde{c}(k) < \alpha$  holds. Subtracting the term  $\tilde{c}(k) \int_0^T \|\nabla \dot{\varphi}(\tau)\|_2^2 d\tau$  on both sides in (2.33) then gives

$$(\underbrace{\alpha - \widetilde{c}(k)}_{>0}) \int_0^T \|\nabla \dot{\varphi}(\tau)\|_2^2 d\tau \le (\widetilde{c}(k) + k) \int_0^T \|\dot{\varphi}(\tau)\|_2^2 d\tau + Ck.$$

Since the (above fixed) value of k depends only on  $\tilde{c}$  and  $\alpha$ , and since  $\tilde{c}$  is independent of  $\beta$  cf. Lemma 2.13, and C as well, the proof is now complete.

**Lemma 2.15** (Boundedness of the  $L^2$ -component). Let Assumptions 1.17, 1.47, 1.56 and 2.8 hold. Then there exists a constant C > 0 independent of  $\beta$  such that

$$\int_0^T \|\dot{\varphi}(\tau)\|_2^2 d\tau \le C.$$

*Proof.* We begin with dividing (2.24) by  $\beta$  and bringing the term  $\int_0^T \|\dot{\varphi}(\tau)\|_2^2 d\tau$  on the left-hand side. This results in

$$\underbrace{\frac{\alpha}{\beta} \int_0^T \|\nabla \dot{\varphi}(\tau)\|_2^2 d\tau}_{\geq 0} + \int_0^T \|\dot{\varphi}(\tau)\|_2^2 d\tau = \int_0^T (\dot{d}(\tau), \dot{\varphi}(\tau))_2 d\tau - \frac{1}{\beta} \int_0^T \langle \frac{d}{dt} F(\tau, \varphi(\tau)), \dot{\varphi}(\tau) \rangle d\tau,$$

whence the estimate

$$\int_0^T \|\dot{\varphi}(\tau)\|_2^2 d\tau \le \int_0^T (\dot{d}(\tau), \dot{\varphi}(\tau))_2 d\tau - \frac{1}{\beta} \int_0^T \langle \frac{d}{dt} F(\tau, \varphi(\tau)), \dot{\varphi}(\tau) \rangle d\tau. \tag{2.34}$$

Using generalized Young inequality for the first term on the right-hand side in (2.34) further yields

$$\int_0^T (\dot{d}(\tau), \dot{\varphi}(\tau))_2 d\tau \le j \int_0^T \|\dot{d}(\tau)\|_2^2 d\tau + \frac{1}{4j} \int_0^T \|\dot{\varphi}(\tau)\|_2^2 d\tau \quad \forall j > 0.$$
 (2.35)

In order to estimate the second term on the right-hand side in (2.34), we apply Lemma 2.13 for some fixed k > 0. Thanks to Lemma 2.14, we then arrive at

$$-\frac{1}{\beta} \int_{0}^{T} \langle \frac{d}{dt} F(\tau, \varphi(\tau)), \dot{\varphi}(\tau) \rangle d\tau \leq \frac{C_{1}}{\beta} \int_{0}^{T} \|\nabla \dot{\varphi}(\tau)\|_{2}^{2} d\tau + \frac{C_{2}}{\beta} \int_{0}^{T} \|\dot{\varphi}(\tau)\|_{2}^{2} d\tau + \frac{C_{3}}{\beta}$$

$$\leq \frac{C_{1}}{\beta} \left( C_{4} \int_{0}^{T} \|\dot{\varphi}(\tau)\|_{2}^{2} d\tau + C_{5} \right) + \frac{C_{2}}{\beta} \int_{0}^{T} \|\dot{\varphi}(\tau)\|_{2}^{2} d\tau + \frac{C_{3}}{\beta}$$

$$\leq \frac{\widetilde{C}}{\beta} \int_{0}^{T} \|\dot{\varphi}(\tau)\|_{2}^{2} d\tau + \frac{C}{\beta},$$
(2.36)

where we abbreviate  $\widetilde{C} := C_1C_4 + C_2$  and  $C := C_1C_5 + C_3$ . Note that C and  $\widetilde{C}$  are independent of  $\beta$  on account of Lemmata 2.13 and 2.14. Further, inserting (2.35) and (2.36) in (2.34) implies

$$\int_{0}^{T} \|\dot{\varphi}(\tau)\|_{2}^{2} d\tau \leq j \int_{0}^{T} \|\dot{d}(\tau)\|_{2}^{2} d\tau + \left(\frac{1}{4j} + \frac{\widetilde{C}}{\beta}\right) \int_{0}^{T} \|\dot{\varphi}(\tau)\|_{2}^{2} d\tau + \frac{C}{\beta} \quad \forall j > 0. \quad (2.37)$$

Now, we set j > 1/2 and impose that  $\beta$  satisfies

$$\beta > 2\widetilde{C},\tag{2.38}$$

cf. Assumption 1.17.2, such that  $1/4j + \tilde{C}/\beta < 1$ , which in view of (2.37) gives in turn

$$\int_0^T \|\dot{\varphi}(\tau)\|_2^2 d\tau \le K_1 \int_0^T \|\dot{d}(\tau)\|_2^2 d\tau + K_2,$$

where  $K_1, K_2 > 0$  are constants independent of  $\beta$ . Due to Lemma 2.11, the desired estimate is now proven.

As a result of Lemmata 2.14 and 2.15, and by relying on Poincaré-Friedrich's inequality and (2.17a), see also (5.39), we can now conclude

Corollary 2.16 (Boundedness of the nonlocal damage). Under Assumptions 1.17, 1.47, 1.56 and 2.8, there exists a constant C > 0 independent of  $\beta$  such that

$$\|\varphi\|_{H^1(0,T;H^1(\Omega))} \le C,$$

where  $\varphi$  is the optimal nonlocal damage associated to  $(P_{\beta})$ .

#### 2.2.2 Passage to the limit in the elliptic system

We begin studying the behaviour of the penalized model  $(P_{\beta})$  as  $\beta$  approaches  $\infty$  with the elliptic system given by (2.16a)-(2.16b). In order to emphasize the presence of the penalty term, we denote from now on by  $(\mathbf{u}_{\beta}, \varphi_{\beta}, d_{\beta})$  the unique solution of the problem  $(P_{\beta})$ .

**Proposition 2.17** (Passage to the limit in (2.16a)). Let Assumptions 1.17.1, 1.47, 1.56 and 2.8 hold. Then, for every sequence  $\beta_n \to \infty$  as  $n \to \infty$ , there exists a (not relabeled) subsequence  $\{\varphi_{\beta_n}\}_{n\in\mathbb{N}}$  and  $\varphi \in H^1(0,T;H^1(\Omega))$  such that the following converges hold

$$\varphi_{\beta_n} \rightharpoonup \varphi \quad in \ H^1(0, T; H^1(\Omega)),$$
 (2.39)

$$\mathbf{u}_{\beta_n} \to \mathcal{U}(\cdot, \varphi(\cdot)) =: \mathbf{u} \quad \text{in } C([0, T]; V)$$
 (2.40)

as  $n \to \infty$ , where  $\mathbf{u}_{\beta_n}$  and  $\varphi_{\beta_n}$  stand for the optimal displacement and optimal nonlocal damage, respectively, associated to the problem  $(P_{\beta_n})$ .

Proof. Since  $H^1(0,T;H^1(\Omega))$  is a reflexive Banach space, see e.g. [81, Theorem 3.1.36], there exists in view of Corollary 2.16 a (not relabeled) weakly convergent subsequence of  $\{\varphi_{\beta_n}\}_{n\geq m}$ , i.e., there exists  $\varphi\in H^1(0,T;H^1(\Omega))$  such that (2.39) holds. Here  $m\in\mathbb{N}$  denotes an index so that  $\beta_n$  fulfills Assumption 1.17.2 for all  $n\geq m$ . Let us now abbreviate r:=2p/(p-2) such that 1/2=1/p+1/r holds. Note that due to Assumption 1.17.1, see (1.32), we have the compact embedding  $H^1(\Omega)\hookrightarrow L^r(\Omega)$ , which by [81, Corollary 3.1.42] implies the compact embedding  $H^1(0,T;H^1(\Omega))\hookrightarrow C([0,T];L^r(\Omega))$ . Therewith we deduce from (2.39) the convergence

$$\varphi_{\beta_n} \to \varphi \text{ in } C([0,T];L^r(\Omega)) \quad \text{as } n \to \infty.$$
 (2.41)

Additionally, as a result of Proposition 1.10, we have at all  $t \in [0, T]$  the estimate

$$\|\mathcal{U}(t,\varphi_{\beta_n}(t)) - \mathcal{U}(t,\varphi(t))\|_V \le L\|\varphi_{\beta_n}(t) - \varphi(t)\|_r \quad \forall n \in \mathbb{N}, \tag{2.42}$$

where L > 0 is a constant independent of n, since  $\beta$  does not appear in the equation associated to the solution operator  $\mathcal{U}$ , i.e., equation (1.48a), see also the proof of Proposition 1.10. Since  $\mathbf{u}_{\beta_n} = \mathcal{U}(\cdot, \varphi_{\beta_n}(\cdot))$ , (2.40) follows now from (2.41) and (2.42).

**Proposition 2.18** (Passage to the limit in (2.16b)). Under Assumptions 1.17.1, 1.47, 1.56 and 2.8 it holds

$$d_{\beta_n} \rightharpoonup \varphi \text{ in } H^1(0,T;L^2(\Omega)) \quad \text{as } n \to \infty,$$

where  $\{\beta_n\}_{n\in\mathbb{N}}$  denotes the subsequence from Proposition 2.17 and  $\varphi$  the corresponding limit, while  $d_{\beta_n}$  stands for the optimal local damage associated to the problem  $(P_{\beta_n})$ . This implies in particular that the local and nonlocal damage coincide in the limit.

*Proof.* Let  $n \in \mathbb{N}$  and  $t \in [0,T]$  be arbitrary, but fixed. We start by showing a convergence result for the sequence given by  $\xi_{\beta_n} := d_{\beta_n} - \varphi_{\beta_n} \in C^1([0,T];L^2(\Omega))$ . From (2.16b) we know that

$$\beta_n \int_{\Omega} \xi_{\beta_n}(t) \psi \ dx = \int_{\Omega} \alpha \nabla \varphi_{\beta_n}(t) \nabla \psi \ dx + \langle F(t, \varphi_{\beta_n}(t)), \psi \rangle \quad \forall \psi \in H^1(\Omega).$$
 (2.43)

Moreover, in view of Assumption 1.17.1 we have  $H^1(\Omega) \hookrightarrow L^{\frac{p}{p-2}}(\Omega)$ , which combined with Hölder's inequality and (1.23) gives in turn

$$\frac{1}{\beta_{n}} \left| \int_{\Omega} \alpha \nabla \varphi_{\beta_{n}}(t) \nabla \psi \, dx \right| + \frac{1}{\beta_{n}} \left| \langle F(t, \varphi_{\beta_{n}}(t)), \psi \rangle_{H^{1}(\Omega)} \right| \\
\leq \frac{1}{\beta_{n}} \left( \alpha \| \nabla \varphi_{\beta_{n}}(t) \|_{2} + \frac{1}{2} \| g'(\varphi_{\beta_{n}}(t)) \|_{\infty} \| \mathbb{C}\varepsilon(\boldsymbol{u}_{\beta_{n}}(t)) : \varepsilon(\boldsymbol{u}_{\beta_{n}}(t)) \|_{\frac{p}{2}} \right) \|\psi\|_{H^{1}(\Omega)} \\
\leq \frac{C}{\beta_{n}} \|\psi\|_{H^{1}(\Omega)} \quad \forall \psi \in H^{1}(\Omega), \tag{2.44}$$

where C > 0 is independent of n. Note that for the last inequality we used the embedding  $H^1(0,T;H^1(\Omega)) \hookrightarrow C([0,T];H^1(\Omega))$  combined with Corollary 2.16, as well as (3) and Corollary 1.9. Now (2.43) and (2.44) lead to

$$|\langle \xi_{\beta_n}(t), \psi \rangle_{H^1(\Omega)}| \le \frac{C}{\beta_n} \|\psi\|_{H^1(\Omega)} \quad \forall \, \psi \in H^1(\Omega),$$

whence

$$\xi_{\beta_n} \to 0 \quad \text{in } C([0, T]; H^1(\Omega)^*) \quad \text{as } n \to \infty.$$
 (2.45)

By employing (2.41) and  $L^{2p/(p-2)}(\Omega) \hookrightarrow L^2(\Omega) \hookrightarrow H^1(\Omega)^*$  we further obtain

$$\varphi_{\beta_n} \to \varphi$$
 in  $C([0,T]; H^1(\Omega)^*)$  as  $n \to \infty$ ,

which together with (2.45) now gives

$$d_{\beta_n} \to \varphi \quad \text{in } C([0,T]; H^1(\Omega)^*) \quad \text{as } n \to \infty.$$
 (2.46)

On the other hand, as a result of Lemma 2.11, there exists a subsequence of  $\{d_{\beta_n}\}_{n\in\mathbb{N}}$  and  $d\in H^1(0,T;L^2(\Omega))$  such that

$$d_{\beta_{n_k}} \rightharpoonup d \text{ in } H^1(0,T;L^2(\Omega)) \quad \text{ as } k \to \infty.$$

Thus, by the embedding  $H^1(0,T;L^2(\Omega)) \hookrightarrow C([0,T];H^1(\Omega)^*)$  and (2.46) we arrive at  $d = \varphi$ . Arguing as above, one can show that any weakly convergent subsequence of  $\{d_{\beta_n}\}_{n\in\mathbb{N}}$  possesses the same limit  $\varphi$ . This completes the proof.

Let us conclude this subsection by pointing out the herein achieved results. It was established that after passing to the limit  $\beta \to \infty$  in (2.16a), i.e., (1.48a), one obtains in light of (2.40) the same equation for the coupling between displacement and nonlocal damage. Further, it turns out that the equation (2.16b) for the nonlocal damage, i.e., (1.48b), reduces to the identity  $d = \varphi$ , where d and  $\varphi$  stand for the local and nonlocal damage after passing to the limit, respectively. Thus, both damage variables become equal when  $\beta \to \infty$ , which shows that the penalty approach makes sense from a mathematical point of view. Keep in mind that (at this point), the limit model depends on the chosen subsequence in Proposition 2.17. In the next section we will see that under special conditions, no matter what subsequence one chooses, the limit is the same and consequently the whole sequences  $\{\varphi_{\beta_n}\}$  (and thus,  $\{d_{\beta_n}\}$ ) and  $\{u_{\beta_n}\}$  converge for any  $\beta_n \to \infty$  as  $n \to \infty$ .

#### 2.2.3 Passage to the limit in the energy identity

After we exploited the behaviour of the elliptic system when  $\beta \to \infty$ , we now turn our attention to the dynamical component of the penalized damage model, i.e., (2.16c). However, as already indicated at the beginning of Section 2.1, we do not possess any useful information about the boundedness of the term  $\beta(d_{\beta} - \varphi_{\beta})$ . That is why we again make use of the characterization of the local damage by means of the energy identity, see

(2.9), which has the advantage of not including this term. Passing to the limit therein will result in an *energy inequality*, where the original functionals slightly change, see Definitions 2.19 and 2.21 below. Moreover, as we will see in Section 2.3 below, this is equivalent to an evolutionary equation. Besides, the resulting *energy inequality* is in fact an *energy identity* as well, where, as in (2.9), one integrates over an arbitrary interval  $[s,t] \subset [0,T]$ , see Remark 2.27 below.

We begin by introducing the functionals that will arise after passing to the limit. We first define the energy functional without penalty  $\widetilde{\mathcal{E}}: [0,T] \times V \times H^1(\Omega) \to \mathbb{R}$  by

$$\widetilde{\mathcal{E}}(t, \boldsymbol{u}, \varphi) := \frac{1}{2} \int_{\Omega} g(\varphi) \mathbb{C}\varepsilon(\boldsymbol{u}) : \varepsilon(\boldsymbol{u}) \, dx - \langle \ell(t), \boldsymbol{u} \rangle_{V} + \frac{\alpha}{2} \|\nabla \varphi\|_{2}^{2}, \tag{2.47}$$

which has rather the purpose of motivating Definition 2.19 below. By comparing Definition 0.2 and (2.47) it is easy to see that at all  $(t, \varphi, d) \in [0, T] \times H^1(\Omega) \times L^2(\Omega)$  it holds

$$\underset{\boldsymbol{u} \in V}{\arg\min} \, \widetilde{\mathcal{E}}(t, \boldsymbol{u}, \varphi) = \underset{\boldsymbol{u} \in V}{\arg\min} \, \mathcal{E}(t, \boldsymbol{u}, \varphi, d).$$

Hence, for a given pair  $(t, \varphi) \in [0, T] \times H^1(\Omega)$ , the functional  $\widetilde{\mathcal{E}}$  reaches its minimum with respect to the variable  $\boldsymbol{u}$  at  $\mathcal{U}(t, \varphi)$ , see also Proposition 1.7 and Definition 1.8. Therefore, after minimizing one obtains

**Definition 2.19** (Reduced energy functional without penalty). We define the functional  $\widetilde{\mathcal{I}}: [0,T] \times H^1(\Omega) \to \mathbb{R}$  as

$$\widetilde{\mathcal{I}}(t,\varphi) := \min_{\boldsymbol{u} \in V} \widetilde{\mathcal{E}}(t,\boldsymbol{u},\varphi) = \widetilde{\mathcal{E}}(t,\mathcal{U}(t,\varphi),\varphi).$$

Note that Assumption 1.5 is not needed in order to define  $\widetilde{\mathcal{I}}$ , as  $\mathcal{E}$ , and thus  $\widetilde{\mathcal{E}}$ , has an unique minimum w.r.t. u in V characterized by (1.14) also when  $\ell$  maps only to  $V^*$ . This is shown by a short inspection of the proof of Proposition 1.7. The only difference is that the solution operator  $\mathcal{U}$  does no longer have range in  $\mathbf{W}_D^{1,p}(\Omega)$ , but only in V.

On account of (2.47) and Definitions 1.2 and 1.8 we can rewrite the reduced energy functional without penalty at all  $(t, \varphi) \in [0, T] \times H^1(\Omega)$  as

$$\widetilde{\mathcal{I}}(t,\varphi) = \frac{1}{2} \langle A_{\varphi} \mathcal{U}(t,\varphi), \mathcal{U}(t,\varphi) \rangle_{V} - \langle \ell(t), \mathcal{U}(t,\varphi) \rangle_{V} + \frac{\alpha}{2} \|\nabla \varphi\|_{2}^{2} 
= -\frac{1}{2} \langle \ell(t), \mathcal{U}(t,\varphi) \rangle_{V} + \frac{\alpha}{2} \|\nabla \varphi\|_{2}^{2}.$$
(2.48)

**Lemma 2.20** (Fréchet-differentiability of  $\widetilde{\mathcal{I}}$ ). Under Assumptions 1.17.1 and 1.47, it holds  $\widetilde{\mathcal{I}} \in C^1([0,T] \times H^1(\Omega))$  and at all  $(t,\varphi) \in [0,T] \times H^1(\Omega)$  we have

$$\partial_t \widetilde{\mathcal{I}}(t,\varphi) = -\langle \dot{\ell}(t), \mathcal{U}(t,\varphi) \rangle_V,$$
 (2.49a)

$$\partial_{\varphi} \widetilde{\mathcal{I}}(t, \varphi) = -\alpha \Delta \varphi + F(t, \varphi).$$
 (2.49b)

*Proof.* The proof is very similar to the proof of Lemma 2.2. We apply product rule for the mapping  $f:[0,T]\times H^1(\Omega)\ni (t,\varphi)\mapsto -1/2\langle \ell(t),\mathcal{U}(t,\varphi)\rangle_V\in\mathbb{R}$ , which yields  $f\in C^1([0,T]\times H^1(\Omega))$ , in view of Assumption 1.47 and Proposition 1.53. The derivative thereof at all  $(t,\varphi)\in[0,T]\times H^1(\Omega)$  reads

$$f'(t,\varphi)(\delta t,\delta\varphi) = -\frac{1}{2} \langle \dot{\ell}(t)\delta t, \mathcal{U}(t,\varphi) \rangle_{V} - \frac{1}{2} \langle \ell(t), \mathcal{U}'(t,\varphi)(\delta t,\delta\varphi) \rangle_{V}$$

$$= -\langle \dot{\ell}(t)\delta t, \mathcal{U}(t,\varphi) \rangle_{V} + \langle F(t,\varphi), \delta\varphi \rangle_{H^{1}(\Omega)} \quad \forall (\delta t,\delta\varphi) \in \mathbb{R} \times H^{1}(\Omega).$$
(2.50)

Note that the last equality from above follows by the arguments used for deriving (2.4) and (2.5), i.e., (1.72), (1.75) and (1.23), as well as Definitions 1.2 and 1.8. From (2.50) combined with (2.48) and  $\|\nabla \cdot\|_2^2 \in C^1(H^1(\Omega))$  we can now deduce that  $\widetilde{\mathcal{I}} \in C^1([0,T] \times H^1(\Omega))$  and that its derivative at all  $(t,\varphi) \in [0,T] \times H^1(\Omega)$  is given by

$$\mathcal{I}'(t,\varphi)(\delta t,\delta \varphi) = -\langle \dot{\ell}(t)\delta t, \mathcal{U}(t,\varphi)\rangle_{V} + \langle F(t,\varphi), \delta \varphi \rangle_{H^{1}(\Omega)} + \alpha(\nabla \varphi, \nabla \delta \varphi)_{2} \quad \forall (\delta t, \delta \varphi) \in \mathbb{R} \times H^{1}(\Omega).$$

This completes the proof.

Next we introduce the viscous dissipation functional corresponding to the situation without penalty.

**Definition 2.21** (Viscous dissipation functional). We define the functional  $\widetilde{\mathcal{R}}_{\delta}: H^1(\Omega) \to [0,\infty]$  by

$$\widetilde{\mathcal{R}}_{\delta}(\eta) := \begin{cases} r \int_{\Omega} \eta \ dx + \frac{\delta}{2} \|\eta\|_{2}^{2} & \text{if } \eta \geq 0 \ a.e. \ in \ \Omega, \\ \infty & \text{otherwise,} \end{cases}$$

that is,  $\widetilde{\mathcal{R}}_{\delta} = \mathcal{R}_{\delta}|_{H^1(\Omega)}$ .

We now begin to analyze the behaviour of the terms in (2.9) as  $\beta \to \infty$ . In the sequel we again denote by  $(\boldsymbol{u}_{\beta}, \varphi_{\beta}, d_{\beta})$  the unique solution of the problem  $(P_{\beta})$ . Recall that, cf. Propositions 2.17 and 2.18 there exists a sequence  $\beta_n \to \infty$  as  $n \to \infty$  and  $\varphi \in H^1(0,T;H^1(\Omega))$  such that the following converges hold

$$\varphi_{\beta_n} \rightharpoonup \varphi \quad \text{in } H^1(0, T; H^1(\Omega)),$$
 (2.51a)

$$d_{\beta_n} \rightharpoonup \varphi \quad \text{in } H^1(0, T; L^2(\Omega)),$$
 (2.51b)

$$\mathbf{u}_{\beta_n} \to \mathcal{U}(\cdot, \varphi(\cdot)) \text{ in } C([0, T]; V)$$
 (2.51c)

as  $n \to \infty$ , provided that Assumptions 1.17.1, 1.47, 1.56 and 2.8 are satisfied. In the rest of the section we denote by  $\{\beta_n\}_{n\in\mathbb{N}}$  such a fixed sequence and by  $\varphi$  the corresponding limit.

**Lemma 2.22.** Under Assumptions 1.17.1, 1.47, 1.56 and 2.8 it holds

$$\int_0^t \widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(\tau)) \ d\tau \le \liminf_{n \to \infty} \int_0^t \mathcal{R}_{\delta}(\dot{d}_{\beta_n}(\tau)) \ d\tau$$

for all  $t \in [0,T]$ .

*Proof.* Let  $t \in [0,T]$  be arbitrary, but fixed. First note that (2.51b) implies

$$\dot{d}_{\beta_n} \rightharpoonup \dot{\varphi} \text{ in } L^2(0,t;L^2(\Omega)),$$
 (2.52)

which further leads to

$$\dot{d}_{\beta_n} \rightharpoonup \dot{\varphi} \text{ in } L^1(0,t;L^1(\Omega))$$
 (2.53)

as  $n \to \infty$ . Note that here we also employed the fact that the operator  $H^1(0,T;L^2(\Omega)) \ni \zeta \mapsto \zeta|_{[0,t]} \in H^1(0,t;L^2(\Omega))$  is linear and continuous. Now, since  $\|\cdot\|_{L^1(0,t;L^1(\Omega))}$  and  $\|\cdot\|_{L^2(0,t;L^2(\Omega))}^2$  are weakly lower semicontinuous, we infer from (2.52) and (2.53)

$$r \int_{0}^{t} \|\dot{\varphi}(\tau)\|_{1} d\tau + \frac{\delta}{2} \int_{0}^{t} \|\dot{\varphi}(\tau)\|_{2}^{2} d\tau \leq \liminf_{n \to \infty} r \int_{0}^{t} \|\dot{d}_{\beta_{n}}(\tau)\|_{1} d\tau + \frac{\delta}{2} \int_{0}^{t} \|\dot{d}_{\beta_{n}}(\tau)\|_{2}^{2} d\tau. \tag{2.54}$$

Further, recall that from (2.16c) one reads  $\dot{d}_{\beta_n} \geq 0$  for all  $n \in \mathbb{N}$ . Since the set  $\{f \in L^2(0,t;L^2(\Omega)): f \geq 0\}$  is weakly closed, one deduces from (2.52) that  $\dot{\varphi}|_{[0,t]} \geq 0$ . Now (2.54) gives in view of Definitions 0.4 and 2.21 the assertion.

**Lemma 2.23.** Let Assumptions 1.17.1, 1.47, 1.56 and 2.8 hold. Then for all  $t \in [0, T]$  we have the convergence

$$\partial_d \mathcal{I}(t, d_{\beta_n}(t)) \rightharpoonup \partial_{\varphi} \widetilde{\mathcal{I}}(t, \varphi(t)) \text{ in } H^1(\Omega)^* \text{ as } n \to \infty.$$

*Proof.* Let  $n \in \mathbb{N}$  and  $t \in [0, T]$  be arbitrary, but fixed. From (2.43) we know in view of (2.2b) and (2.49b) that

$$\partial_d \mathcal{I}(t, d_{\beta_n}(t)) = \partial_{\varphi} \widetilde{\mathcal{I}}(t, \varphi_{\beta_n}(t)) \text{ in } H^1(\Omega)^*.$$
 (2.55)

As a result of Assumption 1.17.1 we have  $H^1(\Omega) \hookrightarrow L^{\frac{2p}{p-2}}(\Omega)$ , see (1.32), so that by the convergence (2.51a) and Lemma 5.9 we get

$$\varphi_{\beta_n}(t) \to \varphi(t) \text{ in } L^{\frac{2p}{p-2}}(\Omega),$$
 (2.56a)

$$\nabla \varphi_{\beta_n}(t) \rightharpoonup \nabla \varphi(t) \text{ in } L^2(\Omega; \mathbb{R}^N)$$
 (2.56b)

as  $n \to \infty$ . Employing again (2.49b) we obtain by means of Lemma 1.18 for all  $v \in H^1(\Omega)$  the estimate

$$|\langle \partial_{\varphi} \widetilde{\mathcal{I}}(t, \varphi_{\beta_{n}}(t)) - \partial_{\varphi} \widetilde{\mathcal{I}}(t, \varphi(t)), v \rangle_{H^{1}(\Omega)}| \leq C \|\varphi_{\beta_{n}}(t) - \varphi(t)\|_{\frac{2p}{p-2}} \|v\|_{\frac{2p}{p-2}} + \alpha |(\nabla \varphi_{\beta_{n}}(t) - \nabla \varphi(t), \nabla v)_{2}|,$$

$$(2.57)$$

where the constant C is independent of n, since  $\beta$  does not appear in the definition of F, see (1.23). We also refer to the proof of Lemma 1.18 for more details. From (2.57) we can now conclude with (2.56a), (2.56b) and (2.55) the desired assertion.

**Lemma 2.24.** Under Assumptions 1.17.1, 1.47, 1.56 and 2.8 it holds

$$\int_0^t \widetilde{\mathcal{R}}_{\delta}^* \left( -\partial_{\varphi} \widetilde{\mathcal{I}}(\tau, \varphi(\tau)) \right) d\tau \leq \liminf_{n \to \infty} \int_0^t \mathcal{R}_{\delta}^* \left( -\partial_{d} \mathcal{I}(\tau, d_{\beta_n}(\tau)) \right) d\tau.$$

for all  $t \in [0, T]$ .

*Proof.* Recall that  $\widetilde{\mathcal{R}}_{\delta} = \mathcal{R}_{\delta}|_{H^1(\Omega)}$ , which by the definition of the Fenchel conjugate gives in turn

$$\widetilde{\mathcal{R}}_{\delta}^{*}(\xi) = \sup_{v \in H^{1}(\Omega)} (\xi(v) - \widetilde{\mathcal{R}}_{\delta}(v)) \le \sup_{v \in L^{2}(\Omega)} (\xi(v) - \mathcal{R}_{\delta}(v)) = \mathcal{R}_{\delta}^{*}(\xi).$$
 (2.58)

for any  $\xi \in L^2(\Omega)$ . Let now again  $t \in [0,T]$  be arbitrary, but fixed. Since  $\widetilde{\mathcal{R}}_{\delta}^*$ :  $H^1(\Omega)^* \to (-\infty,\infty]$  is convex and lower semicontinuous, cf. e.g. [69, Theorem 12.2], and thus, weakly lower semicontinuous, Lemma 2.23 leads to

$$\widetilde{\mathcal{R}}_{\delta}^* \left( -\partial_{\varphi} \widetilde{\mathcal{I}}(\tau, \varphi(\tau)) \right) \leq \liminf_{n \to \infty} \widetilde{\mathcal{R}}_{\delta}^* \left( -\partial_{d} \mathcal{I}(\tau, d_{\beta_n}(\tau)) \right) \quad \forall \tau \in [0, t].$$

By setting  $\xi := -\partial_d \mathcal{I}(\tau, d_{\beta_n}(\tau))$  in (2.58), where  $\tau \in [0, t]$ , the above estimate can be continued as

$$\widetilde{\mathcal{R}}_{\delta}^{*}\left(-\partial_{\varphi}\widetilde{\mathcal{I}}(\tau,\varphi(\tau))\right) \leq \liminf_{n \to \infty} \mathcal{R}_{\delta}^{*}\left(-\partial_{d}\mathcal{I}(\tau,d_{\beta_{n}}(\tau))\right) \quad \forall \, \tau \in [0,t].$$
 (2.59)

Further, one observes in view of Lemma 2.4, that for all  $n \in \mathbb{N}$  the map  $[0,t] \ni \tau \mapsto \mathcal{R}^*_{\delta}(-\partial_d \mathcal{I}(\tau,d_{\beta_n}(\tau))) = \delta/2 \|\dot{d}_{\beta_n}(\tau)\|_2$  is non-negative and continuous, as  $\dot{d}_{\beta_n} \in C([0,T];L^2(\Omega))$ . Moreover,  $\sup_{n\in\mathbb{N}} \|\dot{d}_{\beta_n}\|_{L^2(0,t;L^2(\Omega))} < \infty$ , due to (2.51b). Hence, we are allowed to apply Fatou's lemma, cf. e.g. [7, Lemma 4.1], which tells us that

$$\int_{0}^{t} \liminf_{n \to \infty} \mathcal{R}_{\delta}^{*} \left( -\partial_{d} \mathcal{I}(\tau, d_{\beta_{n}}(\tau)) \right) d\tau \leq \liminf_{n \to \infty} \int_{0}^{t} \mathcal{R}_{\delta}^{*} \left( -\partial_{d} \mathcal{I}(\tau, d_{\beta_{n}}(\tau)) \right) d\tau. \tag{2.60}$$

We also notice that  $\partial_{\varphi} \widetilde{\mathcal{I}}(\cdot, \varphi(\cdot))$  is continuous, thanks to Proposition 2.20 and the embedding  $H^1(0,T;H^1(\Omega)) \hookrightarrow C([0,T];H^1(\Omega))$ . Since  $\widetilde{\mathcal{R}}^*_{\delta}$  is lower semicontinuous, one concludes that the mapping  $[0,t] \ni \tau \mapsto \widetilde{\mathcal{R}}^*_{\delta}(-\partial_{\varphi}\widetilde{\mathcal{I}}(\tau,\varphi(\tau))) \in \mathbb{R}$  is lower semicontinuous as well, and therefore, measurable. Note that  $\widetilde{\mathcal{R}}^*_{\delta}(-\partial_{\varphi}\widetilde{\mathcal{I}}(\cdot,\varphi(\cdot)))$  is real-valued, since  $\widetilde{\mathcal{R}}_{\delta}$  is proper and in view of (2.59),  $\sup_{n\in\mathbb{N}} \|\dot{d}_{\beta_n}\|_{L^2(0,t;L^2(\Omega))} < \infty$ , and Lemma 2.4. Now we can integrate (2.59) over (0,t), which combined with (2.60) finally gives the assertion.

We are now in the position to state the main result of this subsection.

**Proposition 2.25** (The energy inequality without penalty). Let Assumptions 1.17.1, 1.47, 1.56 and 2.8 hold. Then the limit function  $\varphi \in H^1(0,T;H^1(\Omega))$  from Proposition 2.17 fulfills for all  $t \in [0,T]$  the estimate

$$\int_{0}^{t} \widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(\tau)) d\tau + \int_{0}^{t} \widetilde{\mathcal{R}}_{\delta}^{*} \left(-\partial_{\varphi} \widetilde{\mathcal{I}}(\tau, \varphi(\tau))\right) d\tau + \widetilde{\mathcal{I}}(t, \varphi(t)) \leq \widetilde{\mathcal{I}}(0, \varphi(0)) + \int_{0}^{t} \partial_{t} \widetilde{\mathcal{I}}(\tau, \varphi(\tau)) d\tau$$
(2.61)

and at the initial time point it holds  $\varphi(0) = 0$ .

*Proof.* Let  $t \in [0,T]$  be arbitrary, but fixed. Setting s := 0 in (2.9) yields the following identity

$$\int_{0}^{t} \mathcal{R}_{\delta}(\dot{d}_{\beta_{n}}(\tau)) d\tau + \int_{0}^{t} \mathcal{R}_{\delta}^{*}(-\partial_{d}\mathcal{I}(\tau, d_{\beta_{n}}(\tau))) d\tau + \mathcal{I}(t, d_{\beta_{n}}(t))$$

$$= \mathcal{I}(0, d_{\beta_{n}}(0)) + \int_{0}^{t} \partial_{t}\mathcal{I}(\tau, d_{\beta_{n}}(\tau)) d\tau \quad \forall n \in \mathbb{N}, \tag{2.62}$$

where we next pass to the limit  $n \to \infty$ . In view of Lemmata 2.22 and 2.24 we only need to discuss the last three terms in (2.62). To this end, we first notice that due to (2.51a), (2.51c) combined with Lemma 5.9, and the weak lower semicontinuity of  $\|\cdot\|_2^2: L^2(\Omega) \to \mathbb{R}$  we have

$$-\frac{1}{2}\langle \ell(t), \mathcal{U}(t, \varphi(t)) \rangle_{V} + \frac{\alpha}{2} \|\nabla \varphi(t)\|_{2}^{2} \leq \liminf_{n \to \infty} -\frac{1}{2} \langle \ell(t), \boldsymbol{u}_{\beta_{n}}(t) \rangle_{V} + \frac{\alpha}{2} \|\nabla \varphi_{\beta_{n}}(t)\|_{2}^{2} + \liminf_{n \to \infty} \underbrace{\frac{\beta_{n}}{2} \|\varphi_{\beta_{n}}(t) - d_{\beta_{n}}(t)\|_{2}^{2}}_{>0},$$

which in light of (2.1) and (2.48) reads

$$\widetilde{\mathcal{I}}(t,\varphi(t)) \le \liminf_{n \to \infty} \mathcal{I}(t,d_{\beta_n}(t)).$$
 (2.63)

For the right-hand side in (2.62) we obtain by employing (2.1) and (2.2a), as well as Assumption 2.8 and (2.17a) the equality

$$\mathcal{I}(0, d_{\beta_n}(0)) + \int_0^t \partial_t \mathcal{I}(\tau, d_{\beta_n}(\tau)) d\tau = \int_0^t \langle -\dot{\ell}(\tau), \boldsymbol{u}_{\beta_n}(\tau) \rangle d\tau \quad \forall n \in \mathbb{N}.$$
 (2.64)

On the other side, from (2.17a) and (2.51a) together with Lemma 5.9 one deduces  $\varphi(0) = 0$ . By relying again on Assumption 2.8, we infer by (2.48) and (2.49a) that

$$\widetilde{\mathcal{I}}(0,\varphi(0)) + \int_0^t \partial_t \widetilde{\mathcal{I}}(\tau,\varphi(\tau)) d\tau = \int_0^t \langle -\dot{\ell}(\tau), \mathcal{U}(\tau,\varphi(\tau)) \rangle d\tau. \tag{2.65}$$

As a consequence of  $\dot{\ell} \in C([0,T];V^*)$ , by assumption, and (2.51c) one arrives at

$$\int_0^t \langle \dot{\ell}(\tau), \boldsymbol{u}_{\beta_n}(\tau) - \mathcal{U}(\tau, \varphi(\tau)) \rangle d\tau \leq t \|\dot{\ell}\|_{C([0,T];V^*)} \|\boldsymbol{u}_{\beta_n} - \mathcal{U}(\cdot, \varphi(\cdot))\|_{C([0,T];V)} \to 0$$

as  $n \to \infty$ , which on account of (2.64) and (2.65) gives in turn

$$\lim_{n \to \infty} \mathcal{I}(0, d_{\beta_n}(0)) + \int_0^t \partial_t \mathcal{I}(\tau, d_{\beta_n}(\tau)) d\tau = \widetilde{\mathcal{I}}(0, \varphi(0)) + \int_0^t \partial_t \widetilde{\mathcal{I}}(\tau, \varphi(\tau)) d\tau.$$
 (2.66)

By using Lemmata 2.22, 2.24 and the convergences (2.63), (2.66), we ultimately obtain (2.61) from (2.62). The claim is now proven.

In the next sections we employ the *energy inequality* (2.61) to show that the limit in (2.51a)-(2.51c) satisfies a system of equations, namely (2.75) below, which is equivalent to a classical viscous partial damage model containing only one damage variable, cf. Section 2.5 below. As a secondary result, we will also deduce that (2.61) is in fact equivalent to an *energy identity*, see Remark 2.27 below.

### 2.3 A single-field gradient damage model

The main purpose of this section is to show that the limit function  $\varphi$  from Proposition 2.17 satisfies an evolutionary equation, i.e., (2.67) below. This will then allow us to write our single-field limit damage model as a PDE system, which as it will turn out in Section 2.5 below, falls into the category of models analyzed in [41]. The section ends with the discussion of the unique solvability thereof.

By mainly following the lines of the proof of [41, Proposition 3.2], it can be shown that the *energy inequality* (2.61) is equivalent to the above mentioned evolutionary equation:

**Proposition 2.26** (Evolution without penalty). Let Assumptions 1.17.1 and 1.47 hold true. Then, any  $\varphi \in H^1(0,T;H^1(\Omega))$  which fulfills for all  $t \in [0,T]$  the energy inequality (2.61) satisfies

$$-\partial_{\varphi}\widetilde{\mathcal{I}}(t,\varphi(t)) \in \partial \widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(t)) \quad \text{f.a.a. } t \in (0,T).$$
 (2.67)

The reverse assertion is true as well.

*Proof.* The proof is based on the same arguments as the proof of Proposition 2.5. We begin by proving two auxiliary results that will be needed for both implications. To this end, let  $\varphi \in H^1(0,T;H^1(\Omega))$  be arbitrary, but fixed. Since  $\widetilde{\mathcal{R}}_{\delta}$  is convex and proper, a well known convex analysis result, see e.g. [69, Theorem 23.5], leads to

$$\widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(t)) + \widetilde{\mathcal{R}}_{\delta}^{*}(-\partial_{\varphi}\widetilde{\mathcal{I}}(t,\varphi(t))) = -\langle \partial_{\varphi}\widetilde{\mathcal{I}}(t,\varphi(t)), \dot{\varphi}(t) \rangle_{H^{1}(\Omega)} \\ \iff (2.68)$$
$$-\partial_{\varphi}\widetilde{\mathcal{I}}(t,\varphi(t)) \in \partial \widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(t))$$

f.a.a.  $t \in (0,T)$ . Further, note that by (2.49), Assumption 1.47 and Corollary 1.9, in combination with (1.23) and (3), we have

$$|\partial_t \widetilde{\mathcal{I}}(t,\phi)|, \|\partial_\varphi \widetilde{\mathcal{I}}(t,\phi)\|_{H^1(\Omega)^*} \le C \|\phi\|_{H^1(\Omega)} + c \quad \forall (t,\phi) \in [0,T] \times H^1(\Omega),$$

where C, c > 0 are independent of  $(t, \phi)$ . We observe that this implies (5.4) for  $\widetilde{\mathcal{I}}$ . Thus, in view of Lemma 2.20, we can apply Lemma 5.5, which gives in turn that the function  $[0,T] \ni t \mapsto \widetilde{\mathcal{I}}(t,\varphi(t)) \in \mathbb{R}$  belongs to  $H^1(0,T)$ . Moreover, the derivative thereof is given by

$$\frac{d}{dt}\widetilde{\mathcal{I}}(t,\varphi(t)) = \partial_t \widetilde{\mathcal{I}}(t,\varphi(t)) + \langle \partial_\varphi \widetilde{\mathcal{I}}(t,\varphi(t)), \dot{\varphi}(t) \rangle_{H^1(\Omega)} \quad \text{f.a.a. } t \in (0,T).$$
 (2.69)

Let us now assume that  $\varphi$  fulfills (2.61) for all  $t \in [0, T]$ . Due to  $\widetilde{\mathcal{I}}(\cdot, \varphi(\cdot)) \in H^1(0, T)$ , the energy inequality reads for t = T as follows

$$\int_0^T \widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(\tau)) d\tau + \int_0^T \widetilde{\mathcal{R}}_{\delta}^*(-\partial_{\varphi}\widetilde{\mathcal{I}}(\tau,\varphi(\tau))) d\tau \leq -\int_0^T \frac{d}{dt}\widetilde{\mathcal{I}}(\tau,\varphi(\tau)) - \partial_t \widetilde{\mathcal{I}}(\tau,\varphi(\tau)) d\tau,$$

which in light of (2.69) has as consequence

$$\int_{0}^{T} \widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(\tau)) d\tau + \int_{0}^{T} \widetilde{\mathcal{R}}_{\delta}^{*}(-\partial_{\varphi}\widetilde{\mathcal{I}}(\tau,\varphi(\tau))) d\tau \leq -\int_{0}^{T} \langle \partial_{\varphi}\widetilde{\mathcal{I}}(\tau,\varphi(\tau)), \dot{\varphi}(\tau) \rangle_{H^{1}(\Omega)} d\tau.$$
(2.70)

On the other side, one deduces from the definition of the Fenchel conjugate the estimate

$$\widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(t)) + \widetilde{\mathcal{R}}_{\delta}^{*}(-\partial_{\varphi}\widetilde{\mathcal{I}}(t,\varphi(t))) \ge -\langle \partial_{\varphi}\widetilde{\mathcal{I}}(t,\varphi(t)), \dot{\varphi}(t) \rangle_{H^{1}(\Omega)} \quad \text{f.a.a. } t \in (0,T). \quad (2.71)$$

Combining (2.70) and (2.71) now yields

$$\widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(t)) + \widetilde{\mathcal{R}}_{\delta}^*(-\partial_{\varphi}\widetilde{\mathcal{I}}(t,\varphi(t))) = -\langle \partial_{\varphi}\widetilde{\mathcal{I}}(t,\varphi(t)), \dot{\varphi}(t) \rangle_{H^1(\Omega)}$$
 f.a.a.  $t \in (0,T)$ 

and with (2.68) we arrive at

$$-\partial_{\varphi}\widetilde{\mathcal{I}}(t,\varphi(t)) \in \partial \widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(t))$$
 f.a.a.  $t \in (0,T)$ .

The reverse assertion can be concluded by following the lines of the proof of Proposition 2.5. To this end, assume that  $\varphi$  satisfies (2.67). From (2.68) and (2.69) we then obtain

$$\widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(t)) + \widetilde{\mathcal{R}}_{\delta}^{*}(-\partial_{\varphi}\widetilde{\mathcal{I}}(t,\varphi(t))) = -\frac{d}{dt}\widetilde{\mathcal{I}}(t,\varphi(t)) + \partial_{t}\widetilde{\mathcal{I}}(t,\varphi(t)) \quad \text{f.a.a. } t \in (0,T). \quad (2.72)$$

Note that any  $\varphi$  which fulfills (2.67) satisfies automatically  $\dot{\varphi}(t) \geq 0$  f.a.a.  $t \in (0,T)$ , in view of Definition 2.21. The latter one also accounts for the integrability of  $\widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(\cdot))$ , as  $\dot{\varphi} \in L^2(0,T;L^2(\Omega))$ . For the terms on the right-hand side in (2.72) we have due to Lemma 2.20, the embedding  $H^1(0,T;H^1(\Omega)) \hookrightarrow C([0,T];H^1(\Omega))$  and since  $\widetilde{\mathcal{I}}(\cdot,\varphi(\cdot)) \in H^1(0,T)$  that  $\partial_t \widetilde{\mathcal{I}}(\cdot,\varphi(\cdot)) \in C[0,T]$  and  $\frac{d}{dt} \widetilde{\mathcal{I}}(\cdot,\varphi(\cdot)) \in L^2(0,T)$ , respectively. Thus, we are allowed to integrate (2.72) in time, which now implies for all  $t \in [0,T]$  the following

$$\begin{split} \int_0^t \widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(\tau)) \ d\tau + \int_0^t \widetilde{\mathcal{R}}_{\delta}^*(-\partial_{\varphi}\widetilde{\mathcal{I}}(\tau,\varphi(\tau))) \ d\tau \\ &= \widetilde{\mathcal{I}}(0,\varphi(0)) - \widetilde{\mathcal{I}}(t,\varphi(t)) + \int_0^t \partial_t \widetilde{\mathcal{I}}(\tau,\varphi(\tau)) \ d\tau. \end{split}$$

This gives the desired assertion.

Remark 2.27. An inspection of the proof of Proposition 2.26 shows that, in order to prove (2.67), it suffices that (2.61) holds only at t = T. In addition, at the end of the proof it can be seen that (2.67) leads to an energy equality where one can actually integrate over an arbitrary interval  $[s,t] \subset [0,T]$ . Altogether, this means that the energy inequality (2.61) for t = T is equivalent to the evolution (2.67), which is further equivalent to the corresponding energy identity on some arbitrary interval  $[s,t] \subset [0,T]$ . We refer here also to [41, Proposition 3.2], where a very similar result is proven.

Thus, the passage to the limit  $\beta \to \infty$  in (2.9) preserves the structure of the energy identity with penalty, in light of Proposition 2.25 and the above comments.

We summarize our results so far in the following

**Theorem 2.28** (Single-field gradient damage model). Let Assumptions 1.17.1, 1.47, 1.56 and 2.8 hold and  $\{\beta_n\}_{n\in\mathbb{N}}$  be a sequence with  $\beta_n \to \infty$  as  $n \to \infty$ . Then there

is a subsequence (denoted by the same symbol) and a pair  $(\boldsymbol{u},\varphi) \in C([0,T];V) \times H^1(0,T;H^1(\Omega))$  such that

$$\boldsymbol{u}_{\beta_n} \to \boldsymbol{u} \text{ in } C([0,T];V),$$
  
 $\varphi_{\beta_n} \rightharpoonup \varphi \text{ in } H^1(0,T;H^1(\Omega)), \quad d_{\beta_n} \rightharpoonup \varphi \text{ in } H^1(0,T;L^2(\Omega)),$ 

$$(2.73)$$

as  $n \to \infty$ , where  $(\mathbf{u}_{\beta_n}, \varphi_{\beta_n}, d_{\beta_n})$  is the unique solution of the problem  $(P_{\beta_n})$ . Moreover, every limit of such a sequence satisfies  $\mathbf{u} = \mathcal{U}(\cdot, \varphi(\cdot))$  and

$$\alpha \Delta \varphi(t) - F(t, \varphi(t)) - \delta \dot{\varphi}(t) \in \partial \widetilde{\mathcal{R}}_1(\dot{\varphi}(t)) \quad \text{f.a.a. } t \in (0, T), \quad \varphi(0) = 0, \tag{2.74}$$

which is equivalent to the following PDE system:

$$-\operatorname{div} g(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{u}(t)) = \ell(t) \quad \text{in } \boldsymbol{W}^{-1,p}(\Omega) \quad \forall t \in [0,T],$$

$$\alpha\Delta\varphi(t) - \frac{1}{2}g'(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{u}(t)) : \varepsilon(\boldsymbol{u}(t)) - \delta\dot{\varphi}(t) \in \partial\widetilde{\mathcal{R}}_{1}(\dot{\varphi}(t)) \quad \text{f.a.a. } t \in (0,T),$$

$$\varphi(0) = 0,$$

$$(2.75)$$

with the (non-viscous) dissipation functional  $\widetilde{\mathcal{R}}_1: H^1(\Omega) \to [0,\infty]$  defined by

$$\widetilde{\mathcal{R}}_{1}(\eta) := \begin{cases} r \int_{\Omega} \eta \ dx & \text{if } \eta \geq 0 \text{ a.e. in } \Omega, \\ \infty & \text{otherwise.} \end{cases}$$
 (2.76)

*Proof.* The first assertion, as well as  $\mathbf{u} = \mathcal{U}(\cdot, \varphi(\cdot))$ , were already proven in Propositions 2.17 and 2.18, while Propositions 2.25 and 2.26, combined with (2.49b) yield

$$\alpha \Delta \varphi(t) - F(t, \varphi(t)) \in \partial \widetilde{\mathcal{R}}_{\delta}(\dot{\varphi}(t)) \quad \text{f.a.a. } t \in (0, T), \quad \varphi(0) = 0.$$
 (2.77)

The above mentioned results also show that for any given sequence with (2.73), the limit satisfies (2.77). In view of the sum rule for convex subdifferentials, Definition 2.21 and the Fréchet-differentiability of  $\|\cdot\|_2^2: L^2(\Omega) \to \mathbb{R}$ , (2.77) is (2.74). The equivalence with (2.75) follows by the definitions of  $\mathcal{U}$  and F, see Definition 1.8 and (1.23), respectively.

The above theorem shows that (2.75) admits at least one solution. Of course, it would be desirable to prove its uniqueness too, since this guarantees in particular that, no matter which subsequence one chooses in Theorem 2.28, the limit function is the same, whence the (weak) convergence of the whole sequence. Unfortunately, for this purpose, we have to impose an additional assumption on the index p, which is rather restrictive. We underline that this is only needed to show the unique solvability of (2.75), while the rest of the analysis remains unaffected. For proving uniqueness for the evolution equation (2.74), it is necessary that the Sobolev embedding  $H^1(\Omega) \hookrightarrow L^{\frac{2p}{p-4}}(\Omega)$  holds true, as a closer inspection of the proof of Proposition 2.31 below shows. This is ensured if

**Assumption 2.29.** In the rest of the section we require that the assertion in Lemma 1.3 on page 12 holds for some p > 4 in the two-dimensional case and  $p \ge 6$  in the three-dimensional case.

Remark 2.30. Although restrictive, Assumption 2.29 is fulfilled cf. Remark 1.27 on page 25, provided that the domain possesses  $C^1$ -boundary, no mixed boundary conditions are present, and the difference between the boundedness and monotonicity constants of the stress-strain relation (1.7) is sufficiently small. Adapted to our situation, this means that the values  $\epsilon \gamma_{\mathbb{C}}$  and  $\|\mathbb{C}\|_{\infty}$  have to be close enough to each other, which is clearly rather restrictive (beside the assumptions on the domain). We refer here again to [29] for more details.

These assumptions on the data can be weakened, if one proceeds as in [41, Section 2.4] and uses the Sobolev-Slobodeckij space  $H^s(\Omega)$  with  $s \in (N/2,2)$ , instead of  $H^1(\Omega)$ , as space for the nonlocal damage in the penalized model  $(P_{\beta})$ . To this end, one replaces the gradient term in the energy functional given in Definition 0.2 by the seminorm generated by [41, (2.4b)], see also Remark 1.28 on page 25. Note that this tells us that the additional assumptions on p in Chapter 1 are no longer needed. A close inspection of the preceding analysis then shows that analogous results to the throughout this chapter proven results can be reached for the modified damage model. In particular, the limit function  $\varphi$  will then belong to the space  $H^1(0,T;H^s(\Omega))$ . The advantage thereof is that  $H^s(\Omega) \hookrightarrow C^{0,\zeta}(\bar{\Omega})$ for some  $\zeta \in (0,1]$  in both space dimensions, see e.g. [79]. In this case, it suffices to impose that p = 4 in both dimensions instead of Assumption 2.29. This new assumption is fulfilled e.g. by imposing smoothness assumptions on the domain and on  $\mathbb C$  and if no mixed boundary conditions are present, see [41, (2.40) and (2.43)]. One then obtains by [20, Theorem 10.17] and a classical density argument that for any  $\varrho \in [2, \infty)$ , the operator  $A_{\varphi(t)}: \mathbf{W}_0^{1,\varrho}(\Omega) \to \mathbf{W}_0^{1,\varrho'}(\Omega)^*$  is continuously invertible for any  $t \in [0,T]$ . However, unlike in Lemma 1.3,  $\|A_{\varphi(t)}^{-1}\|$  is now bounded by the term  $c\|\varphi\|_{H^1(0,T;H^s(\Omega))}$ (for all  $t \in [0,T]$ ), where c > 0 is a constant depending only on the given data. This is due to the fact that  $||A_{\varphi(t)}^{-1}||$  is bounded by  $c ||g(\varphi(t))||_{C^{0,\zeta}(\bar{\Omega})}$  and thus by  $c ||\varphi(t)||_{H^{s}(\Omega)}$ , in view of the Lipschitz continuity of g and  $H^s(\Omega) \hookrightarrow C^{0,\zeta}(\bar{\Omega})$ . This implies that the constant C in (2.79) below depends on  $\|\varphi_i\|_{H^1(0,T;H^s(\Omega))}$ , i=1,2. An inspection of the proof of Proposition 2.31 below shows however that this does not affect the uniqueness result for (2.67). We also refer here to [41, Sections 2.4 and 3.2], where one deals with a very similar situation. Since the bilinear form associated with  $H^s(\Omega)$  is harder to realize in numerical practice, we do not follow this approach.

**Proposition 2.31** (Unique solvability of (2.75)). Suppose that Assumptions 1.47, 1.56, 2.8 and 2.29 hold true. Then, the evolutionary equation (2.67) admits a unique solution  $\varphi \in H_0^1(0,T;H^1(\Omega))$  and thus, (2.75) is uniquely solvable.

*Proof.* The proof is similar to the proof of [41, Proposition 3.6.]. We notice that it suffices to show that (2.74) is uniquely solvable. This is due to its equivalence with (2.67) supplemented with  $\varphi(0) = 0$  and since (2.74) is just (2.75) written in compact from. To this end, let  $\varphi_1, \varphi_2 \in H^1(0, T; H^1(\Omega))$  be two solutions of (2.74). Note that their

existence is given by Theorem 2.28. Therefore, it holds  $\partial \widetilde{\mathcal{R}}_1(\dot{\varphi}_i(t)) \neq \emptyset$  f.a.a.  $t \in (0,T)$ , whence  $\dot{\varphi}_i(t) \geq 0$  f.a.a.  $t \in (0,T)$ , for any i=1,2. By testing (2.74) for i=1 with  $\dot{\varphi}_2$  and vice versa, we arrive at

$$\langle \alpha \Delta \varphi_1(t) - F(t, \varphi_1(t)) - \delta \dot{\varphi}_1(t), \dot{\varphi}_2(t) - \dot{\varphi}_1(t) \rangle_{H^1(\Omega)} \leq r \int_{\Omega} \dot{\varphi}_2(t) - \dot{\varphi}_1(t) \, dx$$
$$\langle \alpha \Delta \varphi_2(t) - F(t, \varphi_2(t)) - \delta \dot{\varphi}_2(t), \dot{\varphi}_1(t) - \dot{\varphi}_2(t) \rangle_{H^1(\Omega)} \leq r \int_{\Omega} \dot{\varphi}_1(t) - \dot{\varphi}_2(t) \, dx$$

f.a.a.  $t \in (0,T)$ . Adding the above estimates and then inserting the term  $\alpha(\varphi_1(t) - \varphi_2(t), \dot{\varphi}_1(t) - \dot{\varphi}_2(t))_2$  on both sides leads to

$$\delta \|\dot{\varphi}_{1}(t) - \dot{\varphi}_{2}(t)\|_{2}^{2} + \alpha(\varphi_{1}(t) - \varphi_{2}(t), \dot{\varphi}_{1}(t) - \dot{\varphi}_{2}(t))_{H^{1}(\Omega)} 
\leq \langle F(t, \varphi_{2}(t)) - F(t, \varphi_{1}(t)), \dot{\varphi}_{1}(t) - \dot{\varphi}_{2}(t) \rangle_{H^{1}(\Omega)} 
+ \alpha(\varphi_{1}(t) - \varphi_{2}(t), \dot{\varphi}_{1}(t) - \dot{\varphi}_{2}(t))_{2} \quad \text{f.a.a. } t \in (0, T).$$
(2.78)

Thanks to Assumption 2.29, we can apply to the first term on the right-hand side in (2.78) Lemma 1.18 with r:=2p/(p-4), so that s=2. By Cauchy-Schwarz inequality and the embeddings  $H^1(\Omega) \hookrightarrow L^r(\Omega)$  and  $H^1(\Omega) \hookrightarrow L^2(\Omega)$  we then find f.a.a.  $t \in (0,T)$  the following

$$\delta \|\dot{\varphi}_{1}(t) - \dot{\varphi}_{2}(t)\|_{2}^{2} + \alpha(\varphi_{1}(t) - \varphi_{2}(t), \dot{\varphi}_{1}(t) - \dot{\varphi}_{2}(t))_{H^{1}(\Omega)} 
\leq C \|\varphi_{1}(t) - \varphi_{2}(t)\|_{H^{1}(\Omega)} \|\dot{\varphi}_{1}(t) - \dot{\varphi}_{2}(t)\|_{2} 
\leq \frac{C}{4\varepsilon} \|\varphi_{1}(t) - \varphi_{2}(t)\|_{H^{1}(\Omega)}^{2} + C\varepsilon \|\dot{\varphi}_{1}(t) - \dot{\varphi}_{2}(t)\|_{2}^{2} \quad \forall \varepsilon > 0,$$
(2.79)

where the last estimate follows from the generalized Young inequality. Note that C > 0 denotes a constant independent of  $t, \varphi_1$  and  $\varphi_2$ , in view of Lemma 1.18. Further, we set  $\varepsilon$  (small enough) in (2.79) such that  $C\varepsilon < \delta$  holds, e.g.  $\varepsilon := \delta/(2C)$ , which now gives in turn

$$\alpha(\varphi_1(t) - \varphi_2(t), \dot{\varphi}_1(t) - \dot{\varphi}_2(t))_{H^1(\Omega)} \le \frac{C^2}{2\delta} \|\varphi_1(t) - \varphi_2(t)\|_{H^1(\Omega)}^2 \quad \text{f.a.a. } t \in (0, T).$$
 (2.80)

On the other side, by e.g. [81, Lemma 3.1.43], we have for all  $t \in [0, T]$ 

$$\int_0^t (\varphi_1(\tau) - \varphi_2(\tau), \dot{\varphi}_1(\tau) - \dot{\varphi}_2(\tau))_{H^1(\Omega)} d\tau = \frac{1}{2} \|\varphi_1(t) - \varphi_2(t)\|_{H^1(\Omega)}^2 - \frac{1}{2} \|\varphi_1(0) - \varphi_2(0)\|_{H^1(\Omega)}^2$$

and due to  $\varphi_1(0) = \varphi_2(0)$ , we arrive after integrating (2.80) at

$$\frac{\alpha}{2} \|\varphi_1(t) - \varphi_2(t)\|_{H^1(\Omega)}^2 \le \frac{C^2}{2\delta} \int_0^t \|\varphi_1(\tau) - \varphi_2(\tau)\|_{H^1(\Omega)}^2 d\tau \quad \forall t \in [0, T].$$
 (2.81)

In light of  $H^1(0,T;H^1(\Omega)) \hookrightarrow C([0,T];H^1(\Omega))$ , the map  $t \mapsto \|\varphi_1(t) - \varphi_2(t)\|_{H^1(\Omega)}^2$  is continuous and applying Gronwall's lemma for (2.81) leads to

$$\|\varphi_1(t) - \varphi_2(t)\|_{H^1(\Omega)}^2 \le 0 \quad \forall t \in [0, T],$$

whence  $\varphi_1 = \varphi_2$ . This completes the proof.

As an immediate consequence of Proposition 2.31 combined with Theorem 2.28 we infer

Corollary 2.32. If in addition to Assumptions 1.47, 1.56 and 2.8, Assumption 2.29 is fulfilled, then the convergence in (2.73) is not only valid for a subsequence, but for the whole sequence  $\{(u_{\beta_n}, \varphi_{\beta_n}, d_{\beta_n})\}$ .

## 2.4 $L^{\infty}$ -bound for the limit damage variable

The last essential step before proving that the damage model without penalty is equivalent to a version of the model analyzed by [41] consists of showing that the limit function  $\varphi$  in (2.75) belongs to  $L^{\infty}((0,T)\times\Omega)$ . This is needed to perform the transformation of one model into the other in the upcoming section.

To this end, we turn back to the problem  $(P_{\beta})$  and show that, under mild assumptions, the local and nonlocal damage are bounded a.e. in  $(0,T) \times \Omega$  by a constant independent of  $\beta$ . The convergence (2.73) will then ensure that at least one of the solutions of (2.74) is essentially bounded. For the optimal damage variables associated to  $(P_{\beta})$ , the existence of  $L^{\infty}$ -bounds was indeed already established in case of N=2, see Remark 1.46, but we emphasize that these are dependent on  $\beta$ . To show the desired boundedness, we mostly follow the ideas of [41], where the authors derive a similar result in two dimensions under comparative assumptions, see [41, Proposition 4.5]. We prove the claim via a time-discretization procedure, by passing to the limit in a suitable time-discretization scheme for the problem  $(P_{\beta})$  and by showing that the discrete local and nonlocal damage variables possess (a.e. in  $(0,T) \times \Omega$ ) a uniform (w.r.t.  $\beta$  and time-step size) bound, see Lemma 2.38 below. As it will turn out, the value of the bound can be ultimately read from the properties of the function g.

In the sequel,  $\beta$  is fixed and large enough cf. Assumption 1.17.2. We omit for the sake of convenience the symbol for the dependence on the penalty term of the solution of  $(P_{\beta})$  and of the associated discrete solutions. We only highlight the dependence of the latter ones on the different time-step sizes.

Recall that, by Proposition 2.7, the damage model with penalty reduces to the operator differential equation

$$\dot{d}(t) = \frac{1}{\delta} \max \left( -\beta (d(t) - \varphi(t)) - r \right) \quad \forall t \in [0, T], \quad d(0) = d_0, \tag{2.82}$$

where d and  $\varphi = \Phi(\cdot, d(\cdot))$  stand for the local and nonlocal damage, respectively. This notation and the upcoming ones are valid throughout this section.

Starting from (2.82) we introduce the following time-discrete incremental problem: Given the number of time-steps  $n \in \mathbb{N}^+$ , we set  $\tau := T/n$  and denote by  $\{t_k^{\tau} = k\tau\}_{k=0,...,n}$  the corresponding partition of the time interval [0,T]. Further we define, beginning with  $d_0^{\tau} := d_0$ , the approximation of the local damage at time point  $t_{k+1}^{\tau}$ ,  $k \in \{0,...,n-1\}$ , as the unique solution of the fixed-point equation

$$d_{k+1}^{\tau} = d_k^{\tau} + \frac{\tau}{\delta} \max \left( -\beta (d_{k+1}^{\tau} - \Phi(t_{k+1}^{\tau}, d_{k+1}^{\tau})) - r \right). \tag{$P_k^{\beta, \tau}$}$$

Note that the unique solvability of  $(P_k^{\beta,\tau})$  can be easily concluded for any  $k \in \{0,...,n-1\}$  from Banach fixed-point theorem. To see this, we observe that

$$\frac{\tau}{\delta} \| \max(-\beta(z_1 - \Phi(t_{k+1}^{\tau}, z_1)) - r) - \max(-\beta(z_2 - \Phi(t_{k+1}^{\tau}, z_2)) - r) \|_2$$

$$\leq \frac{\tau}{\delta} \| - \beta(z_1 - \Phi(t_{k+1}^{\tau}, z_1)) + \beta(z_2 - \Phi(t_{k+1}^{\tau}, z_2)) \|_2$$

$$\leq \frac{\tau}{\delta} (\beta + \beta L) \|z_1 - z_2\|_2 \quad \forall z_1, z_2 \in L^2(\Omega),$$

by the Lipschitz continuity of max and  $\Phi$  (with constant L), cf. Lemma 5.6(i) and (1.38), respectively. Hence, if  $\tau < \delta/(\beta + \beta L)$ , then the function  $L^2(\Omega) \ni z \mapsto d_k^{\tau} + \frac{\tau}{\delta} \max\left(-\beta(z - \Phi(t_{k+1}^{\tau}, z)) - r\right) \in L^2(\Omega)$  is a contraction. This is also well defined, and thus,  $(P_k^{\beta,\tau})$  admits an unique solution  $d_{k+1}^{\tau} \in L^2(\Omega)$ , provided that  $\tau$  is small enough. The latter one is supposed to hold throughout this section, as we however aim to pass to the limit  $\tau \searrow 0$ .

#### 2.4.1 Passage to the limit in the time-discrete problem

Let the number of steps n be fixed and large enough. Before we proceed with analyzing the behaviour of the time-discrete solutions as  $\tau \searrow 0$ , we introduce, similarly to [41, Section 4], the notations

$$\bar{t}_{\tau}(t) := t_{k+1}^{\tau} \text{ for } t \in (t_k^{\tau}, t_{k+1}^{\tau}], \ \underline{t}_{\tau}(t) := t_k^{\tau} \quad \text{for } t \in [t_k^{\tau}, t_{k+1}^{\tau}), \quad k \in \{0, ..., n-1\},$$

 $\bar{t}_{\tau}(0) := 0$  and  $\underline{t}_{\tau}(T) := T$ . We define the piecewise constant interpolation functions  $\bar{d}_{\tau}, \underline{d}_{\tau} : [0, T] \to L^2(\Omega)$  by

$$\bar{d}_{\tau}(t) := d_{k+1}^{\tau} \text{ for } t \in (t_k^{\tau}, t_{k+1}^{\tau}], \ \underline{d}_{\tau}(t) := d_k^{\tau} \quad \text{for } t \in [t_k^{\tau}, t_{k+1}^{\tau}), \quad k \in \{0, ..., n-1\},$$

 $\bar{d}_{\tau}(0) := d_0$  and  $\underline{d}_{\tau}(T) := d_n$ , as well as the piecewise linear interpolation function  $d_{\tau} : [0,T] \to L^2(\Omega)$  as

$$d_{\tau}(t) := d_k^{\tau} + \frac{t - t_k^{\tau}}{\tau} (d_{k+1}^{\tau} - d_k^{\tau}) \quad \text{for } t \in [t_k^{\tau}, t_{k+1}^{\tau}], \quad k \in \{0, ..., n-1\}.$$

Notice that  $d_{\tau}$  is differentiable on  $[0,T]\setminus\{t_0,t_1,...,t_n\}$  with

$$\dot{d}_{\tau}(t) = \frac{d_{k+1}^{\tau} - d_{k}^{\tau}}{\tau} \quad \text{for } t \in (t_{k}^{\tau}, t_{k+1}^{\tau}), \quad k \in \{0, ..., n-1\},$$
(2.83)

which implies in view of  $(P_k^{\beta,\tau})$  that for  $t \in [0,T] \setminus \{t_0,t_1,...,t_n\}$  it holds

$$\dot{d}_{\tau}(t) = \frac{1}{\delta} \max\left(-\beta \left(\bar{d}_{\tau}(t) - \Phi(\bar{t}_{\tau}(t), \bar{d}_{\tau}(t))\right) - r\right). \tag{2.84}$$

To prove convergence in the above time-discretization scheme, we need the following

**Lemma 2.33.** Let Assumptions 1.5 and 1.17 hold true and suppose that  $\tau < \delta/(\beta + \beta L)$  is small enough, depending only on the given data. Then, there exists a constant C > 0, independent of  $\tau$ , such that for all  $t \in [0,T] \setminus \{t_0,t_1,...,t_n\}$  it holds

$$\|\dot{d}_{\tau}(t)\|_{2} \le C.$$

*Proof.* The proof is inspired by the proof of [41, Proposition 4.2]. We aim to estimate  $\|\dot{d}_{\tau}(t)\|_{2}^{2}$  in such a way that (the discrete version of) Gronwall's lemma can be applied. Let  $k \in \{1, ..., n-1\}$  be arbitrary, but fixed, where n denotes the number of time steps. Then by testing (2.84) for fixed  $t \in (t_{k}^{\tau}, t_{k+1}^{\tau})$  and fixed  $s \in (t_{k-1}^{\tau}, t_{k}^{\tau})$  with  $\dot{d}_{\tau}(t)$  we have

$$\begin{aligned} \|\dot{d}_{\tau}(t)\|_{2}^{2} &= \frac{1}{\delta} \left( \max \left( -\beta \left( \bar{d}_{\tau}(t) - \Phi(\bar{t}_{\tau}(t), \bar{d}_{\tau}(t)) \right) - r \right), \dot{d}_{\tau}(t) \right)_{2}, \\ (\dot{d}_{\tau}(s), \dot{d}_{\tau}(t))_{2} &= \frac{1}{\delta} \left( \max \left( -\beta \left( \bar{d}_{\tau}(s) - \Phi(\bar{t}_{\tau}(s), \bar{d}_{\tau}(s)) \right) - r \right), \dot{d}_{\tau}(t) \right)_{2}. \end{aligned}$$

By subtracting the second identity from the first one and by using

$$\|\dot{d}_{\tau}(t)\|_{2}^{2} - \|\dot{d}_{\tau}(s)\|_{2}^{2} \le 2\|\dot{d}_{\tau}(t)\|_{2}^{2} - 2(\dot{d}_{\tau}(s), \dot{d}_{\tau}(t))_{2},$$

we arrive at

$$\delta/2(\|\dot{d}_{\tau}(t)\|_{2}^{2} - \|\dot{d}_{\tau}(s)\|_{2}^{2}) 
\leq (\max(-\beta(\bar{d}_{\tau}(t) - \Phi(\bar{t}_{\tau}(t), \bar{d}_{\tau}(t))) - r) - \max(-\beta(\bar{d}_{\tau}(s) - \Phi(\bar{t}_{\tau}(s), \bar{d}_{\tau}(s))) - r), \dot{d}_{\tau}(t))_{2} 
\leq (\beta\|\bar{d}_{\tau}(t) - \bar{d}_{\tau}(s)\|_{2} + \beta L(\tau + \|\bar{d}_{\tau}(t) - \bar{d}_{\tau}(s)\|_{2}))\|\dot{d}_{\tau}(t)\|_{2} 
= (c\tau + C\|\bar{d}_{\tau}(t) - \bar{d}_{\tau}(s)\|_{2})\|\dot{d}_{\tau}(t)\|_{2},$$
(2.85)

where  $c = \beta L$  and  $C = \beta(L+1)$  are constants independent of  $\tau$ . Note that for the last inequality we used the Lipschitz continuity of max and  $\Phi$ , cf. Lemma 5.6(i) and (1.38), respectively, as well as  $\bar{t}_{\tau}(t) - \bar{t}_{\tau}(s) = \tau$ . Relying on (2.83),  $\bar{d}_{\tau}(t) = d_{k+1}^{\tau}$  and  $\bar{d}_{\tau}(s) = d_{k}^{\tau}$ , (2.85) can be continued as

$$\delta/2(\|\dot{d}_{\tau}(t)\|_{2}^{2} - \|\dot{d}_{\tau}(s)\|_{2}^{2}) \leq (c\tau + C\tau \|\dot{d}_{\tau}(t)\|_{2}) \|\dot{d}_{\tau}(t)\|_{2} 
\leq C\tau + C\tau \|\dot{d}_{\tau}(t)\|_{2}^{2} \quad \forall t \in (t_{k}^{\tau}, t_{k+1}^{\tau}), \ \forall s \in (t_{k-1}^{\tau}, t_{k}^{\tau}),$$
(2.86)

where for the last estimate we employed Young's inequality. Further, notice that as a consequence of (2.83), we can write

$$\dot{d}_{\tau}(\rho) = \dot{d}_{\tau} \left( \frac{t_j^{\tau} + t_{j+1}^{\tau}}{2} \right) \quad \forall \, \rho \in (t_j^{\tau}, t_{j+1}^{\tau})_{j=0,\dots,n-1}.$$
 (2.87)

Now let  $t \in [t_1, T] \setminus \{t_1, ..., t_n\}$  be arbitrary, but fixed, which implies that there exists some  $m \in \{1, ..., n-1\}$  such that  $t \in (t_m, t_{m+1})$ . By adding (2.86) for k = 1, ..., m we arrive at

$$\|\dot{d}_{\tau}(t)\|_{2}^{2} \leq \left\|\dot{d}_{\tau}\left(\frac{\tau}{2}\right)\right\|_{2}^{2} + mC\tau + C\tau \sum_{k=1}^{m} \left\|\dot{d}_{\tau}\left(\frac{t_{k}^{\tau} + t_{k+1}^{\tau}}{2}\right)\right\|_{2}^{2},\tag{2.88}$$

in view of (2.87). From the estimate (2.88) we want to conclude with the discrete version of Gronwall's lemma the assertion. Therefor, we need to find an independent of  $\tau$  bound

for  $\|\dot{d}_{\tau}(\frac{\tau}{2})\|_{2}$ . To this end, we test (2.84) at time point  $\tau/2$  with  $\dot{d}_{\tau}(\tau/2)$  and deduce  $\|\dot{d}_{\tau}(\tau/2)\|_{2}^{2} = \frac{1}{\delta} \left( \max\left(-\beta \left(\bar{d}_{\tau}(\tau/2) - \Phi(\bar{t}_{\tau}(\tau/2), \bar{d}_{\tau}(\tau/2)\right)\right) - r\right), \dot{d}_{\tau}(\tau/2) \right)_{2}$   $= \frac{1}{\delta} \left( \max\left(-\beta \left(\bar{d}_{\tau}(\tau/2) - \Phi(\bar{t}_{\tau}(\tau/2), \bar{d}_{\tau}(\tau/2)\right)\right) - r\right) - \max\left(-\beta \left(d_{0} - \varphi_{0}\right) - r\right), \dot{d}_{\tau}(\tau/2) \right)_{2}$   $+ \frac{1}{\delta} \left( \max(-\beta (d_{0} - \varphi_{0}) - r), \dot{d}_{\tau}(\tau/2) \right)_{2},$ 

where  $\varphi_0 := \Phi(0, d_0)$ . By relying on the same arguments used for deriving (2.85) and (2.86), as well as Cauchy-Schwarz inequality, we then infer

$$\|\dot{d}_{\tau}(\tau/2)\|_{2}^{2} \leq c\tau + C\tau \|\dot{d}_{\tau}(\tau/2)\|_{2}^{2} + C\|\beta(d_{0} - \varphi_{0}) + r\|_{2}\|\dot{d}_{\tau}(\tau/2)\|_{2}$$

$$\leq c\tau + C\tau \|\dot{d}_{\tau}(\tau/2)\|_{2}^{2},$$
(2.89)

with c, C > 0 independent of  $\tau$ . Note that for the last estimate we employed again Young's inequality. Since  $\tau \searrow 0$  later anyway, we may choose  $\tau \le \frac{1}{2C}$  in (2.89). This leads to  $\|\dot{d}_{\tau}(\tau/2)\|_2^2 \le 2c\tau \le c/C$ , which we insert in (2.88), giving in turn

$$\|\dot{d}_{\tau}(t)\|_{2}^{2} \le c + C\tau \sum_{k=1}^{m} \left\|\dot{d}_{\tau}\left(\frac{t_{k}^{\tau} + t_{k+1}^{\tau}}{2}\right)\right\|_{2}^{2} \quad \text{for } t \in (t_{m}, t_{m+1}),$$

where we used  $m \leq n-1$  and  $\tau = T/n$ . Since  $\dot{d}_{\tau}(t) = \dot{d}_{\tau}\left(\frac{t_m + t_{m+1}}{2}\right)$ , cf. (2.87), and  $\tau \leq \frac{1}{2C}$ , we further have

$$\left\| \dot{d}_{\tau} \left( \frac{t_m + t_{m+1}}{2} \right) \right\|_2^2 \le 2c + 2C\tau \sum_{k=1}^{m-1} \left\| \dot{d}_{\tau} \left( \frac{t_k^{\tau} + t_{k+1}^{\tau}}{2} \right) \right\|_2^2$$
 (2.90)

with the convention  $\sum_{k=1}^{0} = 0$ . Note that (2.90) holds for any  $m \in \{1, ..., n-1\}$ , since  $t \in [t_1, T] \setminus \{t_1, ..., t_n\}$  was arbitrary. Thus, the discrete version of Gronwall's lemma applied for the sequence  $\{\omega_k\}$  with  $\omega_k := \left\|\dot{d}_{\tau}\left(\frac{t_k^{\tau} + t_{k+1}^{\tau}}{2}\right)\right\|_2^2$  now tells us that

$$\left\|\dot{d}_{\tau}\left(\frac{t_{m}+t_{m+1}}{2}\right)\right\|_{2}^{2} \le c \exp\left(\sum_{k=1}^{m-1} C\tau\right) \le c \exp(CT) \quad \forall m \in \{1,...,n-1\},$$

in view of  $\tau = T/n$ . We observe that the constant on the right-hand side is independent of  $\tau$ . Thanks to (2.87), this means that

$$\|\dot{d}_{\tau}(t)\|_{2}^{2} \leq C$$
 for all  $t \in [t_{1}, T] \setminus \{t_{1}, ..., t_{n}\},$ 

which together with  $\|\dot{d}_{\tau}(\tau/2)\|_2^2 \leq c/C$  (see above), gives the assertion.

The main result in this subsection is given by

**Proposition 2.34** (Convergence of the time-discretization). Suppose that Assumptions 1.5 and 1.17 are fulfilled and let d be the unique solution of (2.82) and  $\varphi := \Phi(\cdot, d(\cdot))$ . Then, the following convergence holds true

$$d_{\tau} \to d$$
 in  $W^{1,\infty}(0,T;L^2(\Omega))$  as  $\tau \searrow 0$ .

Moreover,

$$\bar{d}_{\tau} \to d \text{ in } L^{\infty}(0,T;L^{2}(\Omega)) \text{ and } \bar{\varphi}_{\tau} \to \varphi \text{ in } L^{\infty}(0,T;H^{1}(\Omega)) \text{ as } \tau \searrow 0,$$
 (2.91)  
where  $\bar{\varphi}_{\tau}(t) := \Phi(\bar{t}_{\tau}(t),\bar{d}_{\tau}(t)) \text{ for all } t \in [0,T].$ 

Proof. Let  $\tau < \delta/(\beta + \beta L)$  be arbitrary, but fixed. To make sure that the desired convergences make sense, let us first investigate the regularity of  $d_{\tau}$ ,  $\bar{d}_{\tau}$  and  $\bar{\varphi}_{\tau}$ . Clearly,  $\bar{d}_{\tau}: [0,T] \to L^2(\Omega)$  and  $\bar{\varphi}_{\tau}: [0,T] \to H^1(\Omega)$  are Bochner simple functions, and thus, they belong to  $L^{\infty}(0,T;L^2(\Omega))$  and  $L^{\infty}(0,T;H^1(\Omega))$ , respectively. Since  $L^2(\Omega)$  is reflexive Banach space and  $d_{\tau}$  is almost everywhere differentiable with  $\dot{d}_{\tau} = 1/\tau(\bar{d}_{\tau} - \underline{d}_{\tau})$  a.e. in (0,T), we obtain that  $d_{\tau} \in W^{1,\infty}(0,T;L^2(\Omega))$ , see e.g. [81, p. 58]. Note that here we used that  $\bar{d}_{\tau} - \underline{d}_{\tau} \in L^{\infty}(0,T;L^2(\Omega))$  as difference of Bochner simple functions. Thereby, we deduce that indeed

$$d_{\tau} \in W^{1,\infty}(0,T;L^2(\Omega)), \quad \bar{d}_{\tau} \in L^{\infty}(0,T;L^2(\Omega)) \text{ and } \bar{\varphi}_{\tau} \in L^{\infty}(0,T;H^1(\Omega)).$$
 (2.92)

We begin by deriving an estimate which will be very useful in what follows. First note that for all  $t \in [0, T]$  we have

$$d_{\tau}(t) - \bar{d}_{\tau}(t) = \left(1 - \frac{t - \underline{t}_{\tau}(t)}{\tau}\right) (\underline{d}_{\tau}(t) - \bar{d}_{\tau}(t)).$$

Hence, due to  $\frac{t-\underline{t}_{\tau}(t)}{\tau} \in [0,1]$  for all  $t \in [0,T]$  and (2.83) it holds

$$||d(t) - \bar{d}_{\tau}(t)||_{2} \leq ||d(t) - d_{\tau}(t)||_{2} + ||d_{\tau}(t) - \bar{d}_{\tau}(t)||_{2}$$

$$\leq ||d(t) - d_{\tau}(t)||_{2} + ||\bar{d}_{\tau}(t) - \underline{d}_{\tau}(t)||_{2},$$

$$= ||d(t) - d_{\tau}(t)||_{2} + \tau ||\dot{d}_{\tau}(t)||_{2}$$

$$\leq ||d(t) - d_{\tau}(t)||_{2} + \tau C \quad \text{f.a.a. } t \in (0, T),$$

$$(2.93)$$

where C > 0 is the constant from Lemma 2.33.

Now, subtracting (2.84) from (2.82) yields f.a.a.  $t \in (0,T)$ 

$$\dot{d}(t) - \dot{d}_{\tau}(t) = \frac{1}{\delta} \left( \max \left( -\beta (d(t) - \varphi(t)) - r \right) - \max \left( -\beta \left( \bar{d}_{\tau}(t) - \Phi(\bar{t}_{\tau}(t), \bar{d}_{\tau}(t)) \right) - r \right) \right). \tag{2.94}$$

By making again use of the Lipschitz continuity of max and  $\Phi$  we obtain from (2.94) and (2.93) the estimate

$$\|\dot{d}(t) - \dot{d}_{\tau}(t)\|_{2} \leq \frac{1}{\delta} \left(\beta \|d(t) - \bar{d}_{\tau}(t)\|_{2} + \beta L(|t - \bar{t}_{\tau}(t)| + \|d(t) - \bar{d}_{\tau}(t)\|_{2})\right)$$

$$\leq \beta (L+1)/\delta \|d(t) - \bar{d}_{\tau}(t)\|_{2} + \tau \beta L/\delta$$

$$\leq c \|d(t) - d_{\tau}(t)\|_{2} + \tau c \quad \text{f.a.a. } t \in (0, T),$$

$$(2.95)$$

where c > 0 is independent of  $\tau$ . Note that for the second inequality we employed the definition of  $\bar{t}_{\tau}$ . Since  $d(0) = d_{\tau}(0)$  and  $d - d_{\tau} \in W^{1,\infty}(0,T;L^2(\Omega))$ , we can apply Lemma 5.10 for (2.95), which results in

$$||d - d_{\tau}||_{W^{1,\infty}(0,T;L^{2}(\Omega))} \le K(c,T)c\tau,$$

whence the convergence

$$d_{\tau} \to d$$
 in  $W^{1,\infty}(0,T;L^2(\Omega))$  as  $\tau \searrow 0$ .

From (2.93) we then conclude

$$\bar{d}_{\tau} \to d \text{ in } L^{\infty}(0,T;L^{2}(\Omega)) \text{ as } \tau \searrow 0.$$

The Lipschitz continuity of  $\Phi$ , see (1.38), leads to

$$\|\bar{\varphi}_{\tau}(t) - \Phi(t, d(t))\|_{H^{1}(\Omega)} \le L(\tau + \|\bar{d}_{\tau} - d\|_{L^{\infty}(0, T; L^{2}(\Omega))})$$
 f.a.a.  $t \in (0, T)$ ,

and thus,  $\bar{\varphi}_{\tau} \to \varphi$  in  $L^{\infty}(0,T;H^{1}(\Omega))$  as  $\tau \searrow 0$ . This completes the proof.

#### 2.4.2 Uniform estimates for the discrete damage variables

In this subsection we focus on proving that the piecewise constant interpolation function  $\bar{d}_{\tau}$  is bounded a.e. in  $(0,T)\times\Omega$  by a constant independent of  $\beta$  and of  $\tau$ . This constant can be a priori chosen by imposing corresponding assumptions on the data, from which one can read its precise value. As already mentioned above, this will also bound (a.e. in  $(0,T)\times\Omega$ ) the limit function  $\varphi$  in (2.75), which is our final goal in this section.

To prove the result, we follow the ideas of the proof of [41, Proposition 4.5], that is, we show by induction on the index k that the solution of  $(P_k^{\beta,\tau})$ , i.e.,  $d_{k+1}^{\tau}$ , satisfies the desired boundedness condition for any  $k \in \{0, ..., n-1\}$ , see proof of Lemma 2.38 below. Therefor we first need to rewrite  $(P_k^{\beta,\tau})$  as an equivalent (uniquely solvable) minimization problem:

**Lemma 2.35.** Suppose that Assumptions 1.5, 1.17 and 1.56 are fulfilled. Let the number of time-steps n be large enough, fixed and let  $k \in \{0, ..., n-1\}$  be given. Then, the solution of  $(P_k^{\beta, \tau})$  is the unique minimizer of

$$\min_{\substack{v \in L^2(\Omega), \\ v \ge d_k^{\tau}}} \mathcal{I}(t_{k+1}^{\tau}, v) + \tau \mathcal{R}_{\delta}\left(\frac{v - d_k^{\tau}}{\tau}\right). \tag{2.96}$$

*Proof.* (i) We begin by addressing the existence of solutions for (2.96). To this end, we define for simplicity  $f: \mathcal{C} \to \mathbb{R}$  as

$$f(v) := \mathcal{I}(t_{k+1}^{\tau}, v) + r \int_{\Omega} v - d_k^{\tau} \, dx + \frac{\delta}{2} \frac{\|v - d_k^{\tau}\|_2^2}{\tau}, \tag{2.97}$$

where  $\mathcal{C} := \{v \in L^2(\Omega) : v \geq d_k^{\tau} \text{ a.e. in } \Omega\}$ . Note that f coincides with the objective in (2.96) on  $\mathcal{C}$ , as a result of Definition 0.4. We also observe that

$$f(v) = -\frac{1}{2} \langle \ell(t_{k+1}^{\tau}), \mathcal{U}(t_{k+1}^{\tau}, \Phi(t_{k+1}^{\tau}, v)) \rangle_{V} + \frac{\alpha}{2} \|\nabla \Phi(t_{k+1}^{\tau}, v)\|_{2}^{2} + \frac{\beta}{2} \|\Phi(t_{k+1}^{\tau}, v) - v\|_{2}^{2} + r \int_{\Omega} v - d_{k}^{\tau} dx + \frac{\delta}{2} \frac{\|v - d_{k}^{\tau}\|_{2}^{2}}{\tau} \quad \text{for all } v \in \mathcal{C},$$

$$(2.98)$$

in view of (2.1). The existence of solutions for (2.96) follows by classical arguments of the direct method of variational calculus. We begin by noticing that  $\mathcal{C}$  is convex and closed, and thus, weakly closed, while f is radially unbounded, in light of (2.98) combined with Corollary 1.9. Moreover, it is weakly lower semicontinuous, as we will next establish.

To see this, we first prove that  $\Phi(t_{k+1}^{\tau},\cdot)$  is weakly continuous. We consider a sequence  $\{v_j\}\subset\mathcal{C}$  and  $v\in\mathcal{C}$  such that  $v_j\rightharpoonup v$  in  $L^2(\Omega)$  as  $j\to\infty$  and we abbreviate  $\varphi_j:=\Phi(t_{k+1}^{\tau},v_j)$  for the sake of convenience. Since  $\{v_j\}$  is bounded by a constant independent of j and since  $\Phi$  is Lipschitz continuous, see (1.38), we get by the reflexivity of  $H^1(\Omega)$  that there exists a subsequence of  $\{\varphi_j\}$ , denoted by the same symbol, and  $\widetilde{\varphi}\in H^1(\Omega)$ , with

$$\varphi_j \to \widetilde{\varphi} \quad \text{in } H^1(\Omega) \quad \text{as } j \to \infty.$$
 (2.99)

Due to Assumption 1.17.1, the compact embedding  $H^1(\Omega) \hookrightarrow \hookrightarrow L^{2p/(p-2)}(\Omega)$  holds true, cf. (1.32). Then, we obtain in view of Lemma 1.18 that

$$||F(t_{k+1}^{\tau}, \varphi_j) - F(t_{k+1}^{\tau}, \widetilde{\varphi})||_{H^1(\Omega)^*} \le C||\varphi_j - \widetilde{\varphi}||_{2p/(p-2)} \to 0 \quad \text{as } j \to \infty,$$

whence

$$B\varphi_i + F(t_{k+1}^{\tau}, \varphi_i) \rightharpoonup B\widetilde{\varphi} + F(t_{k+1}^{\tau}, \widetilde{\varphi}) \quad \text{in } H^1(\Omega)^* \text{ as } j \to \infty,$$

where we also used (1.22) and (2.99). Further, the definition of  $\varphi_j$  and Definition 1.24 imply that  $\beta v_j = B\varphi_j + F(t_{k+1}^\tau, \varphi_j)$ , while the uniqueness of the above weak limit yields  $\beta v = B\widetilde{\varphi} + F(t_{k+1}^\tau, \widetilde{\varphi})$ , since  $v_j \rightharpoonup v$  in  $L^2(\Omega)$  by assumption. However, by Definition 1.24, this leads to  $\widetilde{\varphi} = \Phi(t_{k+1}^\tau, v)$ . From (2.99) we now have

$$\Phi(t_{k+1}^{\tau}, v_j) \rightharpoonup \Phi(t_{k+1}^{\tau}, v) \quad \text{in } H^1(\Omega) \quad \text{as } j \to \infty.$$
(2.100)

Note that (2.100) holds for the entire sequence, as the limit  $\Phi(t_{k+1}^{\tau}, v)$  is independent of the chosen subsequence. Hence,  $L^2(\Omega) \ni v \mapsto \Phi(t_{k+1}^{\tau}, v) \in H^1(\Omega)$  is weakly continuous, as claimed.

To ultimately conclude the weak lower semicontinuity of f we rely in view of (2.98) on (2.100) combined with the arguments used in the proof of Proposition 1.12, and the weak lower semicontinuity of the norm. As  $L^2(\Omega)$  is reflexive Banach space, we finally have all the necessary tools for applying the direct method of variational calculus, from which we deduce that (2.96) admits solutions.

(ii) We now turn our attention to the unique solvability of (2.96). To this end, let z be an arbitrary, but fixed, solution thereof. By taking a look at (2.97), one sees in

view of Lemma 2.2 that f is Fréchet-differentiable at  $z \in \mathcal{C}$ . Notice that here we need Assumption 1.56 to be fulfilled, as this guarantees the existence of the derivative of  $\Phi$ , and thus of  $\mathcal{I}$ , w.r.t. the  $L^2$ -variable, as shown by a short inspection of Subsection 1.3.2 and the proof of Lemma 2.2, respectively. Thus, since  $\mathcal{C}$  is convex, z satisfies the following necessary optimality condition

$$f'(z; v - z) \ge 0 \quad \forall v \in \mathcal{C},$$

which in view of (2.2b) and (2.97) reads

$$\left(\beta(z - \Phi(t_{k+1}^{\tau}, z)) + r + \frac{\delta}{\tau}(z - d_k^{\tau}), v - z\right)_2 \ge 0 \quad \text{for all } v \in \mathcal{C}.$$
 (2.101)

Observe that  $2z-d_k^{\tau},\,d_k^{\tau}\in\mathcal{C}$  and by testing (2.101) therewith, one has

$$\left(\beta(z - \Phi(t_{k+1}^{\tau}, z)) + r + \frac{\delta}{\tau}(z - d_k^{\tau}), z - d_k^{\tau}\right)_2 = 0. \tag{2.102}$$

Now let  $w \in L^2(\Omega)$  with  $w \ge 0$  be arbitrary, but fixed and test (2.101) by  $w + z \in \mathcal{C}$ . Then, fundamental lemma of the calculus of variations yields

$$\beta(z - \Phi(t_{k+1}^{\tau}, z)) + r + \frac{\delta}{\tau}(z - d_k^{\tau}) \ge 0$$
 a.e. in  $\Omega$ ,

which together with (2.102) and  $z \geq d_k^{\tau}$  gives

$$0 \le \frac{\delta}{\tau}(z - d_k^{\tau}) \perp \beta(z - \Phi(t_{k+1}^{\tau}, z)) + r + \frac{\delta}{\tau}(z - d_k^{\tau}) \ge 0 \quad \text{ a.e. in } \Omega.$$

Since the max-function is a well known complementarity function, we infer that z solves  $(P_k^{\beta,\tau})$  and since the latter one is uniquely solvable, the claim is now proven.

In order to obtain the desired boundedness result, we impose the following

**Assumption 2.36.** There exists M > 0 such that  $g(x) \ge g(M)$  for all  $x \ge M$ .

Recall that the function  $g: \mathbb{R} \to [\epsilon, 1]$  shows in what measure the elastic properties of the body are preserved under the influence of the damage. Thus, Assumption 2.36 says that whenever the damage is larger than M, the material rigidity of the body is at least g(M). Note that from a mechanical point of view, it makes sense to impose (in addition to g monotonically decreasing, see page 6) that if the damage surpasses the value M, then the body preserves the degree of material rigidity g(M), i.e., g'(x) = 0 for all  $x \geq M$ , which is just a special case of Assumption 2.36.

By taking a look at (2.115) below, which shows how the coefficient function g is connected to its counterpart in [41], we see that Assumption 2.36 corresponds to the assumption made on the analogous function in [41, Proposition 4.5], where one shows a similar boundedness result.

The following lemma plays a key role for proving the main result in this section.

Lemma 2.37. Under Assumptions 1.5, 1.17 and 2.36, we have

$$\mathcal{I}(t, \min(z, M)) \le \mathcal{I}(t, z) \quad \forall (t, z) \in [0, T] \times L^2(\Omega).$$

Moreover, if there exists  $z \in L^2(\Omega)$  with  $\min(z, M) = z$ , then  $\min(\Phi(t, z), M) = \Phi(t, z)$  for any  $t \in [0, T]$ . Here  $\min(\cdot, M)$  stands for the Nemytskii operator associated to  $\min\{\cdot, M\} : \mathbb{R} \to \mathbb{R}$ .

*Proof.* Let us first notice that the operator  $\min(\cdot, M)$  maps  $L^2(\Omega)$  to  $L^2(\Omega)$  and is Lipschitz continuous with Lipschitz constant 1. This follows from the identity

$$\min\{a, M\} = -\max\{M - a, 0\} + M \quad \forall \, a \in \mathbb{R}$$
 (2.103)

and the properties of the max-operator, see Lemma 5.6(i). Moreover, in view of [38, Theorem A.1] and (2.103),  $\min(\cdot, M)$  maps  $H^1(\Omega)$  to  $H^1(\Omega)$  and it holds for all  $\phi \in H^1(\Omega)$ 

$$\nabla \min(\phi, M) = -\nabla \max(M - \phi) = \begin{cases} \nabla \phi & \text{a.e. in } \{x \in \Omega : \phi(x) < M\}, \\ 0 & \text{a.e. in } \{x \in \Omega : \phi(x) \ge M\}, \end{cases}$$

whence

$$\|\nabla \min(\phi, M)\|_2^2 \le \|\nabla \phi\|_2^2. \tag{2.104}$$

Let now  $(t, z) \in [0, T] \times L^2(\Omega)$  be arbitrary, but fixed and let us for simplicity abbreviate  $\bar{\varphi} := \Phi(t, z)$  in what follows. By Theorem 1.23 and Definition 2.1 one establishes that

$$\mathcal{I}(t, \min(z, M)) \leq \mathcal{E}(t, \mathcal{U}(t, \bar{\varphi}), \min(\bar{\varphi}, M), \min(z, M)) 
= \frac{1}{2} \int_{\Omega} g(\min(\bar{\varphi}, M)) \mathbb{C}\varepsilon(\mathcal{U}(t, \bar{\varphi})) : \varepsilon(\mathcal{U}(t, \bar{\varphi})) dx - \langle \ell(t), \mathcal{U}(t, \bar{\varphi}) \rangle_{V} 
+ \frac{\alpha}{2} \|\nabla \min(\bar{\varphi}, M)\|_{2}^{2} + \frac{\beta}{2} \|\min(\bar{\varphi}, M) - \min(z, M)\|_{2}^{2},$$
(2.105)

where for the equality we employed Definition 0.2. By Assumption 2.36, we further have  $g(\min(\bar{\varphi}, M)) \leq g(\bar{\varphi})$  a.e. in  $\Omega$ . Thanks to (4), (2.104), and since the function  $\min(\cdot, M) : L^2(\Omega) \to L^2(\Omega)$  is Lipschitz continuous with Lipschitz constant 1, (2.105) can be continued as

$$\mathcal{I}(t, \min(z, M)) \leq \frac{1}{2} \int_{\Omega} g(\bar{\varphi}) \mathbb{C}\varepsilon(\mathcal{U}(t, \bar{\varphi})) : \varepsilon(\mathcal{U}(t, \bar{\varphi})) \, dx - \langle \ell(t), \mathcal{U}(t, \bar{\varphi}) \rangle_{V} 
+ \frac{\alpha}{2} \|\nabla \bar{\varphi}\|_{2}^{2} + \frac{\beta}{2} \|\bar{\varphi} - z\|_{2}^{2} 
= \mathcal{I}(t, z),$$
(2.106)

in view of Definitions 0.2 and 2.1. Therewith the first assertion is proven.

Now, if z satisfies  $\min(z, M) = z$ , then (2.106) holds as an equality and thus, the inequality in (2.105) is an equality. Since the minimization problem in  $(P_{\beta})$  is uniquely solvable, cf. Theorem 1.23, it holds  $\min(\bar{\varphi}, M) = \Phi(t, \min(z, M)) = \Phi(t, z)$ , which completes the proof.

The next result states that the damage variables in the time-discrete problem are bounded a.e. in  $(0,T) \times \Omega$  by a constant independent of  $\beta$  and of  $\tau$ , which is just the constant M from Assumption 2.36, provided that this is fulfilled.

**Lemma 2.38** (Uniform estimates for the discrete local and nonlocal damage). Suppose that Assumptions 1.5, 1.17, 1.56 and 2.36 hold true. Moreover, let the number of timesteps in the discrete problem be large enough. If  $d_0 \in L^{\infty}(\Omega)$  with  $\|d_0\|_{L^{\infty}(\Omega)} \leq M$ , then

$$\bar{d}_{\tau}(t,x) \leq M$$
,  $\bar{\varphi}_{\tau}(t,x) \leq M$  a.e. in  $\Omega$ ,  $\forall t \in [0,T]$ .

Here  $\bar{\varphi}_{\tau}$  stands again for  $\Phi(\bar{t}_{\tau}(\cdot), \bar{d}_{\tau}(\cdot))$ .

*Proof.* Let n denote again the number of time-steps. We first address the discrete local damage by following the lines of the proof of [41, Proposition 4.5], that is, we show that  $d_k^{\tau} \leq M$  for all  $k \in \{0, ..., n\}$  by induction on the index k. Note that for k = 0 the assertion is fulfilled, since  $d_0(x) \leq M$  a.e. in  $\Omega$ , by assumption. Now let  $k \in \{0, ..., n-1\}$  be fixed and assume that

$$d_k^{\tau}(x) \le M$$
 a.e. in  $\Omega$ . (2.107)

The idea of the proof is to show that  $(d_{k+1}^{\tau})^- := \min(d_{k+1}^{\tau}, M)$  solves the problem (2.96), which will give in turn  $d_{k+1}^{\tau} = \min(d_{k+1}^{\tau}, M)$ , in view of Lemma 2.35. From  $(P_k^{\beta, \tau})$  it is clear that  $d_{k+1}^{\tau} \ge d_k^{\tau}$ , and as a result of (2.107) we thus have  $(d_{k+1}^{\tau})^- \ge d_k^{\tau}$ . With Definition 0.4 we then obtain

$$\mathcal{R}_{\delta}\left(\frac{(d_{k+1}^{\tau})^{-} - d_{k}^{\tau}}{\tau}\right) = r \int_{\Omega} \frac{(d_{k+1}^{\tau})^{-} - d_{k}^{\tau}}{\tau} dx + \frac{\delta}{2} \left\| \frac{(d_{k+1}^{\tau})^{-} - d_{k}^{\tau}}{\tau} \right\|_{2}^{2} \\
\leq r \int_{\Omega} \frac{d_{k+1}^{\tau} - d_{k}^{\tau}}{\tau} dx + \frac{\delta}{2} \left\| \frac{d_{k+1}^{\tau} - d_{k}^{\tau}}{\tau} \right\|_{2}^{2} \\
= \mathcal{R}_{\delta}\left(\frac{d_{k+1}^{\tau} - d_{k}^{\tau}}{\tau}\right), \tag{2.108}$$

where for the inequality we used  $0 \le (d_{k+1}^{\tau})^{-} - d_{k}^{\tau} \le d_{k+1}^{\tau} - d_{k}^{\tau}$ . From Lemma 2.37 we further deduce

$$\mathcal{I}(t_{k+1}^{\tau},(d_{k+1}^{\tau})^{-}) \leq \mathcal{I}(t_{k+1}^{\tau},d_{k+1}^{\tau}),$$

which added to (2.108) multiplied by  $\tau$  leads to

$$\mathcal{I}(t_{k+1}^{\tau}, (d_{k+1}^{\tau})^{-}) + \tau \mathcal{R}_{\delta} \Big( \frac{(d_{k+1}^{\tau})^{-} - d_{k}^{\tau}}{\tau} \Big) \leq \mathcal{I}(t_{k+1}^{\tau}, d_{k+1}^{\tau}) + \tau \mathcal{R}_{\delta} \Big( \frac{d_{k+1}^{\tau} - d_{k}^{\tau}}{\tau} \Big).$$

On the other side, Lemma 2.35 tells us that  $d_{k+1}^{\tau}$  is the unique solution of (2.96) and as  $(d_{k+1}^{\tau})^{-}$  is in the admissible set of (2.96), we conclude from the above inequality that  $(d_{k+1}^{\tau})^{-} = d_{k+1}^{\tau}$  must hold. Hence,  $d_{k+1}^{\tau} \leq M$ , which ends the induction step. Therefore, we have

$$\bar{d}_{\tau}(t,x) \leq M$$
 a.e. in  $\Omega, \ \forall t \in [0,T]$ .

Furthermore, according to Lemma 2.37, it also holds  $\bar{\varphi}_{\tau}(t,x) = \Phi(\bar{t}_{\tau}(t),\bar{d}_{\tau}(t))(x) \leq M$  a.e. in  $\Omega$ , for all  $t \in [0,T]$ . The proof is now complete.

With the above results at hand, one can easily show that the local and the nonlocal damage associated to  $(P_{\beta})$ , as well as their limit as  $\beta \to \infty$ , are a.e. in  $(0,T) \times \Omega$  bounded by the constant M from Assumption 2.36. This is covered by the following

**Theorem 2.39** ( $L^{\infty}$ -bound for the limit damage variable). Suppose that Assumptions 1.17, 1.47, 1.56, 2.8 and 2.36 are satisfied. Then, the optimal local and the optimal nonlocal damage associated to  $(P_{\beta})$  fulfill the estimates

$$0 \le d_{\beta}(t, x) \le M, \quad \varphi_{\beta}(t, x) \le M \quad a.e. \text{ in } (0, T) \times \Omega.$$
 (2.109)

Moreover, (2.74) admits at least one solution  $\varphi$  in  $L^{\infty}((0,T)\times\Omega)$  with

$$0 \le \varphi(t, x) \le M \quad a.e. \ in \ (0, T) \times \Omega. \tag{2.110}$$

*Proof.* By (1.40) and Proposition 2.7,  $(i) \Leftrightarrow (iii)$ , we know that

$$d_{\beta}(t) = d_0 + \frac{1}{\delta} \int_0^t \max\left(-\beta(d_{\beta}(s) - \varphi_{\beta}(s)) - r\right) ds \quad \forall t \in [0, T].$$

Thus, the first inequality for the local damage in (2.109) is a result of Assumption 2.8. In addition, we observe that the sets  $\{f \in L^{\infty}(0,T;L^{2}(\Omega)): f(t,x) \leq M \text{ a.e. in } (0,T) \times \Omega\}$  and  $\{f \in L^{\infty}(0,T;H^{1}(\Omega)): f(t,x) \leq M \text{ a.e. in } (0,T) \times \Omega\}$  are closed. The estimate (2.109) now follows from (2.91) and Lemma 2.38, in view of Assumption 2.8.

Furthermore, Theorem 2.28 tells us that (2.74) admits at least one solution  $\varphi \in H^1(0,T;H^1(\Omega))$  and that there is a sequence  $\{\beta_n\}$  with  $\beta_n \to \infty$  as  $n \to \infty$  so that

$$d_{\beta_n} \rightharpoonup \varphi \text{ in } H^1(0,T;L^2(\Omega)) \text{ as } n \to \infty,$$
 (2.111)

where  $d_{\beta_n}$  is the optimal local damage associated to  $(P_{\beta_n})$ . The estimates in (2.110) are thus a consequence of (2.111), the estimate for the local damage in (2.109) and the fact that the set  $\{f \in H^1(0,T;L^2(\Omega)) : 0 \leq f(t,x) \leq M \text{ a.e. in } (0,T) \times \Omega\}$  is convex and closed, and thus, weakly closed. Of course,  $\varphi$  belongs to  $L^{\infty}((0,T) \times \Omega)$ , as it is measurable and satisfies (2.110). This completes the proof.

Remark 2.40 (Non-negative values for the nonlocal damage). To make sure that the nonlocal damage has only non-negative values, as the local damage does, one has to impose, in addition to Assumptions 1.5, 1.17 and  $d_0 = 0$ , that g decreases on the interval  $(-\infty,0)$ , i.e.,  $g'(x) \leq 0$  for all x < 0. Recall that this assumption on g is in fact very reasonable from a practical point of view. Then, by testing (1.48b) with  $\min\{\varphi_{\beta}(t),0\}$  at all  $t \in [0,T]$ , one obtains in (1.48b) a non-negative and a non-positive term on the left-hand side and right-hand side, respectively. Therefrom one concludes that  $\varphi_{\beta} \geq 0$  indeed holds true.

### 2.5 Comparison to classical partial damage models

In this section we show that the single-field gradient damage model given by (2.74) falls into the category of classical partial damage models. To be more specific, we prove that in the two-dimensional case (2.74) is equivalent to a version of the viscous damage model studied in [41] for which the body is sound at the beginning of the process. In this situation, the viscous problem in [41] reads

$$-\partial_z \overline{\mathcal{I}}(t, z(t)) \in \partial \overline{\mathcal{R}}_{\bar{\delta}}(\dot{z}(t)) \quad \text{f.a.a. } t \in (0, T), \quad z(0) = 1. \tag{2.112}$$

To see a first similarity with our damage model, recall that (2.74) is in fact equivalent to (2.67) suplemmented with the initial condition, as shown by the proof of Theorem 2.28.

In order to distinguish between the two models, we add in what follows the symbol  $\bar{\phantom{a}}$  to the notations used for the data, functionals and operators in [41], in case that the therein used notations coincide with ours or if we already used them for data or global operators in the present work. In the sequel, N=2 and Assumptions 1.47, 1.56, 2.8 and 2.36 are supposed to be satisfied the whole time. Observe that Assumption 1.17.1 is automatically fulfilled, in view of Lemma 1.3. Moreover,  $\varphi \in H^1(0,T;H^1(\Omega))$  denotes from now on a solution of (2.74) with

$$0 \le \varphi(t, x) \le M$$
 a.e. in  $(0, T) \times \Omega$ ,

where M is the constant from Assumption 2.36. Note that such a solution exists, thanks to Theorem 2.39, see also the proof thereof.

The main difference between (2.74) and (2.112) consists in the definition of the dissipation functional. To see this, compare Definition 2.21 to [41, (1.3) and (1.9)], which tells us that the viscous dissipation  $\bar{\mathcal{R}}_{\bar{\delta}}: H^1(\Omega) \to [0, \infty]$  is given by  $\bar{\mathcal{R}}_{\bar{\delta}} = \bar{\mathcal{R}}_1 + \frac{\bar{\delta}}{2} ||\cdot||_2^2$ , where

$$\bar{\mathcal{R}}_1(\eta) := \begin{cases} \kappa \int_{\Omega} |\eta| \ dx, & \text{if } \eta \leq 0 \text{ a.e. in } \Omega, \\ \infty, & \text{otherwise.} \end{cases}$$

Here  $\kappa > 0$  and  $\bar{\delta} > 0$  denote the fracture toughness and the viscosity parameter, respectively. Therefore, unlike in our situation, the therein considered damage variable, which is denoted by z, can only decrease in time. This is due to the fact that in [41] the damage variable  $z:[0,T]\times\Omega\to\mathbb{R}$  measures the soundness of the material, not the degree of the material rigidity loss, as in our case. This means that, the larger the values of z, the sounder the body. Moreover, in the designed model, this takes values only in the interval [0,1], so that z(t,x)=0 and z(t,x)=1 when the system is fully damaged and completely sound, respectively. We refer here also to [17], where the gradient damage model which serves as basis for [41] was introduced. In [41, Proposition 4.5] it is shown under additional assumptions that one has  $z(t,x)\in[0,1]$  throughout the whole process, for at least one of the solutions of (2.112), thus proving the viability of the mathematical model in this regard. As shown by the computations below, see (2.115), these assumptions, in particular [41, (4.26)], correspond to those required to show (2.110).

The above motivates the following transformation

$$z := 1 - \frac{\varphi}{M} \in H^1(0, T; H^1(\Omega)). \tag{2.113}$$

The case z=0 corresponds to  $\varphi=M$  (maximal damage), while z=1 corresponds to  $\varphi=0$  (complete soundness), so that (2.113) is a very reasonable starting point for showing the equivalence between the two models.

The function spaces and the assumptions on the data for (2.112) are introduced in [41, Sections 2.1 and 2.2]. We observe that the setting coincides mostly with ours, excepting the following:

- the energy functional  $\widetilde{\mathcal{E}}$  associated to (2.67), see (2.47), does not include a counterpart for the term featuring the nonlinearity f in [41, (1.1)]. For this reason, we consider in the following just the case f = 0. Although the condition [41, (2.8)] does not allow the function f to be the zero function, this is here not problematic, as explained in Remark 2.44 below;
- in the energy functional in [41], see [41, Section 2.2], the degree of gradient regularization is 1. As in (2.47), this can be however replaced by some parameter  $\bar{\alpha} > 0$ , which does not affect the analysis in [41] at all;
- since in our model the Dirichlet boundary remains fixed during the damage process, we consider the special case  $u_D = 0$  in [41], see [41, (2.16)].

Under the above considerations, the energy functional  $\bar{\mathcal{E}}: [0,T] \times V \times H^1(\Omega) \to \mathbb{R}$  cf. [41, (1.1)], is given by

$$\bar{\mathcal{E}}(t, \boldsymbol{u}, z) := \frac{1}{2} \int_{\Omega} \bar{g}(z) \mathbb{C}\varepsilon(\boldsymbol{u}) : \varepsilon(\boldsymbol{u}) \ dx - \langle \bar{\ell}(t), \boldsymbol{u} \rangle_{V} + \frac{\bar{\alpha}}{2} \|\nabla z\|_{2}^{2}, \tag{2.114}$$

which now corresponds entirely to the definition of the energy functional  $\widetilde{\mathcal{E}}$  in (2.47).

We can now proceed towards our goal, which is to show that the function z defined in (2.113) satisfies the considered version of the viscous model (2.112). To this end, we have to transform the function g and resize  $\alpha$ , the viscosity parameter  $\delta$  and the fracture toughness r, see (2.115) and (2.117) below.

In view of (2.113), we define  $\bar{g}: \mathbb{R} \to \mathbb{R}$  by

$$\overline{g}(x) := g(M(1-x)),$$
 (2.115)

such that the following holds true

$$\overline{q}(z(t)) = q(\varphi(t)) \quad \forall t \in [0, T]. \tag{2.116}$$

Note that  $\overline{g}$  satisfies condition [41, (2.10)] due to Assumptions 0.6 and 1.56. Moreover, we observe that this transformation is reversible, in the sense that, given  $\overline{g}$ , one can reobtain g via  $g(x) := \overline{g}(1 - x/M) \ \forall x \in \mathbb{R}$ . Note that the properties of  $\overline{g}$  transfer to g as well. Recall that the coefficient function g assesses the degree of the material

elasticity loss (under the influence of damage). That is why, in practical applications, this is considered to be monotonically decreasing, unlike  $\bar{q}$ , which should monotonically increase cf. [41, Remark 4.6.]. Note that this aspect is also confirmed by (2.115).

In order to obtain (2.112) we still need to resize the following data:

$$\bar{\alpha} := \alpha M^2, \tag{2.117a}$$

$$\bar{\delta} := \delta M^2, \tag{2.117b}$$

$$\kappa := rM. \tag{2.117c}$$

Observe that the transformations in (2.117) are reversible and that the positivity is

Since it is interesting to see how both models behave with respect to each other under the influence of the same external load, we impose

$$\bar{\ell} := \ell. \tag{2.118}$$

We now have all the necessary tools for proving that the limit model (2.74) can be transformed into a version of (2.112) and vice versa.

**Proposition 2.41.** Let N=2 and suppose that Assumptions 1.47, 1.56 and 2.8 are satisfied. Moreover, let  $\varphi$  be a solution of (2.74) and M > 0 be given. Then, the function  $z := 1 - \varphi/M$  solves the problem (2.112) with  $f = 0, \mathbf{u}_D = 0$  and parameters  $\bar{g}, \bar{\alpha}, \bar{\delta}, \kappa \text{ and } \bar{\ell}, \text{ given by (2.115), (2.117a), (2.117b), (2.117c) and (2.118), respectively.}$ If  $\varphi$  is obtained via (2.73) and Assumption 2.36 is fulfilled, then  $z(t,x) \in [0,1]$  f.a.a.  $(t,x) \in (0,T) \times \Omega.$ 

*Proof.* We begin by enumerating some results which will be very useful in what follows. Firstly, as a consequence of (2.113), we have

$$\varphi(t) = M(1 - z(t)) \quad \forall t \in [0, T],$$
(2.119a)

$$\dot{\varphi}(t) = -M \,\dot{z}(t) \qquad \text{f.a.a. } t \in (0, T), \tag{2.119b}$$

$$\dot{\varphi}(t) = -M \, \dot{z}(t) \qquad \text{f.a.a. } t \in (0, T),$$

$$\nabla \varphi(t) = -M \nabla z(t) \qquad \forall t \in [0, T].$$

$$(2.119b)$$

$$(2.119c)$$

Secondly, since in the two-dimensional case the operator  $g: H^1(\Omega) \to L^{\tau}(\Omega)$  is continuously Fréchet-differentiable for  $\tau \in [1, \infty)$ , see Lemma 5.3, the same holds for the operator  $\bar{g}$ , as this preserves the properties of g. Thus, on account of (2.115) and (2.119a) it holds

$$\overline{q}'(z(t)) = -Mq'(\varphi(t)) \quad \forall t \in [0, T]. \tag{2.120}$$

Further, notice that [41, (2.28)] (adapted to the here considered situation) reads

$$\partial_{z}\overline{\mathcal{I}}(t,z(t)) = -\bar{\alpha}\Delta z(t) + \frac{1}{2}\overline{g}'(z(t))\mathbb{C}\varepsilon(\bar{\boldsymbol{u}}(t)) : \varepsilon(\bar{\boldsymbol{u}}(t)) \quad \forall t \in [0,T],$$
 (2.121)

where  $\bar{\boldsymbol{u}}(t)$  solves at  $t \in [0,T]$  the balance of momentum equation

$$-\operatorname{div}\left(\bar{g}(z(t))\mathbb{C}\varepsilon(\bar{\boldsymbol{u}}(t))\right) = \bar{\ell}(t) \quad \text{in } \boldsymbol{W}^{-1,p}(\Omega). \tag{2.122}$$

To see this, we first recall that g and  $\bar{g}$  have the same properties, and thus, Proposition 1.7 tells us that for given  $(t,z) \in [0,T] \times H^1(\Omega)$ , the functional  $\bar{\mathcal{E}}(t,\cdot,z)$  is minimized on V by the unique solution u of

$$-\operatorname{div}\left(\bar{g}(z)\mathbb{C}\varepsilon(\boldsymbol{u})\right) = \ell(t) \quad \text{in } \boldsymbol{W}^{-1,p}(\Omega), \tag{2.123}$$

in view of (2.118). Analogously to Definitions 1.8 and 2.19 we introduce the solution operator  $\overline{\mathcal{U}}:[0,T]\times H^1(\Omega)\ni (t,z)\mapsto \boldsymbol{u}\in \boldsymbol{W}^{1,p}(\Omega)$  associated with (2.123) and the reduced energy functional  $\overline{\mathcal{I}}:[0,T]\times H^1(\Omega)\ni (t,z)\mapsto \overline{\mathcal{E}}(t,\overline{\mathcal{U}}(t,z),z)$ , respectively. We refer here also to [41, Lemma 2.4]. Following the lines of Lemma 2.20, one then shows that  $\overline{\mathcal{I}}$  is Fréchet-differentiable w.r.t. z and the identity (2.121) can be deduced in the exact same way as (2.49b). Observe that by relying on (2.116) and (2.118), (2.122) gives in turn

$$\bar{\boldsymbol{u}}(t) = \mathcal{U}(t, \varphi(t)) \quad \forall t \in [0, T],$$
 (2.124)

on account of Definition 1.8. With (2.117a), (2.119c), (2.120) and (2.124), the identity (2.121) then becomes

$$\partial_z \overline{\mathcal{I}}(t, z(t)) = M\left(\alpha \Delta \varphi(t) - \frac{1}{2}g'(\varphi(t))\mathbb{C}\varepsilon\left(\mathcal{U}(t, \varphi(t))\right) : \varepsilon\left(\mathcal{U}(t, \varphi(t))\right)\right),$$

i.e.,

$$\partial_z \overline{\mathcal{I}}(t, z(t)) = -M \, \partial_\omega \widetilde{\mathcal{I}}(t, \varphi(t)) \quad \forall t \in [0, T].$$
 (2.125)

Note that (2.125) is a result of (2.49b) and (1.23). Further, by comparing (2.76) and [41, (1.3)], we find in view of (2.119b) f.a.a.  $t \in (0, T)$  the following

$$\xi \in \partial \widetilde{\mathcal{R}}_{1}(\dot{\varphi}(t))$$

$$\iff \langle \xi, v - \dot{\varphi}(t) \rangle \leq \widetilde{\mathcal{R}}_{1}(v) - \widetilde{\mathcal{R}}_{1}(\dot{\varphi}(t)) = r/\kappa \left( \overline{\mathcal{R}}_{1}(-v) - \overline{\mathcal{R}}_{1}(-\dot{\varphi}(t)) \right)$$

$$\iff \langle -\xi, v - M\dot{z}(t) \rangle \leq r/\kappa \left( \overline{\mathcal{R}}_{1}(v) - \overline{\mathcal{R}}_{1}(M\dot{z}(t)) \right) \quad \forall v \in H^{1}(\Omega)$$

$$\iff -\kappa \xi/r \in \partial \overline{\mathcal{R}}_{1}(M\dot{z}(t)) \iff -M\xi \in \partial \overline{\mathcal{R}}_{1}(\dot{z}(t)),$$

where for the last equivalence we used (2.117c) and the positive homogeneity of  $\overline{\mathcal{R}}_1$ . Thus, we have

$$\partial \overline{\mathcal{R}}_1(\dot{z}(t)) = -M \, \partial \widetilde{\mathcal{R}}_1(\dot{\varphi}(t)) \quad \text{f.a.a. } t \in (0, T).$$
 (2.126)

Now, by means of (2.125), (2.119b), (2.117b), (2.126) and (2.113), (2.74) can be rewritten as

$$1/M\big(\partial_z\overline{\mathcal{I}}(t,z(t))+\,\bar{\delta}\dot{z}(t)\big)\in -1/M\partial\overline{\mathcal{R}}_1(\dot{z}(t))\quad \text{ f.a.a. } t\in(0,T),\quad z(0)=1,$$

where we kept (2.49b) in mind. Thereby we deduce that z solves

$$-\partial_z \overline{\mathcal{I}}(t, z(t)) \in \partial \overline{\mathcal{R}}_1(\dot{z}(t)) + \bar{\delta}\dot{z}(t)$$
 f.a.a.  $t \in (0, T), \quad z(0) = 1$ 

and by applying sum rule for convex subdifferentials we ultimately obtain (2.112), in view of the definition of the viscous dissipation functional in [41, (1.9)].

The proof of Theorem 2.39 shows that if  $\varphi$  is obtained via (2.73) and Assumption 2.36 is fulfilled, then  $z(t,x) \in [0,1]$  f.a.a.  $(t,x) \in (0,T) \times \Omega$ , as a result of (2.113). The proof is now complete.

A short inspection of the proof of Proposition 2.41 (in particular, (2.125) and (2.126)) shows that (2.74) can be deduced from (2.112) as well:

Corollary 2.42. Let N=2 and suppose that [41, (2.10), (2.16)] are satisfied. Moreover, let M>0 be given. If (2.112) with f=0,  $\bar{\alpha}>0$  and  $\mathbf{u}_D=0$  admits a solution  $z\in H^1(0,T;H^1(\Omega))$ , then  $\varphi:=M(1-z)$  solves (2.74) with parameters  $g(\cdot)=\bar{g}(1-\cdot/M)$ ,  $\alpha=\bar{\alpha}/M^2$ ,  $\delta=\bar{\delta}/M^2$ ,  $r=\kappa/M$  and  $\ell=\bar{\ell}$ .

Remark 2.43. Note that if one replaces the initial condition in (2.112) by  $z(0) = z_0$  a.e. in  $\Omega$ , where  $z_0$  is a positive constant function, then (2.74) and (2.112) are still equivalent in the sense of Proposition 2.41 and Corollary 2.42 via the transformation  $z := z_0 \left(1 - \frac{\varphi}{M}\right)$ . Of course, this calls for redefining  $\bar{g}$  in (2.115) accordingly. However, if  $z_0 \not\equiv 1$ , then the models are no longer compatible from a practical point of view, as the soundness of the body  $(\varphi(t, x) = 0)$  would correspond to  $z(t, x) = z_0 \neq 1$ .

**Remark 2.44.** Proposition 2.41 tells us that, by means of the penalization approach, the existence of viscous solutions for the here considered version of the model in [41] can also be established when  $f \equiv 0$  in the energy given by [41, (1.1)]. Note that this case is excluded in [41] because of the growth condition in [41, (2.8)].

Anyway, the nonlinearity f can be easily incorporated in the problem  $(P_{\beta})$  as a function acting on the nonlocal damage. A closer inspection of the preceding analysis then shows that [41, (2.8)] is indeed needed to prove the essential results leading to Theorem 2.28. The result in Proposition 2.41 then holds true with  $\overline{f}(\cdot) := f(M(1-\cdot))$ . We however decided not to consider this additional term and to rely on the energy from [12] instead, as this serves as basis for our penalized damage model. Note that if we include f in our investigations, the transformed system (2.74), i.e., (2.112), is just a special case of [41, (1.8)] in two dimensions. Thus, the crucial result in [41] concerning the vanishing viscosity analysis can be applied to our viscous limit model.

# Chapter 3

# Optimal control of the damage model with penalty

In this chapter we turn our attention to the damage model with penalty, this time in the context of optimal control. Recall that this reads as follows

for almost all  $t \in (0, T)$ . Moreover, recall that  $(P_{\ell})$  describes in terms of the displacement u, the nonlocal damage  $\varphi$  and the local damage d the effect of a force  $\ell$  on an elastic body. For more details, see the introduction of the thesis. In many practical applications it is of interest to gain information about those loads which (locally) minimize a given cost functional, e.g. one may be interested to minimize the damage or/and the distance to a desired displacement. This motivates our goal in this chapter: to derive necessary optimality conditions for an optimal control problem governed by the damage model with penalty, where the load is used as control. The throughout this chapter studied optimization problem is given by

$$\min_{\ell \in \mathfrak{L}} \quad \mathcal{J}(\boldsymbol{u}, \varphi, d, \ell) 
\text{s.t.} \quad (\boldsymbol{u}, \varphi, d) \text{ solves } (P_{\ell}) \text{ with right-hand side } \ell,$$

$$(P_{min})$$

where the assumptions on the objective  $\mathcal{J}$  and the control set  $\mathfrak{L}$  are to be introduced and motivated in Section 3.4 below.

Recall that in Chapter 1 we already studied the damage model with penalty for a fixed, smooth in time  $\ell$ . It turned out that  $(P_{\ell})$  is uniquely solvable and that the continuous differentiability in time of the load transfers to the unique solution  $(\boldsymbol{u}, \varphi, d)$ , see Theorem 1.62, p. 52. This was ultimately essential for the limit analysis  $\beta \to \infty$  in the previous chapter, e.g. as we employed the chain rule identity. By contrast, in this chapter we work with *variable*, *nonsmooth in time* loads, which calls for a careful reinvestigation of the constraint in  $(P_{min})$ . The reason is two-fold:

- (I) as the natural way to approach  $(P_{min})$  is to define the control-to-state operator, we have to ask the question if the main results in Chapter 1 remain unaffected by the fact that  $\ell$  variates. The answer is negative, as we will see in Section 3.1 below;
- (II) we will not restrict only on smooth loads, but we choose the admissible set as general as possible (so that unique solvability of  $(P_{\ell})$  is still guaranteed), see also Remark 1.26, p. 25. In this case, the unique solution possesses much less regularity, insufficient for performing the limit analysis in Chapter 2. We refer here to Section 3.2 below.

As we will see, the arguments employed to establish unique solvability for  $(P_{\ell})$  in this new context are very similar to those in Chapter 1 and that is why we will not entirely repeat them. However, because of the reasons enumerated above, an accurate inspection of  $(P_{\ell})$  from this new perspective cannot be omitted.

The problem  $(P_{min})$  falls into the class of optimal control problems governed by time-dependent VIs. To be more precise,  $(P_{\ell})$  can be formulated (under suitable assumptions) as a time-dependent VI of first and second kind, namely (2.15), see Proposition 2.7 on page 59. While the optimal control of time-dependent VIs of second kind was very little investigated, there are many contributions in the field of time-dependent VIs of first kind. We mention [2, 3, 15, 18, 25, 34, 35], which focus on the optimal control of the parabolic obstacle problem. In [81] the optimal control problem of quasistatic plasticity is analyzed, while [15] and [33] deal with the optimal control of the Allen-Cahn and Cahn-Hilliard VIs, respectively. In all these contributions nonsmooth constraints are considered, and thus, the standard method of deriving necessary optimality conditions in the form of Karush-Kuhn-Tucker conditions is not applicable. As we will see, this is also the case when it comes to the optimal control of the damage model with penalty.

Deriving necessary optimality conditions is a challenging issue even in the finite dimensional case, where a special attention is given to MPECs (mathematical programs with equilibrium constraints). We refer here only to [37, 65, 66, 72] and the references therein. In [72] a detailed overview of various optimality conditions of different strength was introduced for a special class of MPECs, namely MPCCs (mathematical programs with complementarity constraints). In the spirit thereof, stationarity concepts for the infinite dimensional case are defined in [32]. The most rigorous stationarity concept is strong stationarity. Roughly speaking, the strong stationarity conditions involve an optimality system, which is equivalent to the purely primal conditions. Other stationarity concepts such as Clarke (C) stationarity are less rigorous compared to strong stationarity. The weakest concept is weak stationarity, which involves the existence of multipliers, but no sign conditions for the multipliers at all.

The most common way to overcome the lack of differentiability of the control-tostate operator consists of employing regularization and relaxation techniques, see [2, 3, 15, 18, 25, 33, 34, 35, 81]. The thereby derived optimality systems are in the best case of intermediate strength such as C stationarity, see e.g. [33] (time-dependent) and [30, 34] (time-independent). For the optimal control of the parabolic obstacle problem a strong stationarity system can be found in [62], but no rigorous proof is given there. Recently, an optimality system of strong stationary type was derived in [50] for an optimal control problem governed by a nonsmooth parabolic PDE. This was possible due to the presence of so-called 'ample controls', which are in most of the existing contributions necessary for deriving strong stationarity, see e.g. [10, 31, 63] (elliptic VIs). To the best of our knowledge, [82] is the only paper where a strong stationary optimality system is derived in the absence thereof. We refer here to Remark 3.47 below for more details. Hence, it is not surprising, that without regularizing, additional assumptions are needed in order to derive an optimality system for  $(P_{min})$ . Therefor we also make use of the special structure of the constraint in  $(P_{min})$ . To be more precise, we employ the fact that  $(P_{\ell})$  can be reduced to an ordinary differential equation in Banach space (see Theorem 1.29 on page 26 and (3.38) below).

#### Outline of the chapter

The chapter is organized as follows. Section 3.1 focuses on reanalyzing the minimization problem in  $(P_{\ell})$  in the context where  $\ell$  is a (at first time-independent) variable. In Section 3.2 we introduce the control-to-state operator by carefully choosing its domain of definition, while in Section 3.3 we investigate its differentiability, as a preparatory step for deriving first order necessary optimality conditions for  $(P_{min})$ . These are established in Section 3.4, where our focus lies on deriving an optimality system in a rather straight forward way. As already mentioned above, this can be done only under additional assumptions. As always, Assumption 1.17.1 on page 19 is supposed to hold throughout this chapter.

# 3.1 The dependence of the elliptic system on the load

The purpose of this section is to gain information about the dependence of the displacement u and of the nonlocal damage  $\varphi$  on the load  $\ell$ . Recall that, in Subsection 1.1.1 we approached the minimization problem in  $(P_{\ell})$  in the following way. We investigated

$$\min_{(\boldsymbol{u},\varphi)\in V\times H^1(\Omega)} \mathcal{E}(t,\boldsymbol{u},\varphi,d) \tag{3.1}$$

for fixed  $(t, d) \in [0, T] \times L^2(\Omega)$  and showed that it admits a unique solution characterized by (1.25) on page 19, provided that Assumptions 1.5, p. 13 and 1.17, p. 19 hold. We refer here to Theorem 1.23, p. 23. This allowed us to define the solution operators  $\mathcal{U}$  and  $\Phi$ , see Definitions 1.8, p. 15 and 1.24, p. 24, respectively. The optimal displacement and optimal nonlocal damage for  $(P_{\ell})$  were then ultimately given by  $(\mathcal{U}(\cdot, \Phi(\cdot, d(\cdot))), \Phi(\cdot, d(\cdot)))$  where d denotes the unique solution of the evolutionary equation in  $(P_{\ell})$ .

In this section we present an alternative approach, which has the advantage of highlighting how the minimizer of (3.1) is related to the load. We begin by considering the following optimization problem

$$\min_{(\boldsymbol{u},\varphi)\in V\times H^1(\Omega)} \bar{\mathcal{E}}(\ell,\boldsymbol{u},\varphi,d), \tag{3.2}$$

where  $(\ell, d) \in W^{-1,p}(\Omega) \times L^2(\Omega)$  is fixed. The functional  $\bar{\mathcal{E}} : W^{-1,p}(\Omega) \times V \times H^1(\Omega) \times L^2(\Omega) \to \mathbb{R}$  is just a slight modification of the energy functional  $\mathcal{E}$  on page 4 and it is given by

$$\bar{\mathcal{E}}(\ell, \boldsymbol{u}, \varphi, d) := \frac{1}{2} \int_{\Omega} g(\varphi) \mathbb{C}\varepsilon(\boldsymbol{u}) : \varepsilon(\boldsymbol{u}) \ dx - \langle \ell, \boldsymbol{u} \rangle_{V} + \frac{\alpha}{2} \|\nabla \varphi\|_{2}^{2} + \frac{\beta}{2} \|\varphi - d\|_{2}^{2}.$$

In view of Definition 0.2, p. 4 one sees that at all  $(t, \boldsymbol{u}, \varphi, d) \in [0, T] \times V \times H^1(\Omega) \times L^2(\Omega)$  it holds

$$\bar{\mathcal{E}}(\ell(t), \boldsymbol{u}, \varphi, d) = \mathcal{E}(t, \boldsymbol{u}, \varphi, d) \tag{3.3}$$

for some given  $\ell: [0,T] \to \mathbf{W}^{-1,p}(\Omega)$ . Since we know that for a fixed  $(t,d) \in [0,T] \times L^2(\Omega)$ , (3.1) admits a unique solution, it is clear in view of (3.3) that (3.2) is uniquely solvable for a fixed pair  $(\ell,d) \in \mathbf{W}^{-1,p}(\Omega) \times L^2(\Omega)$ . Moreover, we can deduce from Theorem 1.23, p. 23, that the minimizer  $(\bar{\mathbf{u}},\bar{\varphi})$  of (3.2) is characterized by

$$-\operatorname{div} g(\bar{\varphi})\mathbb{C}\varepsilon(\bar{\boldsymbol{u}}) = \ell \quad \text{in } \boldsymbol{W}^{-1,p}(\Omega)$$
 (3.4a)

$$-\alpha \Delta \bar{\varphi} + \beta \,\bar{\varphi} + \frac{1}{2} \,g'(\bar{\varphi}) \mathbb{C} \,\varepsilon(\bar{\boldsymbol{u}}) : \varepsilon(\bar{\boldsymbol{u}}) = \beta d \quad \text{in } H^1(\Omega)^*. \tag{3.4b}$$

This can be seen by imposing that the load in Assumption 1.5, p. 13 has at any time point the same value  $\ell \in W^{-1,p}(\Omega)$ . Of course, the above statements are true as long as Assumption 1.17 on page 19 is satisfied (with the threshold for  $\beta$  depending on  $\|\ell\|_{W^{-1,p}(\Omega)}$ , cf. Remark 1.22, p. 23).

The above discussion calls for redefining the solution operators  $\mathcal{U}$  and  $\Phi$ , by dropping the time dependence and replacing it with dependence on some (time-independent) load. Although the domain changes (from  $[0,T]\times L^2(\Omega)$  to  $\mathbf{W}^{-1,p}(\Omega)\times L^2(\Omega)$ ), we choose for convenience to remain by the same notations. Most of the properties of the new solution operators, such as continuity, boundedness, can be easily transferred from Subsection 1.1.1 by readapting the corresponding estimates. By using arguments analogous to those employed in Section 1.3, Fréchet-differentiability of the new solution operators is proven in the sequel as well. This serves as important tool in the upcoming section. However, one has to keep in mind that in all the previous sections the load was a smooth in time given data, whereas in what follows the load  $\ell$  is a time-independent variable. Therefore, not all results in Sections 1.1 and 1.3 can be readily transferred for the whole space  $\mathbf{W}^{-1,p}(\Omega)$ . As it will turn out, it is sometimes required that one works with loads from a bounded subset thereof.

We begin by defining the new solution operators and replacing the needed properties from Subsection 1.1.1 accordingly.

**Definition 3.1** (Solution operator of (3.4a)). We define the operator  $\mathcal{U}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to \mathbf{W}^{1,p}_D(\Omega)$  as

 $\mathcal{U}(\ell,\varphi) := A_{\varphi}^{-1}\ell,$ 

where  $A_{\varphi}$  is given by Definition 1.2 on page 12. We emphasize that  $\mathcal{U}$  is well defined in view of Lemma 1.3, p. 12.

Lemma 1.3 also gives immediately

$$\|\mathcal{U}(\ell,\varphi)\|_{\boldsymbol{W}_{p}^{1,p}(\Omega)} \leq c\|\ell\|_{\boldsymbol{W}^{-1,p}(\Omega)} \quad \forall (\ell,\varphi) \in \boldsymbol{W}^{-1,p}(\Omega) \times H^{1}(\Omega), \tag{3.5}$$

where c > 0 is independent of  $\ell$  and  $\varphi$ .

The next result follows by using the exact same arguments as in the proof of Lemma 1.11, p. 16.

**Lemma 3.2** (Continuity of  $(\ell, \varphi) \mapsto \mathcal{U}(\ell, \varphi)$ ). Let  $\{(\ell_n, \varphi_n)\} \subset \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$  and  $(\ell, \varphi) \in \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$  be given such that  $(\ell_n, \varphi_n) \to (\ell, \varphi)$  in  $\mathbf{W}^{-1,p}(\Omega) \times L^1(\Omega)$  as  $n \to \infty$ . Then it holds  $\mathcal{U}(\ell_n, \varphi_n) \to \mathcal{U}(\ell, \varphi)$  in  $\mathbf{W}_{D}^{1,p}(\Omega)$  as  $n \to \infty$  for every  $s \in [2, p)$ .

As for the Lipschitz continuity of  $\mathcal{U}$ , we are interested in an estimate of the type (1.15), p. 15, for r:=2p/(p-2). This will ultimately lead to an estimate for all  $\varphi_1, \varphi_2 \in H^1(\Omega)$ , as a consequence of the Sobolev embedding  $H^1(\Omega) \hookrightarrow L^r(\Omega)$ . Note that the latter one is guaranteed in the two-dimensional case thanks to p>2, and since throughout this section Assumption 1.17.1 on page 19 is supposed to hold, the embedding holds true in the three-dimensional case as well. By taking a closer look at the proof of Proposition 1.10 on page 15, we see that all the arguments employed therein can be readily transferred. However, due to (3.5) and  $H^1(\Omega) \hookrightarrow L^r(\Omega)$ , the estimate (1.18), p. 16, is now replaced by

$$\|(A_{\varphi_{2}} - A_{\varphi_{1}})\mathcal{U}(\ell_{2}, \varphi_{2})\|_{V^{*}} \leq C \|g(\varphi_{1}) - g(\varphi_{2})\|_{r} \|\mathcal{U}(\ell_{2}, \varphi_{2})\|_{\mathbf{W}_{D}^{1,p}(\Omega)}$$

$$\leq C \|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)} \|\ell_{2}\|_{\mathbf{W}^{-1,p}(\Omega)}$$

$$\forall (\ell_{i}, \varphi_{i})_{i=1,2} \in \mathbf{W}^{-1,p}(\Omega) \times H^{1}(\Omega).$$
(3.6)

Note that the constant C > 0 is independent of  $(\ell_i, \varphi_i)_{i=1,2}$ . We point out that the modified estimates hold true for the same integrability exponents as in the proof of Proposition 1.10, which in particular means that  $\pi = 2$ , if r = 2p/(p-2) (see page 16). By further using the exact same arguments as in the proof of Proposition 1.10, p. 15, (3.6) together with  $\mathbf{W}^{-1,p}(\Omega) \hookrightarrow V^*$  yields in the end for all  $(\ell_i, \varphi_i) \in \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$ 

$$\|\mathcal{U}(\ell_{1},\varphi_{1}) - \mathcal{U}(\ell_{2},\varphi_{2})\|_{V} \leq L(\|\ell_{1} - \ell_{2}\|_{\mathbf{W}^{-1,p}(\Omega)} + \|\ell_{2}\|_{\mathbf{W}^{-1,p}(\Omega)} \|\varphi_{1} - \varphi_{2}\|_{r})$$

$$\leq L(\|\ell_{1} - \ell_{2}\|_{\mathbf{W}^{-1,p}(\Omega)} + \|\ell_{2}\|_{\mathbf{W}^{-1,p}(\Omega)} \|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)}),$$
(3.7)

where r = 2p/(p-2) and L > 0 is a constant which depends only on the given data. Although  $\mathcal{U}(\cdot,\varphi)$  and  $\mathcal{U}(\ell,\cdot)$  are globally Lipschitz continuous for any fixed  $(\ell,\varphi) \in \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$ , the same holds no longer for  $\mathcal{U}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to V$ . In view of (3.7), we introduce

**Definition 3.3** (Time-independent admissible loads). For a given M > 0 we define the open ball

$$\mathcal{B}_M := B_{\mathbf{W}^{-1,p}(\Omega)}(0,M).$$

**Remark 3.4.** For the moment we deliberately choose to work with open (bounded) subsets of  $\mathbf{W}^{-1,p}(\Omega)$ , instead of closed ones, since in the upcoming section this turns out to be more convenient, see for example the proof of Lemma 3.25 on page 121.

Based on the available control space, an appropriate value for the constant M will be determined in Section 3.4 below. Until then, M is allowed to take any positive value.

The estimate (3.7) now results in

**Lemma 3.5** (Lipschitz continuity of  $(\ell, \varphi) \mapsto \mathcal{U}(\ell, \varphi)$ ). Let Assumption 1.17.1 on page 19 hold. Then, for every M > 0, there exists a constant L(M) > 0 so that for all  $\ell_1, \ell_2 \in \mathcal{B}_M$  and all  $\varphi_1, \varphi_2 \in H^1(\Omega)$  it holds

$$\|\mathcal{U}(\ell_1, \varphi_1) - \mathcal{U}(\ell_2, \varphi_2)\|_V \le L(M)(\|\ell_1 - \ell_2\|_{\mathbf{W}^{-1,p}(\Omega)} + \|\varphi_1 - \varphi_2\|_{H^1(\Omega)}).$$

Remark 3.6. Notice that the operators in Definitions 1.8 (page 15) and 3.1 both map to  $W_D^{1,p}(\Omega)$ . Further, notice that the solution operator  $\mathcal U$  in Definition 3.1 is bounded and satisfies continuity and Lipschitz continuity with the same integrability exponents as its corresponding solution operator in Subsection 1.1.1, see Corollary 1.9 on page 15, Proposition 1.10 on page 15 and Lemma 1.11 on page 16. This means that replacing estimates which feature  $\mathcal U$  with domain  $[0,T]\times H^1(\Omega)$ , by estimates which feature  $\mathcal U$  considered with domain  $W^{-1,p}(\Omega)\times H^1(\Omega)$ , does not affect any of the integrability exponents involved.

We now turn our attention to (3.4b). Similarly to (1.23), p. 18 we introduce the following useful

**Definition 3.7** (The nonlinear part of (3.4b)). Suppose that Assumption 1.14 on page 18 is fulfilled. Then we define the mapping  $F: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to H^1(\Omega)^*$  by

$$\langle F(\ell,\varphi),\psi\rangle_{H^1(\Omega)}:=\frac{1}{2}\int_{\Omega}g'(\varphi)\mathbb{C}\varepsilon(\mathcal{U}(\ell,\varphi)):\varepsilon(\mathcal{U}(\ell,\varphi))\psi\ dx\quad\forall\,\psi\in H^1(\Omega).$$

Notice that F is well defined for the same reasons why the nonlinearity in Definition 1.15, p. 18 is well defined. We recall here for convenience that this follows from Hölder's inequality combined with (3), Assumption 0.7, Definition 3.1 and the Sobolev embedding  $H^1(\Omega) \hookrightarrow L^{p/(p-2)}(\Omega)$ . Observe that the latter one is guaranteed by Assumption 1.14.

In view of Definitions 3.1 and 3.7, the elliptic system (3.4) can now be written, similarly to (1.24), p. 19 in the compact form

$$B\bar{\varphi} + F(\ell, \bar{\varphi}) = \beta d, \tag{3.8}$$

where  $(\ell, d) \in W^{-1,p}(\Omega) \times L^2(\Omega)$  is fixed, B is given by (1.22) on page 18 and  $\bar{\varphi}$  is the second component of the unique minimizer of (3.2). At the beginning of the section, we argued that (3.4) is uniquely solvable under Assumption 1.17, p. 19 (where this time the threshold for  $\beta$  depends on  $\|\ell\|_{W^{-1,p}(\Omega)}$ ). Hence, (3.8) is uniquely solvable as well. Thus, the operator  $B + F(\ell, \cdot)$  is invertible, however for fixed  $\ell \in W^{-1,p}(\Omega)$ . This can

also be seen by taking a look at the proofs of Propositions 1.12 and Lemma 1.21 (see pages 17 and 22), which ultimately led to the unique solvability of (1.24), p. 19.

Further, note that the reduced functional in the proof of Proposition 1.12 is now replaced by

$$f:H^1(\Omega)\to\mathbb{R},\quad f(\varphi):=-\frac{1}{2}\langle\ell,\mathcal{U}(\ell,\varphi)\rangle+\frac{\alpha}{2}\|\nabla\varphi\|_2^2+\frac{\beta}{2}\|\varphi-d\|_2^2,$$

where  $(\ell, d) \in \mathbf{W}^{-1,p}(\Omega) \times L^2(\Omega)$ . One sees that working with an arbitrary  $\ell$  does not affect the radially unboundedness and the weakly lower semicontinuity of f, which are crucial for proving existence of solutions for  $\min_{\varphi \in H^1(\Omega)} f(\varphi)$  and thus, for (3.8). Notice that the latter one is the necessary optimality condition for a local optimum of the above minimizing problem. We refer here to the proof of Proposition 1.16 on page 19. Thus, as the proof of Proposition 1.12 shows, the existence of solutions for (3.8) remains unaffected when  $\ell$  is a variable. That is no longer the case when it comes to the uniqueness thereof. The latter one was established in Section 1.1.1 by means of Lemma 1.21, p. 22, where Assumption 1.17.2 on page 20 played an essential role, since it guaranteed that  $\beta$  exceeds a given threshold. However, as already stated at the beginning of the present section on page 101 (see also Remark 1.22, p. 23), the threshold for  $\beta$  depends on the load  $\ell$ , which implies that the strong monotonicity of  $B + F(\ell, \cdot)$ , see Lemma 1.21, p. 22, is no longer ensured when  $\ell$  is a variable in  $\mathbf{W}^{-1,p}(\Omega)$ . This is shown by the following

**Lemma 3.8.** Suppose that Assumption 1.17.1 on page 19 holds true. Then, for all  $\ell_1, \ell_2 \in W^{-1,p}(\Omega)$  and all  $\varphi_1, \varphi_2 \in H^1(\Omega)$ ,  $\varphi_1 \neq \varphi_2$  we have

$$\frac{\langle B(\varphi_{1} - \varphi_{2}) + F(\ell_{1}, \varphi_{1}) - F(\ell_{2}, \varphi_{2}), \varphi_{1} - \varphi_{2} \rangle_{H^{1}(\Omega)}}{\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)}}$$

$$\geq \left(\alpha - \widetilde{c}(k)L(\ell_{1}, \ell_{2})^{2}\right)\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)} - CL(\ell_{1}, \ell_{2})\|\ell_{1} - \ell_{2}\|$$

$$+ \left(\beta - \alpha - kL(\ell_{1}, \ell_{2})^{2}\right)\frac{\|\varphi_{1} - \varphi_{2}\|_{2}^{2}}{\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)}} \quad \forall k > 0, \tag{3.9}$$

where we use the notation  $\|\cdot\|$  for the  $\mathbf{W}^{-1,p}(\Omega)$ -norm and abbreviate  $L(\ell_1,\ell_2) := \|\ell_1\| + \|\ell_2\|$ . In the above estimate,  $\widetilde{c} : \mathbb{R}^+ \to \mathbb{R}^+$  is a function which depends only on the given data and which satisfies  $\widetilde{c}(k) \searrow 0$  as  $k \nearrow \infty$ .

*Proof.* We address the properties of the function F considered with domain  $\mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$ , by going through the properties of its corresponding function in Lemmata 1.18 and 1.21 (see pages 20 and 22). We do so by writing down only those estimates which are directly affected by the new definition of the solution operator of (3.4a). We point out that all the integrability exponents in the proofs of Lemmata 1.18 and 1.21 remain unchanged in view of Remark 3.6.

Let  $\ell_1, \ell_2 \in W^{-1,p}(\Omega)$  and  $\varphi_1, \varphi_2 \in H^1(\Omega)$  be arbitrary, but fixed. We define  $\mathbf{u}_i := \mathcal{U}(\ell_i, \varphi_i)$  for i = 1, 2 and set r := 2p/(p-2) such that  $H^1(\Omega) \hookrightarrow L^r(\Omega)$  holds true,

in view of Assumption 1.17.1. We begin by noticing that the estimate (1.28), p. 20 is replaced in view of (3.5) by

$$\|\mathbb{C}\varepsilon(\boldsymbol{u}_1):\varepsilon(\boldsymbol{u}_1)\|_{\frac{p}{2}} \le C\|\ell_1\|^2. \tag{3.10}$$

Instead of the estimate (1.30), p. 21, we have due to triangle inequality, (3.5) and (3.7) the following

$$\|\mathbb{C}\varepsilon(\boldsymbol{u}_{1}):\varepsilon(\boldsymbol{u}_{1})-\mathbb{C}\varepsilon(\boldsymbol{u}_{2}):\varepsilon(\boldsymbol{u}_{2})\|_{\omega} \leq C\|\boldsymbol{u}_{1}+\boldsymbol{u}_{2}\|_{\boldsymbol{W}_{D}^{1,p}(\Omega)}\|\boldsymbol{u}_{1}-\boldsymbol{u}_{2}\|_{V}$$

$$\leq CL(\ell_{1},\ell_{2})(\|\ell_{2}\|\|\varphi_{1}-\varphi_{2}\|_{r}+\|\ell_{1}-\ell_{2}\|),$$
(3.11)

where  $\omega = 2p/(p+2)$ , so that Hölder's inequality with  $1/\omega = 1/p + 1/2$  can be applied. By further following the lines of the proof of Lemma 1.18, we deduce that (3.10) and (3.11) lead to modifying the estimates (1.29), p. 21 and (1.31), p. 21, such that in the end it holds for all  $\psi \in H^1(\Omega)$ 

$$\begin{aligned} |\langle F(\ell_{1}, \varphi_{1}) - F(\ell_{2}, \varphi_{2}), \psi \rangle| &\leq CL(\ell_{1}, \ell_{2}) \left( \|\ell_{2}\| \|\varphi_{1} - \varphi_{2}\|_{r} + \|\ell_{1} - \ell_{2}\| \right) \|\psi\|_{r} \\ &+ C\|\ell_{1}\|^{2} \|\varphi_{1} - \varphi_{2}\|_{r} \|\psi\|_{r} \\ &\leq C \left( L(\ell_{1}, \ell_{2})^{2} \|\varphi_{1} - \varphi_{2}\|_{r} + L(\ell_{1}, \ell_{2}) \|\ell_{1} - \ell_{2}\| \right) \|\psi\|_{r}. \end{aligned}$$

$$(3.12)$$

Notice that for r=2p/(p-2) we have s=2p/(p-2) in Lemma 1.18, in view of 1/s+2/p+1/r=1 (see page 20). From (3.12) we infer by Lemma 1.20 on page 22

$$\begin{split} |\langle F(\ell_{1},\varphi_{1}) - F(\ell_{2},\varphi_{2}),\varphi_{1} - \varphi_{2}\rangle| &\leq L(\ell_{1},\ell_{2})^{2} \big(k\|\varphi_{1} - \varphi_{2}\|_{2}^{2} + \widetilde{c}(k)\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)}^{2}\big) \\ &+ C L(\ell_{1},\ell_{2}) \|\ell_{1} - \ell_{2}\| \|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)} \quad \forall \, k > 0. \end{split}$$

$$(3.13)$$

Note that, in view of Lemma 1.20,  $\tilde{c}: \mathbb{R}^+ \to \mathbb{R}^+$  depends on the constant C from (3.12), on p, and on N, and thus, only on the given data, see also Remark 1.27, p. 25. Now, the definition of B in (1.22), p. 18 and (3.13) finally give the assertion.

Let us now take a look at  $B + F(\ell, \cdot)$  regarding its strong monotonicity for some arbitrary  $\ell \in W^{-1,p}(\Omega)$ . By setting  $\ell_1 = \ell_2 = \ell$  in (3.9) we obtain for all  $\varphi_1, \varphi_2 \in H^1(\Omega)$ 

$$\langle B(\varphi_{1} - \varphi_{2}) + F(\ell, \varphi_{1}) - F(\ell, \varphi_{2}), \varphi_{1} - \varphi_{2} \rangle \geq (\alpha - 4\widetilde{c}(k)\|\ell\|^{2})\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)}^{2} + (\beta - \alpha - 4k\|\ell\|^{2})\|\varphi_{1} - \varphi_{2}\|_{2}^{2} \quad \forall k > 0,$$
(3.14)

where  $\tilde{c}: \mathbb{R}^+ \to \mathbb{R}^+$  has the same properties as in Lemma 3.8. We use for convenience again the notation  $\|\cdot\|$  for the  $W^{-1,p}(\Omega)$ -norm. From the estimate (3.14) we read that, in order to ensure the strong monotonicity of  $B + F(\ell, \cdot)$ , we must firstly choose k large enough such that  $C_1 := \alpha - 4\tilde{c}(k)\|\ell\|^2 > 0$ . As in the proof of Lemma 1.21, p. 22, we again choose k such that  $C_1 \geq \alpha/2$ , i.e., such that  $4\tilde{c}(k)\|\ell\|^2 \leq \alpha/2$  holds. Note that such a k exists in view of the properties of  $\tilde{c}$  and it depends on the given data and  $\|\ell\|$ . Further, it is crucial that

$$\beta \ge \alpha + 4k \|\ell\|^2, \tag{3.15}$$

which in view of (3.14) would give in turn the strong monotonicity of  $B + F(\ell, \cdot)$ . Of course, (3.15) is fulfilled for a fixed load, if  $\beta$  is chosen sufficiently large, but, if the loads vary, as in case of optimal control problems, one cannot guarantee (3.15) with a fixed value of  $\beta$  when we allow the loads to be arbitrary functions in  $\mathbf{W}^{-1,p}(\Omega)$ . The situation changes however, if we turn to sets of admissible controls that are bounded in  $\mathbf{W}^{-1,p}(\Omega)$ , as we will next see. Observe that for  $\ell \in \mathcal{B}_M$ , k can be chosen independent of  $\ell$ , such that in view of the properties of the function  $\widetilde{c}$ , it holds

$$\alpha/2 \ge 4\widetilde{c}(k)M^2 > 4\widetilde{c}(k)\|\ell\|^2. \tag{3.16}$$

Then, if additionally,

$$\beta \ge \alpha + 4kM^2 > \alpha + 4k\|\ell\|^2 \tag{3.17}$$

holds true by Assumption 1.17.2, p. 20, the estimate (3.14) can be continued in view of (3.16) as

$$\langle B(\varphi_1 - \varphi_2) + F(\ell, \varphi_1) - F(\ell, \varphi_2), \varphi_1 - \varphi_2 \rangle_{H^1(\Omega)} \ge \alpha/2 \|\varphi_1 - \varphi_2\|_{H^1(\Omega)}^2$$
 (3.18)

for all  $\varphi_1, \varphi_2 \in H^1(\Omega)$  and all  $\ell \in \mathcal{B}_M$ .

Remark 3.9. We point out that the threshold for  $\beta$  in (3.17) depends now only on the given data, and in particular on M, instead of the (variable) load, see also (3.16) and Lemma 3.8. The constant M, although fixed, may take different values depending on the context. That is why we will sometimes emphasize the dependency of  $\beta$  on M by writing  $\beta = \beta(M)$ , which basically means that  $\beta$  satisfies the first inequality in (3.17), where k is given by the first inequality in (3.16).

Since we already established above that (3.8) admits solutions for  $(\ell, d) \in W^{-1,p} \times L^2(\Omega)$ , (3.18) together with Cauchy-Schwarz inequality yields now, similarly to Theorem 1.23 on page 23, the existence of the solution operator

**Definition 3.10** (Solution operator of (3.8)). Let Assumption 1.17 on page 19 hold and let M > 0 be given. Moreover, let  $\beta = \beta(M)$ . Then we define the operator  $\Phi : \mathcal{B}_M \times L^2(\Omega) \to H^1(\Omega)$  as

$$\Phi(\ell, d) := (B + F(\ell, \cdot))^{-1}(\beta d),$$

where B is given by (1.22) on page 18.

**Lemma 3.11** (Lipschitz continuity of  $(\ell, d) \mapsto \Phi(\ell, d)$ ). Let Assumption 1.17 on page 19 hold. Moreover, let M > 0 be given and  $\beta = \beta(M)$ . Then, there exists a constant L(M) > 0 so that for all  $\ell_1, \ell_2 \in \mathcal{B}_M$  and all  $d_1, d_2 \in L^2(\Omega)$  it holds

$$\|\Phi(\ell_1, d_1) - \Phi(\ell_2, d_2)\|_{H^1(\Omega)} \le L(M)(\|\ell_1 - \ell_2\|_{\boldsymbol{W}^{-1, p}(\Omega)} + \|d_1 - d_2\|_2).$$

*Proof.* Let M > 0 and  $(\ell_i, d_i) \in \mathcal{B}_M \times L^2(\Omega)$ , i = 1, 2, be arbitrary, but fixed. Further, let us abbreviate  $\varphi_i := \Phi(\ell_i, d_i)$ , i = 1, 2. Note that the existence of the solution operator

 $\Phi$  is guaranteed by assumption. We suppose that  $\varphi_1 \neq \varphi_2$ , by keeping in mind that, otherwise, the Lipschitz continuity of  $\Phi$  is trivial. We prove the result by using the exact same arguments as in the proof of Theorem 1.23, p. 23. To this end, we first have to derive an estimate similar to the one in Lemma 1.21, p. 22. Since Assumption 1.17.2, p. 20 is fulfilled with  $\beta = \beta(M)$ , we know in view of Remark 3.9 that  $\beta \geq \alpha + 4kM^2$ , where k is computed such that  $\alpha/2 \geq 4\tilde{c}(k)M^2$  holds. Since  $\ell_1, \ell_2 \in \mathcal{B}_M$ , we have

$$\alpha/2 > \widetilde{c}(k)L(\ell_1, \ell_2)^2, \quad \beta > \alpha + kL(\ell_1, \ell_2)^2.$$

Here, the symbols  $\tilde{c}$  and  $L(\ell_1, \ell_2)$  have the same meaning as in Lemma 3.8, from which we now obtain

$$\frac{\langle B(\varphi_{1} - \varphi_{2}) + F(\ell_{1}, \varphi_{1}) - F(\ell_{2}, \varphi_{2}), \varphi_{1} - \varphi_{2} \rangle}{\|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)}} \ge \alpha/2 \|\varphi_{1} - \varphi_{2}\|_{H^{1}(\Omega)} - C L(\ell_{1}, \ell_{2}) \|\ell_{1} - \ell_{2}\|_{\mathbf{W}^{-1, p}(\Omega)}, \tag{3.19}$$

where C > 0 depends only on the given data. The rest of the proof is entirely analogous to the proof of Theorem 1.23, p. 23, where instead of (1.37) we now have, in view of (3.19), Definition 3.10, and Cauchy-Schwarz inequality, the estimate

$$\alpha/2\|\varphi_{1}-\varphi_{2}\|_{H^{1}(\Omega)}^{2} \leq \langle B(\varphi_{1}-\varphi_{2})+F(\ell_{1},\varphi_{1})-F(\ell_{2},\varphi_{2}),\varphi_{1}-\varphi_{2}\rangle_{H^{1}(\Omega)}$$

$$+CL(\ell_{1},\ell_{2})\|\ell_{1}-\ell_{2}\|_{\mathbf{W}^{-1,p}(\Omega)}\|\varphi_{1}-\varphi_{2}\|_{H^{1}(\Omega)}$$

$$=\beta(d_{1}-d_{2},\varphi_{1}-\varphi_{2})_{2}+CL(\ell_{1},\ell_{2})\|\ell_{1}-\ell_{2}\|_{\mathbf{W}^{-1,p}(\Omega)}\|\varphi_{1}-\varphi_{2}\|_{H^{1}(\Omega)}$$

$$\leq (\beta\|d_{1}-d_{2}\|_{2}+CL(\ell_{1},\ell_{2})\|\ell_{1}-\ell_{2}\|_{\mathbf{W}^{-1,p}(\Omega)})\|\varphi_{1}-\varphi_{2}\|_{H^{1}(\Omega)}.$$

Dividing by  $\|\varphi_1 - \varphi_2\|_{H^1(\Omega)}$  finally gives the assertion with  $L(M) := 2/\alpha \max\{\beta, 2CM\}$ .

Remark 3.12. Actually, the solution operator  $\Phi$  maps  $\mathcal{B}_M \times L^2(\Omega)$  to  $\mathbf{W}^{1,q}(\Omega)$ , where M > 0 is fixed and q > 2 is given by Theorem 1.37 on page 31. Moreover,  $\Phi$  satisfies the corresponding Lipschitz continuity condition. This is due to the fact that all the results in Section 1.2, p. 29, can be readily transferred for the operator in Definition 3.10.

**Remark 3.13.** So far, we have seen that, if  $(\ell, d) \in \mathcal{B}_M \times L^2(\Omega)$ , where M > 0 is fixed, then the dependence on  $(\ell, d)$  of the optimal displacement and optimal nonlocal damage in (3.2) can be expressed under Assumption 1.17, p. 19 (where  $\beta = \beta(M)$ ), via the mapping  $\mathcal{B}_M \times L^2(\Omega) \ni (\ell, d) \mapsto (\mathcal{U}(\ell, \Phi(\ell, d)), \Phi(\ell, d)) \in \mathbf{W}_D^{1,p}(\Omega) \times H^1(\Omega)$ . In other words, the latter one is the solution operator of the system (3.4).

The rest of the section is devoted to proving that, as their corresponding operators in Subsection 1.1.1, p. 11, the new solution operators  $\mathcal{U}$  and  $\Phi$  are also Fréchet-differentiable. This is an essential tool in the upcoming section. We begin with the differentiability result for the operator  $\mathcal{U}$ , which is covered by the following

**Lemma 3.14** (Fréchet-differentiability of  $(\ell, \varphi) \mapsto \mathcal{U}(\ell, \varphi)$ ). Under Assumption 1.17.1 on page 19 it holds  $\mathcal{U} \in C^1(\mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega); V)$ .

*Proof.* We proceed as in Subsection 1.3.1, p. 41, that is, we show that  $\mathcal{U}$  is partially continuously Fréchet-differentiable, which in view of [9, Theorem 3.7.1] will give the assertion. To this end, first notice that the operator  $\mathcal{U}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to \mathbf{W}^{1,p}_D(\Omega)$  is partially Fréchet-differentiable with respect to  $\ell$ . This is a consequence of Definition 3.1 combined with Lemma 1.3, p. 12, which tells us that  $A_{\varphi}^{-1} \in \mathcal{L}(\mathbf{W}^{-1,p}(\Omega), \mathbf{W}_D^{1,p}(\Omega))$  for every  $\varphi \in H^1(\Omega)$ . Therefrom we also deduce

$$\partial_{\ell} \mathcal{U}(\ell, \varphi) = A_{\varphi}^{-1} \quad \forall (\ell, \varphi) \in \mathbf{W}^{-1, p}(\Omega) \times H^{1}(\Omega).$$
 (3.20)

Now let  $\{\varphi_n\} \subset H^1(\Omega)$  and  $\varphi \in H^1(\Omega)$  be given such that  $\varphi_n \to \varphi$  in  $H^1(\Omega)$  as  $n \to \infty$ . Since Assumption 1.17.1 on page 19 is supposed to hold, we can apply (3.7), which together with Definition 3.1 leads to

$$||A_{\varphi_n}^{-1}\ell - A_{\varphi}^{-1}\ell||_V \le L||\ell||_{\boldsymbol{W}^{-1,p}(\Omega)}||\varphi_n - \varphi||_{H^1(\Omega)} \quad \forall \, \ell \in \boldsymbol{W}^{-1,p}(\Omega), \, \, \forall \, n \in \mathbb{N},$$

whence

$$||A_{\varphi_n}^{-1} - A_{\varphi}^{-1}||_{\mathcal{L}(\mathbf{W}^{-1,p}(\Omega),V)} \le L||\varphi_n - \varphi||_{H^1(\Omega)} \to 0 \text{ as } n \to \infty.$$

Thus, in view of (3.20),  $\mathcal{U}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to V$  is partially continuously Fréchet-differentiable with respect to  $\ell$ .

By comparing Definition 1.8 on page 15 with Definition 3.1, one sees that the partial Fréchet-differentiability with respect to  $\varphi$  of the operator  $\mathcal{U}$  in Definition 3.1 is also given by Lemma 1.49 on page 42. To be more precise, the latter one guarantees that  $\mathcal{U}(\ell,\cdot)$ :  $H^1(\Omega) \to W^{1,\nu}_D(\Omega)$  is Fréchet-differentiable for every  $\ell \in W^{-1,p}(\Omega)$  for some  $\nu \in (2,p)$  (see (1.76), p. 42). Moreover, the partial derivative at  $(\ell,\varphi) \in W^{-1,p}(\Omega) \times H^1(\Omega)$  in direction  $\delta \varphi \in H^1(\Omega)$  fulfills

$$A_{\varphi}(\partial_{\varphi}\mathcal{U}(\ell,\varphi)(\delta\varphi)) = \operatorname{div}\left(g'(\varphi)(\delta\varphi)\mathbb{C}\varepsilon(\mathcal{U}(\ell,\varphi))\right) \text{ in } \mathbf{W}^{-1,\nu}(\Omega). \tag{3.21}$$

Let us now derive from the above identity an estimate which will be useful later in the ongoing proof. To this end, let  $\varrho$  and  $\kappa$  be the indices defined in the proof of Lemma 1.49, p. 42. Note that, due to  $H^1(\Omega) \hookrightarrow L^{\varrho}(\Omega)$  and  $\kappa < \varrho$ , the embedding  $H^1(\Omega) \hookrightarrow L^{\kappa}(\Omega)$  holds true. From Lemma 1.3, p. 12, an estimate similar to (1.78) on page 42, and (3.5) we then infer

$$\|\partial_{\varphi}\mathcal{U}(\ell,\varphi)(\delta\varphi)\|_{\mathbf{W}_{D}^{1,\nu}(\Omega)} \leq C \|\operatorname{div}\left(g'(\varphi)(\delta\varphi)\mathbb{C}\varepsilon(\mathcal{U}(\ell,\varphi))\right)\|_{\mathbf{W}^{-1,\nu}(\Omega)}$$

$$\leq C \|\ell\|_{\mathbf{W}^{-1,p}(\Omega)}\|\delta\varphi\|_{\kappa}$$

$$\leq C \|\ell\|_{\mathbf{W}^{-1,p}(\Omega)}\|\delta\varphi\|_{H^{1}(\Omega)} \quad \forall \ell \in \mathbf{W}^{-1,p}(\Omega), \ \forall \varphi, \delta\varphi \in H^{1}(\Omega).$$
(3.22)

In order to prove the continuity of  $\partial_{\varphi}\mathcal{U}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to \mathcal{L}(H^1(\Omega), V)$ , we follow the lines of the proof of Lemma 1.51, p. 44. We emphasize that the arguments are exactly the same and, in view of Remark 3.6, there is no need to discuss the integrability exponents. These readily transfer and are denoted here by the same symbols as in the proof of Lemma 1.51, p. 44. Let  $(\ell_i, \varphi_i)_{i=1,2} \in \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$  and  $\delta \varphi \in H^1(\Omega)$  be arbitrary, but fixed with  $\delta \varphi \neq 0$ . Further, let us abbreviate  $\mathbf{u}_i' := \partial_{\varphi} \mathcal{U}(\ell_i, \varphi_i) \delta \varphi$  and

 $u_i := \mathcal{U}(\ell_i, \varphi_i)$  for i = 1, 2. Moreover, we define the linear forms  $f_1 := A_{\varphi_2} u_2' - A_{\varphi_1} u_2' \in V^*$  and  $f_2 := A_{\varphi_1} u_1' - A_{\varphi_2} u_2' \in V^*$ , which we now intend to estimate by relying on the arguments used in the proof of Lemma 1.51, p. 44. Hence, the estimate for  $f_1$  is, in view of (3.22), given by

$$||f_1||_{V^*} \le C_1 ||g(\varphi_2) - g(\varphi_1)||_{\rho} ||u_2'||_{W_D^{1,\nu}(\Omega)}$$
  
$$\le C_1 ||g(\varphi_2) - g(\varphi_1)||_{\rho} ||\ell_2||_{W^{-1,p}(\Omega)} ||\delta\varphi||_{H^1(\Omega)},$$

while, due to (3.5), the estimate for  $f_2$  reads

$$||f_{2}||_{V^{*}} \leq C_{2}||g'(\varphi_{1})(\delta\varphi)||_{\varrho}||\mathbf{u}_{1} - \mathbf{u}_{2}||_{\mathbf{W}_{D}^{1,s}(\Omega)} + ||(g'(\varphi_{2}) - g'(\varphi_{1}))(\delta\varphi)||_{r}||\mathbf{u}_{2}||_{\mathbf{W}_{D}^{1,p}(\Omega)}$$

$$\leq C_{2}||\delta\varphi||_{H^{1}(\Omega)} (||\mathbf{u}_{1} - \mathbf{u}_{2}||_{\mathbf{W}_{D}^{1,s}(\Omega)} + ||g'(\varphi_{2}) - g'(\varphi_{1})||_{\mathcal{L}(H^{1}(\Omega),L^{r}(\Omega))} ||\ell_{2}||_{\mathbf{W}^{-1,p}(\Omega)}).$$

By employing Lemmata 5.2, 3.2 and 5.3, the new estimates for  $f_1$  and  $f_2$  then give

$$\sup_{\delta\varphi\in H^{1}(\Omega), \delta\varphi\neq 0} \frac{\|f_{1}\|_{V^{*}} + \|f_{2}\|_{V^{*}}}{\|\delta\varphi\|_{H^{1}(\Omega)}} \to 0 \quad \text{as } (\ell_{1}, \varphi_{1}) \to (\ell_{2}, \varphi_{2}) \text{ in } \mathbf{W}^{-1, p}(\Omega) \times H^{1}(\Omega).$$

This allows us, in the exact same way as at the end of the proof of Lemma 1.51 on page 44, to conclude that  $\partial_{\varphi}\mathcal{U}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to \mathcal{L}(H^1(\Omega),V)$  is continuous. The above results now give the assertion in view of [9, Theorem 3.7.1].

Notice that  $\mathcal{U}$  can be differentiated on the whole space  $\mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$ , because the Lipschitz continuity thereof (see Lemma 3.5) is not needed in the above proof. This is of course no longer the case when it comes to  $\Phi$ , as this is not even defined on  $\mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$ .

Before addressing the differentiability thereof, we deduce by means of the arguments used for deriving Lemma 1.50 on page 44 and (3.5) that

$$\|\partial_{\varphi}\mathcal{U}(\ell,\varphi)(\delta\varphi)\|_{V} \leq C \|\ell\|_{\boldsymbol{W}^{-1,p}(\Omega)} \|\delta\varphi\|_{r} \quad \forall \, \ell \in \boldsymbol{W}^{-1,p}(\Omega), \, \, \forall \, \varphi, \delta\varphi \in H^{1}(\Omega), \quad (3.23)$$

where r = 2p/(p-2). This estimate turns out to be useful for proving the following

**Lemma 3.15** (Fréchet-differentiability of  $(\ell, d) \mapsto \Phi(\ell, d)$ ). Let M > 0 be given and suppose that Assumptions 1.17, p. 19 (with  $\beta = \beta(M)$ ) and 1.56, p. 46 hold. Then  $\Phi \in C^1(\mathcal{B}_M \times L^2(\Omega); H^1(\Omega))$ . Moreover, its derivative satisfies at all  $(\ell, d) \in \mathcal{B}_M \times L^2(\Omega)$  and all  $(\delta \ell, \delta d) \in W^{-1,p}(\Omega) \times L^2(\Omega)$ 

$$-\alpha \Delta \delta \phi + \beta \delta \phi + \frac{1}{2} g''(\Phi(\ell, d)) \delta \phi \mathbb{C} \varepsilon (\mathcal{U}(\ell, \Phi(\ell, d))) : \varepsilon (\mathcal{U}(\ell, \Phi(\ell, d)))$$

$$= \beta \delta d - g'(\Phi(\ell, d)) \mathbb{C} \varepsilon (\mathcal{U}(\ell, \Phi(\ell, d))) : \varepsilon (\delta \mathbf{u}) \quad \text{in } H^{1}(\Omega)^{*},$$
(3.24)

where  $\delta \phi$  and  $\delta u$  stand for  $\Phi'(\ell, d)(\delta \ell, \delta d)$  and  $\mathcal{U}'(\ell, \Phi(\ell, d))(\delta \ell, \delta \phi)$ , respectively.

Proof. In order to differentiate the operator  $\Phi$  with domain  $\mathcal{B}_M \times L^2(\Omega)$  we carefully follow the lines of the proofs on pages 47-51, which, by the implicit function theorem, ultimately led to the continuously Fréchet-differentiability of the operator from Definition 1.24, p. 24. When transferring the results, we use for the sake of convenience the same notations for the involved functions, by keeping in mind that the time dependence is now replaced by the dependence on  $\ell$ . In view of Remark 3.6 and since the new solution operator  $\Phi$  has values in the same space as its corresponding operator in Chapter 1 (cf. Definitions 1.24 and 3.10), there is no need to discuss the integrability exponents. These will be in the following denoted by the same symbols as in the proofs from which they readily transfer. When readapting the results from Subsection 1.3.2 (pages 47-51) we go into detail only when it comes to results which are affected by the fact that  $\mathcal{U}$  and  $\Phi$  are now considered with other domains as in Subsection 1.1.1.

The ongoing proof is structured as follows. In the first part we establish the continuously Fréchet-differentiability of the mapping

$$\Psi: \mathbf{W}^{-1,p}(\Omega) \times L^2(\Omega) \times H^1(\Omega) \to H^1(\Omega)^*, \quad \Psi(\ell,d,\varphi) := B\varphi + F(\ell,\varphi) - \beta d, \quad (3.25)$$

where B and F are given by (1.22), p. 18 and Definition 3.7, respectively. For purposes of comparison we refer here to Definition 1.55, Lemma 1.57 and Corollary 1.58 (pages 46-49). In the second part we show that  $\partial_{\varphi}\Psi(\ell,d,\varphi):H^1(\Omega)\to H^1(\Omega)^*$  is bijective for all  $(\ell,d,\varphi)\in\mathcal{B}_M\times L^2(\Omega)\times H^1(\Omega)$ , compare with Lemma 1.59 on page 49. As in Proposition 1.60, p. 51, these two results will finally give the assertion.

(I) Next we focus on proving that  $\Psi \in C^1(\mathbf{W}^{-1,p}(\Omega) \times L^2(\Omega) \times H^1(\Omega); H^1(\Omega)^*)$ . As in Subsection 1.3.2, the challenging part is to show that  $F : \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to H^1(\Omega)^*$  is continuously Fréchet-differentiable. To differentiate F, we proceed as in the proof of Lemma 1.57, p. 47, i.e., we reformulate F by means of two products and apply Lemma 5.4 for these. To this end, we introduce

$$\mathcal{H}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to L^{p/2}(\Omega), \quad \mathcal{H}(\ell,\varphi) := \mathbb{C}\varepsilon(\mathcal{U}(\ell,\varphi)) : \varepsilon(\mathcal{U}(\ell,\varphi))$$

such that

$$F: (\ell, \varphi) \mapsto P_1(g'(\varphi), \mathcal{H}(\ell, \varphi)),$$
 (3.26)

where  $P_1$  is the product defined in (1.88) on page 47. By doing so, we see that the domain of  $\mathcal{H}$  changes in the proof of Lemma 1.57, p. 47. We now intend to show that the therein made statements about  $\mathcal{H}$  still hold true, when this is considered with domain  $\mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$ . In other words, we verify that  $\mathcal{H}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to L^{p/2}(\Omega)$  is well defined, continuous and continuously Fréchet-differentiable when considered as operator with ranges in the same spaces as in the proof of Lemma 1.57, p. 47.

- (i) In view of Hölder's inequality and Definition 3.1, the operator  $\mathcal{H}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to L^{p/2}(\Omega)$  is well defined.
  - (ii) Regarding the continuity of  $\mathcal{H}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to L^{\omega}(\Omega)$ , we estimate again

similarly to (1.30), p. 21, by using (1.90), p. 47, which yields

$$\begin{split} \|\mathcal{H}(\ell_{1},\varphi_{1}) - \mathcal{H}(\ell_{2},\varphi_{2})\|_{\omega} &\leq C \|\mathcal{U}(\ell_{1},\varphi_{1}) + \mathcal{U}(\ell_{2},\varphi_{2})\|_{\mathbf{W}_{D}^{1,p}(\Omega)} \|\mathcal{U}(\ell_{1},\varphi_{1}) - \mathcal{U}(\ell_{2},\varphi_{2})\|_{\mathbf{W}_{D}^{1,s}(\Omega)} \\ &\leq C \left( \|\ell_{1}\|_{\mathbf{W}^{-1,p}(\Omega)} + \|\ell_{2}\|_{\mathbf{W}^{-1,p}(\Omega)} \right) \|\mathcal{U}(\ell_{1},\varphi_{1}) - \mathcal{U}(\ell_{2},\varphi_{2})\|_{\mathbf{W}_{D}^{1,s}(\Omega)} \\ &\qquad \qquad \forall (\ell_{i},\varphi_{i})_{i=1,2} \in \mathbf{W}^{-1,p}(\Omega) \times H^{1}(\Omega). \end{split}$$

Note that for the last inequality we used (3.5). Lemma 3.2 then gives the desired continuity of  $\mathcal{H}$ , since  $s \in (2, p)$  in the proof of Lemma 1.57, p. 47.

(iii) Let us now check if  $\mathcal{H}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to L^{\varrho}(\Omega)$  is continuously Fréchetdifferentiable. For this purpose, we write  $\mathcal{H}$  as

$$\mathcal{H}: (\ell, \varphi) \mapsto P_2(\mathcal{U}(\ell, \varphi), \mathcal{U}(\ell, \varphi)),$$
 (3.27)

where  $P_2$  is the product defined in (1.91) on page 48. As in the proof of Lemma 1.57, p. 47, we then fix the setting in Lemma 5.4 accordingly. That is, we define

$$U := X := \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega), \quad W := L^{\varrho}(\Omega),$$
  
$$P := P_2, \quad f_i := \mathcal{U}, \quad Y_i := \mathbf{W}_D^{1,s}(\Omega), \quad Z_i := V, \quad i = 1, 2.$$

From Lemmata 3.2 and 3.14 we know that  $\mathcal{U}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to \mathbf{W}^{1,s}_D(\Omega)$  is continuous and  $\mathcal{U}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to V$  is continuously Fréchet-differentiable, respectively. Hence, we can apply Lemma 5.4 to (3.27), giving in turn that  $\mathcal{H}: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to L^{\varrho}(\Omega)$  is continuously Fréchet-differentiable with

$$\mathcal{H}'(\ell,\varphi)(\delta\ell,\delta\varphi) := 2\mathbb{C}\varepsilon(\mathcal{U}(\ell,\varphi)) : \varepsilon(\mathcal{U}'(\ell,\varphi)(\delta\ell,\delta\varphi))$$
(3.28)

for all  $(\ell, \varphi) \in W^{-1,p}(\Omega) \times H^1(\Omega)$  and all  $(\delta \ell, \delta \varphi) \in W^{-1,p}(\Omega) \times H^1(\Omega)$ .

Altogether, we have shown that  $\mathcal{H}$  with domain  $\mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$  has the same properties as  $\mathcal{H}$  with domain  $(0,T) \times H^1(\Omega)$  in Lemma 1.57, p. 47. This brings us to the second step therein, where the product rule from Lemma 5.4 is this time applied to (3.26). But since g' does not depend on  $\ell$  and since  $\mathcal{H}$  with domain  $\mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$  maintains the same properties with the exact same integrability exponents, all the results in the second step readily transfer. Of course, when fixing the setting in Lemma 5.4, one defines this time  $U := X := \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$ , which does not affect the final result at all. All this being said, we can now conclude that  $F : \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to H^1(\Omega)^*$  is continuously Fréchet-differentiable and thus, in view of (3.25), it holds  $\Psi \in C^1(\mathbf{W}^{-1,p}(\Omega) \times L^2(\Omega) \times H^1(\Omega); H^1(\Omega)^*)$ .

(II) By readapting the proof of Lemma 1.59 on page 49, we prove in what follows that  $\partial_{\varphi}\Psi(\ell,d,\varphi):H^1(\Omega)\to H^1(\Omega)^*$  is bijective for every  $(\ell,d,\varphi)\in\mathcal{B}_M\times L^2(\Omega)\times H^1(\Omega)$ . We point out that this is not expected to hold for all  $(\ell,d,\varphi)\in \mathbf{W}^{-1,p}(\Omega)\times L^2(\Omega)\times H^1(\Omega)$ . This is due to the fact that, for the coercivity of  $\partial_{\varphi}\Psi(\ell,d,\varphi)$ , and thus for its invertibility, it is crucial that  $\beta$  exceeds a given threshold, which depends on  $\|\ell\|_{\mathbf{W}^{-1,p}(\Omega)}$ , see (3.31) below. On the other side, the value of  $\beta$  is fixed, and thus, it may only depend on the given data. As in a previous situation (see (3.17)),  $\beta$  fulfills these two conditions when

 $\ell \in \mathcal{B}_M$ , as we will next see in (3.32). This is not surprising at all, since the unique solvability of (3.8) can be alternatively followed from the implicit function theorem applied for the exact same setting as in the present proof.

On account of (3.25), we intend to show that for every  $h \in H^1(\Omega)^*$ , the equation

$$B\delta\varphi + \partial_{\varphi}F(\ell,\varphi)\delta\varphi = h \tag{3.29}$$

admits a unique solution  $\delta \varphi \in H^1(\Omega)$ , where  $(\ell, \varphi) \in \mathcal{B}_M \times H^1(\Omega)$ . First notice that, as a result of (3.26), (1.88) (see page 47) and (3.28), it can be deduced in view of (5.1), that the partial derivative of F at  $(\ell, \varphi) \in W^{-1,p}(\Omega) \times H^1(\Omega)$  in direction  $\delta \varphi \in H^1(\Omega)$  reads

$$\langle \partial_{\varphi} F(\ell, \varphi)(\delta \varphi), z \rangle_{H^{1}(\Omega)} = \frac{1}{2} \int_{\Omega} g''(\varphi)(\delta \varphi) \mathbb{C} \varepsilon (\mathcal{U}(\ell, \varphi)) : \varepsilon (\mathcal{U}(\ell, \varphi)) z \, dx + \int_{\Omega} g'(\varphi) \mathbb{C} \varepsilon (\mathcal{U}(\ell, \varphi)) : \varepsilon (\partial_{\varphi} \mathcal{U}(\ell, \varphi)(\delta \varphi)) z \, dx \quad \forall z \in H^{1}(\Omega).$$
(3.30)

In order to solve (3.29) we employ again Lax-Milgram's lemma. The boundedness of  $B + \partial_{\varphi} F(\ell, \varphi)$  is guaranteed for any  $(\ell, \varphi) \in \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$ , by the exact same arguments as in the proof of Lemma 1.59, p. 49. Further, the estimate (1.100) on page 50 is in view of (3.30) replaced by

$$\begin{aligned} |\langle \partial_{\varphi} F(\ell, \varphi) z, z \rangle_{H^{1}(\Omega)}| &\leq C \big( \|g''(\varphi)\|_{\infty} \|z\|_{r} \|\mathbb{C}\varepsilon(\mathcal{U}(\ell, \varphi)) : \varepsilon(\mathcal{U}(\ell, \varphi))\|_{\frac{p}{2}} \|z\|_{r} \\ &+ \|g'(\varphi)\|_{\infty} \|\varepsilon(\mathcal{U}(\ell, \varphi))\|_{p} \|\varepsilon(\partial_{\varphi}\mathcal{U}(\ell, \varphi)(z))\|_{2} \|z\|_{r} \big) \\ &\leq C \|\ell\|_{\mathbf{W}^{-1, p}(\Omega)}^{2} \|z\|_{r}^{2} \quad \forall (\ell, \varphi) \in \mathbf{W}^{-1, p}(\Omega) \times H^{1}(\Omega), \ \forall z \in H^{1}(\Omega), \end{aligned}$$

where for the last inequality we used (3.5) and (3.23). In the exact same way as in the proof of Lemma 1.59, p. 49, we then arrive for all k > 0 at

$$\langle Bz + \partial_{\varphi} F(\ell, \varphi) z, z \rangle \ge \left(\alpha - \widetilde{c}(k) \|\ell\|_{\boldsymbol{W}^{-1, p}(\Omega)}^{2}\right) \|z\|_{H^{1}(\Omega)}^{2} + \left(\beta - \alpha - k \|\ell\|_{\boldsymbol{W}^{-1, p}(\Omega)}^{2}\right) \|z\|_{2}^{2}$$

$$\forall (\ell, \varphi) \in \boldsymbol{W}^{-1, p}(\Omega) \times H^{1}(\Omega), \ \forall z \in H^{1}(\Omega).$$
(3.31)

where  $\tilde{c}: \mathbb{R}^+ \to \mathbb{R}^+$  depends only on the given data and satisfies  $\tilde{c}(k) \searrow 0$  as  $k \nearrow \infty$ . This brings us to a situation similar to (3.14), where the uniformly boundedness of the loads was essential for the coercivity. Hence, from now on, let  $(\ell, \varphi) \in \mathcal{B}_M \times H^1(\Omega)$  be arbitrary, but fixed. Since  $\ell \in \mathcal{B}_M$  and since Assumption 1.17.2, p. 20 holds true with  $\beta = \beta(M)$ , we have in view of Remark 3.9 that

$$\beta \ge \alpha + 4kM^2 > \alpha + kM^2 > \alpha + k\|\ell\|_{\mathbf{W}^{-1,p}(\Omega)}^2,$$
 (3.32)

where k is large enough such that  $\alpha/2 \geq 4\widetilde{c}(k)M^2 > \widetilde{c}(k)M^2 > \widetilde{c}(k)\|\ell\|_{W^{-1,p}(\Omega)}^2$ . In view of (3.31) and (3.32) we finally have

$$\langle Bz + \partial_{\varphi} F(\ell, \varphi)z, z \rangle_{H^{1}(\Omega)} \ge \alpha/2 \|z\|_{H^{1}(\Omega)}^{2} \quad \forall z \in H^{1}(\Omega),$$
 (3.33)

i.e., the coercivity of  $B + \partial_{\varphi} F(\ell, \varphi)$  for any  $(\ell, \varphi) \in \mathcal{B}_M \times H^1(\Omega)$ . Due to the boundedness thereof, we obtain by Lax-Milgram's lemma that  $\partial_{\varphi} \Psi(\ell, d, \varphi) : H^1(\Omega) \to H^1(\Omega)^*$  is bijective for all  $(\ell, d, \varphi) \in \mathcal{B}_M \times L^2(\Omega) \times H^1(\Omega)$ .

The continuously Fréchet-differentiability of  $\Phi: \mathcal{B}_M \times L^2(\Omega) \to H^1(\Omega)$  is ultimately given, as in Proposition 1.60, p. 51, by applying the implicit function theorem, this time for  $\Psi$  in (3.25). Therefor we need the results shown in steps (I) and (II) above, as well as Definition 3.10. The identity (3.24) is proven in the same manner as (1.104), p. 51. From e.g. [84, Section 4.7] we know that

$$\Phi'(\ell, d) = -[\partial_{\varphi} \Psi(\ell, d, \Phi(\ell, d))]^{-1} \partial_{(\ell, d)} \Psi(\ell, d, \Phi(\ell, d)) \quad \forall (\ell, d) \in \mathcal{B}_M \times L^2(\Omega),$$

which, due to (3.25), can be continued as

$$\Phi'(\ell,d) = -[B + \partial_{\varphi} F(\ell,\Phi(\ell,d))]^{-1} \begin{pmatrix} \partial_{\ell} \Psi(\ell,d,\Phi(\ell,d)) \\ \partial_{d} \Psi(\ell,d,\Phi(\ell,d)) \end{pmatrix} \quad \forall (\ell,d) \in \mathcal{B}_{M} \times L^{2}(\Omega). \quad (3.34)$$

In order to write down the equation which characterizes  $\delta \phi$  it remains us to determine the derivative of  $\Psi$  with respect to  $\ell$ . To this end, notice that in view of (3.25) we have

$$\partial_{\ell}\Psi(\ell,d,\varphi) = \partial_{\ell}F(\ell,\varphi) \quad \forall (\ell,d,\varphi) \in \mathbf{W}^{-1,p}(\Omega) \times L^{2}(\Omega) \times H^{1}(\Omega).$$

Thanks to (3.26), the product rule (5.1) together with (3.28) then gives for all  $(\ell, d, \varphi) \in \mathbf{W}^{-1,p}(\Omega) \times L^2(\Omega) \times H^1(\Omega)$ 

$$\partial_{\ell}\Psi(\ell,d,\varphi) = P_1(g'(\varphi),\partial_{\ell}\mathcal{H}(\ell,\varphi)(\cdot)) = P_1(g'(\varphi),2\mathbb{C}\varepsilon(\mathcal{U}(\ell,\varphi)):\varepsilon(\partial_{\ell}\mathcal{U}(\ell,\varphi)(\cdot))), (3.35)$$

where  $P_1$  is the product defined in (1.88) on page 47. Using (3.35) in (3.34) and taking (3.30) into account then yields that  $\delta \phi$  is the unique solution of (3.24). Note that since  $\mathcal{B}_M \times L^2(\Omega)$  is an open set, we entirely skip the end of the proof of Proposition 1.60, p. 51.

We finalize the discussion of the elliptic system (3.4) by pointing out that, given M > 0, we have for all  $(\ell, \varphi, d) \in \mathcal{B}_M \times H^1(\Omega) \times L^2(\Omega)$  the estimates

$$\|\mathcal{U}'(\ell,\varphi)\|_{\mathcal{L}(\boldsymbol{W}^{-1,p}(\Omega)\times H^{1}(\Omega);V)} \leq L(M), \quad \|\Phi'(\ell,d)\|_{\mathcal{L}(\boldsymbol{W}^{-1,p}(\Omega)\times L^{2}(\Omega);H^{1}(\Omega))} \leq L(M), \tag{3.36}$$

where L(M) is the maximum of the Lipschitz constants of  $\mathcal{U}$  and  $\Phi$ . These estimates are given by Lemma 5.8, where we employed Lemmata 3.14, 3.5 and 3.15, 3.11, respectively (provided that Assumptions 1.17, p. 19 (with  $\beta = \beta(M)$ ) and 1.56, p. 46 hold).

### 3.2 The control-to-state operator S

In this section we turn our attention to the constraint in  $(P_{min})$ , i.e., the problem  $(P_{\ell})$ . Recall that the unique solvability thereof was already established in Chapter 1, under Assumption 1.17, p. 19, in the context where  $\ell \in C^{0,1}([0,T]; \mathbf{W}^{-1,p}(\Omega))$  was fixed. We refer here to the end of Section 1.1 and Theorem 1.62 on page 52, where

 $\ell \in C^1([0,T]; \mathbf{W}^{-1,p}(\Omega))$ . In the previous chapters it was reasonable to focus on the time dependence of the solution operators, not on the dependence on the load, which was just one of the given data. In particular, we were interested in the differentiability with respect to time of the solution operators, see Section 1.3, which in combination with the smoothness in time of the load played an essential role when deriving the *energy identity*. We refer here to Lemma 2.2. However, for the sole purpose of showing unique solvability of  $(P_{\ell})$  it suffices to consider some fixed load in  $L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega))$ , as the next results show. The situation changes when we want to control the load in the context of analyzing the problem  $(P_{min})$ . As we will see, this calls for working with (variable) loads in a bounded subset of  $L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega))$ , see Definition 3.21 below.

The aim of this section is to introduce the control-to-state operator, that is, the solution operator associated to the constraint in  $(P_{min})$ , where nonsmooth in time loads are allowed. To this end, we first discuss the transition from smooth to nonsmooth (fixed) loads. We end the section by addressing the transition from fixed to variable (nonsmooth) loads.

#### Unique solvability of $(P_{\ell})$ for a nonsmooth load

From Remark 3.13 we deduce: given  $\ell:[0,T]\to W^{-1,p}(\Omega)$  and  $d:[0,T]\to L^2(\Omega)$ , the minimization problem in  $(P_\ell)$  has a unique solution  $(\bar{\boldsymbol{u}},\bar{\varphi})$ , which satisfies

$$\bar{\boldsymbol{u}}(t) = \mathcal{U}(\ell(t), \bar{\varphi}(t)), \quad \bar{\varphi}(t) = \Phi(\ell(t), d(t)) \quad \text{f.a.a. } t \in (0, T),$$
 (3.37)

provided that there exists M > 0 so that  $\ell(t) \in \mathcal{B}_M$  f.a.a.  $t \in (0,T)$  and Assumption 1.17, p. 19 with  $\beta = \beta(M)$  is satisfied. In view thereof, we work in what follows with a load in  $L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega))$  and impose that  $\beta = \beta(\|\ell\|+1)$  in Assumption 1.17.2, p. 20 (see also Remark 3.9) in order to ensure that  $\bar{\varphi}(t)$ , and consequently,  $\bar{\mathbf{u}}(t)$ , exist for almost all  $t \in (0,T)$ . Note that we have to set  $M > \|\ell\|_{L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega))} (\geq \|\ell(t)\|_{\mathbf{W}^{-1,p}(\Omega)}$  f.a.a.  $t \in (0,T)$ ) in Definition 3.10 only because  $\Phi$  is defined on an open ball. Due to the characterization of the optimal displacement and optimal nonlocal damage in (3.37),  $(P_{\ell})$  reduces to

$$\dot{d}(t) = \frac{1}{\delta} \max \left( -\beta(d(t) - \Phi(\ell(t), d(t)) - r \right) \quad \text{f.a.a.} \ t \in (0, T), \quad d(0) = d_0, \quad (3.38)$$

in view of Theorem 1.29 on page 26. Recall that the unique solvability of (3.38) was already addressed in Theorem 1.32 on page 28, however for  $\ell \in C^{0,1}([0,T]; \mathbf{W}^{-1,p}(\Omega))$ . As nonsmooth loads are now allowed, we have to readdress (3.38), this time by employing Picard-Lindelöf's theorem in abstract function spaces (Lemma 5.7). Prior to this, let us give some comments on the expected time regularity of  $(\bar{\mathbf{u}}, \bar{\varphi})$ , which will be useful when formulating Lemma 3.17 below.

Remark 3.16 (Expected time regularity). Since  $(\bar{\boldsymbol{u}}(t), \bar{\varphi}(t))$  solves the elliptic system (3.4) with right-hand side  $(\ell(t), d(t))$  for almost all  $t \in (0, T)$ , one does not expect  $\bar{\boldsymbol{u}}$  and  $\bar{\varphi}$  to possess more time regularity than  $\ell$  does, regardless of how much time regularity d will turn out to have. One sees that  $\bar{\boldsymbol{u}}$  and  $\bar{\varphi}$  are differentiable in some sense with

respect to time, if  $\ell$  does so. This was the case in Section 1.3, where  $\ell$  belonged to  $C^1([0,T]; \mathbf{W}^{-1,p}(\Omega))$ . So, before having any information about the time regularity of the local damage, it is sufficient to investigate the time regularity of the solutions of the minimization problem in  $(P_{\ell})$  for  $d \in L^{\infty}(0,T;L^2(\Omega))$ .

**Lemma 3.17.** Suppose that Assumption 1.17 on page 19 is satisfied, with  $\beta = \beta(\|\ell\|+1)$ , where  $\ell \in L^{\infty}(0,T; \mathbf{W}^{-1,p}(\Omega))$  is fixed. Then the solution  $(\bar{\mathbf{u}}, \bar{\varphi})$  of the optimization problem in  $(P_{\ell})$  belongs to  $L^{\infty}(0,T; \mathbf{W}_{D}^{1,s}(\Omega)) \times L^{\infty}(0,T; H^{1}(\Omega))$ , for any  $s \in [2,p)$ , provided that  $d \in L^{\infty}(0,T; L^{2}(\Omega))$ .

Proof. As already mentioned at the beginning of the present section, the unique solution of the minimization problem in  $(P_{\ell})$  satisfies (3.37) for almost all  $t \in (0,T)$  (if Assumption 1.17, p. 19, with  $\beta = \beta(\|\ell\| + 1)$  holds). Since  $\ell$  and d are Bochner measurable by assumption, the mapping  $t \mapsto (\ell(t), d(t))$  is Bochner measurable as well. Thanks to  $\ell \in L^{\infty}(0,T; \mathbf{W}^{-1,p}(\Omega))$ , it holds  $\ell(t) \in B_{\mathbf{W}^{-1,p}(\Omega)}(0,\|\ell\| + 1)$  f.a.a.  $t \in (0,T)$ , so that we can combine Lemma 3.11 with Lemma 5.11, where we set  $f = \Phi$ ,  $U = B_{\mathbf{W}^{-1,p}(\Omega)}(0,\|\ell\| + 1) \times L^2(\Omega)$  and  $Y = H^1(\Omega)$ . This yields the Bochner measurability of  $\bar{\varphi} : [0,T] \to H^1(\Omega)$ . Hence,  $t \mapsto (\ell(t),\bar{\varphi}(t))$  is Bochner measurable. By applying Lemma 5.11 again, this time for  $f = \mathcal{U}, X = \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$  and  $Y = \mathbf{W}_D^{1,s}(\Omega)$ , where we take Lemma 3.2 into account, we obtain the Bochner measurability of  $\bar{\mathbf{u}} : [0,T] \to \mathbf{W}_D^{1,s}(\Omega)$ . Further, (3.5) gives immediately that  $\bar{\mathbf{u}} \in L^{\infty}(0,T; \mathbf{W}_D^{1,s}(\Omega))$ . From the Lipschitz continuity of the operator  $\Phi : B_{\mathbf{W}^{-1,p}(\Omega)}(0,\|\ell\| + 1) \times L^2(\Omega) \to H^1(\Omega)$ , cf. Lemma 3.11, one deduces that f.a.a.  $t \in (0,T)$  it holds

$$\|\Phi(\ell(t),d(t))\|_{H^1(\Omega)} \le L(\|\ell\|+1)(\|\ell(t)\|_{\boldsymbol{W}^{-1,p}(\Omega)} + \|d(t)\|_2) + \|\Phi(0,0)\|_{H^1(\Omega)}.$$

The fact that  $(\ell, d) \in L^{\infty}(0, T; \mathbf{W}^{-1,p}(\Omega) \times L^2(\Omega))$  ensures that  $\bar{\varphi} \in L^{\infty}(0, T; H^1(\Omega))$ , which completes the proof.

Remark 3.18. Note that one cannot expect in Lemma 3.17 that  $\bar{\boldsymbol{u}}$  belongs to the space  $L^{\infty}(0,T;\boldsymbol{W}_{D}^{1,p}(\Omega))$ . In view of Definition 3.1, one needs for the Bochner measurability of  $\bar{\boldsymbol{u}}:[0,T]\to\boldsymbol{W}_{D}^{1,p}(\Omega)$  that  $t\mapsto A_{\bar{\varphi}(t)}^{-1}\in\mathcal{L}(\boldsymbol{W}^{-1,p}(\Omega),\boldsymbol{W}_{D}^{1,p}(\Omega))$  is Bochner measurable. Since  $\bar{\varphi}\in L^{\infty}(0,T;H^{1}(\Omega))$ , this asks for the continuity of  $H^{1}(\Omega)\ni\varphi\mapsto A_{\varphi}^{-1}$ , for which a norm gap is needed, on account of Lemma 5.2. In order to see this, observe that one can deduce as in Lemma 1.11 on page 16 that

$$||A_{\varphi_n}^{-1} - A_{\varphi}^{-1}||_{\mathcal{L}(\mathbf{W}^{-1,p}(\Omega),\mathbf{W}_D^{1,s}(\Omega))} \le C ||g(\varphi_n) - g(\varphi)||_{\varrho} \quad \forall n \in \mathbb{N},$$

$$(3.39)$$

with  $\varrho \in [1, \infty)$  such that  $1/\varrho + 1/p + 1/s' = 1$ , where the sequence  $\{\varphi_n\} \subset H^1(\Omega)$  and  $\varphi \in H^1(\Omega)$  are arbitrary. Thus, by (3.39) and Lemma 5.2, the operator

$$H^1(\Omega) \ni \varphi \mapsto A_{\varphi}^{-1} \in \mathcal{L}(\boldsymbol{W}^{-1,p}(\Omega), \boldsymbol{W}_D^{1,s}(\Omega))$$
 is continuous

only when s < p, which ultimately highlights why  $\bar{\boldsymbol{u}} : [0,T] \to \boldsymbol{W}_D^{1,p}(\Omega)$  is not necessarily Bochner measurable.

The unique solvability of the evolutionary equation in  $(P_{\ell})$  (for nonsmooth in time loads) is given by the next

**Theorem 3.19.** Let  $\ell \in L^{\infty}(0,T; \mathbf{W}^{-1,p}(\Omega))$  be fixed. Under Assumption 1.17, p. 19, where  $\beta = \beta(\|\ell\| + 1)$ , there exists a unique function  $d \in W^{1,\infty}(0,T;L^2(\Omega))$  satisfying (3.38).

*Proof.* As in Section 1.1 on page 28 we intend to solve (3.38) by means of Picard-Lindelöf's theorem, except that this time we need to apply the version for Bochner-Sobolev spaces, i.e., Lemma 5.7. To this end, we rewrite (3.38) as

$$\dot{d}(t) = f(t, d(t))$$
 f.a.a.  $t \in (0, T)$ ,  $d(0) = d_0$ , (3.40)

where  $f:(0,T)\times L^2(\Omega)\to L^2(\Omega)$  is defined as

$$f(t,d) := \frac{1}{\delta} \max \left( -\beta \left( d - \Phi(\ell(t), d) \right) - r \right). \tag{3.41}$$

Note that for any  $d \in L^2(\Omega)$ , the value  $\Phi(\ell(t), d) \in H^1(\Omega)$  exists f.a.a.  $t \in (0, T)$ , by assumption (see also Definition 3.10). Since max maps  $L^2(\Omega)$  to  $L^2(\Omega)$  in view of Lemma 5.6.(i), the value  $f(t, d) \in L^2(\Omega)$  is well defined f.a.a.  $t \in (0, T)$  and for any  $d \in L^2(\Omega)$ . Let us now check if the other two assumptions on f in Lemma 5.7 are satisfied. Lemma 3.17 tells us that for  $d \in L^\infty(0, T; L^2(\Omega))$  the mapping  $d - \Phi(\ell(\cdot), d(\cdot))$  belongs to  $L^\infty(0, T; L^2(\Omega))$ , so that from Lemma 5.6.(ii) we deduce

$$f(\cdot, d(\cdot)) \in L^{\infty}(0, T; L^{2}(\Omega))$$
 for  $d \in L^{\infty}(0, T; L^{2}(\Omega))$ ,

on account of (3.41). Thereby the second assumption on f in Lemma 5.7 is fulfilled. From Lemma 5.6.(i) we know that the operator max :  $L^2(\Omega) \to L^2(\Omega)$  is Lipschitz continuous with Lipschitz constant 1, which in combination with Lemma 3.11 yields the estimate

$$||f(t,d_1) - f(t,d_2)||_2 \le \frac{1}{\delta} || \max \left( -\beta \left( d_1 - \Phi(\ell(t), d_1) \right) - r \right) - \max \left( -\beta \left( d_2 - \Phi(\ell(t), d_2) \right) - r \right) ||_2 \le \frac{\beta}{\delta} \left( ||d_1 - d_2||_2 + ||\Phi(\ell(t), d_1) - \Phi(\ell(t), d_2)||_2 \right)$$

$$\le \frac{\beta}{\delta} (L+1) ||d_1 - d_2||_2 \quad \forall d_1, d_2 \in L^2(\Omega), \text{ f.a.a. } t \in (0, T),$$

where  $L = L(\|\ell\| + 1) > 0$  is independent of time. Therewith all the assumptions in Lemma 5.7 are verified. Now we can conclude the unique solvability of (3.40), and thus of (3.38), as well as the desired regularity of d.

Notice that Theorem 3.19 basically covers the unique solvability of the problem  $(P_{\ell})$  (for a nonsmooth load), since as already mentioned above, (3.38) is just the reduced form thereof. In view of (3.37), the unique solution of the problem  $(P_{\ell})$  is given by

 $(\mathcal{U}(\ell(\cdot), \Phi(\ell(\cdot), d(\cdot))), \Phi(\ell(\cdot), d(\cdot)), d)$ , where d solves (3.38). The regularity of the optimal displacement and optimal nonlocal damage can be read from Lemma 3.17 and Theorem 3.19, see also Remark 3.16. In light of Definitions 3.1 and 3.10, we can formulate the new solvability result as follows

**Theorem 3.20.** Let  $\ell \in L^{\infty}(0,T; \mathbf{W}^{-1,p}(\Omega))$  be fixed. Moreover, let Assumption 1.17 on page 19 hold true with  $\beta = \beta(\|\ell\|+1)$ . Then the problem  $(P_{\ell})$  admits a unique solution  $(\mathbf{u}, \varphi, d)$ , satisfying  $\mathbf{u} \in L^{\infty}(0,T; \mathbf{W}_D^{1,s}(\Omega))$ , for every  $s \in [2,p)$ ,  $\varphi \in L^{\infty}(0,T; H^1(\Omega))$ ,  $d \in W^{1,\infty}(0,T;L^2(\Omega))$  and the system of differential equations f.a.a.  $t \in (0,T)$ 

$$-\operatorname{div} g(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{u}(t)) = \ell(t) \quad \text{in } \boldsymbol{W}^{-1,p}(\Omega),$$
$$-\alpha\Delta\varphi(t) + \beta\,\varphi(t) + \frac{1}{2}\,g'(\varphi(t))\mathbb{C}\,\varepsilon(\boldsymbol{u}(t)) : \varepsilon(\boldsymbol{u}(t)) = \beta d(t) \quad \text{in } H^1(\Omega)^*,$$
$$\dot{d}(t) - \frac{1}{\delta}\max\left(-\beta(d(t) - \varphi(t)) - r\right) = 0, \quad d(0) = d_0.$$

#### The solution operator S

Within our scope of analyzing  $(P_{min})$ , we are interested in defining the control-tostate operator associated to  $(P_{\ell})$ , i.e., the solution operator that maps  $\ell$  to  $(\boldsymbol{u}, \varphi, d)$ . This means that the load can no longer be seen as given data, but as variable. On the other side, as already pointed out above,  $(P_{\ell})$  is uniquely solvable provided that  $\beta = \beta(\|\ell\| + 1)$ , see also Theorem 3.20 and Remark 3.9. Since  $\beta$  is fixed and may only depend on the given data, we introduce, as in Section 3.1 on page 103, the following

**Definition 3.21** (Time-dependent admissible loads). For a given M > 0 we define

$$\mathfrak{B}_M := B_{L^{\infty}(0,T;\boldsymbol{W}^{-1,p}(\Omega))}(0,M).$$

Note that this implies  $\mathfrak{B}_M \subset \{\ell \in L^{\infty}(0,T; \mathbf{W}^{-1,p}(\Omega)) : \ell(t) \in \mathcal{B}_M \text{ f.a.a. } t \in (0,T)\}.$ 

Thus, given M > 0 and some variable  $\ell \in \mathfrak{B}_M$ , we need to require that  $\beta = \beta(M+1)$  in Assumption 1.17, p. 19. Then,  $\beta$  depends only on the given data and the unique solvability of  $(P_\ell)$  is ensured, since for all  $\ell \in \mathfrak{B}_M$  we have

$$\beta \ge \alpha + 4k(M+1)^2 > \alpha + 4k(\|\ell\|_{L^{\infty}(0,T;\boldsymbol{W}^{-1,p}(\Omega))} + 1)^2,$$
 (3.43)

where k is chosen (large enough, depending only on M,  $\alpha$ , p and N) such that

$$\alpha/2 \ge 4\widetilde{c}(k)(M+1)^2 > 4\widetilde{c}(k)(\|\ell\|_{L^{\infty}(0,T;\boldsymbol{W}^{-1,p}(\Omega))} + 1)^2.$$

Note that  $\beta$  satisfying (3.43) surpasses the threshold needed in Theorem 3.20, so that we can introduce

**Definition 3.22** (Control-to-state operator associated to  $(P_{\ell})$ ). Let M > 0 be given and suppose that Assumption 1.17, p.19, with  $\beta = \beta(M+1)$  holds true. Then we define  $S: \mathfrak{B}_M \to L^{\infty}(0,T; \mathbf{W}_D^{1,s}(\Omega)) \times L^{\infty}(0,T;H^1(\Omega)) \times W^{1,\infty}(0,T;L^2(\Omega))$ , with  $s \in [2,p)$ ,

$$S(\ell) := (\boldsymbol{u}, \varphi, d),$$

where  $(\mathbf{u}, \varphi, d)$  is the unique solution of  $(P_{\ell})$ , i.e., it satisfies f.a.a.  $t \in (0, T)$  the system

$$\mathbf{u}(t) = \mathcal{U}(\ell(t), \varphi(t)), \quad \varphi(t) = \Phi(\ell(t), d(t)),$$

$$\dot{d}(t) = \frac{1}{\delta} \max(-\beta(d(t) - \varphi(t)) - r), \quad d(0) = d_0,$$
(3.44)

with right-hand side  $\ell \in \mathfrak{B}_M$ .

For  $i \in \{1, 2, 3\}$  we denote by  $S_i$  the operator which associates to any  $\ell \in \mathfrak{B}_M$  the *i-th* component of  $S(\ell)$ .

Remark 3.23. Note that in the two-dimensional case the control-to-state operator S has actually range in  $L^{\infty}(0,T;\mathbf{W}_{D}^{1,p}(\Omega)) \times L^{\infty}(0,T;W^{1,q}(\Omega)) \times W^{1,\infty}(0,T;L^{2}(\Omega))$ , with q > 2 given by Theorem 1.37, p. 31. In order to see this, observe that in view of Remark 3.12 one can show that  $\bar{\varphi} \in L^{\infty}(0,T;W^{1,q}(\Omega))$  in Lemma 3.17 by using the exact same arguments. On the other side, in the two-dimensional case the embedding  $W^{1,q}(\Omega) \hookrightarrow L^{\infty}(\Omega)$  holds true, and thanks to Lemma 5.1, one obtains as in Remark 3.18 that

$$W^{1,q}(\Omega) \ni \varphi \mapsto A_{\varphi}^{-1} \in \mathcal{L}(\boldsymbol{W}^{-1,p}(\Omega), \boldsymbol{W}_{D}^{1,p}(\Omega))$$
 is continuous.

Consequently,  $t \mapsto A_{\bar{\varphi}(t)}^{-1} \in \mathcal{L}(\mathbf{W}^{-1,p}(\Omega), \mathbf{W}_D^{1,p}(\Omega))$  is Bochner measurable, which accounts for the Bochner measurability of  $\bar{\mathbf{u}} : [0,T] \to \mathbf{W}_D^{1,p}(\Omega)$  in Lemma 3.17. For more details, we refer to Remark 3.18. Altogether, the improved regularity results in Lemma 3.17 yield in combination with Theorem 3.19 that, in two dimensions, the unique solution of  $(P_{\ell})$  belongs to  $L^{\infty}(0,T;\mathbf{W}_D^{1,p}(\Omega)) \times L^{\infty}(0,T;\mathbf{W}_A^{1,q}(\Omega)) \times \mathbf{W}^{1,\infty}(0,T;L^2(\Omega))$ , with q > 2 given by Theorem 1.37, p. 31.

Remark 3.24 (Lipschitz continuity of S). It is easy to see that the right-hand side in the operator differential equation in (3.44) is Lipschitz continuous w.r.t.  $(\ell(t), d(t)) \in \mathcal{B}_M \times L^2(\Omega)$  f.a.a.  $t \in (0,T)$ , where M>0 is given. This follows from the Lipschitz continuity of max:  $L^2(\Omega) \to L^2(\Omega)$  and  $\Phi: \mathcal{B}_M \times L^2(\Omega) \to H^1(\Omega)$ , see Lemmata 5.6.(i) and 3.11, respectively. In light of Lemma 5.10, it can then be shown that  $S_3: \mathfrak{B}_M \to W^{1,\infty}(0,T;L^2(\Omega))$  is Lipschitz continuous, which as a result of the Lipschitz continuity of  $\mathcal{U}$  and  $\Phi$  (Lemmata 3.5 and 3.11) gives in turn the Lipschitz continuity of  $\mathcal{S}: \mathfrak{B}_M \to L^\infty(0,T;V) \times L^\infty(0,T;H^1(\Omega)) \times W^{1,\infty}(0,T;L^2(\Omega))$ .

### 3.3 The directional differentiability of S

Since in the next section we derive necessary optimality conditions in primal form for  $(P_{min})$ , we address in what follows the differentiability of the control-to-state operator S. Notice that this is not expected to be Gâteaux-differentiable, since in (3.44) the evolution of the local damage d is described via the max-function, which is not Gâteaux-differentiable at 0 (see also (5.18) in Chapter 5).

For the sake of convenience we work in the rest of the section with a fixed pair  $(\ell, \delta\ell) \in \mathfrak{B}_M \times L^{\infty}(0, T; \mathbf{W}^{-1,p}(\Omega))$ , where M > 0 is given. Throughout this section Assumptions 1.17, p. 19 with  $\beta = \beta(M+1)$  and 1.56, p. 46 are supposed to hold. We

prove the directional differentiability of S at  $\ell$  in direction  $\delta \ell$  in a standard manner, by deriving estimates for the components of  $\left\| \frac{S(\ell+\tau\delta\ell)-S(\ell)}{\tau} - S'(\ell;\delta\ell) \right\|$ , which will ultimately converge towards 0 as  $\tau \searrow 0$ . Of course, in this context  $S'(\ell;\delta\ell)$  does not denote the derivative, but the candidate therefor. Note that this can be provided by 'linearizing' (3.44) at  $\ell$  in direction  $\delta \ell$ . In order to have a better overview of the upcoming results, we use in what follows the abbreviations

$$\ell_{\tau} := \ell + \tau \delta \ell, \quad (\boldsymbol{u}, \varphi, d) := \mathcal{S}(\ell), \quad (\boldsymbol{u}_{\tau}, \varphi_{\tau}, d_{\tau}) := \mathcal{S}(\ell_{\tau}),$$

$$\boldsymbol{\delta u} := \mathcal{U}'(\ell(\cdot), \varphi(\cdot)) (\delta \ell(\cdot), \delta \varphi(\cdot)), \quad \delta \varphi := \Phi'(\ell(\cdot), d(\cdot)) (\delta \ell(\cdot), \delta d(\cdot)),$$
(3.45)

where  $\delta d \in L^{\infty}(0,T;L^2(\Omega))$  is arbitrary, but fixed and  $\tau > 0$  is small enough such that  $\ell_{\tau} \in \mathfrak{B}_M$ . Notice that such a  $\tau$  exists, since  $\mathfrak{B}_M$  is open, and all the operators in (3.45) are well defined. In particular, we remark that  $\delta \varphi$  is a.e. in (0,T) well defined, in view of Definition 3.21 and Lemma 3.15.

Before we proceed with the first estimates, we establish the time-integrability of  $\boldsymbol{\delta u}$  and  $\delta \varphi$ . We know that  $\mathcal{U}' \in C(\boldsymbol{W}^{-1,p}(\Omega) \times H^1(\Omega); \mathcal{L}(\boldsymbol{W}^{-1,p}(\Omega) \times H^1(\Omega); V))$  and  $\Phi' \in C(\mathcal{B}_M \times L^2(\Omega); \mathcal{L}(\boldsymbol{W}^{-1,p}(\Omega) \times L^2(\Omega); H^1(\Omega)))$ , from Lemmata 3.14 and 3.15, respectively. Thus, with Lemma 5.11 we can deduce the measurability of  $\mathcal{U}'(\ell(\cdot), \varphi(\cdot))$  and  $\Phi'(\ell(\cdot), d(\cdot))$ . Moreover, in view of (3.36), it is clear that  $\mathcal{U}'(\ell(\cdot), \varphi(\cdot))$  belongs to  $L^{\infty}(0, T; \mathcal{L}(\boldsymbol{W}^{-1,p}(\Omega) \times H^1(\Omega); V))$  and  $\Phi'(\ell(\cdot), d(\cdot)) \in L^{\infty}(0, T; \mathcal{L}(\boldsymbol{W}^{-1,p}(\Omega) \times L^2(\Omega); H^1(\Omega)))$ . Since  $(\delta \ell, \delta d) \in L^{\infty}(0, T; \boldsymbol{W}^{-1,p}(\Omega) \times L^2(\Omega))$ , we establish as in [81, Lemma 3.1.19] that

$$\delta \varphi \in L^{\infty}(0, T; H^{1}(\Omega)). \tag{3.46}$$

Thereby we have  $(\delta \ell, \delta \varphi) \in L^{\infty}(0, T; \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega))$  and as in [81, Lemma 3.1.19] we also obtain

$$\delta u \in L^{\infty}(0, T; V). \tag{3.47}$$

**Lemma 3.25.** Let  $(\ell, \delta \ell) \in \mathfrak{B}_M \times L^{\infty}(0, T; \mathbf{W}^{-1,p}(\Omega))$ , where M > 0 is given and let Assumptions 1.17, p. 19 with  $\beta = \beta(M+1)$  and 1.56, p. 46 hold true. Then, there exist constants C, c > 0, so that for the quantities defined in (3.45), we have for almost all  $t \in (0,T)$  the estimates

$$\left\| \frac{\boldsymbol{u}_{\tau}(t) - \boldsymbol{u}(t)}{\tau} - \boldsymbol{\delta} \boldsymbol{u}(t) \right\|_{V} \le C \left\| \frac{d_{\tau}(t) - d(t)}{\tau} - \delta d(t) \right\|_{2} + cR_{\Phi}(t, \tau) + R_{\mathcal{U}}(t, \tau), \quad (3.48a)$$

$$\left\| \frac{\varphi_{\tau}(t) - \varphi(t)}{\tau} - \delta \varphi(t) \right\|_{H^{1}(\Omega)} \le C \left\| \frac{d_{\tau}(t) - d(t)}{\tau} - \delta d(t) \right\|_{2} + R_{\Phi}(t, \tau), \quad (3.48b)$$

where  $\tau > 0$  is small enough, independent of t, and  $R_{\mathcal{U}}, R_{\Phi} : (0,T) \times (0,1) \to [0,\infty)$  are mappings which satisfy

$$R_{\mathcal{U}}(\cdot,\tau), R_{\Phi}(\cdot,\tau) \to 0 \text{ in } L^{\varrho}(0,T) \quad \text{as } \tau \searrow 0,$$
 (3.49)

for any  $\varrho \in [1, \infty)$ .

*Proof.* We intend to prove both estimates by making use of Lemma 5.12. In order to apply it for deriving (3.48a), we set

$$X := \mathbf{W}^{-1,p}(\Omega), \ Y := \mathbf{W}^{-1,p}(\Omega) \times H^{1}(\Omega), \ Z := V,$$

$$\mathfrak{U} := \mathfrak{B}_{M}, \ U := \mathcal{B}_{M} \times H^{1}(\Omega),$$

$$\mathfrak{F} := (\cdot, \mathcal{S}_{2}(\cdot)), \ \mathcal{F} := \mathcal{U},$$

$$x := \ell, \ \delta x := \delta \ell, \ \delta y := (\delta \ell, \delta \varphi).$$

Note that  $\mathfrak U$  is open in  $L^{\infty}(0,T;\mathbf W^{-1,p}(\Omega))$  in view of Definition 3.21. On account of Definition 3.3, U is a product of open sets, and thus, it is open as well. In view of Definition 3.22,  $\mathfrak F$  maps  $\mathfrak B_M$  to  $\mathfrak B_M \times L^{\infty}(0,T;H^1(\Omega)) \subset L^{\infty}(0,T;\mathbf W^{-1,p}(\Omega) \times H^1(\Omega))$ . As a result of Lemmata 3.5 and 3.14,  $U:\mathcal B_M \times H^1(\Omega) \to V$  is Lipschitz continuous and directionally differentiable. Due to  $\ell \in \mathfrak B_M$  and Definition 3.21,  $\varepsilon_1 > 0$  may be chosen small enough such that for any  $z \in B_{L^{\infty}(0,T;\mathbf W^{-1,p}(\Omega))}(\ell,\varepsilon_1\|\delta\ell\|_{L^{\infty}(0,T;\mathbf W^{-1,p}(\Omega))}) \subset \mathfrak B_M$  it holds  $z(t) \in \mathcal B_M$  f.a.a.  $t \in (0,T)$ . Thus,  $\mathfrak F(B_{L^{\infty}(0,T;X)}(\ell,\varepsilon_1\|\delta\ell\|_{L^{\infty}(0,T;\mathbf W^{-1,p}(\Omega))}))(t) \subset \mathcal B_M \times H^1(\Omega)$  f.a.a.  $t \in (0,T)$ . Further, thanks to (3.46), it holds  $(\delta\ell,\delta\varphi) \in L^{\infty}(0,T;\mathbf W^{-1,p}(\Omega) \times H^1(\Omega))$ . In view of Definition 3.21, there exists  $\varepsilon > 0$ , independent of t, such that  $\|\ell(t)\|_{\mathbf W^{-1,p}(\Omega)} \leq M-\varepsilon$  f.a.a.  $t \in (0,T)$ . Then, by choosing  $\varepsilon_2 := \varepsilon/(2\|\delta\ell\|_{L^{\infty}(0,T;\mathbf W^{-1,p}(\Omega)})$  (provided that  $\delta\ell \neq 0$ ) we have f.a.a.  $t \in (0,T)$ 

 $B_{\mathbf{W}^{-1,p}(\Omega)\times H^1(\Omega)}\big((\ell(t),\mathcal{S}_2(\ell)(t)), \varepsilon_2\|(\delta\ell,\delta\varphi)\|_{L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega)\times H^1(\Omega))}\big)\subset \mathcal{B}_M\times H^1(\Omega),$  while for  $\delta\ell=0$  the above inclusion holds for any  $\varepsilon_2>0$ . Let now  $\tau\in(0,\min\{\varepsilon_1,\varepsilon_2\})$  be arbitrary, but fixed. We point out that  $\tau$  is independent of t, since  $\varepsilon_1$  and  $\varepsilon_2$  do so. Since above we covered all the assumptions in Lemma 5.12, we are now in the position to apply the latter one, which in view of (3.45) yields

$$\left\| \frac{\boldsymbol{u}_{\tau}(t) - \boldsymbol{u}(t)}{\tau} - \boldsymbol{\delta} \boldsymbol{u}(t) \right\|_{V} \\
= \left\| \frac{\mathcal{U}(\ell_{\tau}(t), \varphi_{\tau}(t)) - \mathcal{U}(\ell(t), \varphi(t))}{\tau} - \mathcal{U}'(\ell(t), \varphi(t)) \left(\delta \ell(t), \delta \varphi(t)\right) \right\|_{V} \\
\leq L_{\mathcal{U}} \left\| \frac{(\ell_{\tau}(t), \varphi_{\tau}(t)) - (\ell(t), \varphi(t))}{\tau} - (\delta \ell(t), \delta \varphi(t)) \right\|_{\boldsymbol{W}^{-1, p}(\Omega) \times H^{1}(\Omega)} + R_{\mathcal{U}}(t, \tau) \\
= L_{\mathcal{U}} \left\| \frac{\varphi_{\tau}(t) - \varphi(t)}{\tau} - \delta \varphi(t) \right\|_{H^{1}(\Omega)} + R_{\mathcal{U}}(t, \tau) \quad \text{f.a.a. } t \in (0, T),$$
(3.50)

where  $L_{\mathcal{U}} > 0$  and  $R_{\mathcal{U}} : (0,T) \times (0,1) \to [0,\infty)$  satisfies

$$R_{\mathcal{U}}(\cdot,\tau) \to 0 \text{ in } L^{\varrho}(0,T) \quad \text{as } \tau \searrow 0,$$

for any  $\varrho \in [1, \infty)$ .

We now proceed in the exact same way in order to derive (3.48b), that is, we begin

by fixing the setting in Lemma 5.12:

$$X := \mathbf{W}^{-1,p}(\Omega), \ Y := \mathbf{W}^{-1,p}(\Omega) \times L^{2}(\Omega), \ Z := H^{1}(\Omega),$$

$$\mathfrak{U} := \mathfrak{B}_{M}, \ U := \mathcal{B}_{M} \times L^{2}(\Omega),$$

$$\mathfrak{F} := (\cdot, \mathcal{S}_{3}(\cdot)), \ \mathcal{F} := \Phi,$$

$$x := \ell, \ \delta x := \delta \ell, \ \delta y := (\delta \ell, \delta d).$$

We verify if the above setting fulfills the assumptions in Lemma 5.12, by addressing only those which haven't been already checked in the first part of the proof. From Definition 3.22 we know that  $\mathfrak{F}$  maps  $\mathfrak{B}_M$  to  $L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega)\times L^2(\Omega))$ . Due to Lemmata 3.11 and 3.15,  $\Phi:\mathcal{B}_M\times L^2(\Omega)\to H^1(\Omega)$  is Lipschitz continuous and directionally differentiable. Note that we can choose the exact same value for  $\varepsilon_1>0$  as in the first part of the proof. Further, the value of  $\varepsilon_2>0$  can be again chosen independent of t, so that f.a.a.  $t\in(0,T)$  we have

$$B_{\mathbf{W}^{-1,p}(\Omega)\times L^2(\Omega)}\big((\ell(t),\mathcal{S}_3(\ell)(t)),\varepsilon_2\|(\delta\ell,\delta d)\|_{L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega)\times L^2(\Omega))}\big)\subset \mathcal{B}_M\times L^2(\Omega).$$

Hence, for  $\tau \in (0, \min\{\varepsilon_1, \varepsilon_2\})$ , independent of t, we can now apply Lemma 5.12, which in view of (3.45) allows us to finally deduce

$$\left\| \frac{\varphi_{\tau}(t) - \varphi(t)}{\tau} - \delta\varphi(t) \right\|_{H^{1}(\Omega)}$$

$$= \left\| \frac{\Phi(\ell_{\tau}(t), d_{\tau}(t)) - \Phi(\ell(t), d(t))}{\tau} - \Phi'(\ell(t), d(t)) \left( \delta\ell(t), \delta d(t) \right) \right\|_{H^{1}(\Omega)}$$

$$\leq L_{\Phi} \left\| \frac{(\ell_{\tau}(t), d_{\tau}(t)) - (\ell(t), d(t))}{\tau} - (\delta\ell(t), \delta d(t)) \right\|_{\mathbf{W}^{-1, p}(\Omega) \times L^{2}(\Omega)} + R_{\Phi}(t, \tau)$$

$$= L_{\Phi} \left\| \frac{d_{\tau}(t) - d(t)}{\tau} - \delta d(t) \right\|_{2} + R_{\Phi}(t, \tau) \quad \text{f.a.a. } t \in (0, T),$$

$$(3.51)$$

where  $L_{\Phi} > 0$  and  $R_{\Phi} : (0,T) \times (0,1) \to [0,\infty)$  fulfills

$$R_{\Phi}(\cdot,\tau) \to 0 \text{ in } L^{\varrho}(0,T) \quad \text{ as } \tau \searrow 0,$$

for any  $\varrho \in [1, \infty)$ . Inserting estimate (3.51) in (3.50) now gives (3.48a). Since (3.51) is (3.48b), the proof is now complete.

In view of the operator differential equation in (3.44), the next lemma provides the candidate for the derivative of  $S_3$  at  $\ell$  in direction  $\delta\ell$ .

**Lemma 3.26.** Let  $(\ell, \delta\ell) \in \mathfrak{B}_M \times L^{\infty}(0, T; \mathbf{W}^{-1,p}(\Omega))$ , where M > 0 is given. Suppose that Assumptions 1.17, p. 19 with  $\beta = \beta(M+1)$  and 1.56, p. 46 hold true. Then the equation

$$\dot{\eta}(t) = \frac{1}{\delta} \max' \left( -\beta(d(t) - \varphi(t)) - r; -\beta(\eta(t) - \Phi'(\ell(t), d(t))) (\delta \ell(t), \eta(t))) \right) 
f.a.a. \ t \in (0, T),$$

$$\eta(0) = 0$$
(3.52)

admits a unique solution  $\eta \in W_0^{1,\infty}(0,T;L^2(\Omega))$ , where  $d = \mathcal{S}_3(\ell)$  and  $\varphi = \mathcal{S}_2(\ell)$ .

*Proof.* We solve the operator differential equation by means of Lemma 5.7. In view of (3.52), we define  $f:(0,T)\times L^2(\Omega)\to L^2(\Omega)$  as

$$f(t,\eta) = \frac{1}{\delta} \max' \left( -\beta(d(t) - \varphi(t)) - r; -\beta \left( \eta - \Phi' \left( \ell(t), d(t) \right) \left( \delta \ell(t), \eta \right) \right) \right). \tag{3.53}$$

Note that for any  $\eta \in L^2(\Omega)$ , the value  $f(t,\eta) \in L^2(\Omega)$  is well defined f.a.a.  $t \in (0,T)$ , as a consequence of Lemmata 5.6.(i) and 3.15. Let us now check if the other assumptions on f in Lemma 5.7 are satisfied. Definition 3.22 tells us that  $d - \varphi \in L^{\infty}(0,T;L^2(\Omega))$  and for  $\eta \in L^{\infty}(0,T;L^2(\Omega))$  we deduce from (3.45) and (3.46) that

$$\eta - \Phi'\big(\ell(\cdot), d(\cdot)\big) \big(\delta\ell(\cdot), \eta(\cdot)\big) \in L^{\infty}(0, T; L^{2}(\Omega)).$$

Now Lemma 5.6.(ii) gives  $\max' \left(-\beta(d(\cdot)-\varphi(\cdot))-r; -\beta(\eta(\cdot)-\Phi'(\ell(\cdot),d(\cdot))(\delta\ell(\cdot),\eta(\cdot)))\right) \in L^{\infty}(0,T;L^{2}(\Omega))$ , and hence

$$f(\cdot, \eta(\cdot)) \in L^{\infty}(0, T; L^{2}(\Omega)) \text{ for } \eta \in L^{\infty}(0, T; L^{2}(\Omega)).$$

Let now  $\eta_i \in L^2(\Omega)$  be arbitrary, but fixed and let us abbreviate

$$\delta \varphi_i := \Phi'(\ell(\cdot), d(\cdot))(\delta \ell(\cdot), \eta_i)$$
 a.e. in  $(0, T)$ 

for i=1,2. From Lemma 5.6.(i) we know that for any  $y\in L^2(\Omega)$  the operator  $\max'(y;\cdot):L^2(\Omega)\to L^2(\Omega)$  is Lipschitz continuous with Lipschitz constant 1 and in view of (3.53) we have as a consequence of (3.36) the estimate

$$||f(t,\eta_{1}) - f(t,\eta_{2})||_{2} \leq \frac{1}{\delta} ||-\beta(\eta_{1} - \delta\varphi_{1}(t)) + \beta(\eta_{2} - \delta\varphi_{2}(t))||_{2}$$

$$\leq \frac{\beta}{\delta} (||\eta_{1} - \eta_{2}||_{2} + ||\Phi'(\ell(t), d(t))||_{\mathcal{L}(\mathbf{W}^{-1,p}(\Omega) \times L^{2}(\Omega); H^{1}(\Omega))} ||\eta_{1} - \eta_{2}||_{2})$$

$$\leq \frac{\beta}{\delta} (L+1) ||\eta_{1} - \eta_{2}||_{2} \quad \text{f.a.a. } t \in (0,T),$$

where L = L(M) > 0 depends only on the given data. Therewith all the assumptions in Lemma 5.7 are verified. Now we can conclude the unique solvability of (3.52), as well as the desired regularity of  $\eta$ .

With the candidate for the derivative of  $S_3$  at hand, we can now prove its directional differentiability at  $\ell$  in direction  $\delta \ell$ , which is covered by the following

**Lemma 3.27** (Directional differentiability of  $S_3$ ). Let  $\varrho \in [1, \infty)$  and  $(\ell, \delta \ell) \in \mathfrak{B}_M \times L^{\infty}(0, T; \mathbf{W}^{-1,p}(\Omega))$ , where M > 0 is given. Under Assumptions 1.17, p. 19, where  $\beta = \beta(M+1)$ , and 1.56, p. 46, the following convergence holds true

$$\frac{\mathcal{S}_3(\ell + \tau \delta \ell) - \mathcal{S}_3(\ell)}{\tau} \to \eta \quad in \ W^{1,\varrho}(0,T;L^2(\Omega)) \quad as \ \tau \searrow 0,$$

where  $\eta \in W_0^{1,\infty}(0,T;L^2(\Omega))$  is the unique solution of (3.52).

*Proof.* For simplicity we use again the notations  $d_{\tau} = \mathcal{S}_3(\ell + \tau \delta \ell)$  and  $d = \mathcal{S}_3(\ell)$ , where  $\tau > 0$  is small enough such that  $\ell + \tau \delta \ell \in \mathfrak{B}_M$ . The result is proven in two steps. We first derive estimates for the term  $\left\| \frac{\dot{d}_{\tau}(t) - \dot{d}(t)}{\tau} - \dot{\eta}(t) \right\|_2$  a.e. in (0, T), by employing Lemma 5.12. This will lead to an estimate of the type (5.38) and Lemma 5.10 will ultimately give the assertion.

(i) In view of (3.45) and (3.52) we fix the following setting in Lemma 5.12:

$$X := \mathbf{W}^{-1,p}(\Omega), \ Y := Z := L^{2}(\Omega),$$

$$\mathfrak{F} := \mathfrak{B}_{M}, \ U := L^{2}(\Omega),$$

$$\mathfrak{F} := -\beta \big( \mathcal{S}_{3}(\cdot) - \mathcal{S}_{2}(\cdot) \big) - r, \ \mathcal{F} := \max,$$

$$x := \ell, \ \delta x := \delta \ell, \ \delta y := -\beta \big( \eta - \Phi' \big( \ell(\cdot), d(\cdot) \big) \big( \delta \ell(\cdot), \eta(\cdot) \big) \big).$$

Clearly,  $\mathfrak U$  and U are open. In view of Definition 3.22,  $\mathfrak F$  maps  $\mathfrak B_M$  to  $L^\infty(0,T;L^2(\Omega))$ , and according to Lemma 5.6.(i), max :  $L^2(\Omega) \to L^2(\Omega)$  is Lipschitz continuous and directionally differentiable. Moreover, in view of Lemma 3.26 and (3.46) it holds  $\delta y \in L^\infty(0,T;L^2(\Omega))$ . Since  $U=L^2(\Omega)$ , the assumptions on  $\varepsilon_1$  and  $\varepsilon_2$  in Lemma 5.12 are automatically fulfilled. Hence, for  $\tau>0$  small enough, depending on  $\ell$  and  $\delta\ell$ , Lemma 5.12 together with the operator differential equation in (3.44) and (3.52) now gives

$$\delta \left\| \frac{\dot{d}_{\tau}(t) - \dot{d}(t)}{\tau} - \dot{\eta}(t) \right\|_{2} = \left\| \frac{\max\left(\mathfrak{F}(\ell_{\tau})(t)\right) - \max\left(\mathfrak{F}(\ell)(t)\right)}{\tau} - \max'\left(\mathfrak{F}(\ell)(t); \delta y(t)\right) \right\|_{2}$$

$$\leq \left\| \frac{\mathfrak{F}(\ell_{\tau})(t) - \mathfrak{F}(\ell)(t)}{\tau} - \delta y(t)\right\|_{2} + R_{\max}(t, \tau) \quad \text{f.a.a. } t \in (0, T),$$

$$(3.54)$$

where  $R_{\text{max}}:(0,T)\times(0,1)\to[0,\infty)$  satisfies

$$R_{\max}(\cdot,\tau) \to 0 \text{ in } L^{\varrho}(0,T) \quad \text{as } \tau \searrow 0.$$

Further, in view of the definition of  $\mathfrak{F}$  and  $\delta y$ , the estimate (3.54) can be continued, for  $\tau > 0$  small enough, as follows

$$\delta \left\| \frac{\dot{d}_{\tau}(t) - \dot{d}(t)}{\tau} - \dot{\eta}(t) \right\|_{2}$$

$$\leq \beta \left\| \frac{d_{\tau}(t) - d(t)}{\tau} - \eta(t) \right\|_{2} + \beta \left\| \frac{\varphi_{\tau}(t) - \varphi(t)}{\tau} - \Phi'(\ell(t), d(t)) \left( \delta \ell(t), \eta(t) \right) \right\|_{2} + R_{\max}(t, \tau)$$

$$\leq 2\beta \left\| \frac{d_{\tau}(t) - d(t)}{\tau} - \eta(t) \right\|_{2} + \beta R_{\Phi}(t, \tau) + R_{\max}(t, \tau) \quad \text{f.a.a. } t \in (0, T),$$

$$(3.55)$$

where the last inequality is ensured by (3.48b) and the embedding  $H^1(\Omega) \hookrightarrow L^2(\Omega)$ . Note that  $R(\cdot,\tau) := \beta R_{\Phi}(\cdot,\tau) + R_{\max}(\cdot,\tau) \geq 0$  f.a.a.  $t \in (0,T)$  and  $R(\cdot,\tau) \to 0$  in  $L^{\varrho}(0,T)$  as  $\tau \searrow 0$ .

(ii) Furthermore, in view of Definition 3.22 and Lemma 3.26,

$$\frac{d_{\tau}-d}{\tau}-\eta\in W^{1,\infty}_0(0,T;L^2(\Omega))\hookrightarrow W^{1,\varrho}_0(0,T;L^2(\Omega)).$$

Thanks to the properties of R and (3.55) we can now apply Lemma 5.10, which gives in turn

$$\left\| \frac{d_{\tau} - d}{\tau} - \eta \right\|_{W^{1,\varrho}(0,T;L^2(\Omega))} \le C \|R(\cdot,\tau)\|_{L^{\varrho}(0,T)} \to 0 \text{ as } \tau \searrow 0,$$

whence the desired convergence. Note that the constant C > 0 depends only on  $\beta, \delta, \varrho$  and T, in view of Lemma 5.10.

We now have all the necessary tools in order to conclude the main result of this section.

**Proposition 3.28** (Directional differentiability of the control-to-state operator). Let M>0 and  $\varrho\in[1,\infty)$  be given. Under Assumptions 1.17, p. 19, where  $\beta=\beta(M+1)$ , and 1.56, p. 46, the operator  $\mathcal{S}:\mathfrak{B}_M\to L^\varrho(0,T;V)\times L^\varrho(0,T;H^1(\Omega))\times W^{1,\varrho}(0,T;L^2(\Omega))$  is directionally differentiable. The derivative at  $\ell\in\mathfrak{B}_M$  in direction  $\delta\ell\in L^\infty(0,T;\mathbf{W}^{-1,p}(\Omega))$ , which we denote by  $\mathcal{S}'(\ell;\delta\ell):=(\delta\mathbf{u},\delta\varphi,\delta d)$ , belongs to  $L^\infty(0,T;V)\times L^\infty(0,T;H^1(\Omega))\times W_0^{1,\infty}(0,T;L^2(\Omega))$ . Moreover, this satisfies f.a.a.  $t\in(0,T)$  the following system

$$\begin{cases}
\delta \boldsymbol{u}(t) = \mathcal{U}'(\ell(t), \varphi(t))(\delta \ell(t), \delta \varphi(t)), \\
\delta \varphi(t) = \Phi'(\ell(t), d(t))(\delta \ell(t), \delta d(t)), \\
\dot{\delta d}(t) = \frac{1}{\delta} \max' \left( -\beta(d(t) - \varphi(t)) - r; -\beta(\delta d(t) - \delta \varphi(t)) \right), \\
\delta d(0) = 0,
\end{cases}$$
(3.56)

where we abbreviate  $\varphi := \mathcal{S}_2(\ell)$  and  $d := \mathcal{S}_3(\ell)$ .

Proof. Let  $\ell \in \mathfrak{B}_M$  and  $\delta \ell \in L^{\infty}(0,T; \mathbf{W}^{-1,p}(\Omega))$  be arbitrary, but fixed. From Lemma 3.27 we know that  $\mathcal{S}_3: \mathfrak{B}_M \to W^{1,\varrho}(0,T;L^2(\Omega))$  is directionally differentiable and  $\mathcal{S}'_3(\ell;\delta\ell) \in W_0^{1,\infty}(0,T;L^2(\Omega))$  is the unique solution of the operator differential equation in (3.56). Hence, it remains to address the directional differentiability of  $\mathcal{S}_1$  and  $\mathcal{S}_2$ , for which we employ Lemma 3.25. Recall that throughout this section,  $\delta d$  denoted some arbitrary element in  $L^{\infty}(0,T;L^2(\Omega))$ , which we now set  $\delta d:=\mathcal{S}'_3(\ell;\delta\ell)$  for the rest of this proof. By building  $L^{\varrho}(0,T)$ -norms in (3.48), we obtain the estimates

$$\left\| \frac{\boldsymbol{u}_{\tau} - \boldsymbol{u}}{\tau} - \boldsymbol{\delta} \boldsymbol{u} \right\|_{L^{\varrho}(0,T;V)} \leq C \left\| \frac{d_{\tau} - d}{\tau} - \delta d \right\|_{L^{\varrho}(0,T;L^{2}(\Omega))} + \|cR_{\Phi}(\cdot,\tau) + R_{\mathcal{U}}(\cdot,\tau)\|_{L^{\varrho}(0,T)},$$

$$(3.57a)$$

$$\left\| \frac{\varphi_{\tau} - \varphi}{\tau} - \delta \varphi \right\|_{L^{\varrho}(0,T;H^{1}(\Omega))} \leq C \left\| \frac{d_{\tau} - d}{\tau} - \delta d \right\|_{L^{\varrho}(0,T;L^{2}(\Omega))} + \|R_{\Phi}(\cdot,\tau)\|_{L^{\varrho}(0,T)},$$

$$(3.57b)$$

where  $\tau > 0$  is small enough,  $R_{\mathcal{U}}$ ,  $R_{\Phi}$  fulfill (3.49) and C, c > 0 are the constants from Lemma 3.25. Note that the abbreviations in (3.57) have the same meaning as in (3.45), where one sets  $\delta d := \mathcal{S}'_3(\ell; \delta \ell)$ . Moreover, note that the fact that  $\tau$  is independent of t in Lemma 3.25 is crucial for deriving (3.57), and that, due to Definition 3.22, (3.47) and (3.46), the norms in (3.57) are well defined. In view of Lemma 3.27, combined with

 $W^{1,\varrho}(0,T;L^2(\Omega)) \hookrightarrow L^{\varrho}(0,T;L^2(\Omega))$  and (3.49), we finally deduce from (3.57) that  $S_1: \mathfrak{B}_M \to L^{\varrho}(0,T;V)$  and  $S_2: \mathfrak{B}_M \to L^{\varrho}(0,T;H^1(\Omega))$  are directionally differentiable with

$$\mathcal{S}'_{1}(\ell;\delta\ell) = \boldsymbol{\delta u} = \mathcal{U}'\big(\ell(\cdot),\varphi(\cdot)\big)\big(\delta\ell(\cdot),\delta\varphi(\cdot)\big) \in L^{\infty}(0,T;V),$$
  
$$\mathcal{S}'_{2}(\ell;\delta\ell) = \delta\varphi = \Phi'\big(\ell(\cdot),d(\cdot)\big)\big(\delta\ell(\cdot),\delta d(\cdot)\big) \in L^{\infty}(0,T;H^{1}(\Omega)),$$

where we relied on (3.45). Note that the regularity thereof is given by (3.47) and (3.46). This completes the proof.

We point out that  $\mathcal{S}:\mathfrak{B}_M\to L^\infty(0,T;V)\times L^\infty(0,T;H^1(\Omega))\times W^{1,\infty}(0,T;L^2(\Omega))$  is not expected to be directionally differentiable without norm gap. In order to see this, we first take a look at the last estimate in the proof of Lemma 3.27. Therein one observes that the convergence of  $\frac{\mathcal{S}_3(\ell+\tau\delta\ell)-\mathcal{S}_3(\ell)}{\tau}-\mathcal{S}_3'(\ell;\delta\ell)$  in  $W^{1,\varrho}(0,T;L^2(\Omega))$  is ultimately given by the convergence in  $L^\varrho(0,T)$  of  $R_\Phi$  and  $R_{\max}$ , where  $\varrho<\infty$ . We refer here also to the proof of Lemma 5.10. Nevertheless, one could make use of  $W^{1,\varrho}(0,T;L^2(\Omega))\hookrightarrow L^\infty(0,T;L^2(\Omega))$  in order to deduce from (3.48) convergence in  $L^\infty$ -spaces, but this is not the case, since  $R_\mathcal{U}$  and  $R_\Phi$  do not necessarily converge in  $L^\infty(0,T)$ . Therefore, the norm gap is altogether caused by the fact that the convergence of  $R_\mathcal{U}$ ,  $R_\Phi$  and  $R_{\max}$  was shown only in  $L^\varrho(0,T)$  with  $\varrho<\infty$ . Recall that each of these convergences was deduced in Lemma 5.12 by means of Lebesgue's dominated convergence theorem, which does not hold true for  $L^\infty$ -spaces, whence the norm gap.

#### 3.4 Optimality system

The last part of this chapter is entirely concerned with deriving necessary optimality conditions for the optimal control problem governed by the damage model with penalty. For convenience let us recall that this reads as follows

$$\begin{aligned} & \min_{\ell \in \mathfrak{L}} & & \mathcal{J}(\boldsymbol{u}, \varphi, d, \ell) \\ & \text{s.t.} & & (\boldsymbol{u}, \varphi, d) \text{ solves } (P_{\ell}), \end{aligned}$$

where  $\mathfrak{L}$  denotes the set of admissible loads and  $\mathcal{J}$  is the objective functional. In this section we rely on the fact that  $(P_{\ell})$  can be reduced to an ordinary differential equation in Banach space. This was also earlier employed, in the context of defining and differentiating the control-to-state operator. With the results from Section 3.3 at hand, we shall first derive a purely primal optimality condition for  $(P_{min})$ , which is given by (VI) below. At the beginning of the last section we already mentioned that the control-to-state operator is not expected to be Gâteaux-differentiable, while Proposition 3.28 states only its directional differentiability. Indeed, as it will turn out, the Gâteaux-differentiability may be violated, if the *strict complementarity* condition (see Assumption 3.38 below) is not fulfilled. Therefore, the standard adjoint calculus for the derivation of optimality conditions from (VI) below is without further ado not applicable. By means of Assumption 3.38, one ultimately obtains an optimality system equivalent to the classical purely

primal optimality condition. The section ends with a short discussion regarding the existence of solutions for  $(P_{min})$  and with some remarks concerning *strict complementarity*.

We begin by fixing the assumptions on the optimal control problem. We choose to work with a very general setting, in the sense that, in view of practical applications, it would suffice for example that the objective is smoother as in Assumption 3.33 below.

**Assumption 3.29** (The control set).  $\mathfrak{L}$  is a nonempty, convex and bounded subset of  $L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega))$ .

**Remark 3.30.** Note that the boundedness assumption on the control set implies that there exists some M > 0 so that  $\mathfrak{L} \subset \mathfrak{B}_M$ . Hence, the results in Section 3.3 ensure that the control-to-state operator associated to  $(P_{\ell})$  is defined and directionally differentiable on  $\mathfrak{L}$  (provided that  $\beta = \beta(M+1)$ ). Notice that a reasonable choice for the set of admissible loads is

$$\mathfrak{L} := \{ \ell \in L^{\infty}(0, T; \mathbf{W}^{-1, p}(\Omega)) : \|\ell(t)\|_{\mathbf{W}^{-1, p}(\Omega)} \le b \text{ f.a.a. } t \in (0, T) \},$$

where b > 0 is a given bound.

Of course, throughout this section Assumptions 1.17, p. 19 and 1.56, p. 46 are supposed to further hold. Given the set  $\mathfrak{L}$ , the value of  $\beta$  in Assumption 1.17.2, p. 20, is supposed to satisfy from now on  $\beta = \beta(M+1)$  (see Remark 3.9), where M > 0 is chosen so that  $\mathfrak{L} \subset \mathfrak{B}_M$ .

The assumption on the objective  $\mathcal{J}$  features the notion of Hadamard-differentiability, which is a concept of differentiability often found in the literature (see e.g. [75], [73]) and will be useful when applying chain rule in the proof of Lemma 3.35 below. For the convenience of the reader we give here the definition thereof, as well as an appropriate chain rule.

**Definition 3.31** (Hadamard (directional) differentiability). [73, Definition 3.1.1] Let X and Y be normed vector spaces. A function  $f: X \to Y$  is said to be Hadamard directionally differentiable at  $x \in X$ , if the following limit exists

$$f'_H(x; \delta x) := \lim_{\substack{\tau \searrow 0 \\ z \to \delta x}} \frac{f(x + \tau z) - f(x)}{\tau} \quad \forall \, \delta x \in X.$$

In this situation,  $f'_H(x;\cdot)$  is called Hadamard directional derivative of f at x. If  $f'_H(x;\cdot)$  is linear, then f is called Hadamard-differentiable at x, in view of [73, Lemma 3.1.3 and Proposition 3.2.4 (ii)]. If existent, the Hadamard derivative of f at x is denoted by  $f'_H(x)$  and belongs to  $\mathcal{L}(X,Y)$ , as a result of [73, Lemma 3.1.3]. Note that any Hadamard-differentiable function is Gâteaux-differentiable.

**Lemma 3.32** (Chain rule for Hadamard directionally differentiable functions). [75, Proposition 3.6 (i)] Let X, Y and Z be normed vector spaces and let  $f_1: X \to Y$ ,  $f_2: Y \to Z$  be some mappings. Assume that  $f_1$  is directionally differentiable at  $x \in X$  and

 $f_2$  is Hadamard directionally differentiable at  $f_1(x) \in Y$ . Then the composite mapping  $f_2 \circ f_1 : X \to Z$  is directionally differentiable at  $x \in X$  and the following chain rule holds

$$(f_2 \circ f_1)'(x; \delta x) = f_2'(f_1(x); (f_1'(x; \delta x))) \quad \forall \, \delta x \in X.$$

Now we can state the assumption on the functional  $\mathcal{J}$ :

**Assumption 3.33** (The objective functional).  $\mathcal{J}: L^2(0,T; \boldsymbol{W}_D^{1,\nu}(\Omega)) \times L^2(0,T; H^1(\Omega)) \times L^2(0,T; L^2(\Omega)) \times L^2(0,T; \boldsymbol{W}^{-1,p}(\Omega)) \to \mathbb{R}$  is Hadamard-differentiable, where  $\nu \in (1,2)$  is given.

The Hadamard-differentiability in Assumption 3.33 is self-evident in practice, since most of the objectives are even continuously Fréchet-differentiable, and hence Hadamard-differentiable, in view of [73, Proposition 3.4.2].

An example therefor is the functional  $\mathcal{J}_{ex}:L^2(0,T;L^2(\Omega))\times L^2(0,T;L^2(\Omega))\times L^2(0,T;V^*)$  given by

$$\mathcal{J}_{ex}(\boldsymbol{u},d,\ell) := \frac{1}{2} \|\boldsymbol{u} - \boldsymbol{u}_d\|_{L^2(0,T;L^2(\Omega))}^2 + \frac{\alpha_1}{2} \|d\|_{L^2(0,T;L^2(\Omega))}^2 + \frac{\alpha_2}{2} \|\ell\|_{L^2(0,T;V^*)}^2,$$

where  $\mathbf{u}_d \in L^2(0,T;L^2(\Omega))$  is a desired displacement and  $\alpha_1,\alpha_2 > 0$ . Note that  $\mathcal{J}_{ex}$  fulfills Assumption 3.33 for  $\nu \in (1,2)$  and  $\nu \in [6/5,2)$  in the two-dimensional and three-dimensional case, respectively. This is due to  $\mathbf{W}_D^{1,\nu}(\Omega) \hookrightarrow L^2(\Omega)$  (see e.g. [80, Theorem 7.1]) and  $\mathbf{W}^{-1,p}(\Omega) \hookrightarrow V^*$ . Moreover note that  $\mathcal{J}$  is continuously Fréchet-differentiable.

Remark 3.34. Although it might seem more reasonable to consider in Assumption 3.33 the Hilbert space  $L^2(0,T;V)$  for the displacement instead of  $L^2(0,T;\mathbf{W}_D^{1,\nu}(\Omega))$ , where  $\nu \in (1,2)$ , we choose to work with the latter one in order to obtain  $L^2(0,T;V)$ -regularity in both dimensions for the adjoint state  $\mathbf{w}$  in Lemma 3.40 below. Otherwise, the adjoint state belongs to  $L^2(0,T;\mathbf{W}_D^{1,\zeta}(\Omega))$ , where  $\zeta < 2$ , in the three-dimensional case, while in the two-dimensional case it belongs to  $L^2(0,T;V)$ . For the sake of convenience we choose not to make this distinction. We point out that the condition for  $\nu$  in Assumption 3.33 is essential only for the Bochner measurability of  $\mathbf{w} : [0,T] \to V$ . For the desired integrability of the latter one it suffices to work with  $L^2(0,T;V)$  as space for the displacement. This is due to the estimate (3.65) in the proof of Lemma 3.40 below.

We explain in what follows why the choice of  $\nu$  in Assumption 3.33 is crucial for the Bochner measurability of  $\mathbf{w}:[0,T]\to V$ . Assuming that the space for the displacement is  $L^2(0,T;\mathbf{W}_D^{1,\nu}(\Omega))$ , where  $\nu\in(1,p)$  is to be determined, we have in view of the Hadamard-differentiability of  $\mathcal{J}$  and [14, Theorem 7.1.23 (vi)] that  $\partial_{\mathbf{u}}\mathcal{J}(\cdot)\in L^2(0,T;\mathbf{W}^{-1,\nu'}(\Omega))$ . Hence,  $t\mapsto\partial_{\mathbf{u}}\mathcal{J}(\cdot)(t)\in\mathbf{W}^{-1,\nu'}(\Omega)$  is Bochner measurable. Note that here  $(\cdot)$  denotes a point in the domain of  $\mathcal{J}$ . In view of (3.64) below, one needs for the Bochner measurability of  $\mathbf{w}:[0,T]\to V$  that  $t\mapsto A_{\varphi(t)}^{-1}\in\mathcal{L}(\mathbf{W}^{-1,\nu'}(\Omega),V)$  is Bochner measurable, where  $\varphi\in L^\infty(0,T;H^1(\Omega))$  and  $\nu'\in[2,p]$ . This asks for the continuity of  $H^1(\Omega)\ni\varphi\mapsto A_{\varphi}^{-1}\in\mathcal{L}(\mathbf{W}^{-1,\nu'}(\Omega),V)$ , for which a norm gap is needed, on account of Lemma 5.2. We point out that this is very similar to the situation described in Remark 3.18. Thus,

$$H^1(\Omega) \ni \varphi \mapsto A_{\varphi}^{-1} \in \mathcal{L}(\mathbf{W}^{-1,\nu'}(\Omega), V) \text{ is continuous}$$
 (3.58)

only when  $\nu' > 2$ , which ultimately motivates the assumption on  $\mathcal{J}$ . Similarly, it can be deduced that, working with  $\nu = 2$  in Assumption 3.33 leads to  $\mathbf{w} \in L^2(0,T;\mathbf{W}_D^{1,\zeta}(\Omega))$ , where  $\zeta < 2$ . For this, one also needs Remark 1.4 on page 13.

As mentioned above, in the two-dimensional case it actually suffices to work with  $L^2(0,T;V)$  as space for the displacement in order to obtain the Bochner measurability of  $\mathbf{w}:[0,T]\to V$ . This is due to the fact that, in Lemma 3.40 below,  $\varphi=\mathcal{S}_2(\ell)$ , where  $\ell\in\mathfrak{L}$  and cf. Remark 3.23, there exists q>2 so that  $\varphi\in L^\infty(0,T;W^{1,q}(\Omega))$ . The Bochner measurability of  $\mathbf{w}:[0,T]\to V$  follows then by the arguments employed in Remark 3.23.

With the properties of  $\mathfrak{L}$  and  $\mathcal{J}$  at hand, we can now derive

**Lemma 3.35** (Purely primal necessary optimality conditions). Let Assumptions 1.17, p. 19 and 1.56, p. 46 hold. Moreover, suppose that Assumptions 3.29 and 3.33 are fulfilled. Then, any local solution  $\bar{\ell}$  of the problem  $(P_{min})$  satisfies

$$\partial_{(\boldsymbol{u},\varphi,d)} \mathcal{J}(\mathcal{S}(\bar{\ell}),\bar{\ell}) \left( \mathcal{S}'(\bar{\ell};\delta\ell-\bar{\ell}) \right) + \partial_{\ell} \mathcal{J}(\mathcal{S}(\bar{\ell}),\bar{\ell}) (\delta\ell-\bar{\ell}) \ge 0 \quad \forall \, \delta\ell \in \mathfrak{L}. \tag{VI}$$

*Proof.* The proof is standard. In view of Definition 3.22 and Remark 3.30, the optimal control problem  $(P_{min})$  can be rewritten as

$$\min_{\ell \in \mathfrak{L}} f(\ell),$$

where  $f: \ell \mapsto \mathcal{J}(\mathcal{S}(\ell), \ell)$  is the reduced objective. Due to Proposition 3.28 and Assumption 3.33 we can apply Lemma 3.32 for f. This gives in turn that f is directionally differentiable at  $\bar{\ell} \in \mathfrak{L}$  with

$$f'(\bar{\ell};\delta\ell-\bar{\ell}) = \mathcal{J}'(\mathcal{S}(\bar{\ell}),\bar{\ell})(\mathcal{S}'(\bar{\ell};\delta\ell-\bar{\ell}),\delta\ell-\bar{\ell}) \quad \forall \, \delta\ell \in L^{\infty}(0,T;\boldsymbol{W}^{-1,p}(\Omega)).$$

Note that the Hadamard derivative of  $\mathcal{J}$  coincides with its Gâteaux derivative. From the convexity of  $\mathfrak{L}$  we then deduce that any local minimizer of  $(P_{min})$  satisfies (VI), which completes the proof.

It is easy to see, in view of the Hadamard-differentiability of  $\mathcal{J}$  and (3.56), that, if  $\mathcal{S}_3$  is Gâteaux-differentiable at  $\bar{\ell}$ , then all the terms in (VI) are linear in  $\delta\ell - \bar{\ell}$ . Since, within our scope of deriving an optimality system, we want to write the left-hand side in (VI) as a linear form in  $\delta\ell - \bar{\ell}$ , we investigate in the following under which conditions the nonlinearity of  $\mathcal{S}_3'(\bar{\ell};\cdot)$  can be overcome. Therefor, a closer inspection of the operator differential equation in (3.56) is required.

Prior to this, let us define for simplicity

**Definition 3.36.** For a given  $\ell \in \mathfrak{L}$  we define at  $t \in (0,T)$  the following sets:

- $\Omega_t^+ := \{ x \in \Omega : -\beta(d(t, x) \varphi(t, x)) r > 0 \},$
- $\Omega_t^0 := \{ x \in \Omega : -\beta(d(t, x) \varphi(t, x)) r = 0 \},$
- $\Omega_t^- := \{ x \in \Omega : -\beta(d(t, x) \varphi(t, x)) r < 0 \},$

where we abbreviate  $\varphi := S_2(\ell)$  and  $d := S_3(\ell)$ . Note that, for almost all  $t \in (0,T)$ , each of the above subsets of  $\Omega$  is uniquely specified up to a set of measure zero (in  $\Omega$ ). We emphasize that the above defined sets ultimately depend on  $\ell$  and the given data.

Let now  $\ell \in \mathfrak{L}$  and  $\delta \ell \in L^{\infty}(0,T; \boldsymbol{W}^{-1,p}(\Omega))$  be arbitrary, but fixed. In view of Proposition 3.28 and (5.17) combined with Definition 3.36,  $\delta d := \mathcal{S}'_3(\ell; \delta \ell)$  is characterized as the unique solution of the operator differential equation

$$\dot{\delta d}(t) = \begin{cases} -\frac{\beta}{\delta} (\delta d(t) - \delta \varphi(t)) & \text{a.e. in } \Omega_t^+ \\ \max \left( -\frac{\beta}{\delta} (\delta d(t) - \delta \varphi(t)) \right) & \text{a.e. in } \Omega_t^0 & \text{f.a.a. } t \in (0, T), \quad \delta d(0) = 0. \end{cases}$$

$$0 \quad \text{a.e. in } \Omega_t^-$$

Here we use again the notations  $\delta \varphi := \Phi'(\ell(\cdot), d(\cdot))(\delta \ell(\cdot), \delta d(\cdot))$  and  $d := S_3(\ell)$ .

Remark 3.37. From (3.59) we read, in view of the nonlinearity of  $\max\{\cdot,0\}$ , that as long as there exist  $0 \le t_1 < t_2 \le T$  such that  $\mu(\Omega_t^0) > 0$  f.a.a.  $t \in (t_1,t_2)$ , the operator  $S_3$  is not necessarily Gâteaux-differentiable at  $\ell$ . This is also shown by straight forward computation.

Indeed, as we will next see, the linearity of  $S_3'(\ell;\cdot)$  is ensured if

**Assumption 3.38** (Strict complementarity). The set  $\Omega_t^0$  associated to  $\ell \in \mathfrak{L}$  has measure zero for almost all  $t \in (0,T)$ , i.e.,

$$\mu(\Omega_t^0) = 0$$
 f.a.a.  $t \in (0, T)$ .

To prove that  $S_3'(\ell;\cdot)$  is linear under Assumption 3.38, note that (3.59) reads

$$\dot{\delta d}(t) = f_t(\delta \ell(t), \delta d(t)) \quad \text{f.a.a. } t \in (0, T), \quad \delta d(0) = 0, \tag{3.60}$$

where  $f_t: \mathbf{W}^{-1,p}(\Omega) \times L^2(\Omega) \to L^2(\Omega)$  is given by

$$f_t(\delta\ell, \delta d) := -\frac{\beta}{\delta} \chi_{\Omega_t^+} (\delta d - \Phi'(\ell(t), d(t))(\delta\ell, \delta d)).$$

Observe that, since  $\Phi$  is Gâteaux-differentiable at  $(\ell(t), d(t))$ ,  $f_t$  is linear for almost all  $t \in (0,T)$ . Now let  $\delta \ell_1, \delta \ell_2 \in L^{\infty}(0,T; \mathbf{W}^{-1,p}(\Omega))$  and  $a,b \in \mathbb{R}$ . With the notation  $\delta d_i := \mathcal{S}_3'(\ell; \delta \ell_i)$ , where i = 1, 2, (3.60) leads to

$$a \, \dot{\delta d}_1(t) + b \, \dot{\delta d}_2(t) = a f_t(\delta \ell_1(t), \delta d_1(t)) + b f_t(\delta \ell_2(t), \delta d_2(t))$$
  
=  $f_t(a \, \delta \ell_1(t) + b \, \delta \ell_2(t), a \, \delta d_1(t) + b \, \delta d_2(t))$  f.a.a.  $t \in (0, T)$ 

and  $a\delta d_1(0) + b\delta d_2(0) = 0$ . Hence,  $a\delta d_1 + b\delta d_2$  is the unique solution of (3.60) associated to  $a\delta \ell_1 + b\delta \ell_2$ , that is,  $a\mathcal{S}_3'(\ell;\delta\ell_1) + b\mathcal{S}_3'(\ell;\delta\ell_2) = \mathcal{S}_3'(\ell;a\delta\ell_1 + b\delta\ell_2)$ .

At the end of this chapter we make a few more comments on the *strict complementarity* assumption, including possible alternative approaches, such as regularization, see Remark 3.46 below.

Remark 3.39. We emphasize that the above discussion concerning the linearity of the derivative has the sole purpose of giving an idea under which conditions one should expect an optimality system. The linearity result is rather indirectly used, e.g. by replacing the operator differential equation in (3.56) by the linear equation (3.60), for constructing an 'artificial adjoint operator', see Lemma 3.40 and Remark 3.41 below. By [73, Lemma 3.1.2(b)] we infer from Remark 3.24 and Proposition 3.28 that S is Hadamard directionally differentiable, and hence, Hadamard differentiable, provided that Assumption 3.38 is satisfied. However, the standard approach of deducing that the solution operator of the adjoint equation is  $S'(\ell)^*$  cannot be employed here. That is so because, in view of Proposition 3.28, if existent, the adjoint operator

$$S'(\ell)^*: L^{\varrho}(0,T;V)^* \times L^{\varrho}(0,T;H^1(\Omega))^* \times W^{1,\varrho}(0,T;L^2(\Omega))^* \to L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega))^*$$

with  $\varrho \in [1, \infty)$ , gives in turn one adjoint state instead of three. This problem may be overcome by considering three controls instead of one, in which case the adjoint operator has range in  $L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega))^* \times X^* \times Y^*$ , where X and Y are the spaces for the auxiliary controls. However, in this situation, a system of the type (3.61) cannot be fulfilled by the adjoint states, as long as the first adjoint state does not possess higher regularity. This is due to the mere fact that  $L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega))^*$  cannot be identified with a Bochner-Lebesgue space. Altogether it does not make sense to adjoint  $S'(\ell)$ , but rather to derive, for example via the formal Lagrange method, a candidate for the adjoint system. The latter one is a system with right-hand side  $(\partial_{\mathbf{u}} \mathcal{J}(S(\ell), \ell), \partial_{\varphi} \mathcal{J}(S(\ell), \ell), \partial_{d} \mathcal{J}(S(\ell), \ell))$ , which is uniquely solved by what will turn out to be the adjoint states associated to  $\ell$  in Theorem 3.42 below.

The following result is an essential tool for deriving the main result of this chapter, as it provides the candidates for the adjoint states associated to  $\ell \in \mathfrak{L}$ .

**Lemma 3.40** (Adjoint equation). Let Assumptions 1.17, p. 19, 1.56, p. 46, 3.29 and 3.33 hold. Moreover, let  $\ell \in \mathfrak{L}$  be given and define  $(\boldsymbol{u}, \varphi, d) := \mathcal{S}(\ell)$ . Then there exists a unique  $(\boldsymbol{w}, v, \xi) \in L^2(0, T; V) \times L^2(0, T; H^1(\Omega)) \times W_T^{1,2}(0, T; L^2(\Omega))$  which fulfills f.a.a.  $t \in (0, T)$  the following system of equations

$$-\operatorname{div}\left(g(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{w}(t)) + g'(\varphi(t))\upsilon(t)\mathbb{C}\varepsilon(\boldsymbol{u}(t))\right) = \partial_{\boldsymbol{u}}\mathcal{J}(\cdot)(t) \quad \text{in } V^*, \tag{3.61a}$$

$$-\alpha \Delta v(t) + \beta \left( v(t) - \frac{1}{\delta} \chi_{\Omega_t^+} \xi(t) \right) + g'(\varphi(t)) \mathbb{C} \varepsilon(\boldsymbol{u}(t)) : \varepsilon(\boldsymbol{w}(t))$$

$$+ \frac{1}{2} g''(\varphi(t)) v(t) \mathbb{C} \varepsilon(\boldsymbol{u}(t)) : \varepsilon(\boldsymbol{u}(t)) = \partial_{\varphi} \mathcal{J}(\cdot)(t) \quad in \ H^1(\Omega)^*,$$
(3.61b)

$$-\dot{\xi}(t) = \beta \left( \upsilon(t) - \frac{1}{\delta} \chi_{\Omega_t^+} \xi(t) \right) + \partial_d \mathcal{J}(\cdot)(t) \quad in \ L^2(\Omega),$$

$$\xi(T) = 0,$$
(3.61c)

where  $(\cdot)$  stands for  $(\mathcal{S}(\ell), \ell)$ .

Proof. We begin by noticing that (3.61) has a similar structure to (1.48) on page 29, as well as to its 'linearized' counterpart. Thus, well-known arguments are expected to be encountered throughout this proof. We approach the system (3.61) as in Section 1.1, in the sense that we first solve the elliptic system (3.61a)-(3.61b). Notice that this features the quantities  $\boldsymbol{u}$  and  $\varphi$ , which are not well defined at any  $t \in [0,T]$ , but only a.e. in (0,T) (see Definitions 3.22, 3.21 and 3.10). Hence, its solvability is addressed f.a.a.  $t \in (0,T)$  and for any  $\xi \in L^2(\Omega)$ . Then, with the resulting 'solution operators' at hand, we rewrite the operator differential equation in (3.61) as depending only on  $\xi$ . The latter one is then solved by using Lemma 5.7, which practically yields the unique solvability of (3.61). The time regularity of  $\xi$  ultimately gives the time regularity of  $\boldsymbol{w}$  and  $\boldsymbol{v}$ . For the sake of convenience we will denote the 'solution operators' involved in the elliptic system also by  $\boldsymbol{w}$  and  $\boldsymbol{v}$ . Before we begin to discuss the solvability of (3.61), let us mention here that, since  $\boldsymbol{W}_D^{1,\nu}(\Omega)$ ,  $H^1(\Omega)$  and  $L^2(\Omega)$  are reflexive Banach spaces, the partial derivatives  $\partial_{\boldsymbol{u}} \mathcal{J}(\cdot)$ ,  $\partial_{\varphi} \mathcal{J}(\cdot)$  and  $\partial_{d} \mathcal{J}(\cdot)$  belong to  $L^2(0,T;\boldsymbol{W}^{-1,\nu'}(\Omega))$ ,  $L^2(0,T;H^1(\Omega)^*)$  and  $L^2(0,T;L^2(\Omega))$ , respectively, see e.g. [14, Theorem 7.1.23 (vi)]. Note that, due to  $\nu' > 2$ , we have  $\boldsymbol{W}^{-1,\nu'}(\Omega) \hookrightarrow V^*$ , whence  $\partial_{\boldsymbol{u}} \mathcal{J}(\cdot) \in L^2(0,T;V^*)$ .

(i) Solvability of (3.61a). We search f.a.a.  $t \in (0,T)$  and for any  $v \in H^1(\Omega)$  for  $\boldsymbol{w}$  such that

$$-\operatorname{div}\left(g(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{w})\right) = \partial_{\boldsymbol{u}}\mathcal{J}(\cdot)(t) + \operatorname{div}\left(g'(\varphi(t))\upsilon\mathbb{C}\varepsilon(\boldsymbol{u}(t))\right) \quad \text{in } V^*. \tag{3.62}$$

The first thing to observe is that the right-hand side belongs to  $V^*$ . This follows from Assumption 3.33 and Hölder's inequality with (p-2)/2p+1/p=1/2, combined with Lemma 5.1, Definition 3.1 and the embedding  $H^1(\Omega) \hookrightarrow L^{2p/(p-2)}(\Omega)$ . Recall that the latter one is a consequence of Assumption 1.17.1, p. 19. Moreover, f.a.a.  $t \in (0,T)$  and for any  $v \in H^1(\Omega)$  it holds

$$\|\operatorname{div}\left(g'(\varphi(t))\upsilon\mathbb{C}\varepsilon(\boldsymbol{u}(t))\right)\|_{V^*} \leq C\|\upsilon\|_{2p/(p-2)}\|\mathcal{U}(\ell(t),\varphi(t))\|_{\boldsymbol{W}_{D}^{1,p}(\Omega)} \leq C\|\upsilon\|_{H^{1}(\Omega)}.$$
(3.63)

Note that the last inequality was deduced from (3.5) and that C > 0 depends only on the given data, including the constant M from Remark 3.30. Further, in view of Definition 1.2, Lemma 1.3 on page 12 tells us that (3.62) is uniquely solvable at almost all  $t \in (0,T)$  and for any  $v \in H^1(\Omega)$  with

$$\boldsymbol{w}(t,v) = A_{\varphi(t)}^{-1} \left( \partial_{\boldsymbol{u}} \mathcal{J}(\cdot)(t) + \operatorname{div} \left( g'(\varphi(t)) v \mathbb{C} \varepsilon(\boldsymbol{u}(t)) \right) \in V.$$
 (3.64)

We now apply estimate (1.6) to  $A_{\varphi(t)}$  and in view of (3.63) we get

$$\|\boldsymbol{w}(t,v)\|_{V} \le C(\|\partial_{\boldsymbol{u}}\mathcal{J}(\cdot)(t)\|_{V^*} + \|v\|_{H^1(\Omega)}) \text{ f.a.a. } t \in (0,T), \ \forall v \in H^1(\Omega),$$
 (3.65)

where again C > 0 depends only on the given data.

We now show that the Nemytskii operator associated to  $\boldsymbol{w}$  maps  $L^2(0,T;H^1(\Omega))$  to  $L^2(0,T;V)$ . This result will be useful at the end of the proof, after the time regularity of the solution of the operator differential equation (3.61c) is established. Let  $v \in L^2(0,T;H^1(\Omega))$  be arbitrary, but fixed. In view of Remark 3.34, see also (3.58) and (3.64), it is necessary for the Bochner measurability of  $t \mapsto \boldsymbol{w}(t,v(t)) \in V$  that

$$t \mapsto \partial_{\boldsymbol{u}} \mathcal{J}(\cdot)(t) + \operatorname{div}\left(g'(\varphi(t))v(t)\mathbb{C}\varepsilon(\boldsymbol{u}(t))\right)$$

is Bochner measurable as a function mapping to  $W^{-1,\rho}(\Omega)$ , with some  $\rho > 2$ . Hence, we prove next that there exists  $\rho > 2$  such that at almost all  $t \in (0,T)$  it holds

$$\operatorname{div}\left(g'(\varphi(t))\upsilon(t)\mathbb{C}\varepsilon(\boldsymbol{u}(t))\right)\in\boldsymbol{W}^{-1,\rho}(\Omega).$$

To see this, recall that in view of Assumption 1.17.1, there exists  $\kappa > 2p/(p-2)$  so that the embedding  $H^1(\Omega) \hookrightarrow L^{\kappa}(\Omega)$  holds true. By choosing  $\varrho < \infty$  and  $s \in [2,p)$  large enough such that  $1/\varrho + 1/\kappa + 1/s < (p-2)/2p + 1/p = 1/2$ , we can then define the index  $\varrho > 2$  via

$$1/\rho := 1/\varrho + 1/\kappa + 1/s < 1/2.$$

Hölder's inequality with  $1/\varrho + 1/\kappa + 1/s = 1/\rho$  now gives the desired regularity and from the boundedness of the operator div :  $L^{\rho}(\Omega; \mathbb{R}^{N \times N}) \to \mathbf{W}^{-1,\rho}(\Omega)$  we deduce

$$\|\operatorname{div}\left(g'(\varphi(t))v(t)\mathbb{C}\varepsilon(\boldsymbol{u}(t))\right)\|_{\boldsymbol{W}^{-1,\rho}(\Omega)} \leq \|g'(\varphi(t))\|_{\varrho}\|v(t)\|_{\kappa}\|\mathbb{C}\varepsilon(\boldsymbol{u}(t))\|_{s}$$
(3.66)

for almost all  $t \in (0, T)$ . Further, we know from Lemmata 5.2 and 5.11 that the mapping  $t \mapsto g'(\varphi(t)) \in L^{\varrho}(\Omega)$  is Bochner measurable, while  $t \mapsto \mathbb{C}\varepsilon(\boldsymbol{u}(t)) \in L^{s}(\Omega; \mathbb{R}^{N \times N})$  is Bochner measurable as well, as a result of Definition 3.22. Thus, on account of the definition of the Bochner measurability and (3.66), we obtain the Bochner measurability of

$$t \mapsto \operatorname{div} \left( g'(\varphi(t)) v(t) \mathbb{C} \varepsilon(\boldsymbol{u}(t)) \right) \in \boldsymbol{W}^{-1,\rho}(\Omega),$$

where we also employed that  $\upsilon:[0,T]\to L^\kappa(\Omega)$  is Bochner measurable, by assumption. Besides,  $t\mapsto \partial_{\boldsymbol{u}}\mathcal{J}(\cdot)(t)\in \boldsymbol{W}^{-1,\nu'}(\Omega)$  is Bochner measurable as well, in view of Assumption 3.33, whence we conclude that

$$t \mapsto \partial_{\boldsymbol{u}} \mathcal{J}(\cdot)(t) + \operatorname{div} \left( g'(\varphi(t)) v(t) \mathbb{C} \varepsilon(\boldsymbol{u}(t)) \right) \in \boldsymbol{W}^{-1,\omega}(\Omega)$$

is Bochner measurable, where  $\omega := \min\{\rho, \nu'\} > 2$ . From (3.58) and (3.64) we now deduce the Bochner measurability of  $t \mapsto \boldsymbol{w}(t, v(t)) \in V$ . Assumption 3.33 together with (3.65) ultimately yields

$$\boldsymbol{w}(\cdot, v(\cdot)) \in L^2(0, T; V) \text{ for } v \in L^2(0, T; H^1(\Omega)). \tag{3.67}$$

(ii) Solvability of (3.61b). For almost all  $t \in (0,T)$  and for any  $\xi \in L^2(\Omega)$ , we now search for  $v \in H^1(\Omega)$  that solves

$$Bv + g'(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{u}(t)) : \varepsilon(\boldsymbol{w}(t,v)) + \frac{1}{2}g''(\varphi(t))v\mathbb{C}\varepsilon(\boldsymbol{u}(t)) : \varepsilon(\boldsymbol{u}(t))$$

$$= \partial_{\varphi}\mathcal{J}(\cdot)(t) + \frac{\beta}{\delta}\chi_{\Omega_{t}^{+}}\xi \quad \text{in } H^{1}(\Omega)^{*}.$$
(3.68)

Recall that the linearity B is given by (1.22) on page 18. Let us begin by rewriting (3.68) in a compact form. To this end, notice that in view of (3.64) and (3.21) it holds

$$\boldsymbol{w}(t,\upsilon) = A_{\varphi(t)}^{-1} \partial_{\boldsymbol{u}} \mathcal{J}(\cdot)(t) + \partial_{\varphi} \mathcal{U}(\ell(t),\varphi(t))(\upsilon) \quad \text{f.a.a. } t \in (0,T), \ \forall \, \upsilon \in H^1(\Omega),$$

where we relied on the linearity of  $A_{\varphi(t)}^{-1}$ . By taking a look at (3.30) one sees now that (3.68) is equivalent to

$$Bv + \partial_{\varphi} F(\ell(t), \varphi(t))v = \iota(t, \xi) \text{ in } H^{1}(\Omega)^{*},$$

where the mapping  $\iota:(0,T)\times L^2(\Omega)\to H^1(\Omega)^*$  is given by

$$\iota(t,\xi) := \partial_{\varphi} \mathcal{J}(\cdot)(t) + \frac{\beta}{\delta} \chi_{\Omega_{t}^{+}} \xi - g'(\varphi(t)) \mathbb{C}\varepsilon(\boldsymbol{u}(t)) : \varepsilon(A_{\varphi(t)}^{-1}(\partial_{\boldsymbol{u}} \mathcal{J}(\cdot)(t))). \tag{3.69}$$

In order to see that  $\iota(t,\xi)$  is well defined f.a.a.  $t \in (0,T)$  and all  $\xi \in L^2(\Omega)$  we argue as follows. From part (i) we recall that  $A_{\varphi(t)}^{-1}(\partial_{\boldsymbol{u}}\mathcal{J}(\cdot)(t)) \in V$ , since  $A_{\varphi(t)}^{-1} \in \mathcal{L}(V^*,V)$ , f.a.a.  $t \in (0,T)$ . Then, on account of Lemma 5.1, Definition 3.1 and the embedding  $H^1(\Omega) \hookrightarrow L^{2p/(p-2)}(\Omega)$ , we can apply Hölder's inequality with 1/p+1/2+(p-2)/2p=1 for the last term on the right-hand side in (3.69), which gives in turn the  $H^1(\Omega)^*$ -regularity thereof. Additionally, by employing again (3.5) and (1.6), we deduce here

$$\|g'(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{u}(t)) : \varepsilon(A_{\varphi(t)}^{-1}(\partial_{\boldsymbol{u}}\mathcal{J}(\cdot)(t)))\|_{H^{1}(\Omega)^{*}} \leq C\|\partial_{\boldsymbol{u}}\mathcal{J}(\cdot)(t)\|_{V^{*}} \text{ f.a.a. } t \in (0,T),$$
(3.70)

which is needed only later in the proof. Note that C>0 depends only on the given data, including the constant M from Remark 3.30. As  $\chi_{\Omega_t^+} \in L^{\infty}(\Omega)$  f.a.a.  $t \in (0,T)$ , we finally conclude with Assumption 3.33 and  $L^2(\Omega) \hookrightarrow H^1(\Omega)^*$  that  $\iota(t,\xi)$  is well defined f.a.a.  $t \in (0,T)$  and all  $\xi \in L^2(\Omega)$ . On the other side,  $B + \partial_{\varphi} F(\ell(t), \varphi(t)) \in \mathcal{L}(H^1(\Omega), H^1(\Omega)^*)$  is f.a.a.  $t \in (0,T)$  continuously invertible with

$$\|(B + \partial_{\varphi} F(\ell(t), \varphi(t)))^{-1}\|_{\mathcal{L}(H^{1}(\Omega)^{*}, H^{1}(\Omega))} \le 2/\alpha \quad \text{f.a.a. } t \in (0, T).$$
 (3.71)

We refer here to page 113, where the unique solvability of (3.29) is shown. Note that the estimate (3.71) follows from (3.33). Hence, (3.68) is uniquely solvable f.a.a.  $t \in (0, T)$  and for any  $\xi \in L^2(\Omega)$  with

$$v(t,\xi) = (B + \partial_{\varphi} F(\ell(t), \varphi(t)))^{-1} \iota(t,\xi) \in H^{1}(\Omega).$$
(3.72)

Moreover, due to (3.72), (3.71) and (3.69) we have f.a.a.  $t \in (0,T)$  and for all  $\xi_1, \xi_2 \in L^2(\Omega)$  the estimate

$$\|v(t,\xi_1) - v(t,\xi_2)\|_{H^1(\Omega)} \le \frac{2}{\alpha} \|\iota(t,\xi_1) - \iota(t,\xi_2)\|_{H^1(\Omega)^*} \le \frac{2\beta}{\alpha\delta} \|\xi_1 - \xi_2\|_2.$$
 (3.73)

In preparation for the next part of the proof, we prove in the following that v belongs to  $L^2(0,T;H^1(\Omega))$  if  $\xi \in L^2(0,T;L^2(\Omega))$ . Note that the above time regularity of v does not improve, even when  $\xi$  turns out to have higher regularity than  $L^2(0,T;L^2(\Omega))$ . This can be seen by taking a look at (3.72), (3.71) and (3.69), while keeping in mind that  $\partial_{\varphi} \mathcal{J}(\cdot) \in L^2(0,T;H^1(\Omega)^*)$ . We do not go here into any more details and refer to Remark 3.16, where a similar observation was made in the context of establishing the

time regularity of the solutions of the minimization problem in  $(P_{\ell})$ . Firstly, we want to prove that

$$(B + \partial_{\varphi} F(\ell(t), \varphi(t)))^{-1} \in L^{\infty}(0, T; \mathcal{L}(H^{1}(\Omega)^{*}, H^{1}(\Omega))). \tag{3.74}$$

To this end, recall that  $\partial_{\varphi}F: \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega) \to \mathcal{L}(H^1(\Omega), H^1(\Omega)^*)$  is continuous. This is stated at the end of part (I) of the proof of Lemma 3.15. Since  $t \mapsto (\ell(t), \varphi(t)) \in \mathbf{W}^{-1,p}(\Omega) \times H^1(\Omega)$  is Bochner measurable, Lemma 5.11 yields the Bochner measurability of  $t \mapsto B + \partial_{\varphi}F(\ell(t), \varphi(t)) \in \mathcal{L}(H^1(\Omega), H^1(\Omega)^*)$ . Cf. e.g. [74, Ch. III.8], the inversion  $A \mapsto A^{-1}$  is continuous from the set of isomorphisms in  $\mathcal{L}(H^1(\Omega), H^1(\Omega)^*)$  to  $\mathcal{L}(H^1(\Omega)^*, H^1(\Omega))$ . This together with the definition of Bochner measurability then gives the measurability of the mapping in (3.74). The desired Bochner integrability thereof follows now from estimate (3.71). Secondly, it holds

$$\iota(\cdot, \xi(\cdot)) \in L^2(0, T; H^1(\Omega)^*) \quad \text{for } \xi \in L^2(0, T; L^2(\Omega)).$$
 (3.75)

To see this, we address the time regularity of each term in (3.69) separately. We already know that  $\partial_{\varphi} \mathcal{J}(\cdot) \in L^2(0,T;H^1(\Omega)^*)$ , in view of Assumption 3.33. Further, Definition 3.36 allows us to write

$$\chi_{\Omega_{+}^{+}}(x)\xi(t)(x) = (\chi_{Q_{+}}\xi)(t,x)$$
 for almost all  $(t,x) \in (0,T) \times \Omega$ ,

where  $Q_+ := \{(t,x) \in (0,T) \times \Omega : -\beta(d(t,x) - \varphi(t,x)) - r > 0\}$  and  $\xi \in L^2(0,T;L^2(\Omega))$ . Note that  $Q_+$  depends only on  $\ell$  and it is well defined up to a set of measure zero in  $(0,T) \times \Omega$ . Since  $\chi_{Q_+} \in L^{\infty}((0,T) \times \Omega)$  and  $L^2((0,T) \times \Omega) = L^2(0,T;L^2(\Omega))$ , see e.g. [14, Theorem 7.1.24], we have  $\chi_{Q_+} \xi \in L^2((0,T) \times \Omega)$ , whence

$$t \mapsto \chi_{\Omega_t^+} \xi(t) \in L^2(0, T; L^2(\Omega)) \text{ for } \xi \in L^2(0, T; L^2(\Omega)).$$
 (3.76)

Furthermore, recall that in view of Assumption 1.17.1, there exists  $\kappa > 2p/(p-2)$  so that  $H^1(\Omega) \hookrightarrow L^{\kappa}(\Omega)$ . In light of  $1/2 + 1/\kappa < 1 - 1/p$ , we find  $s \in [2, p)$  and  $\varrho < \infty$  large enough, so that

$$1/\rho + 1/s + 1/2 + 1/\kappa = 1. \tag{3.77}$$

The mapping  $t \mapsto g'(\varphi(t)) \in L^{\varrho}(\Omega)$  is Bochner measurable thanks to Lemmata 5.2 and 5.11, while  $t \mapsto \mathbb{C}\varepsilon(\boldsymbol{u}(t)) \in L^s(\Omega; \mathbb{R}^{N \times N})$  is Bochner measurable as well, as a result of Definition 3.22. Arguing as at the end of part (i), it follows further that  $t \mapsto \varepsilon(A_{\varphi(t)}^{-1}(\partial_{\boldsymbol{u}}\mathcal{J}(\cdot)(t))) \in L^2(\Omega; \mathbb{R}^{N \times N})$  is also Bochner measurable. Altogether we deduce from the above that

$$t\mapsto g'(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{u}(t)):\varepsilon\big(A_{\varphi(t)}^{-1}\big(\partial_{\boldsymbol{u}}\mathcal{J}(\cdot)(t)\big)\big)\in H^1(\Omega)^*$$

is Bochner measurable, on account of Hölder's inequality with (3.77) and the definition of Bochner measurability. From (3.70) we then obtain

$$t\mapsto g'(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{u}(t)):\varepsilon\big(A_{\varphi(t)}^{-1}\big(\partial_{\boldsymbol{u}}\mathcal{J}(\cdot)(t)\big)\big)\in L^2(0,T;H^1(\Omega)^*).$$

Therewith we addressed all the terms in (3.69), and thus, we can now infer (3.75). By means of (3.74) and (3.75) one can finally establish as in [81, Lemma 3.1.19] that

$$v(\cdot, \xi(\cdot)) \in L^2(0, T; H^1(\Omega)) \text{ for } \xi \in L^2(0, T; L^2(\Omega)),$$
 (3.78)

where we relied on (3.72).

(iii) Solvability of (3.61c). By means of the 'solution operator'  $(t, \xi) \mapsto v(t, \xi)$  of the elliptic system, (3.61) reduces to

$$-\dot{\xi}(t) = \beta \left( \upsilon(t, \xi(t)) - \frac{1}{\delta} \chi_{\Omega_t^+} \xi(t) \right) + \partial_d \mathcal{J}(\cdot)(t) \quad \text{f.a.a. } t \in (0, T),$$

$$\xi(T) = 0.$$
(3.79)

We intend to solve (3.79) by employing again Lemma 5.7. For this purpose, note first that via the (at this point informal) transformation

$$\widetilde{\xi}(\cdot) = \xi(T - \cdot),$$

(3.79) is equivalent to

$$\dot{\widetilde{\xi}}(t) = f(t, \widetilde{\xi}(t)) \quad \text{f.a.a.} \quad t \in (0, T), \quad \widetilde{\xi}(0) = 0, \tag{3.80}$$

where  $f:(0,T)\times L^2(\Omega)\to L^2(\Omega)$  is given by

$$f(t,\widetilde{\xi}) = \beta \left( \upsilon(T - t, \widetilde{\xi}) - \frac{1}{\delta} \chi_{\Omega_{T-t}^{+}} \widetilde{\xi} \right) + \partial_{d} \mathcal{J}(\cdot)(T - t). \tag{3.81}$$

Note that for any  $\widetilde{\xi} \in L^2(\Omega)$ , the value  $f(t,\widetilde{\xi}) \in L^2(\Omega)$  is well defined f.a.a.  $t \in (0,T)$ , as a consequence of (3.72), Definition 3.36 and Assumption 3.33. Let now  $\widetilde{\xi} \in L^2(0,T;L^2(\Omega))$  be arbitrary, but fixed. It is easy to see that this implies  $t \mapsto \widetilde{\xi}(T-t) \in L^2(0,T;L^2(\Omega))$ . According to (3.78), we then have  $v(\cdot,\widetilde{\xi}(T-\cdot)) \in L^2(0,T;H^1(\Omega))$ , which in particular means that

$$t \mapsto \upsilon(T - t, \widetilde{\xi}(t)) \in L^2(0, T; H^1(\Omega)). \tag{3.82}$$

With (3.76) we establish that  $t \mapsto \chi_{\Omega_t^+} \widetilde{\xi}(T-t) \in L^2(0,T;L^2(\Omega))$ , and consequently

$$t \mapsto \chi_{\Omega_{T-t}^{+}} \widetilde{\xi}(t) \in L^{2}(0, T; L^{2}(\Omega)). \tag{3.83}$$

From Assumption 3.33 we infer  $t \mapsto \partial_d \mathcal{J}(\cdot)(T-t) \in L^2(0,T;L^2(\Omega))$ , which combined with (3.82) and (3.83) now yields

$$f(\cdot, \widetilde{\xi}(\cdot)) \in L^2(0, T; L^2(\Omega))$$
 for  $\widetilde{\xi} \in L^2(0, T; L^2(\Omega))$ ,

in view of (3.81). Therewith the second assumption on f in Lemma 5.7 is satisfied. As (3.81) and (3.73) imply

$$\begin{split} \|f(t,\widetilde{\xi}_1) - f(t,\widetilde{\xi}_2)\|_2 &\leq \beta \|\upsilon(T - t,\widetilde{\xi}_1) - \upsilon(T - t,\widetilde{\xi}_2)\|_2 + \beta/\delta \|\widetilde{\xi}_1 - \widetilde{\xi}_2\|_2 \\ &\leq L \|\widetilde{\xi}_1 - \widetilde{\xi}_2\|_2 \quad \forall \, \widetilde{\xi}_1,\widetilde{\xi}_2 \in L^2(\Omega), \quad \text{f.a.a.} \ t \in (0,T), \end{split}$$

the third assumption on f in Lemma 5.7 is satisfied as well. Note that L > 0 depends only on  $\alpha, \beta$  and  $\delta$ . Lemma 5.7 now tells us that (3.80) is uniquely solvable with  $\widetilde{\xi} \in W_0^{1,2}(0,T;L^2(\Omega))$ , whence the unique solvability of (3.79) with

$$t \mapsto \xi(t) = \widetilde{\xi}(T - t) \in W_T^{1,2}(0, T; L^2(\Omega)).$$

Note that the time regularity of  $\xi$  is due to  $t \mapsto \widetilde{\xi}(T-t) \in L^2(0,T;L^2(\Omega))$  and  $t \mapsto \dot{\xi}(t) = -\dot{\widetilde{\xi}}(T-t) \in L^2(0,T;L^2(\Omega))$ . This gives the unique solvability of (3.61) with the desired regularity of solutions, in view of (3.67) and (3.78). The proof is now complete.

Remark 3.41. Clearly, given  $\ell \in \mathfrak{L}$ , the system (3.61) admits a unique solution (with the exact same regularity) for an arbitrary right-hand side  $h \in L^2(0,T;\mathbf{W}^{-1,\nu'}(\Omega)) \times L^2(0,T;H^1(\Omega)^*) \times L^2(0,T;L^2(\Omega))$  (instead of  $\partial_{(\mathbf{u},\varphi,d)}\mathcal{J}(\mathcal{S}(\ell),\ell)$ ), where  $\nu \in (1,2)$  is given. The thereby induced solution operator, which we here call  $\Upsilon$ , can be interpreted as an 'artificial adjoint operator' of  $\mathcal{S}'(\ell)$ , provided that Assumption 3.38 holds true for  $\ell$ . This can be seen by linearizing the state equation (3.84) below at  $\ell$  in some arbitrary direction  $\delta \ell \in L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega))$  and by arguing as in the proof of Theorem 3.42 below, where we replace the partial derivatives of  $\mathcal{J}$  in (3.85) by  $\ell$  as above. Then, one has instead of (3.93) below, the equality

$$\int_0^T \sum_{i=1}^3 \langle h_i(t), \mathcal{S}_i'(\ell)(\delta\ell)(t) \rangle \ dt = \int_0^T \langle \delta\ell(t), \Upsilon_1(h)(t) \rangle_V \ dt,$$

i.e.,

$$\langle h, \mathcal{S}'(\ell)(\delta\ell) \rangle_{L^2(0,T;\boldsymbol{W}_D^{1,\nu}(\Omega) \times H^1(\Omega) \times L^2(\Omega))} = \langle \Upsilon_1 h, \delta\ell \rangle_{L^2(0,T;\boldsymbol{W}^{-1,p}(\Omega))}$$

for any  $h \in L^2(0,T; \mathbf{W}^{-1,\nu'}(\Omega) \times H^1(\Omega)^* \times L^2(\Omega))$  and any  $\delta \ell \in L^{\infty}(0,T; \mathbf{W}^{-1,p}(\Omega))$ . Here  $\Upsilon_1$  stands for the first component of the operator  $\Upsilon$ .

We point out that  $\Upsilon_1$  is the classical adjoint operator of  $\mathcal{S}'(\ell): L^{\infty}(0,T; \mathbf{W}^{-1,p}(\Omega)) \to L^2(0,T; \mathbf{W}_D^{1,\nu}(\Omega) \times H^1(\Omega) \times L^2(\Omega))$ , as a result of the above identity and as it maps as follows

$$L^2(0,T;\boldsymbol{W}^{-1,\nu'}(\Omega)\times H^1(\Omega)^*\times L^2(\Omega))\stackrel{\Upsilon_1}{\longmapsto} L^2(0,T;V)\subset L^\infty(0,T;\boldsymbol{W}^{-1,p}(\Omega))^*,$$

in view of Lemma 3.40 and  $L^{\infty}(0,T; \mathbf{W}^{-1,p}(\Omega)) \hookrightarrow L^2(0,T;V^*)$ .

The main result of this chapter is covered by the following

**Theorem 3.42** (Optimality system). Let Assumptions 1.17, p. 19, 1.56, p. 46, 3.29 and 3.33 hold. Moreover, let  $\bar{\ell}$  be a local solution of  $(P_{min})$  with associated states

$$(\bar{\boldsymbol{u}},\bar{\varphi},\bar{d}) = \mathcal{S}(\bar{\ell}) \in L^{\infty}(0,T;\boldsymbol{W}_{D}^{1,s}(\Omega)) \times L^{\infty}(0,T;H^{1}(\Omega)) \times W^{1,\infty}(0,T;L^{2}(\Omega)),$$

where  $s \in [2, p)$ , and suppose that Assumption 3.38 is fulfilled for  $\bar{\ell}$ . Then there exist unique adjoint states

$$(\boldsymbol{w}, v, \xi) \in L^2(0, T; V) \times L^2(0, T; H^1(\Omega)) \times W_T^{1,2}(0, T; L^2(\Omega))$$

so that the following optimality system is f.a.a.  $t \in (0,T)$  satisfied:

$$-\operatorname{div} g(\bar{\varphi}(t))\mathbb{C}\varepsilon(\bar{\boldsymbol{u}}(t)) = \bar{\ell}(t) \quad \text{in } \boldsymbol{W}^{-1,p}(\Omega), \qquad (3.84a)$$

$$-\alpha \Delta \bar{\varphi}(t) + \beta \bar{\varphi}(t) + \frac{1}{2} g'(\bar{\varphi}(t)) \mathbb{C} \varepsilon(\bar{\boldsymbol{u}}(t)) : \varepsilon(\bar{\boldsymbol{u}}(t)) = \beta \bar{d}(t) \quad in \ H^{1}(\Omega)^{*}, \tag{3.84b}$$

$$\dot{\bar{d}}(t) = \frac{1}{\delta} \max(-\beta(\bar{d}(t) - \bar{\varphi}(t)) - r), \quad \bar{d}(0) = d_0, \tag{3.84c}$$

$$-\operatorname{div}\left(g(\bar{\varphi}(t))\mathbb{C}\varepsilon(\boldsymbol{w}(t)) + g'(\bar{\varphi}(t))\upsilon(t)\mathbb{C}\varepsilon(\bar{\boldsymbol{u}}(t))\right) = \partial_{\boldsymbol{u}}\mathcal{J}(\cdot)(t) \quad \text{in } V^*, \tag{3.85a}$$

$$-\alpha \Delta v(t) + \beta \left( v(t) - \frac{1}{\delta} \chi_{\Omega_t^+} \xi(t) \right) + g'(\bar{\varphi}(t)) \mathbb{C} \varepsilon(\bar{\boldsymbol{u}}(t)) : \varepsilon(\boldsymbol{w}(t))$$

$$+ \frac{1}{2} g''(\bar{\varphi}(t)) v(t) \mathbb{C} \varepsilon(\bar{\boldsymbol{u}}(t)) : \varepsilon(\bar{\boldsymbol{u}}(t)) = \partial_{\varphi} \mathcal{J}(\cdot)(t) \quad in \ H^1(\Omega)^*,$$

$$(3.85b)$$

$$-\dot{\xi}(t) = \beta \left( \upsilon(t) - \frac{1}{\delta} \chi_{\Omega_t^+} \xi(t) \right) + \partial_d \mathcal{J}(\cdot)(t) \quad in \ L^2(\Omega),$$

$$\xi(T) = 0,$$
(3.85c)

$$\langle \boldsymbol{w} + \partial_{\ell} \mathcal{J}(\cdot), \delta \ell - \bar{\ell} \rangle_{L^{2}(0,T;\boldsymbol{W}^{-1,p}(\Omega))} \ge 0 \quad \forall \, \delta \ell \in \mathfrak{L},$$
 (3.86)

where  $(\cdot)$  denotes  $(S(\bar{\ell}), \bar{\ell})$ .

*Proof.* The state equation (3.84) is automatically fulfilled in view of Definition 3.22, combined with Definitions 3.1 and 3.10, while from Lemma 3.40 we know that (3.85) admits a unique solution  $(\boldsymbol{w}, v, \xi)$  with the desired regularity. Hence, it remains to prove the gradient inequality (3.86), for which we employ standard arguments. Roughly speaking, we test the linearized counterpart of (3.84) with the adjoint states and show that the (integrated over time) sum of the resulting equations is the (integrated over time) sum of the equations in (3.85) tested with the directional derivatives of  $\mathcal{S}$ . Using this in (VI) will ultimately give the claim. For a better overview, we split the rest of the proof in two parts: in the first one we derive a linearization for the state equation, while in the second one we test as depicted above and finalize the proof.

(i) Let  $\delta\ell \in L^{\infty}(0,T; \boldsymbol{W}^{-1,p}(\Omega))$  be arbitrary, but fixed. In view of Assumption 3.38, (3.84) can be linearized, such that the resulting system is uniquely solved by  $\mathcal{S}'(\bar{\ell}; \delta\ell - \bar{\ell})$ . For the sake of simplicity we denote the latter one by  $(\delta \boldsymbol{u}, \delta \varphi, \delta d)$  in what follows. To be more precise, Proposition 3.28 and Assumption 3.38 ensure that  $(\delta \boldsymbol{u}, \delta \varphi, \delta d)$  satisfies f.a.a.  $t \in (0,T)$  the system

$$\delta u(t) = \mathcal{U}'(\bar{\ell}(t), \bar{\varphi}(t))(\delta \ell(t) - \bar{\ell}(t), \delta \varphi(t)), \tag{3.87a}$$

$$\delta\varphi(t) = \Phi'(\bar{\ell}(t), \bar{d}(t))(\delta\ell(t) - \bar{\ell}(t), \delta d(t)), \tag{3.87b}$$

$$\dot{\delta d}(t) = -\frac{\beta}{\delta} \chi_{\Omega_t^+}(\delta d(t) - \delta \varphi(t)), \quad \delta d(0) = 0.$$
 (3.87c)

From the differentiability results in Section 3.1, we can deduce the equations which characterize  $\boldsymbol{\delta u}(t)$  and  $\delta \varphi(t)$  f.a.a.  $t \in (0,T)$ . Firstly, from (3.20) and (3.21) we infer that

$$\boldsymbol{\delta u}(t) = A_{\bar{\varphi}(t)}^{-1} \Big( \delta \ell(t) - \bar{\ell}(t) + \operatorname{div} \big( g'(\bar{\varphi}(t)) \delta \varphi(t) \mathbb{C} \varepsilon \big( \mathcal{U}(\bar{\ell}(t), \bar{\varphi}(t)) \big) \big) \Big) \quad \text{f.a.a. } t \in (0, T),$$

on account of (3.87a). By Definition 1.2 on page 12 we then obtain that  $\boldsymbol{\delta u}(t)$  is the unique solution of

$$-\operatorname{div}\left(g(\bar{\varphi}(t))\mathbb{C}\varepsilon(\boldsymbol{\delta}\boldsymbol{u}(t))\right) = \delta\ell(t) - \bar{\ell}(t) + \operatorname{div}\left(g'(\bar{\varphi}(t))\delta\varphi(t)\mathbb{C}\varepsilon(\bar{\boldsymbol{u}}(t))\right) \quad \text{in } V^* \quad (3.88)$$

f.a.a.  $t \in (0,T)$ . Secondly, using (3.87b) in (3.24) yields that  $\delta \varphi(t)$  is the unique solution of

$$-\alpha \Delta \delta \varphi(t) + \beta \delta \varphi(t) + \frac{1}{2} g''(\bar{\varphi}(t)) \delta \varphi(t) \mathbb{C} \varepsilon(\bar{\boldsymbol{u}}(t)) : \varepsilon(\bar{\boldsymbol{u}}(t))$$

$$= \beta \delta d(t) - g'(\bar{\varphi}(t)) \mathbb{C} \varepsilon(\bar{\boldsymbol{u}}(t)) : \varepsilon(\boldsymbol{\delta} \boldsymbol{u}(t)) \quad \text{in } H^{1}(\Omega)^{*}$$
(3.89)

f.a.a.  $t \in (0, T)$ .

To summarize, the linearized counterpart of (3.84) (at  $\bar{\ell}$  in direction  $\delta\ell-\bar{\ell}$ ) consists of (3.88), (3.89) and (3.87c). Let us recall here that  $(\delta \boldsymbol{u}, \delta \varphi, \delta d)$  belongs to  $L^{\infty}(0, T; V) \times L^{\infty}(0, T; H^{1}(\Omega)) \times W_{0}^{1,\infty}(0, T; L^{2}(\Omega))$ , cf. Proposition 3.28.

(ii) We test (3.88), (3.89) and (3.87c) with  $\boldsymbol{w}(t) \in V$ ,  $v(t) \in H^1(\Omega)$  and  $\xi(t) \in L^2(\Omega)$ , respectively, at almost all  $t \in (0,T)$ . For the sake of convenience, we bring all the resulting terms containing  $\boldsymbol{\delta u}$  and  $\delta \varphi$  together, by abbreviating

$$\iota(t) := -\langle \operatorname{div} \left( g(\bar{\varphi}(t)) \mathbb{C} \varepsilon(\boldsymbol{\delta u}(t)) + g'(\bar{\varphi}(t)) \delta \varphi(t) \mathbb{C} \varepsilon(\bar{\boldsymbol{u}}(t)) \right), \boldsymbol{w}(t) \rangle_{V}$$

$$+ \langle -\alpha \Delta \delta \varphi(t) + \beta \delta \varphi(t) + \frac{1}{2} g''(\bar{\varphi}(t)) \delta \varphi(t) \mathbb{C} \varepsilon(\bar{\boldsymbol{u}}(t)) : \varepsilon(\bar{\boldsymbol{u}}(t)), \upsilon(t) \rangle_{H^{1}(\Omega)}$$

$$+ \langle g'(\bar{\varphi}(t)) \mathbb{C} \varepsilon(\bar{\boldsymbol{u}}(t)) : \varepsilon(\boldsymbol{\delta u}(t)), \upsilon(t) \rangle_{H^{1}(\Omega)} - \frac{\beta}{\delta} (\chi_{\Omega_{t}^{+}} \delta \varphi(t), \xi(t))_{2} \text{ f.a.a. } t \in (0, T).$$

Thereby, the sum of the above tested equations reads f.a.a.  $t \in (0,T)$  as follows

$$\iota(t) + (\dot{\delta}d(t), \xi(t))_2 = \beta(\delta d(t), \upsilon(t))_2 - \frac{\beta}{\delta} \chi_{\Omega_t^+} \delta d(t), \xi(t))_2 + \langle \delta \ell(t) - \bar{\ell}(t), \boldsymbol{w}(t) \rangle_V.$$
(3.90)

Furthermore, as a result of (3.85a) and (3.85b), combined with Definition 3.36, we observe that

$$\iota(t) = \langle \partial_{\boldsymbol{u}} \mathcal{J}(\cdot)(t), \boldsymbol{\delta u}(t) \rangle_{V} + \langle \partial_{\varphi} \mathcal{J}(\cdot)(t), \delta \varphi(t) \rangle_{H^{1}(\Omega)} \quad \text{f.a.a. } t \in (0, T),$$
 (3.91)

while testing (3.85c) with  $\delta d(t)$  gives in turn

$$(\partial_d \mathcal{J}(\cdot)(t), \delta d(t))_2 = -(\dot{\xi}(t), \delta d(t))_2 - \beta(\upsilon(t), \delta d(t))_2 + \frac{\beta}{\delta} (\chi_{\Omega_t^+} \xi(t), \delta d(t))_2 \quad \text{f.a.a.} \quad t \in (0, T).$$

Assumption 3.33 and the regularity of  $\delta d$ , see Proposition 3.28, imply further that  $t \mapsto (\partial_d \mathcal{J}(\cdot)(t), \delta d(t))_2 \in L^2(0,T)$ . Hence, we can integrate the above identity over time,

which results in

$$\begin{split} \int_{0}^{T} (\partial_{d} \mathcal{J}(\cdot)(t), \delta d(t))_{2} \, dt &= -\int_{0}^{T} (\dot{\xi}(t), \delta d(t))_{2} \, dt - \int_{0}^{T} \beta(\upsilon(t), \delta d(t))_{2} - \frac{\beta}{\delta} (\chi_{\Omega_{t}^{+}} \xi(t), \delta d(t))_{2} \, dt \\ &= -(\xi(T), \delta d(T))_{2} + (\xi(0), \delta d(0))_{2} + \int_{0}^{T} (\dot{\delta} d(t), \xi(t))_{2} \\ &- \int_{0}^{T} \beta(\upsilon(t), \delta d(t))_{2} - \frac{\beta}{\delta} (\chi_{\Omega_{t}^{+}} \xi(t), \delta d(t))_{2} \, dt \\ &= \int_{0}^{T} (\dot{\delta} d(t), \xi(t))_{2} - \int_{0}^{T} \beta(\upsilon(t), \delta d(t))_{2} - \frac{\beta}{\delta} (\chi_{\Omega_{t}^{+}} \xi(t), \delta d(t))_{2} \, dt, \\ &(3.92) \end{split}$$

where we employed  $\xi \in W_T^{1,2}(0,T;L^2(\Omega))$ ,  $\delta d \in W_0^{1,\infty}(0,T;L^2(\Omega))$  and integrated by parts cf. [81, Lemma 3.1.43]. Moreover, note that the terms on the right-hand side in (3.91) are integrable over time, by Assumption 3.33 and Proposition 3.28. Inserting (3.91) and (3.92) in (3.90) leads to

$$\int_{0}^{T} \langle \partial_{\boldsymbol{u}} \mathcal{J}(\cdot)(t), \boldsymbol{\delta u}(t) \rangle_{V} + \langle \partial_{\varphi} \mathcal{J}(\cdot)(t), \delta \varphi(t) \rangle_{H^{1}(\Omega)} + (\partial_{d} \mathcal{J}(\cdot)(t), \delta d(t))_{2} dt$$

$$= \int_{0}^{T} \langle \delta \ell(t) - \bar{\ell}(t), \boldsymbol{w}(t) \rangle_{V} dt.$$
(3.93)

On the other side, by Assumption 3.33, we are allowed to write

$$\partial_{(\boldsymbol{u},\varphi,d)} \mathcal{J}(\mathcal{S}(\bar{\ell}),\bar{\ell}) \left( \mathcal{S}'(\bar{\ell};\delta\ell-\bar{\ell}) \right) 
= \langle \partial_{\boldsymbol{u}} \mathcal{J}(\cdot), \boldsymbol{\delta} \boldsymbol{u} \rangle_{L^{2}(0,T;V)} + \langle \partial_{\varphi} \mathcal{J}(\cdot), \delta \varphi \rangle_{L^{2}(0,T;H^{1}(\Omega))} + \langle \partial_{d} \mathcal{J}(\cdot), \delta d \rangle_{L^{2}(0,T;L^{2}(\Omega))} 
= \int_{0}^{T} \langle \partial_{\boldsymbol{u}} \mathcal{J}(\cdot)(t), \boldsymbol{\delta} \boldsymbol{u}(t) \rangle_{V} + \langle \partial_{\varphi} \mathcal{J}(\cdot)(t), \delta \varphi(t) \rangle_{H^{1}(\Omega)} + (\partial_{d} \mathcal{J}(\cdot)(t), \delta d(t))_{2} dt,$$
(3.94)

due to the reflexivity of V,  $H^1(\Omega)$  and  $L^2(\Omega)$ , see e.g. [14, Theorem 7.1.23(vi)]. As  $\bar{\ell}$  is a local minimizer of  $(P_{min})$ , it satisfies the variational inequality in Lemma 3.35, which in light of (3.93), (3.94) and Assumption 3.33 now reads

$$\int_{0}^{T} \langle \delta \ell(t) - \bar{\ell}(t), \boldsymbol{w}(t) \rangle_{V} dt + \langle \partial_{\ell} \mathcal{J}(\mathcal{S}(\bar{\ell}), \bar{\ell}), \delta \ell - \bar{\ell} \rangle_{L^{2}(0,T;\boldsymbol{W}^{-1,p}(\Omega))} 
= \langle \boldsymbol{w}, \delta \ell - \bar{\ell} \rangle_{L^{2}(0,T;\boldsymbol{W}^{-1,p}(\Omega))} + \langle \partial_{\ell} \mathcal{J}(\mathcal{S}(\bar{\ell}), \bar{\ell}), \delta \ell - \bar{\ell} \rangle_{L^{2}(0,T;\boldsymbol{W}^{-1,p}(\Omega))} \geq 0 \quad \forall \delta \ell \in \mathfrak{L}, \tag{3.95}$$

where we employed the reflexivity of V, the embedding  $V \hookrightarrow \mathbf{W}^{-1,p}(\Omega)^*$ , as well as the regularity of  $\mathbf{w}$ . Since (3.95) is (3.86), the proof is now complete.

With the next result we can conclude that the optimality system (3.84)-(3.86) is in fact equivalent to the first order necessary optimality condition in Lemma 3.35, provided that the *strict complementarity* assumption is fulfilled:

**Proposition 3.43.** Suppose that Assumptions 1.17, p. 19, 1.56, p. 46, 3.29 and 3.33 hold true. Moreover, let Assumption 3.38 be fulfilled for some  $\bar{\ell} \in \mathfrak{L}$ . Then, if  $\bar{\ell}$  together with its states

$$(\bar{\boldsymbol{u}}, \bar{\varphi}, \bar{d}) \in L^{\infty}(0, T; \boldsymbol{W}_{D}^{1,s}(\Omega)) \times L^{\infty}(0, T; H^{1}(\Omega)) \times W^{1,\infty}(0, T; L^{2}(\Omega)),$$

where  $s \in [2, p)$ , and adjoint states

$$(\boldsymbol{w}, v, \xi) \in L^2(0, T; V) \times L^2(0, T; H^1(\Omega)) \times W_T^{1,2}(0, T; L^2(\Omega)),$$

satisfies the optimality system (3.84)-(3.86), it also satisfies the variational inequality (VI).

*Proof.* In view of (VI), it suffices to show

$$\langle \boldsymbol{w}, \delta \ell - \bar{\ell} \rangle_{L^{2}(0,T;\boldsymbol{W}^{-1,p}(\Omega))} = \partial_{(\boldsymbol{u},\varphi,d)} \mathcal{J}(\mathcal{S}(\bar{\ell}),\bar{\ell}) \mathcal{S}'(\bar{\ell};\delta \ell - \bar{\ell}) \quad \forall \delta \ell \in \mathfrak{L}.$$
(3.96)

Since in the proof of Theorem 3.42 one derives (3.93) and (3.94) regardless of the (therein imposed) local optimality assumption on  $\bar{\ell}$ , these two equalities can be used here to infer (3.96). Note that therefor we employ again the reflexivity of V, the embedding  $V \hookrightarrow W^{-1,p}(\Omega)^*$ , as well as the regularity of w. Moreover, note that it is essential that the *strict complementarity* assumption on  $\bar{\ell}$  is fulfilled in order to conclude (3.96), see the proof of Theorem 3.42 for more details. The proof is now complete.

**Remark 3.44.** Notice that (3.86) holds as an equality on  $\mathfrak{L}$  if the local optimum  $\overline{\ell}$  is an inner point of  $\mathfrak{L}$ , and in particular, if  $\mathfrak{L}$  is an open set. To be more precise, (3.86) is then replaced by

$$\langle \boldsymbol{w} + \partial_{\ell} \mathcal{J}(\cdot), \delta \ell \rangle_{L^{2}(0,T;\boldsymbol{W}^{-1,p}(\Omega))} = 0 \quad \forall \, \delta \ell \in \mathfrak{L}.$$
 (3.97)

In order to see this, we argue as follows. Since  $\bar{\ell} \in int(\mathfrak{L})$ , there exists  $\tau > 0$  small enough such that  $\bar{\ell} \pm \tau \delta \ell \in \mathfrak{L}$  for all  $\delta \ell \in \mathfrak{L}$ . Testing in (3.86) then yields

$$\langle \boldsymbol{w} + \partial_{\ell} \mathcal{J}(\cdot), \pm \tau \delta \ell \rangle_{L^{2}(0,T;\boldsymbol{W}^{-1,p}(\Omega))} \geq 0 \quad \forall \, \delta \ell \in \mathfrak{L},$$

whence (3.97).

Let us now make some comments regarding the existence of solutions for  $(P_{min})$ . To be more precise, we derive a setting so that the direct method of variational calculus can be applied for  $(P_{min})$ . To this end, let us recall that the latter one can be written

$$\min_{\ell \in \mathfrak{L}} f(\ell), \tag{3.98}$$

where  $f: \ell \mapsto \mathcal{J}(\mathcal{S}(\ell), \ell)$  is the reduced objective functional. The first thing to observe is that the direct method of variational calculus cannot be applied to solve (3.98) without further ado. The control set  $\mathfrak{L}$  is indeed a bounded subset of the reflexive Banach space  $L^2(0,T;V^*)$ , in light of  $L^\infty(0,T;\mathbf{W}^{-1,p}(\Omega)) \hookrightarrow L^2(0,T;V^*)$ . Since cf. Assumption 3.29,  $\mathfrak{L}$  is convex, it would suffice to impose that it is also closed in  $L^2(0,T;V^*)$ , in order to

obtain its weak compactness (in  $L^2(0,T;V^*)$ ). However,  $f: \mathfrak{L} \subset L^2(0,T;V^*) \to \mathbb{R}$  is not necessarily weakly lower semicontinuous. This would be the case if  $\mathcal{S}$  were (at least) weakly continuous, which is not to be expected due to the structure of (3.44), see also Theorem 3.20. As cf. Remark 3.24,  $\mathcal{S}$  is Lipschitz continuous, it makes sense to require that the control set is first of all a (bounded) subset of a (reflexive Banach) space which compactly embeds in  $L^{\infty}(0,T;\mathbf{W}^{-1,p}(\Omega))$ . An example for such a control set, which as we will see, satisfies all the conditions needed for showing existence of solutions, is

$$\mathfrak{L} := \{ \ell \in H_0^1(0, T; L^p(\Omega; \mathbb{R}^N)) : \|\dot{\ell}\|_{L^2(0, T; L^p(\Omega; \mathbb{R}^N))} \le b \}, \tag{3.99}$$

where b > 0 is a given bound. Clearly,  $\mathfrak{L}$  satisfies Assumption 3.29 and although for the sole purpose of finding global minimizers for (3.98), weaker assumptions for the functional  $\mathcal{J}$  would suffice, we let Assumption 3.33 further hold, such that all the results proven in this section still apply.

We now prove that for  $\mathfrak L$  given by (3.99) (and  $\mathcal J$  as in Assumption 3.33) the optimal control problem  $(P_{min})$  admits solutions. We begin by noticing that  $\mathfrak L$  is a nonempty, convex, closed and bounded subset of the reflexive Banach space  $H^1(0,T;L^p(\Omega;\mathbb R^N))$ , which in particular means that  $\mathfrak L$  is weakly compact. Moreover,  $f:\mathfrak L\to\mathbb R$  is weakly lower semicontinuous. To see this, consider a sequence  $\{\ell_n\}\subset\mathfrak L$  with  $\ell_n\to\ell$  in  $H^1(0,T;L^p(\Omega;\mathbb R^N))$  as  $n\to\infty$ . Due to  $\mathbf W_D^{1,p'}(\Omega)\hookrightarrow L^{p'}(\Omega;\mathbb R^N)$  and Schauder's theorem we have the compact embedding  $H^1(0,T;L^p(\Omega;\mathbb R^N))\hookrightarrow L^\infty(0,T;\mathbf W^{-1,p}(\Omega))$ , see e.g. [81, Corollary 3.1.42]. With Remark 3.24 we then infer

$$\mathcal{S}(\ell_n) \to \mathcal{S}(\ell)$$
 in  $L^{\infty}(0,T;V) \times L^{\infty}(0,T;H^1(\Omega)) \times W^{1,\infty}(0,T;L^2(\Omega))$ 

as  $n \to \infty$ . As a Hadamard-differentiable function, the objective  $\mathcal{J}$  is continuous on  $L^2(0,T;\boldsymbol{W}_{D}^{1,\nu}(\Omega)) \times L^2(0,T;H^1(\Omega)) \times L^2(0,T;L^2(\Omega)) \times L^2(0,T;\boldsymbol{W}^{-1,p}(\Omega))$ , cf. [73, Proposition 3.2.5]. This together with the above convergence gives that f is indeed weakly lower semicontinuous. Finally, a standard argument yields that (3.98) admits solutions and as an immediate consequence, so does  $(P_{min})$ .

We end this chapter with some remarks related with Assumption 3.38. We first point out that one can reformulate  $(P_{min})$  as an MPCC, provided that the set  $\mathfrak{L}$  is chosen accordingly.

Remark 3.45 ( $(P_{min})$  as MPCC). In view of Theorem 3.20 and since the max $\{\cdot,0\}$ -function is a complementarity function, the unique solution  $(\mathbf{u},\varphi,d)$  of the problem  $(P_{\ell})$  with right-hand side  $\ell \in \mathfrak{L}$  is characterized by

$$-\operatorname{div} g(\varphi(t))\mathbb{C}\varepsilon(\boldsymbol{u}(t)) = \ell(t) \qquad in \ \boldsymbol{W}^{-1,p}(\Omega),$$

$$-\alpha\Delta\varphi(t) + \beta\,\varphi(t) + \frac{1}{2}\,g'(\varphi(t))\mathbb{C}\,\varepsilon(\boldsymbol{u}(t)) : \varepsilon(\boldsymbol{u}(t)) = \beta d(t) \qquad in \ H^1(\Omega)^*,$$

$$0 \le \dot{d}(t) \perp \beta(d(t) - \varphi(t)) + r + \delta \dot{d}(t) \ge 0 \quad a.e. \ in \ \Omega, \quad d(0) = d_0$$

f.a.a.  $t \in (0,T)$ , see also (1.44), p. 27. Thus, if the control set can be described e.g. only by inequalities, e.g.  $\mathfrak{L} = \overline{\mathfrak{B}}_M$ , then the problem  $(P_{min})$  falls into the class of MPCCs, see [72] for the definition thereof in the finite-dimensional case.

Remark 3.46 (Strict complementarity). In contrast to our approach, the most authors dealing with time-dependent MPCCs make use of regularization and penalization techniques, see e.g. [2, 35] (parabolic obstacle problem), [81] (quasistatic plasticity) and [15, 33] (Allen-Cahn and Cahn-Hilliard VIs). In [81] the complementarity condition is relaxed by regularizing, so that the biactive set, i.e., the set where both inequalities in the complementarity condition hold with equality, vanishes. As already mentioned in the introduction of this chapter, the optimality systems obtained by applying the above mentioned methods are in the best case of intermediate strength. This is not surprising at all, since one always loses information when passing to the limit (in the regularized/penalized problem). Hence, an optimality system of strong stationary type is not to be expected when applying this technique.

Recall that, cf. Remark 3.39, the operator S is Hadamard-differentiable in those points for which Assumption 3.38 is fulfilled. Thus, conditions for Gâteaux-differentiability are in our situation simpler described as in the above contributions, since therein the complementarity conditions feature dual variables, see e.g. [81, (4.18)] and [15, (2.1)-(2.5)] (Allen-Cahn VIs).

Remark 3.47 ('Ample controls'). When it comes to the optimal control of nonsmooth problems, strong stationary optimality systems have been mostly derived in the presence of 'ample controls', i.e., (distributed) controls that are not restricted by additional constraints. The literature here is rather scarce. We refer to [10, 31, 63] (elliptic VIs) and to [50] (nonsmooth parabolic equations). In [31] a modified version of the static plasticity problem is considered, in the context of distributed controls without control constraints. After deriving the strong stationarity conditions for the optimal control thereof, the authors state a strong stationary optimality system for the original problem (without proving its necessity).

Until recently, it was an open question whether such a system can be derived in the absence of 'ample controls', see also [63, Section 4]. It turns out, that indeed, the necessity of strong stationarity can be proven for the obstacle problem with pointwise constraints on the control. This was shown in [82], however by requiring that the (unknown) optimizer satisfies certain assumptions (constraint qualifications). There one obtains a strong stationary optimality system, which is a generalization of the optimality system for the optimal control of the obstacle problem derived by [63] in the more restrictive case of 'ample controls'.

# Chapter 4

# Conclusions and outlook

In the present thesis we obtained satisfying results regarding the viability of the damage model with penalty (P). This is two-fold. Firstly, we proved the well-posedness of the problem from a mathematical point of view. Secondly, we showed that the penalty approach makes sense, as both damage variables become equal in the limiting case  $\beta \to \infty$ , whereas the limit model turns out to be a classical viscous damage model. Finally, we derived an optimality system for an optimal control problem governed by (P). Unfortunately, this was possible only under an additional assumption.

In Chapter 1 we saw that (P) admits an unique solution, which is characterized by a system consisting of two elliptic PDEs and an operator differential equation. This was shown under nonrestrictive assumptions in two dimensions, whereas for the unique solvability of the minimization problem in (P) in case of N=3, additional assumptions on the data are required. Without imposing further assumptions, we established that the  $H^1(\Omega)$ -regularity of the nonlocal damage can be improved, which is useful especially in two dimensions, as it leads to significantly better differentiability results for the solution operators, and thus for the entire solution of (P). All in all, we stated different regularity results for the latter one, which take many forms, depending on the space regularity of  $d_0$ , the smoothness of the coefficient function g and the smoothness in time of the load. We recall here also the findings in Chapter 3, see Section 3.2, where the unique solvability of (P) was addressed in a more general setting.

A noteworthy result in Chapter 1 was the equivalence of the evolution equation in (P) with an ordinary differential equation in Banach space. This is particularly advantageous from a numerical point of view, as well as from a theoretical point of view e.g. for performing the time-discretization in Chapter 2 or for investigating the differentiability of the control-to-state operator in Chapter 3. We saw that, in fact, the evolution of the local damage can be expressed by means of various equivalent formulations. Here we recall the energy identity which was essential for proving the viability of the penalty approach. In this context, we established the following. The two damage variables are equal when  $\beta \to \infty$  and the resulting single-field gradient damage model is just a version of a viscous damage model analyzed in [41]. Of course, this is the case only in two dimensions, as in three dimensions the designated space for the

damage variable in [41] no longer coincides with  $H^1(\Omega)$ . In particular, we deduced that the vanishing viscosity analysis from [41] may be transferred to our limit model, so that there is no need to analyze the behaviour of the non-penalized problem (2.74) as  $\delta \searrow 0$ . An interesting question arises here: what happens to the penalized model (P) when the viscosity vanishes, i.e.,  $\delta \searrow 0$ ? This is of particular interest, since the non-viscous damage model with penalty forms the basis for mechanical applications, see [12].

In Chapter 3 we considered an optimal control problem governed by the penalized damage model, where we used the load as control. A reinvestigation of the constraint in  $(P_{min})$  was necessary, which among others led us to a general suitable admissible set for the loads. The control-to-state operator associated to (P) turned out to be only (Hadamard) directionally differentiable. Therefore, standard adjoint calculus could not be applied for the derivation of necessary optimality conditions in form of an optimality system. This was however possible under the *strict complementarity* assumption. Unfortunately, the latter one cannot be a priori verified, as it depends on the searched local optimum. Anyway, this type of assumption is not surprising at all, if we take a look at the existing results in the literature. Roughly speaking, it is the price one has to pay for not regularizing, and thus for obtaining a stronger optimality system as in the case of regularizing. In what concerns the existence of global solutions for  $(P_{min})$ , we saw that this is guaranteed if the control set is a (bounded) subset of a space which compactly embeds in  $L^{\infty}(0,T; \mathbf{W}^{-1,p}(\Omega))$ . It remains an open question if  $(P_{min})$  possesses solutions in the original control set.

# Chapter 5

# Auxiliary results

### 5.1 Nemytskii operators g and g'

In this section we establish some useful results on the Nemytskii operators associated to the function g and its derivatives, see Assumption 0.6. The latter one is supposed to hold throughout this section.

**Lemma 5.1.** For all  $\rho \in [1, \infty]$ , the Nemytskii operators  $g : L^{\rho}(\Omega) \to L^{\infty}(\Omega)$  and  $g' : L^{\rho}(\Omega) \to L^{\infty}(\Omega)$  are well defined and Lipschitz continuous from  $L^{\rho}(\Omega)$  to  $L^{\rho}(\Omega)$ . If in addition,  $g \in C^{2}(\mathbb{R})$ , then the operator  $g'' : L^{\rho}(\Omega) \to L^{\infty}(\Omega)$  is well defined as well.

*Proof.* We prove the result just for the function g'. The results for g and g'' follow completely analogously. The first thing to notice is that g' transforms measurable functions into measurable functions, since g' is continuous in view of (3). Moreover,  $g' \in L^{\infty}(\mathbb{R})$  and hence,  $g' : L^{\rho}(\Omega) \to L^{\infty}(\Omega)$  is well defined for all  $\rho \in [1, \infty]$ . The Lipschitz continuity from  $L^{\rho}(\Omega)$  to  $L^{\rho}(\Omega)$  is a direct consequence of the Lipschitz continuity of  $g' : \mathbb{R} \to \mathbb{R}$ .

**Lemma 5.2.** The Nemytskii operators  $g, g': L^1(\Omega) \to L^{\rho}(\Omega)$  are continuous for all  $\rho \in [1, \infty)$ .

*Proof.* The functions g and g' are continuous on account of (3) and the associated Nemytskii operators  $g, g' : L^1(\Omega) \to L^{\infty}(\Omega)$  are well defined by means of Lemma 5.1. Thus, the assumptions in [21, Theorem 4] are fulfilled, which gives the assertion.

**Lemma 5.3.** The operator  $g: L^{\rho}(\Omega) \to L^{\tau}(\Omega)$  is continuously Fréchet-differentiable for  $1 \leq \tau < \rho < \infty$ . If we assume that the map g satisfies  $g \in C^2(\mathbb{R})$ , then the operator  $g': L^{\rho}(\Omega) \to L^{\tau}(\Omega)$  is continuously Fréchet-differentiable as well.

Proof. We prove the continuously Fréchet-differentiability by means of [21, Theorem 7]. We address just the second part of the statement, since the first one follows with the exactly same arguments. From Lemma 5.2 and by employing [21, Theorem 4], we obtain that g'' is continuous from  $L^{\rho}(\Omega)$  to  $L^{\frac{\rho\tau}{\rho-\tau}}(\Omega)$  for  $1 \leq \tau < \rho < \infty$ . Since  $g' \in C^1(\mathbb{R})$ , all the assumptions in [21, Theorem 7] are fulfilled. This completes the proof.

### 5.2 Product and chain rule

This section consists of a generalization of the product rule in Banach spaces and a generalization of the chain rule, where the inner function is only weakly differentiable.

**Lemma 5.4** (Product rule). Let X, W and  $Y_i$ ,  $Z_i$ , i=1,2, be Banach spaces. Moreover, let  $U \subset X$  be an open set and  $f_i: U \to Y_i$ , i=1,2, be continuous mappings, which are continuously Fréchet-differentiable, when considered as mappings from U to  $Z_i$ . Additionally, let  $P: Z_1 \times Y_2 \to W$  be a product, i.e., a continuous bilinear mapping, and assume that P possesses the same properties, when considered as a mapping from  $Y_1 \times Z_2$  to W. Then, the map  $h: x \in U \to P(f_1(x), f_2(x)) \in W$  is continuously Fréchet-differentiable with

$$h'(x)(\delta x) = P(f'_1(x)(\delta x), f_2(x)) + P(f_1(x), f'_2(x)(\delta x)) \quad \forall x \in U, \ \forall \delta x \in X.$$
 (5.1)

*Proof.* Let  $x \in U$  be arbitrary, but fixed and  $\delta x \in X$  with  $\|\delta x\|_X \neq 0$  small enough such that  $x + \delta x \in U$ . Straight forward computation yields

$$||R(\delta x)||_{W} :=$$

$$:= ||h(x + \delta x) - h(x) - P(f'_{1}(x)(\delta x), f_{2}(x)) - P(f_{1}(x), f'_{2}(x)(\delta x))||_{W}$$

$$\leq ||P(f_{1}(x + \delta x), f_{2}(x)) - P(f_{1}(x), f_{2}(x)) - P(f'_{1}(x)(\delta x), f_{2}(x))||_{W}$$

$$+ ||P(f_{1}(x + \delta x), f_{2}(x + \delta x)) - P(f_{1}(x + \delta x), f_{2}(x)) - P(f_{1}(x + \delta x), f'_{2}(x)(\delta x))||_{W}$$

$$+ ||P(f_{1}(x + \delta x), f'_{2}(x)(\delta x)) - P(f_{1}(x), f'_{2}(x)(\delta x))||_{W}.$$

Since  $P: Z_1 \times Y_2 \to W$ ,  $P: Y_1 \times Z_2 \to W$  are continuous bilinear mappings, we obtain in view of the Fréchet-differentiability of  $f_i: U \to Z_i$  for every  $i \in \{1, 2\}$ , combined with the continuity of  $f_1: U \to Y_1$  that

$$\frac{\|R(\delta x)\|_{W}}{\|\delta x\|_{X}} \leq C \left( \frac{\|R_{f_{1}}(\delta x)\|_{Z_{1}}}{\|\delta x\|_{X}} \|f_{2}(x)\|_{Y_{2}} + \frac{\|R_{f_{2}}(\delta x)\|_{Z_{2}}}{\|\delta x\|_{X}} \|f_{1}(x+\delta x)\|_{Y_{1}} + \|f_{1}(x+\delta x) - f_{1}(x)\|_{Y_{1}} \frac{\|f'_{2}(x)(\delta x)\|_{Z_{2}}}{\|\delta x\|_{X}} \right) \to 0, \quad \text{as } \|\delta x\|_{X} \to 0,$$

where  $R_{f_i}(\delta x) := f_i(x + \delta x) - f_i(x) - f'_i(x)(\delta x)$  for every  $i \in \{1, 2\}$ . Therefore, h is Fréchet-differentiable at  $x \in U$ , with derivative given by (5.1).

In order to show the continuity thereof, let  $\{x_n\} \subset U$  with  $x_n \xrightarrow{n \to \infty} x$  in X be given. By employing the properties of P we obtain for all  $\delta x \in X$ 

$$||P(f'_{1}(x_{n})(\delta x), f_{2}(x_{n})) - P(f'_{1}(x)(\delta x), f_{2}(x))||_{W}$$

$$\leq ||P(f'_{1}(x_{n})(\delta x) - f'_{1}(x)(\delta x), f_{2}(x_{n}))||_{W} + ||P(f'_{1}(x)(\delta x), f_{2}(x_{n}) - f_{2}(x))||_{W}$$

$$\leq C(||f'_{1}(x_{n})(\delta x) - f'_{1}(x)(\delta x)||_{Z_{1}}||f_{2}(x_{n})||_{Y_{2}} + ||f'_{1}(x)(\delta x)||_{Z_{1}}||f_{2}(x_{n}) - f_{2}(x)||_{Y_{2}})$$

$$\leq C(||f'_{1}(x_{n}) - f'_{1}(x)||_{\mathcal{L}(X,Z_{1})}||\delta x||_{X}||f_{2}(x_{n})||_{Y_{2}}$$

$$+ ||f'_{1}(x)||_{\mathcal{L}(X,Z_{1})}||\delta x||_{X}||f_{2}(x_{n}) - f_{2}(x)||_{Y_{2}}).$$

The continuity of  $f'_1: U \to \mathcal{L}(X, Z_1)$  and  $f_2: U \to Y_2$  thus implies

$$\sup_{\|\delta x\|_{X}=1} \|P(f'_{1}(x_{n})(\delta x), f_{2}(x_{n})) - P(f'_{1}(x)(\delta x), f_{2}(x))\|_{W} 
\leq C(\|f'_{1}(x_{n}) - f'_{1}(x)\|_{\mathcal{L}(X,Z_{1})}\|f_{2}(x_{n})\|_{Y_{2}} 
+ \|f'_{1}(x)\|_{\mathcal{L}(X,Z_{1})}\|f_{2}(x_{n}) - f_{2}(x)\|_{Y_{2}}) \to 0 \quad \text{as } n \to \infty.$$
(5.2)

Completely analogously we obtain

$$\sup_{\|\delta x\|_X = 1} \|P(f_1(x_n), f_2'(x_n)(\delta x)) - P(f_1(x), f_2'(x)(\delta x))\|_W \to 0 \quad \text{as } n \to \infty.$$
 (5.3)

Finally, (5.1), (5.2) and (5.3) result in

$$\sup_{\|\delta x\|_X=1} \|h'(x_n)(\delta x) - h'(x)(\delta x)\|_W \to 0 \quad \text{as } n \to \infty,$$

which completes the proof.

**Lemma 5.5** (Chain rule). Let  $\mathfrak{F} \in C^1([0,T] \times H^1(\Omega))$ . Assume that for any K > 0 there is  $C_K > 0$  such that

$$\|\partial_t \mathfrak{F}(\cdot, \phi(\cdot))\|_{L^2(0,T)}, \|\partial_z \mathfrak{F}(\cdot, \phi(\cdot))\|_{L^\infty(0,T;H^1(\Omega)^*)} \le C_K \tag{5.4}$$

for all  $\phi \in H^1(0,T;H^1(\Omega))$  with  $\|\phi\|_{H^1(0,T;H^1(\Omega))} \leq K$ . Then, for  $z \in H^1(0,T;H^1(\Omega))$ , the map  $t \mapsto \mathfrak{F}(t,z(t))$  belongs to  $H^1(0,T)$  with weak derivative given by

$$\frac{d}{dt}\mathfrak{F}(t,z(t)) = \partial_t\mathfrak{F}(t,z(t)) + \langle \partial_z\mathfrak{F}(t,z(t)),\dot{z}(t)\rangle_{H^1(\Omega)} \quad \text{f.a.a. } t \in (0,T).$$

*Proof.* We intend to prove the assertion by employing a density argument, as well as the fact that chain rule holds true for  $C^1$ -functions. First, we observe that a short inspection of the proof of [70, Lemma 7.2] shows that

$$C^{\infty}([0,T];H^1(\Omega)) \stackrel{d}{\hookrightarrow} H^1(0,T;H^1(\Omega)).$$

In view thereof, there exists a sequence  $\{z_n\} \subset C^{\infty}([0,T];H^1(\Omega))$  such that

$$z_n \to z \text{ in } H^1(0,T;H^1(\Omega)) \text{ as } n \to \infty.$$
 (5.5)

By  $H^1(0,T;H^1(\Omega)) \hookrightarrow C([0,T];H^1(\Omega))$ , cf. [81, Theorem 3.1.41], this leads to

$$z_n(t) \to z(t) \text{ in } H^1(\Omega) \ \forall t \in [0, T] \quad \text{as } n \to \infty.$$
 (5.6)

Moreover, as a result of chain rule, it holds for all  $n \in \mathbb{N}$ 

$$\mathfrak{F}(t,z_n(t)) = \mathfrak{F}(0,z_n(0)) + \int_0^t \frac{d}{dt} \mathfrak{F}(s,z_n(s)) ds$$

$$= \mathfrak{F}(0,z_n(0)) + \int_0^t \partial_t \mathfrak{F}(s,z_n(s)) + \langle \partial_z \mathfrak{F}(s,z_n(s)), \dot{z}_n(s) \rangle_{H^1(\Omega)} ds \quad \forall t \in [0,T].$$
(5.7)

We now focus on passing to the limit in (5.7). The continuity of  $\mathfrak{F}$  combined with (5.6) yields

$$\mathfrak{F}(t, z_n(t)) - \mathfrak{F}(0, z_n(0)) \stackrel{n \to \infty}{\longrightarrow} \mathfrak{F}(t, z(t)) - \mathfrak{F}(0, z(0)) \quad \text{for all } t \in [0, T], \tag{5.8}$$

so that it remains to show the convergence of the last term on the right-hand side in (5.7). Let us first abbreviate for simplicity

$$f_n := \partial_t \mathfrak{F}(\cdot, z_n(\cdot)) + \langle \partial_z \mathfrak{F}(\cdot, z_n(\cdot)), \dot{z}_n(\cdot) \rangle_{H^1(\Omega)} \quad \forall n \in \mathbb{N},$$
  
$$f := \partial_t \mathfrak{F}(\cdot, z(\cdot)) + \langle \partial_z \mathfrak{F}(\cdot, z(\cdot)), \dot{z}(\cdot) \rangle_{H^1(\Omega)}.$$

We observe that due to (5.5) it holds

$$\dot{z}_n \to \dot{z} \quad \text{in } L^2(0, T; H^1(\Omega)) \text{ as } n \to \infty.$$
 (5.9)

Thus, there exists a subsequence  $\{z_{n_k}\}$  so that

$$\dot{z}_{n_k}(t) \to \dot{z}(t)$$
 in  $H^1(\Omega)$  f.a.a.  $t \in (0,T)$  as  $k \to \infty$ . (5.10)

Since  $\mathfrak{F} \in C^1([0,T] \times H^1(\Omega))$ , one further deduces from (5.6) the convergences

$$\partial_t \mathfrak{F}(t, z_n(t)) \xrightarrow{n \to \infty} \partial_t \mathfrak{F}(t, z(t)) \quad \text{for all } t \in [0, T], \\
\partial_z \mathfrak{F}(t, z_n(t)) \xrightarrow{n \to \infty} \partial_z \mathfrak{F}(t, z(t)) \text{ in } H^1(\Omega)^* \quad \text{for all } t \in [0, T].$$
(5.11)

Now, (5.10) and (5.11) give in turn

$$f_{n_k}(t) \to f(t)$$
 f.a.a.  $t \in (0, T)$  as  $k \to \infty$ . (5.12)

Further, for every  $n \in \mathbb{N}$ , Hölder's inequality leads to

$$||f_n||_{L^2(0,T)} \le ||\partial_t \mathfrak{F}(\cdot, z_n(\cdot))||_{L^2(0,T)} + ||\partial_z \mathfrak{F}(\cdot, z_n(\cdot))||_{L^\infty(0,T;H^1(\Omega)^*)} ||\dot{z}_n||_{L^2(0,T;H^1(\Omega))}.$$
(5.13)

In view of (5.5) and (5.9) there exists K > 0, independent of n, so that

$$||z_n||_{H^1(0,T;H^1(\Omega))}, ||\dot{z}_n||_{L^2(0,T;H^1(\Omega))} \leq K \text{ for all } n \in \mathbb{N}.$$

Thanks to (5.4), the estimate (5.13) can be continued as

$$||f_n||_{L^2(0,T)} \le C \quad \text{for all } n \in \mathbb{N}, \tag{5.14}$$

where C > 0 is independent of n. Since  $L^2(0,T)$  is a reflexive Banach space, we deduce from (5.14) that  $\{f_{n_k}\}$  possesses a weakly convergent subsequence, denoted by the same symbol, which together with (5.12) gives in turn

$$f_{n_k} \rightharpoonup f \text{ in } L^2(0,T) \text{ as } k \to \infty.$$
 (5.15)

In particular, we have for all  $t \in [0, T]$  the convergence

$$\int_0^t f_{n_k}(s) \ ds \rightharpoonup \int_0^t f(s) \ ds \quad \text{as } k \to \infty, \tag{5.16}$$

since the integral operator is linear and continuous. Passing to the limit in (5.7), where we rely on (5.8) and (5.16), gives in turn

$$\mathfrak{F}(t,z(t))=\mathfrak{F}(0,z(0))+\int_0^t\partial_t\mathfrak{F}(s,z(s))+\langle\partial_z\mathfrak{F}(s,z(s)),\dot{z}(s)\rangle_{H^1(\Omega)}ds\quad\forall\,t\in[0,T].$$

Since  $f \in L^2(0,T)$ , cf. (5.15), we now deduce that  $\mathfrak{F}(\cdot,z(\cdot)) \in H^1(0,T)$  with  $\frac{d}{dt}\mathfrak{F}(\cdot,z(\cdot)) = f$ , see e.g. [81, Lemma 3.1.37]. The proof is now complete.

# 5.3 Properties of the max-operator

#### Lemma 5.6.

(i) The Nemytskii operator associated to  $\max : \mathbb{R} \to \mathbb{R}$ ,  $\max(\cdot) := \max\{\cdot, 0\}$ , maps  $L^2(\Omega)$  to  $L^2(\Omega)$ . Moreover,  $\max : L^2(\Omega) \to L^2(\Omega)$  is Lipschitz continuous with Lipschitz constant 1 and directionally differentiable. For any  $y, h \in L^2(\Omega)$ , the derivative satisfies

$$\max'(y;h)(x) = \max'(y(x);h(x)) = \begin{cases} h(x) & \text{if } y(x) > 0\\ \max\{h(x),0\} & \text{if } y(x) = 0\\ 0 & \text{if } y(x) < 0 \end{cases}$$
 (5.17)

In addition, at any  $y \in L^2(\Omega)$ , the operator  $\max'(y;\cdot): L^2(\Omega) \to L^2(\Omega)$  is Lipschitz continuous with Lipschitz constant 1.

(ii) The Nemytskii operator associated to  $\max: L^2(\Omega) \to L^2(\Omega)$  maps  $L^{\infty}(0,T;L^2(\Omega))$  to  $L^{\infty}(0,T;L^2(\Omega))$ . Moreover,  $\max: L^{\infty}(0,T;L^2(\Omega)) \to L^{\infty}(0,T;L^2(\Omega))$  is Lipschitz continuous with constant 1. The operator  $\max: L^{\infty}(0,T;L^2(\Omega)) \to L^{\varrho}(0,T;L^2(\Omega))$  is directionally differentiable for any  $\varrho \in [1,\infty)$ , with  $\max'(\cdot;\cdot) \in L^{\infty}(0,T;L^2(\Omega))$ . For any  $y,h \in L^{\infty}(0,T;L^2(\Omega))$  the derivative is given by

$$\max'(y;h)(t,x) = \max'(y(t);h(t))(x) = \max'(y(t,x);h(t,x))$$

$$= \begin{cases} h(t,x) & \text{if } y(t,x) > 0\\ \max\{h(t,x),0\} & \text{if } y(t,x) = 0\\ 0 & \text{if } y(t,x) < 0 \end{cases} f.a.a. \ (t,x) \in (0,T) \times \Omega.$$
(5.18)

*Proof.* (i) The first thing to notice is that  $\mathbb{R} \ni v \mapsto \max\{v,0\} \in \mathbb{R}$  is globally Lipschitz continuous with constant 1 and satisfies  $|\max\{v,0\}| \le |v|$  for all  $v \in \mathbb{R}$ . As a continuous function, max maps measurable functions into measurable functions, and from the above estimate we deduce that the Nemytskii operator  $\max : L^2(\Omega) \to L^2(\Omega)$  is well defined. In particular,

$$\|\max(y)\|_2 \le \|y\|_2 \quad \forall y \in L^2(\Omega).$$
 (5.19)

The desired Lipschitz continuity of max :  $L^2(\Omega) \to L^2(\Omega)$  is an immediate consequence of the Lipschitz continuity of max :  $\mathbb{R} \to \mathbb{R}$ .

Furthermore, straight forward computation shows that  $\max : \mathbb{R} \to \mathbb{R}$  is directionally differentiable with

$$\max'(v; \delta v) = \begin{cases} \delta v & \text{if } v > 0\\ \max\{\delta v, 0\} & \text{if } v = 0 \quad \forall v, \delta v \in \mathbb{R}.\\ 0 & \text{if } v < 0 \end{cases}$$
 (5.20)

As a consequence thereof (or by employing Lemma 5.8), we deduce that

$$|\max'(v; \delta v)| \le |\delta v| \quad \forall v, \delta v \in \mathbb{R}.$$
 (5.21)

With the above results at hand we next show by Lebesgue's dominated convergence theorem that the Nemytskii operator max :  $L^2(\Omega) \to L^2(\Omega)$  is directionally differentiable, too. To this end, let  $y, h \in L^2(\Omega)$  be arbitrary, but fixed. The directional differentiability of max :  $\mathbb{R} \to \mathbb{R}$  yields

$$\left|\frac{\max(y(x)+\tau h(x))-\max(y(x))}{\tau}-\max'(y(x);h(x))\right| \stackrel{\tau\searrow 0}{\longrightarrow} 0 \quad \text{f.a.a. } x\in\Omega.$$

Note that  $\Omega \ni x \mapsto \max(y(x) + \tau h(x)) - \max(y(x)) \in \mathbb{R}$  is measurable for any  $\tau \in \mathbb{R}$  and, as a.e. limit of measurable functions,  $\Omega \ni x \mapsto \max'(y(x); h(x)) \in \mathbb{R}$  is measurable as well. On the other side, the global Lipschitz continuity of  $\max : \mathbb{R} \to \mathbb{R}$  with constant 1 together with (5.21) implies for  $\tau \neq 0$  that

$$\left|\frac{\max(y(x)+\tau h(x))-\max(y(x))}{\tau}-\max'(y(x);h(x))\right|\leq 2\left|h(x)\right| \text{ f.a.a. } x\in\Omega.$$

Now, Lebesgue's dominated convergence theorem gives that max :  $L^2(\Omega) \to L^2(\Omega)$  is directionally differentiable with directional derivative given by (5.17), in view of (5.20).

It now remains to show that at any  $y \in L^2(\Omega)$ , the operator  $\max'(y;\cdot): L^2(\Omega) \to L^2(\Omega)$  is Lipschitz continuous with Lipschitz constant 1. This follows immediately from the definition of the directional derivative and the Lipschitz continuity of the operator max, which imply that

$$\|\max'(y; h_1) - \max'(y; h_2)\|_2 = \lim_{\tau \searrow 0} \left\| \frac{\max(y + \tau h_1) - \max(y + \tau h_2)}{\tau} \right\|_2$$
  
$$\leq \|h_1 - h_2\|_2 \quad \forall h_1, h_2 \in L^2(\Omega).$$

(ii) The desired results follow by very similar arguments to those used in part (i). From the global Lipschitz continuity with constant 1 of max :  $L^2(\Omega) \to L^2(\Omega)$  and from (5.19), we infer that max :  $L^{\infty}(0,T;L^2(\Omega)) \to L^{\infty}(0,T;L^2(\Omega))$  is well defined and Lipschitz continuous with constant 1. We now prove the directional differentiability at  $y \in L^{\infty}(0,T;L^2(\Omega))$  in direction  $h \in L^{\infty}(0,T;L^2(\Omega))$ . We proceed as in part (i). From the directional differentiability of max :  $L^2(\Omega) \to L^2(\Omega)$  we obtain the convergence

$$\left\| \frac{\max(y(t) + \tau h(t)) - \max(y(t))}{\tau} - \max'(y(t); h(t)) \right\|_{2} \xrightarrow{\tau \searrow 0} 0 \quad \text{f.a.a. } t \in (0, T). \quad (5.22)$$

On the other side, by employing e.g. (5.17) and (5.21), we arrive at

$$\|\max'(y(t); h(t))\|_{2} \le \|h(t)\|_{2}$$
 f.a.a.  $t \in (0, T)$ . (5.23)

This leads for any  $\tau \neq 0$  to

$$\left\| \frac{\max(y(t) + \tau h(t)) - \max(y(t))}{\tau} - \max'(y(t); h(t)) \right\|_{2} \le 2 \|h(t)\|_{2} \text{ f.a.a. } t \in (0, T),$$

$$(5.24)$$

where we also used the Lipschitz continuity with constant 1 of max :  $L^2(\Omega) \to L^2(\Omega)$ . The measurability of

$$t \mapsto \left\| \frac{\max(y(t) + \tau h(t)) - \max(y(t))}{\tau} - \max'(y(t); h(t)) \right\|_{2}$$

for any  $\tau \neq 0$  follows with the exactly same arguments from part (i), in combination with e.g. [71, Lemma 1.7]. In view of (5.22) and (5.24), Lebesgue's dominated convergence theorem yields now that, for any  $\varrho \in [1, \infty)$ , max :  $L^{\infty}(0, T; L^{2}(\Omega)) \to L^{\varrho}(0, T; L^{2}(\Omega))$  is directionally differentiable. The directional derivative is given by

$$\max'(y; h) = \max'(y(\cdot); h(\cdot))$$
 a.e. in  $(0, T)$ . (5.25)

and from (5.23) we deduce that  $t \mapsto \max'(y;h)(t) \in L^{\infty}(0,T;L^{2}(\Omega))$ , in view of  $h \in L^{\infty}(0,T;L^{2}(\Omega))$ . Finally, we notice that (5.18) follows from (5.25) and (5.17). The proof is now complete.

## 5.4 Picard-Lindelöf's theorem in abstract function spaces

**Lemma 5.7.** Let  $r \in [1, \infty]$  be given. Moreover, let X be a Banach space and  $z_0 \in X$ . Suppose that the map  $f: (0,T) \times X \to X$  satisfies

- (i) for any  $z \in X$ , the value  $f(t,z) \in X$  is well defined f.a.a.  $t \in (0,T)$ ,
- (ii) the Nemytskii operator associated to f maps  $L^r(0,T;X)$  to  $L^r(0,T;X)$ ,
- (iii)  $f(t,\cdot)$  is globally Lipschitz continuous f.a.a.  $t \in (0,T)$ , with constant  $L_f > 0$  independent of t.

Then, the initial value problem

$$\dot{z}(t) = f(t, z(t))$$
 f.a.a.  $t \in (0, T)$ ,  $z(0) = z_0$  (5.26)

admits a unique global solution  $z \in W^{1,r}(0,T;X)$ .

*Proof.* We intend to find (at first) local solutions for (5.26) by using similar arguments to those used for proving [14, Theorem 7.2.3]. That is, we solve (5.26) by Banach fixed-point theorem, which provides the existence of a unique local solution  $z \in W^{1,r}(0,t;X) \hookrightarrow C([0,t];X)$ , where t may be less than T. However, by considering (5.26) on (t,T) with initial value z(t), one finds a new local solution. One proceeds in this way until a

local solution on an interval containing the point T is achieved. The global solution is then obtained from the local ones. The time regularity of the local solutions ultimately transfers to the global solution of (5.26).

#### (I) Existence of local solutions

We begin by searching for fixed points for the mapping  $L^r(0,t;X)\ni z\mapsto \mathcal{F}(z):[0,t]\to X$ , given by

$$\mathcal{F}(z)(\tau) := z_0 + \int_0^{\tau} f(s, z(s)) \, ds \quad \forall \, \tau \in [0, t], \tag{5.27}$$

where  $t \in (0,T]$  is to be determined so that the assumptions in Banach fixed-point theorem are satisfied. As it will turn out, a fixed point of  $\mathcal{F}$  on  $L^r(0,t;X)$  solves (5.26) on (0,t).

We first show that  $\mathcal{F}$  maps  $L^r(0,t;X)$  to  $W^{1,r}(0,t;X)$  for some arbitrary, but fixed  $t \in (0,T]$ . For  $z \in L^r(0,t;X)$  we have, as a consequence of (ii), that  $f(\cdot,z(\cdot)) \in L^r(0,t;X)$ . In view of (5.27), we obtain from [81, Lemma 3.1.37] that  $\tau \mapsto \mathcal{F}(z)(\tau) \in W^{1,r}(0,t;X)$ . Hence,  $\mathcal{F}: L^r(0,t;X) \to W^{1,r}(0,t;X)$  is indeed well defined. Note that due to  $W^{1,r}(0,t;X) \hookrightarrow C([0,t];X)$ , see e.g. [81, Theorem 3.1.41],  $\mathcal{F}(z)$  is well defined in each  $\tau \in [0,t]$ . Moreover, as a result of [81, Lemma 3.1.37], it holds

$$\frac{d}{dt}\mathcal{F}(z)(\tau) = f(\tau, z(\tau)) \quad \text{f.a.a. } \tau \in (0, t).$$
 (5.28)

We now search for t > 0 such that the function  $\mathcal{F}: L^r(0,t;X) \to L^r(0,t;X)$  is a contraction, i.e., Lipschitz continuous with constant K < 1. Thanks to (5.27) and (iii) we have for all  $z_1, z_2 \in L^r(0,t;X)$  the estimate

$$\|\mathcal{F}(z_1)(\tau) - \mathcal{F}(z_2)(\tau)\|_X \le L_f \int_0^\tau \|z_1(s) - z_2(s)\|_X ds$$

$$\le L_f \|z_1 - z_2\|_{L^1(0,t;X)}$$

$$\le L_f t^{1-1/r} \|z_1 - z_2\|_{L^r(0,t;X)} \quad \text{for all } \tau \in [0,t],$$

whence

$$\|\mathcal{F}(z_1) - \mathcal{F}(z_2)\|_{L^r(0,t;X)} \le L_f t \|z_1 - z_2\|_{L^r(0,t;X)}.$$
 (5.29)

Notice that for the above estimate we employed the definition of the  $L^r(0,t;X)$ -norm. From (5.29) we read that  $\mathcal{F}: L^r(0,t;X) \to L^r(0,t;X)$  is a contraction provided that  $K := L_f t < 1$ . That is why we choose to define

$$t := \begin{cases} T, & \text{if } L_f T < 1, \\ 1/(2L_f), & \text{otherwise.} \end{cases}$$
 (5.30)

Now, by Banach fixed-point theorem, we can finally conclude that, for t as in (5.30), the equation

$$z = \mathcal{F}(z) \tag{5.31}$$

admits a unique solution in  $L^r(0,t;X)$ , which, for later purposes, we denote by  $z_1$ . Since  $\mathcal{F}$  has range in  $W^{1,r}(0,t;X)$ , we obtain as an immediate consequence of (5.31) that  $z_1 \in W^{1,r}(0,t;X)$ . Moreover, (5.28) in combination with (5.31), and (5.27) yield

$$\dot{z}_1(\tau) = f(\tau, z_1(\tau))$$
 f.a.a.  $\tau \in (0, t), \quad z_1(0) = z_0.$  (5.32)

Hence, we have proven that, indeed, the fix point of  $\mathcal{F}$  on  $L^r(0,t;X)$  solves (5.26) on (0,t), where t is given by (5.30). Relying on [81, Corollary 3.1.39] we obtain that any solution in  $W^{1,r}(0,t;X)$  of (5.32) solves (5.31), which in particular means that (5.32) is uniquely solvable. Altogether, we have shown that (5.26) admits a unique local solution in  $W^{1,r}(0,t;X)$ , where t is given by (5.30).

Thus, if  $L_f T < 1$ , this solution is in fact global and the proof is now complete. Otherwise, we further search for local solutions, this time for the initial value problem

$$\dot{z}(\tau) = f(\tau, z(\tau))$$
 f.a.a.  $\tau \in (t, T), \quad z(t) = z_1(t).$  (5.33)

We proceed in the exact same way as above and find a unique  $z_2 \in W^{1,r}(t, 2t; X)$  which solves  $z = \mathcal{F}(z)$ , where  $\mathcal{F}: L^r(t, 2t; X) \to W^{1,r}(t, 2t; X)$  must be accordingly redefined as

$$\mathcal{F}(z)(\tau) := z_1(t) + \int_t^{\tau} f(s, z(s)) ds \quad \forall \tau \in [t, 2t].$$

Then, we deduce as above that (5.33) admits a unique local solution  $z_2 \in W^{1,r}(t, 2t; X)$ . In the end, one finds for each j = 1, ..., n + 1, where  $n = \lfloor 2TL_f \rfloor$ , a unique  $z_j \in W^{1,r}((j-1)t, jt; X)$ , which satisfies

$$\dot{z}_j(\tau) = f(\tau, z_j(\tau))$$
 f.a.a.  $\tau \in ((j-1)t, jt), \quad z_j((j-1)t) = z_{j-1}((j-1)t).$  (5.34)

Here we use the conventions  $z_0(0) = z_0$  and (n+1)t = T. Of course, the number of initial value problems one has to solve, i.e., n+1, can be reduced by setting t larger in (5.30) (such that  $L_f t < 1$ ).

#### (II) Existence of global solutions

In [14, Theorem 7.2.6] the global solvability of (5.26) is shown by constructing a (to  $\|\cdot\|_{C([0,T];X)}$ ) equivalent norm such that  $\mathcal{F}$  is a contraction on the space C([0,T];X) endowed with the new norm. The latter one allows for deriving pointwise in time estimates, which combined with the Lipschitz continuity of f lead to a sharper estimate of the type (5.29). As we work with Lebesgue spaces, we cannot derive pointwise in time estimates in a similar way. That is why we construct a global solution of (5.26) by concatenating the local solutions found in the previous step. That is, we show that the global solution is the function  $z:[0,T] \to X$  given by

$$z(\tau) := \sum_{j=1}^{n+1} \widetilde{z}_j(\tau) \quad \forall \tau \in [0, T], \tag{5.35}$$

where

$$\widetilde{z}_j := \begin{cases} z_j & \text{on } [(j-1)t, jt), \\ 0 & \text{elsewhere,} \end{cases} \quad \text{for } j = 1, ..., n, \qquad \widetilde{z}_{n+1} := \begin{cases} z_{n+1} & \text{on } [nt, T], \\ 0 & \text{elsewhere.} \end{cases}$$

We first prove that  $z \in W^{1,r}(0,T;X)$ . To this end, let  $\tau \in [0,T]$  be arbitrary, but fixed. This means that there exists  $k \in \{1,...,n+1\}$  so that  $\tau \in [(k-1)t,kt]$ . From (5.35) and [81, Corollary 3.1.39] applied for  $z_j \in W^{1,r}((j-1)t,jt;X), j=k,...,1$ , we have

$$z(\tau) = z_k(\tau) = z_k((k-1)t) + \int_{(k-1)t}^{\tau} \dot{z}_k(s) \, ds$$

$$= z_{k-1}((k-1)t) + \int_{(k-1)t}^{\tau} \dot{z}_k(s) \, ds$$

$$= z_0 + \sum_{j=1}^{k-1} \int_{(j-1)t}^{jt} \dot{z}_j(s) \, ds + \int_{(k-1)t}^{\tau} \dot{z}_k(s) \, ds$$

$$= z_0 + \int_0^{\tau} y(s) \, ds,$$

where  $y = \dot{z}_j$  a.e. in ((j-1)t, jt) for any j = 1, ..., n+1. Since  $\dot{z}_j \in L^r((j-1)t, jt; X)$ , the mapping  $\tau \mapsto y(\tau)$  belongs to  $L^r(0,T;X)$ . This follows for example by defining y as the sum of the extensions of  $\dot{z}_j$  on (0,T) by zero, which means that y is sum of functions in  $L^r(0,T;X)$ . Now, [81, Lemma 3.1.37] in combination with the above identity ensures that  $z \in W^{1,r}(0,T;X)$  with  $\dot{z} = y$  a.e. in (0,T). With (5.34) we then infer

$$\dot{z}(\tau) = \dot{z}_j = f(\tau, z_j(\tau))$$
 a.e. in  $((j-1)t, jt), \ j = 1, ..., n+1,$ 

i.e.,

$$\dot{z}(\tau) = f(\tau, z(\tau))$$
 a.e. in  $(0, T)$ ,

where we used (5.35). Therefrom, we also get  $z(0) = z_0$ . In other words, z given by (5.35) is a global solution of (5.26). In order to prove the uniqueness thereof, we argue as follows. Any global solution  $\tilde{z}$  of (5.26) solves (5.32), i.e.,  $\tilde{z}|_{[0,t]} = z_1 \in W^{1,r}(0,t;X)$ . In particular,  $\tilde{z}(t) = z_1(t)$ . Additionally,  $\tilde{z}$  satisfies (5.33) on (t,2t), and thus,  $\tilde{z}|_{[t,2t]} = z_2 \in W^{1,r}(t,2t;X)$ . By further arguing in this way, we obtain  $\tilde{z}|_{[(j-1)t,jt]} = z_j \in W^{1,r}((j-1)t,jt;X)$  for all j=1,...,n+1 and the uniqueness of the local solutions yields the uniqueness of the global one. The proof is now complete.

## 5.5 Miscellaneous

This section collects various general results, of which we often make use in the thesis.

**Lemma 5.8.** Let X, Y, Z be Banach spaces and let  $U \subset X$  be an open set. Moreover, let  $f: U \times Y \to Z$  be directionally differentiable and Lipschitz continuous, i.e., there exists  $L_f > 0$  so that

$$||f(x) - f(y)||_Z \le L_f ||x - y||_{X \times Y} \quad \forall x, y \in U \times Y.$$

Then, for any  $y \in U \times Y$ , we have

$$||f'(y;h)||_Z \le L_f ||h||_{X\times Y} \quad \forall h \in X \times Y.$$

*Proof.* Let  $y \in U \times Y$  and  $h \in X \times Y$  be arbitrary, but fixed. Since  $U \times Y$  is open, there exists  $\tau > 0$  small enough such that  $y + \tau h \in U \times Y$ . By assumption, we then have

$$||f(y+\tau h) - f(y)||_{Z} \le L_{f} ||\tau h||_{X \times Y}.$$
(5.36)

Dividing by  $\tau > 0$  and passing to the limit  $\tau \searrow 0$  in (5.36) gives the claim.

**Lemma 5.9.** Let X be a Banach space. For any  $t \in [0,T]$ , we define the operator  $\Theta_t: W^{1,1}(0,T;X) \to X$  as

$$\Theta_t(\zeta) := \zeta(t).$$

Then,  $\Theta_t \in \mathcal{L}(W^{1,1}(0,T;X),X)$  and there exists C > 0, independent of t, so that  $\|\Theta_t\|_{\mathcal{L}(W^{1,1}(0,T;X),X)} \leq C$  for all  $t \in [0,T]$ .

*Proof.* Let  $t \in [0,T]$  be arbitrary, but fixed. We notice that  $\Theta_t$  is well defined, as a result of

$$W^{1,1}(0,T;X) \hookrightarrow C([0,T];X),$$
 (5.37)

see e.g. [81, Theorem 3.1.41]. We only address its boundedness, as the linearity is obvious. This follows easily, since from (5.37) we have for all  $\zeta \in W^{1,1}(0,T;X)$ 

$$\|\Theta_t(\zeta)\|_X = \|\zeta(t)\|_X \le \|\zeta\|_{C([0,T];X)} \le C\|\zeta\|_{W^{1,1}(0,T;X)},$$

where C > 0 is the embedding constant, which depends only on T, see [81, (3.30)] for more details. Moreover, note that  $\|\Theta_t\|_{\mathcal{L}(W^{1,1}(0,T;X),X)} \leq C$  for all  $t \in [0,T]$ . This completes the proof.

**Lemma 5.10.** Let  $r \in [1, \infty]$  and  $z \in W_0^{1,r}(0, T; X)$  be given, where X is a Banach space. If there exists a non-negative function  $y \in L^r(0,T)$  and a constant  $c \geq 0$  such that

$$\|\dot{z}(t)\|_{X} \le c\|z(t)\|_{X} + y(t)$$
 f.a.a.  $t \in (0, T),$  (5.38)

then

$$||z||_{W^{1,r}(0,T;X)} \le K||y||_{L^r(0,T)},$$

with a positive constant K = K(c, r, T) if  $r < \infty$  and K = K(c, T) if  $r = \infty$ . As a consequence, Poincaré-Friedrich's inequality in abstract function spaces holds true, i.e.,

$$||z||_{L^r(0,T;X)} \le K ||\dot{z}||_{L^r(0,T;X)} \quad \forall z \in W_0^{1,r}(0,T;X),$$
 (5.39)

where K = K(r,T) > 0 is a constant (independent of r if  $r = \infty$ ).

*Proof.* By integrating (5.38) we arrive at

$$||z(t)||_X \le \int_0^t ||\dot{z}(s)||_X ds \le c \int_0^t ||z(s)||_X ds + \int_0^t y(s) ds \quad \forall t \in [0, T],$$

where for the first inequality we employed [81, Corollary 3.1.39], z(0) = 0 and the embedding  $W^{1,r}(0,T;X) \hookrightarrow C([0,T];X)$ . Since  $z \in C([0,T];X)$  and  $[0,T] \ni t \mapsto \int_0^t y(s) ds \in [0,\infty)$  is nondecreasing, Gronwall's lemma yields

$$||z(t)||_X \le \exp(ct) \int_0^t y(s) \ ds \le C||y||_{L^r(0,T)} \quad \forall t \in [0,T].$$

In particular,  $||z||_{L^r(0,T;X)} \le C||y||_{L^r(0,T)}$ . Notice that  $C = \exp(cT)T^{(1-1/r)}$ , whence  $C = \exp(cT)T$  if  $r = \infty$ . By means of (5.38) we now have

$$\|\dot{z}\|_{L^r(0,T;X)} \le c\|z\|_{L^r(0,T;X)} + \|y\|_{L^r(0,T)} \le K(c,C)\|y\|_{L^r(0,T)},$$

which gives the first claim. Poincaré-Friedrich's inequality follows immediately by considering the special case  $y := \|\dot{z}(\cdot)\|_X$  and by setting c := 0 in (5.38). The proof is now complete.

**Lemma 5.11.** Let X,Y be Banach spaces and let U be a nonempty open set in X. Moreover, let  $x:[0,T] \to X$  be a Bochner measurable function, such that  $x(t) \in U$  f.a.a.  $t \in (0,T)$  and let  $f:U \to Y$  be a continuous function. Then, the mapping  $t \mapsto f(x(t)) \in Y$  is Bochner measurable.

*Proof.* In view of the Bochner measurability of x, there exists a sequence of simple functions  $\{x_n\}$  such that

$$x_n(t) \to x(t)$$
 f.a.a.  $t \in (0,T)$  as  $n \to \infty$ .

Since U is open and  $x(t) \in U$  f.a.a.  $t \in (0,T)$ , there exists at almost all  $t \in (0,T)$  some  $m = m(t) \in \mathbb{N}$  large enough such that

$$x_n(t) \in U \quad \text{for all } n \ge m(t).$$
 (5.40)

Applying the continuity of  $f: U \to Y$  gives in turn

$$f(x_n(t)) \to f(x(t))$$
 f.a.a.  $t \in (0,T)$  as  $n \to \infty$ . (5.41)

Let now  $n \in \mathbb{N}$  be arbitrary, but fixed. Since  $x_n$  is a simple function, it holds

$$x_n(t) = \sum_{i=1}^{k_n} \chi_{B_{n,i}}(t) \, x_{n,i}$$
 f.a.a.  $t \in (0,T)$ ,

with Lebesgue measurable subsets  $B_{n,i} \subset [0,T]$  and  $x_{n,i} \in X$ ,  $i = 1, ..., k_n$ . W.l.o.g. we assume that  $\{B_{n,i}\}_i$  is a disjoint partition of [0,T]. Since  $U \neq \emptyset$ , we can fix  $u \in U$ . Then, the function  $f_n : [0,T] \to Y$  given by

$$f_n(t) := \sum_{i=1}^{k_n} \chi_{B_{n,i}}(t) \mathcal{F}(x_{n,i}),$$

where  $\mathcal{F}: X \to Y$  satisfies  $\mathcal{F}(z) = f(z)$  if  $z \in U$  and  $\mathcal{F}(z) = f(u)$  otherwise, is well defined and a simple function as well. In particular,

$$f_n(t) = f(x_n(t)) \text{ if } n \ge m(t), \text{ f.a.a. } t \in (0, T),$$

in view of (5.40). Note that here it was crucial that  $\{B_{n,i}\}_i$  is a disjoint partition of [0,T]. We can now deduce from (5.41) the Bochner measurability of  $f:[0,T] \to Y$ . The proof is now complete.

**Lemma 5.12.** Let X, Y, Z be Banach spaces and  $\mathfrak{U} \subset L^{\infty}(0,T;X)$ ,  $U \subset Y$  be open sets. Moreover, let  $\mathfrak{F}: \mathfrak{U} \to L^{\infty}(0,T;Y)$  and  $\mathcal{F}: U \to Z$  be two given mappings. Suppose that  $\mathcal{F}$  is Lipschitz continuous and directionally differentiable. Further, let  $x \in \mathfrak{U}$ ,  $\delta x \in L^{\infty}(0,T;X)$  be given and choose  $\varepsilon_1 > 0$  such that  $B_{L^{\infty}(0,T;X)}(x,\varepsilon_1\|\delta x\|_{L^{\infty}(0,T;X)}) \subset \mathfrak{U}$ . Assume that

$$\mathfrak{F}\big(B_{L^{\infty}(0,T;X)}(x,\varepsilon_1\|\delta x\|_{L^{\infty}(0,T;X)})\big)(t)\in U \text{ f.a.a. } t\in(0,T).$$

Moreover, let  $\delta y \in L^{\infty}(0,T;Y)$  be given and assume that there exists  $\varepsilon_2 > 0$ , independent of t, such that  $B_Y(\mathfrak{F}(x)(t),\varepsilon_2||\delta y||_{L^{\infty}(0,T;Y)}) \subset U$  f.a.a.  $t \in (0,T)$ . Then, for any  $\tau \in (0,\min\{\varepsilon_1,\varepsilon_2\})$ , we have

$$\left\| \frac{\mathcal{F}(\mathfrak{F}(x+\tau\delta x)(t)) - \mathcal{F}(\mathfrak{F}(x)(t))}{\tau} - \mathcal{F}'(\mathfrak{F}(x)(t);\delta y(t)) \right\|_{Z}$$

$$\leq L_{\mathcal{F}} \left\| \frac{\mathfrak{F}(x+\tau\delta x)(t) - \mathfrak{F}(x)(t)}{\tau} - \delta y(t) \right\|_{Y} + R_{\mathcal{F}}(t,\tau) \quad \text{f.a.a. } t \in (0,T),$$

where  $L_{\mathcal{F}} > 0$  is the Lipschitz constant of  $\mathcal{F}$  and  $R_{\mathcal{F}} : (0,T) \times (0,1) \to \mathbb{R}^+$  satisfies

$$R_{\mathcal{F}}(\cdot,\tau) \to 0 \text{ in } L^{\varrho}(0,T) \quad \text{as } \tau \searrow 0,$$

for any  $\varrho \in [1, \infty)$ .

*Proof.* First notice that due to  $\tau \in (0, \min\{\varepsilon_1, \varepsilon_2\})$  we have

$$x + \tau \delta x \in B_{L^{\infty}(0,T:X)}(x,\varepsilon_1 \|\delta x\|_{L^{\infty}(0,T:X)}) \subset \mathfrak{U}$$

and from  $\mathfrak{F}(B_{L^{\infty}(0,T;X)}(x,\varepsilon_1\|\delta x\|_{L^{\infty}(0,T;X)}))(t) \in U$  f.a.a.  $t \in (0,T)$  we deduce  $\mathfrak{F}(x+\tau\delta x)(t),\mathfrak{F}(x)(t)\in U$  f.a.a.  $t\in (0,T)$ . Moreover, the assumption on  $\varepsilon_2$  ensures that  $\mathfrak{F}(x)(t)+\tau\delta y(t)\in U$  f.a.a.  $t\in (0,T)$ . Therefore, all the quantities we deal with in the next estimate are well defined. Notice that the fact that  $\varepsilon_2>0$  is independent of t implies that  $\tau$  does not depend on t, which is crucial for proving the convergence in the second part of the proof. Now, from the Lipschitz continuity of  $\mathcal{F}$  we obtain for almost all  $t\in (0,T)$  the estimate

$$\begin{split} & \left\| \frac{\mathcal{F} \big( \mathfrak{F} (x + \tau \delta x)(t) \big) - \mathcal{F} \big( \mathfrak{F} (x)(t) \big)}{\tau} - \mathcal{F}' \big( \mathfrak{F} (x)(t); \delta y(t) \big) \right\|_{Z} \\ & \leq \left\| \frac{\mathcal{F} \big( \mathfrak{F} (x + \tau \delta x)(t) \big) - \mathcal{F} \big( \mathfrak{F} (x)(t) + \tau \delta y(t) \big)}{\tau} \right\|_{Z} \\ & + \left\| \frac{\mathcal{F} \big( \mathfrak{F} (x)(t) + \tau \delta y(t) \big) - \mathcal{F} \big( \mathfrak{F} (x)(t) \big)}{\tau} - \mathcal{F}' \big( \mathfrak{F} (x)(t); \delta y(t) \big) \right\|_{Z} \\ & \leq L_{\mathcal{F}} \left\| \frac{\mathfrak{F} (x + \tau \delta x)(t) - \mathfrak{F} (x)(t)}{\tau} - \delta y(t) \right\|_{Y} + R_{\mathcal{F}} (t, \tau), \end{split}$$

where  $R_{\mathcal{F}}(t,\cdot):(0,1)\to\mathbb{R}^+$  is given by

$$R_{\mathcal{F}}(t,\tau) = \begin{cases} \left\| \frac{\mathcal{F}(\mathfrak{F}(x)(t) + \tau \delta y(t)) - \mathcal{F}(\mathfrak{F}(x)(t))}{\tau} - \mathcal{F}'\big(\mathfrak{F}(x)(t); \delta y(t)\big) \right\|_{Z}, & \text{if } \tau \in (0, \min\{\varepsilon_{1}, \varepsilon_{2}\}), \\ 0, & \text{otherwise}, \\ \text{at almost all } t \in (0, T). \end{cases}$$

It now remains to prove the convergence of  $R_{\mathcal{F}}$ . This is shown by means of Lebesgue's dominated convergence theorem. To this end, first note that the mapping

$$t \mapsto \frac{\mathcal{F}(\mathfrak{F}(x)(t) + \tau \delta y(t)) - \mathcal{F}(\mathfrak{F}(x)(t))}{\tau} \in Z$$

is Bochner measurable at all  $\tau \in (0, \min\{\varepsilon_1, \varepsilon_2\})$ , in view of Lemma 5.11. As pointwise (a.e. in (0,T)) limit of Bochner measurable functions, the directional derivative  $t \mapsto \mathcal{F}'(\mathfrak{F}(x)(t); \delta y(t)) \in \mathbb{Z}$  is Bochner measurable as well, cf. [81, Corollary 3.1.5]. Hence, as a result of e.g. [71, Lemma 1.7],  $R_{\mathcal{F}}(\cdot, \tau)$  is measurable for all  $\tau \in (0,1)$ . Further, the directional differentiability of  $\mathcal{F}$  gives in turn

$$R_{\mathcal{F}}(t,\tau) \to 0$$
 as  $\tau \searrow 0$  f.a.a.  $t \in (0,T)$ .

By employing the Lipschitz continuity of  $\mathcal{F}$  and by applying Lemma 5.8 for  $f: U \times \{0\} \to Z$ ,  $f(\cdot,0) := \mathcal{F}$ , we arrive at

$$R_{\mathcal{F}}(t,\tau) \le 2L_{\mathcal{F}} \|\delta y(t)\|_{Y}$$
 f.a.a.  $t \in (0,T), \forall \tau \in (0,1)$ .

Since  $\delta y \in L^{\infty}(0,T;Y)$ , Lebesgue's dominated convergence theorem implies that

$$R_{\mathcal{F}}(\cdot,\tau) \to 0 \text{ in } L^{\varrho}(0,T) \quad \text{ as } \tau \searrow 0,$$

for any  $\varrho \in [1, \infty)$ . This completes the proof.

# **Bibliography**

- [1] J.-F. Babadjian, G. Francfort, and M.G. Mora. Quasistatic evolution in non-associative plasticity the cap model. SIAM J. Math. Anal., 44:245–292, 2012.
- [2] Viorel Barbu. Necessary conditions for distributed control problems governed by parabolic variational inequalities. SIAM Journal on Control and Optimization, 19(1):64–86, 1981.
- [3] M. Bergounioux and H. Zidani. A fully discrete approximation for control problems governed by parabolic variational inequalities. *SIAM J. Numer. Anal.*, 39(6):2014–2033, 2002.
- [4] E. Bonetti and G. Bonfanti. Well-posedness results for a model of damage in thermoviscoelastic materials. Ann. Inst. H. Poincaré Anal. Non Linéaire, 25:1187–1208, 2008.
- [5] E. Bonetti and G. Schimperna. Local existence to Frémond's model for damaging in elastic materials. *Contin. Mech. Thermodyn.*, 16:319–335, 2004.
- [6] E. Bonetti, G. Schimperna, and A. Segatti. On a doubly nonlinear model for the evolution of damaging in viscoelastic materials. J. Differential Equations, 218:91–116, 2005.
- [7] H. Brezis. Functional Analysis, Sobolev Spaces and Partial Differential Equations. Springer, New York, 2011.
- [8] M. Brokate, P. Krejčí, and H. Schnabel. On uniqueness in evolution quasivariational inequalities. *J. Convex Analysis*, 11:111–130, 2004.
- [9] H. Cartan. Differential Calculus. Kershaw Publishing Company, London, 1971.
- [10] J. C. De los Reyes and C. Meyer. Strong stationarity conditions for a class of optimization problems governed by variational inequalities of the second kind. *Journal of Optimization Theory and Applications*, 168(2):375–409, 2016.
- [11] J. Diestel and J. J. Uhl. *Vector Measures*. Mathematical Surveys and Monographs. American Mathematical Society, Providence, 1977.

- [12] B.J. Dimitrijevic and K. Hackl. A method for gradient enhancement of continuum damage models. *Technische Mechanik*, 28(1):43–52, 2008.
- [13] M. Efendiev and A. Mielke. On the rate-independent limit of systems with dry friction and small viscosity. *J. Convex Analysis*, 13:151–167, 2006.
- [14] E. Emmrich. Gewöhnliche und Operator Differentialgleichungen. Vieweg, Wiesbaden, 2004.
- [15] M.H. Farshbaf-Shaker. A penalty approach to optimal control of Allen-Cahn variational inequalities: MPEC-view. Numerical Functional Analysis and Optimization, 33(11):1321–1349, 2012.
- [16] M. Frémond and N. Kenmochi. Damage problems for viscous locking materials. *Adv. Math. Sci. Appl.*, 16(2):697–716, 2006.
- [17] M. Frémond and B. Nedjar. Damage, gradient of damage and principle of virtual power. *Int. J. Solids Struct.*, 33(8):1083–1103, 1996.
- [18] A. Friedman. Optimal control for parabolic variational inequalities. SIAM Journal on Control and Optimization, 25(2):482–497, 1987.
- [19] H. Gajewski, K. Gröger, and K. Zacharias. *Nichtlineare Operatorgleichungen und Operatordifferentialgleichungen*. Akademie-Verlag, Berlin, 1974.
- [20] E. Giusti. *Direct methods in the calculus of variations*. World Scientific Publishing Co. Inc., 2003.
- [21] H. Goldberg, W. Kampowsky, and F. Tröltzsch. On Nemytskij operators in  $L^p$  spaces of abstract functions. *Mathematische Nachrichten*, 155:127–140, 1992.
- [22] P. Grisvard. Elliptic Problems in Nonsmooth Domains. Pitman, Boston, 1985.
- [23] K. Gröger. A  $W^{1,p}$ -estimate for solutions to mixed boundary value problems for second order elliptic differential equations. *Math. Ann.*, 283:679–687, 1989.
- [24] R. Haller-Dintelmann, C. Meyer, J. Rehberg, and A. Schiela. Hölder continuity and optimal control for nonsmooth elliptic problems. Applied Mathematics and Optimization, 60:397–428, 2009.
- [25] Z.X. He. State constrained control problems governed by variational inequalities. SIAM Journal on Control and Optimization, 25:1119–1144, 1987.
- [26] C. Heinemann and C. Kraus. Existence of weak solutions for Cahn-Hilliard systems coupled with elasticity and damage. *Adv. Math. Sci. Appl.*, 21:321–359, 2011.
- [27] C. Heinemann and C. Kraus. *Phase Separation Coupled with Damage Processes*. Springer Spektrum, Wiesbaden, 2014.

- [28] R. Herzog and C. Meyer. Optimal control of static plasticity with linear kinematic hardening. J. Appl. Math. Mech., 91(10):777–794, 2011.
- [29] R. Herzog, C. Meyer, and G. Wachsmuth. Integrability of displacement and stresses in linear and nonlinear elasticity with mixed boundary conditions. *Journal of Mathematical Analysis and Applications*, 382(2):802–813, 2011.
- [30] R. Herzog, C. Meyer, and G. Wachsmuth. C-stationarity for optimal control of static plasticity with linear kinematic hardening. SIAM J. Optim., 50(5):3052–3082, 2012.
- [31] R. Herzog, C. Meyer, and G. Wachsmuth. B- and strong stationarity for optimal control of static plasticity with hardening. SIAM J. Optim., 23(1):321–352, 2013.
- [32] M. Hintermüller and I. Kopacka. Mathematical programs with complementarity constraints in function space: C- and strong stationarity and a path-following algorithm. SIAM Journal on Optimization, 20(2):868–902, 2009.
- [33] M. Hintermüller and D. Wegner. Distributed optimal control of the Cahn-Hilliard system including the case of a double-obstacle homogeneous free energy density. SIAM J. Control and Optimization, 50(1):388–418, 2012.
- [34] M. Hintermüller and D. Wegner. Optimal control of a semi-discrete Cahn-Hilliard-Navier-Stokes system. SIAM J. Control and Optimization, 52(1):747–772, 2014.
- [35] K. Ito and K. Kunisch. Optimal control of parabolic variational inequalities. *Journal de Mathématiques Pures et Appliqués*, 93(4):329–360, 2010.
- [36] W. Kaballo. Aufbaukurs Funktionalanalysis und Operatortheorie. Springer Spektrum, Heidelberg Berlin, 2014.
- [37] C. Kanzow and A. Schwartz. Mathematical programs with equilibrium constraints: enhanced Fritz John-conditions, new constraint qualifications, and improved exact penalty results. SIAM Journal on Optimization, 20(5):2730–2753, 2010.
- [38] D. Kinderlehrer and G. Stampacchia. An introduction to variational inequalities and their applications. SIAM, Philadelphia, 2000.
- [39] D. Knees, A. Mielke, and C. Zanini. On the inviscid limit of a model for crack propagation. *Math. Models Methods Appl. Sci.*, 18(9):1529–1569, 2008.
- [40] D. Knees, R. Rossi, and C. Zanini. A vanishing viscosity approach to a rate-independent damage model. Technical Report 1633, WIAS Berlin, 2011.
- [41] D. Knees, R. Rossi, and C. Zanini. A vanishing viscosity approach to a rate-independent damage model. *Mathematical Models and Methods in Applied Sciences*, 23(4):565–616, 2013.

- [42] D. Knees, R. Rossi, and C. Zanini. A quasilinear differential inclusion for viscous and rate-independent damage systems in non-smooth domains. *Nonlinear Analysis: Real World Applications*, 24:126–162, 2015.
- [43] D. Knees, C. Zanini, and A. Mielke. Crack propagation in polyconvex materials. Physica D, 239:1470–1484, 2010.
- [44] M. Kunze and M.D.P. Monteiro Marques. On parabolic quasi-variational inequalities and state-dependent sweeping processes. *Topol. Methods Nonlinear Anal.*, 12:179–191, 1998.
- [45] G. Lazzaroni and R. Toader. A model of crack propagation based on viscous approximation. *Math. Models Methods Appl. Sci.*, 21:1–29, 2011.
- [46] J. Makowski, K. Hackl, and H. Stumpf. The fundamental role of nonlocal and local balance laws of material forces in finite elastoplasticity and damage mechanics. *International Journal of Solids and Structures*, 43:3940 – 3959, 2006.
- [47] G. Dal Maso, A. DeSimone, and F. Solombrino. Quasistatic evolution for cam-clay plasticity: a weak formulation via viscoplastic regularization and time rescaling. *Calc. Var. Partial Differential Equations*, 40:125–181, 2011.
- [48] G. Dal Maso, A. DeSimone, and F. Solombrino. Quasistatic evolution for cam-clay plasticity: properties of the viscosity solutions. Calc. Var. Partial Differential Equations., 2012. In press. Online published since August 2011, DOI 10.1007/s00526-011-0443-6.
- [49] C. Meyer and L.M. Susu. Analysis of a viscous two-field gradient damage model. Part I: Existence and uniqueness. Technical Report 526, Ergebnisberichte des Instituts für Angewandte Mathematik, TU Dortmund, 2015.
- [50] C. Meyer and L.M. Susu. Optimal control of nonsmooth, semilinear parabolic equations. Technical Report 524, Ergebnisberichte des Instituts für Angewandte Mathematik, TU Dortmund, 2015.
- [51] C. Meyer and L.M. Susu. Analysis of a viscous two-field gradient damage model. Part II: Penalization limit. Technical Report 538, Ergebnisberichte des Instituts für Angewandte Mathematik, TU Dortmund, 2016.
- [52] A. Mielke. Differential, energetic, and metric formulations for rate-independent processes. In *Nonlinear PDE's and Applications*, volume 2028 of *Lect. Notes Math.*, pages 87–167. Springer, Berlin, 2011. C.I.M.E. Summer School, Cetraro, Italy 2008.
- [53] A. Mielke and R. Rossi. Existence and uniqueness results for a class of rate-independent hysteresis problems. *Math. Models Meth. Appl. Sci.*, 17:81–123, 2007.
- [54] A. Mielke, R. Rossi, and G. Savaré. Modeling solutions with jumps for rate-independent systems on metric spaces. *Discr. Cont. Dynam. Systems Ser. A*, 25(2):585–615, 2009.

- [55] A. Mielke, R. Rossi, and G. Savaré. BV solutions and viscosity approximations of rate-independent systems. *ESAIM Control Optim. Calc. Var.*, 18:36–80, 2012.
- [56] A. Mielke, R. Rossi, and G. Savaré. Balanced viscosity (BV) solutions to infinite-dimensional rate-independent systems. *J. Eur. Math. Soc.*, 18(9):2107–2165, 2016.
- [57] A. Mielke and T. Roubíček. *Rate-Independent Systems, Theory and Application*. Springer, Berlin, 2015.
- [58] A. Mielke and F. Theil. A mathematical model for rate-independent phase transformations with hysteresis. In H.-D. Alber, R.M. Balean, and R. Farwig, editors, Proceedings of the Workshop on "Models of Continuum Mechanics in Analysis and Engineering", page 117–129. Shaker- Verlag, Aachen, 1999.
- [59] A. Mielke and F. Theil. On rate-independent hysteresis models. Nonl. Diff. Eqns. Appl. (NoDEA), 11:151–189, 2004.
- [60] A. Mielke, F. Theil, and V.I. Levitas. A variational formulation of rate-independent phase transformations using an extremum principle. Arch. Rational Mech. Anal., 162:137–177, 2002.
- [61] A. Mielke and S. Zelik. On the vanishing viscosity limit in parabolic systems with rate-independent dissipation terms. *Ann. Sc. Norm. Sup. Pisa Cl. Sci.*, XIII(5):67–135, 2014.
- [62] F. Mignot and J.-P. Puel. Contrôle optimal d'un système gouverné par une inéquation variationelle parabolique. Comptes Rendus de l'Académie des Sciences Paris, 298(12):277–280, 1984.
- [63] F. Mignot and J.-P. Puel. Optimal control in some variational inequalities. SIAM Journal on Control and Optimization, 22(3):466–476, 1984.
- [64] B. Nedjar. Elastoplastic-damage modelling including the gradient of damage: formulation and computational aspects. *International Journal of Solids and Structures*, 38:5421 – 5451, 2001.
- [65] J. Outrata. Optimality conditions for a class of mathematical programs with equilibrium constraints. *Mathematics of Operations Research*, 24(3):627–644, 1999.
- [66] J. Outrata, M. Kočvara, and J. Zowe. Nonsmooth approach to optimization problems with equilibrium constraints: Theory, applications, and numerical results. In *Nonconvex Optimization and its Applications*, volume 28. Kluwer Academic Publishers, Dordrecht, The Netherlands, 1998.
- [67] R. Peerlings. Enhanced damage modeling for fracture and fatigue. Dissertation, Technische Universität Eindhoven, 1999.
- [68] E. Rocca and R. Rossi. A degenerating PDE system for phase transitions and damage. Math. Models and Methods in Applied Sciences, 24(7):1265–1341, 2014.

- [69] R.T. Rockafellar. Convex Analysis. Princeton University Press, Princeton, 1970.
- [70] T. Roubíček. Nonlinear Partial Differential Equations with Applications, volume 153 of International Series of Numerical Mathematics. Birkhäuser, Basel, 2005.
- [71] M. Růžička. Nichtlineare Funktionalanalysis. Springer, Berlin Heidelberg, 2004.
- [72] H. Scheel and S. Scholtes. Mathematical programs with complementarity constraints: Stationarity, optimality, and sensitivity. *Mathematics of Operations Research*, 25(1):1–22, 2000.
- [73] W. Schirotzek. Nonsmooth Analysis. Springer, Berlin, 2007.
- [74] L. Schwartz. Analyse Mathematique I. Hermann, Paris, 1967.
- [75] A. Shapiro. On concepts of directional differentiability. *Journal of Optimization Theory and Applications*, 66(3):477–487, 1990.
- [76] A. Simone, H. Askes, R. Peerlings, and L. Sluys. Interpolation requirements for implicit gradient-enhanced continuum damage models. *Communications in Numerical Methods in Engineering*, 19:563–572, 2003.
- [77] M. Sofonea and M. Andaluzia. Variational Inequalities with Applications. Springer, New York, 2009.
- [78] R. Toader and C. Zanini. An artificial viscosity approach to quasistatic crack growth. *Boll. Unione Mat. Ital.*, 2(9):1–35, 2009.
- [79] H. Triebel. Interpolation Theory, Function Spaces, Differential Operators. North-Holland, Amsterdam, 1978.
- [80] F. Tröltzsch. Optimal Control of Partial Differential Equations, volume 112 of Graduate Studies in Mathematics. American Mathematical Society, Providence, 2010. Theory, methods and applications, Translated from the 2005 German original by Jürgen Sprekels.
- [81] G. Wachsmuth. Optimal control of quasistatic plasticity. Dissertation, Chemnitz University of Technology, 2011.
- [82] G. Wachsmuth. Strong stationarity for optimal control of the obstacle problem with control constraints. SIAM Journal on Optimization, 24(4):1914–1932, 2014.
- [83] T. Waffenschmidt, C. Polindara, A. Menzel, and S. Blanco. A gradient-enhanced large-deformation continuum damage model for fibre-reinforced materials. *Comput. Meth. Appl. Mech. Engrg.*, 268:801–842, 2014.
- [84] E. Zeidler. Nonlinear Functional Analysis and its Applications I. Fixed-Point Theorems. Springer, New York, 1986.